## **UNITED STATES** SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, DC 20549

## FORM 10-Q

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended June 30, 2019

OR

TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Commission File Number: 001-36729



Delaware
(State or other jurisdiction of incorporation or organization)
400 Plaza Drive, 1st Floor, Secaucus, New Jersey
(Address of principal executive offices)

20-1884894 (I.R.S. Employer Identification No.) 07094 (Zip Code)

Registrant's telephone number, including area code: (201) 520-4000

Securities registered pursuant to Section 12(b) of the Act:

	Title of each class	Trading Symbol	Name of each exchange on which registered
	Common Stock	FRPT	NASDAQ Global Market
reports),	Indicate by check mark whether the registrant (1) has filed all reports required to be find (2) has been subject to such filing requirements for the past 90 days. Yes $\boxtimes$ N		g the preceding 12 months (or for such shorter period that the registrant was required to file such
for such	Indicate by check mark whether the registrant has submitted electronically if any, horter period that the registrant was required to submit such files). Yes $\boxtimes$ No $\square$	every Interactive Data File required to be submitted and posted pursuant	to Rule 405 of Regulation S-T (§ 232.405 of this chapter) during the preceding 12 months (or
"smaller	Indicate by check mark whether the registrant is a large accelerated filer, an accele eporting company," and "emerging growth company" in Rule 12b-2 of the Exchange		erging growth company. See the definitions of "large accelerated filer," "accelerated filer,"
Large	ccelerated filer		Accelerated filer
Non-ac	celerated filer		Smaller reporting company
Emerg	ng growth company		
Act. □	If an emerging growth company, indicate by check mark if the registrant has elect	d not to use the extended transition period for complying with any new	or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange
	Indicate by check mark whether the registrant is a shell company (as defined in Ru	le 12b-2 of the Exchange Act). Yes □ No ⊠	
As of Au	gust 2, 2019, the registrant had 36,076,406 shares of common stock, \$0.001 par value p	er share, outstanding.	

## TABLE OF CONTENTS

		Page No.
Part I. Financia	l Information	4
Item 1.	Financial Statements	4
	Consolidated Balance Sheets	4
	Consolidated Statements of Operations and Comprehensive Loss	5
	Changes to Statements of Stockholders' Equity	6
	Consolidated Statements of Cash Flows	7
	Notes to Consolidated Financial Statements	8
Item 2.	Management's Discussion and Analysis of Financial Conditions and Results of Operations	16
Item 3.	Quantitative and Qualitative Disclosures About Market Risks	27
Item 4.	Controls and Procedures	28
Part II. Other In		29
Item 1.	Legal Proceedings	29
Item 1A.	Risk Factors	29
Item 6.	Exhibits	30

### Forward-Looking Statements

This report contains forward-looking statements that are subject to risks and uncertainties. All statements other than statements of historical fact included in this report are forward-looking statements. Forward-looking statements discuss our current expectations and projections relating to our financial condition, results of operations, plans, objectives, future performance and business. You can identify forward-looking statements by the fact that they do not relate strictly to historical or current facts. These statements may include words such as "aim," "anticipate," "believe," "estimate," "expect," "forecast," "outlook," "potential," "projectio," "plan," "intend," "seek," "may," "could," "would," "will," "should," "can," "can have," "likely," the negatives thereof and other words and terms of similar meaning in connection with any discussion of the timing or nature of future operating or financial performance or other events. They appear in a number of places throughout this report and include statements regarding our intentions, beliefs or current expectations concerning, among other things, our results of operations, financial condition, liquidity, prospects, growth, strategies and the industry in which we operate. All forward-looking statements are subject to risks and uncertainties that may cause actual results to differ materially from those that we expected, including:

- · our ability to successfully implement our growth strategy;
- · our ability to timely complete the construction of our Freshpet Kitchens 2.0 and achieve the anticipated benefits therefrom;
- our ability to generate sufficient cash flow or raise capital on acceptable terms;
- the loss of key members of our senior management team;
- · allegations that our products cause injury or illness or fail to comply with government regulations;
- the loss of a significant customer;
- the entrance of new competitors into our industry;
- the effectiveness of our marketing and trade spending programs;
- our ability to introduce new products and improve existing products;
- · our limited manufacturing capacity;
- the impact of government regulation, scrutiny, warning and public perception;
- · the effect of false marketing claims;
- adverse weather conditions, natural disasters, pestilences and other natural conditions affecting our operations;
- · our ability to develop and maintain our brand;
- the effect of potential price increases and shortages on the inputs, commodities and ingredients that we require;
- · our ability to manage our supply chain effectively;
- · volatility in the price of our common stock; and
- other factors discussed under the headings "Risk Factors" and "Business" in our Annual Report on Form 10-K and under the heading "Management's Discussion and Analysis of Financial Condition and Results of Operations" in our Annual Report on Form 10-K and in this report.

While we believe that our assumptions are reasonable, we caution that it is very difficult to predict the impact of known factors and it is impossible for us to anticipate all factors that could affect our actual results. All forward-looking statements are expressly qualified in their entirety by these cautionary statements. You should evaluate all forward-looking statements made in this report in the context of these risks and uncertainties.

# CONSOLIDATED BALANCE SHEETS (Unaudited)

	j	lune 30, 2019	I	December 31, 2018
ASSETS		· -		
CURRENT ASSETS:				
Cash and cash equivalents	\$	4,752,836	\$	7,554,388
Accounts receivable, net of allowance for doubtful accounts		19,356,428		12,326,703
Inventories, net		13,225,832		9,317,232
Prepaid expenses		1,286,692		1,078,232
Other current assets		650,482		681,550
Total Current Assets		39,272,270		30,958,105
Property, plant and equipment, net		122,394,944		102,094,248
Deposits on equipment		6,437,602		4,730,176
Operating lease right of use assets		9,763,324		_
Other assets		3,314,858		2,182,329
Total Assets	\$	181,182,998	\$	139,964,858
LIABILITIES AND STOCKHOLDERS' EQUITY				
CURRENT LIABILITIES:				
Accounts payable		15,801,273		9,166,412
Accrued expenses		9,018,582		9,050,551
Current operating lease liabilities		1,068,396		_
Other current liabilities		200,000		
Total Current Liabilities	\$	26,088,251	\$	18,216,963
Long term debt		28,482,000		_
Long term operating lease liabilities		9,020,293		_
Other liabilities				273,420
Total Liabilities	\$	63,590,544	\$	18,490,383
STOCKHOLDERS' EQUITY:				
Common stock — voting, \$0.001 par value, 200,000,000 shares authorized, 36,089,945 issued and 36,075,776 outstanding on June 30, 2019, and 35,556,595 issued and 35,542,426 outstanding on December 31, 2018		36,089		35,556
Additional paid-in capital		328,342,507		323,079,437
Accumulated deficit		(210,436,032)		(201,352,682)
Accumulated other comprehensive income		(93,884)		(31,610)
Treasury stock, at cost $-$ 14,169 shares on June 30, 2019 and on December 31, 2018		(256,226)		(256,226)
Total Stockholders' Equity		117,592,454		121,474,475
Total Liabilities and Stockholders' Equity	\$	181,182,998	\$	139,964,858

## CONSOLIDATED STATEMENTS OF OPERATIONS AND COMPREHENSIVE LOSS (Unaudited)

For the Three Months Ended For the Six Months Ended June 30, June 30, 2019 2018 NET SALES COST OF GOODS SOLD 60,052,179 114,844,381 90,794,557 61,602,819 47,788,647 32,725,598 24,747,064 GROSS PROFIT 27,326,581 22,877,892 53,241,562 43,005,910 SELLING GENERAL AND ADMINISTRATIVE EXPENSES 61,904,534 (8,662,972) 49.825.467 LOSS FROM OPERATIONS OTHER INCOME/(EXPENSES): Other Income/(Expenses), net (20,748) 26,067 (3,453) 3,090 Interest Expense (275,649) (97,923) (378,425) (166,925) (296,397) (71,856) (381,878) (163,835) LOSS BEFORE INCOME TAXES INCOME TAX EXPENSE (5,642,100) 19,250 (3,481,487) 19,032 (9,044,850) 38,500 (6,983,392) 38,064 (7,021,456) LOSS ATTRIBUTABLE TO COMMON STOCKHOLDERS (5,661,350) (3,500,519) (9,083,350) OTHER COMPREHENSIVE LOSS:
Change in foreign currency translation (153,321) (201,187) (62,274) (1,633) TOTAL OTHER COMPREHENSIVE LOSS (153,321) (5,814,671) (201,187) (3,701,706) (62,274) (9,145,624) (1,633) TOTAL COMPREHENSIVE LOSS NET LOSS PER SHARE ATTRIBUTABLE TO COMMON STOCKHOLDERS (0.20) (0.16) (0.10) (0.25)-BASIC -DILUTED (0.10) (0.25) (0.20) (0.16) WEIGHTED AVERAGE SHARES OF COMMON STOCK OUTSTANDING USED IN COMPUTING NET LOSS PER SHARE ATTRIBUTABLE TO COMMON STOCKHOLDERS -BASIC 35,930,350 35,241,198 35,800,061 35,189,636 -DILUTED 35.930.350 35,241,198 35,800,061 35,189,636

# CHANGES TO STATEMENTS OF STOCKHOLDERS' EQUITY (Unaudited)

	Common S	tock - Vo	ting						Treasu	ry Stoc	k		
	Number of Shares Issued		Amount	Ad	dditional Paid-in Capital	Aco	cumulated Deficit	cumulated Other Comprehensive Income	Number of Shares		Amount	Tota	al Stockholders' Equity
BALANCES, December 31, 2018	35,556,595	\$	35,556	\$	323,079,437	\$	(201,352,682)	\$ (31,610)	14,169	\$	(256,226)	\$	121,474,475
Exercise of options to purchase common stock	248,195		248		1,791,420			 	_		_		1,791,668
Issuance of restricted stock units	61,532		62		(673,836)		_	_	_		_		(673,774)
Share-based compensation expense	_		_		1,260,126		_	_	_		_		1,260,126
Foreign Currency Translation	_		_		_		_	91,047	_		_		91,047
Net loss							(3,422,000)						(3,422,000)
BALANCES, March 31, 2019	35,866,322	\$	35,866	\$	325,457,147	\$	(204,774,682)	\$ 59,437	14,169	\$	(256,226)	\$	120,521,542
Exercise of options to purchase common stock	194,497		194		1,983,685								1,983,879
Issuance of restricted stock units	29,126		29		(579,207)		_	_	_		_		(579,178)
Share-based compensation expense	_		_		1,480,882		_	_	_		_		1,480,882
Foreign Currency Translation	_		_		_		_	(153,321)	_		_		(153,321)
Net loss							(5,661,350)						(5,661,350)
BALANCES, June 30, 2019	36,089,945	\$	36,089	\$	328,342,507	\$	(210,436,032)	\$ (93,884)	14,169	\$	(256,226)	\$	117,592,454

	Common St	ock - Vo	ting						Treasur	y Stock	<		
	Number of Shares Issued		Amount	Ad	lditional Paid-in Capital	Acc	umulated Deficit	cumulated Other Comprehensive Income	Number of Shares		Amount	Tota	al Stockholders' Equity
BALANCES, December 31, 2017	35,132,548	\$	35,132	\$	312,783,195	\$	(195,991,478)	\$ 76,129		\$		\$	116,902,978
Exercise of options to purchase common stock	6,696		7		53,855		_	_	_		_		53,862
Share-based compensation expense	_		_		1,092,260		_	_	_		_		1,092,260
Foreign Currency Translation	_		_		_		_	199,554	_		_		199,554
Net loss							(3,520,937)						(3,520,937)
BALANCES, March 31, 2018	35,139,244	\$	35,139	\$	313,929,310	\$	(199,512,415)	\$ 275,683		\$		\$	114,727,717
Exercise of options to purchase common stock	103,693		104		945,324								945,428
Issuance of restricted stock units	74,583		75		(75)		_	_	14,169		(256,226)		(256,226)
Share-based compensation expense	_		_		1,301,774		_	_	_				1,301,774
Foreign Currency Translation	_		_		_		_	(201,187)	_		_		(201,187)
Net loss							(3,500,519)						(3,500,519)
BALANCES, June 30, 2018	35,317,520	\$	35,317	\$	316,176,333	\$	(203,012,934)	\$ 74,496	14,169	\$	(256,226)	\$	113,016,987

# CONSOLIDATED STATEMENTS OF CASH FLOWS (Unaudited)

## For the Six Months Ended

	Jur	June 30,		
	2019		2018	
CASH FLOWS FROM OPERATING ACTIVITIES:				
Net loss	\$ (9,083,350)	\$	(7,021,456)	
Adjustments to reconcile net loss to net cash flows provided by operating activities:				
Provision for loss/(gains) on accounts receivable	(2,520)		(17,443)	
Loss on disposal of equipment	684		76,261	
Share-based compensation	2,630,180		2,394,034	
Inventory obsolescence	105,170		69,074	
Depreciation and amortization	7,643,452		6,795,380	
Amortization of deferred financing costs and loan discount	72,294		57,551	
Changes in operating assets and liabilities:				
Accounts receivable	(7,027,205)		1,618,733	
Inventories	(4,013,770)		(241,151)	
Prepaid expenses and other current assets	(177,392)		(262,533)	
Operating lease right of use	(177,249)		_	
Other assets	(44,498)		(14,371)	
Accounts payable	(158,556)		1,665,187	
Accrued expenses	(31,969)		(1,964,518)	
Other lease liabilities	229,194		(14,994)	
Other current liabilities	200,000		_	
Net cash flows provided by (used in) operating activities	(9,835,535)		3,139,754	
CASH FLOWS FROM INVESTING ACTIVITIES:				
Acquisitions of property, plant and equipment, software and deposits on equipment	(22,888,753)		(8,932,791)	
Net cash flows used in investing activities	(22,888,753)		(8,932,791)	
CASH FLOWS FROM FINANCING ACTIVITIES:	(,,		(0,002,.02)	
Proceeds from exercise of options to purchase common stock	3.775.548		999.289	
Tax withholdings related to net shares settlements of restricted stock units	(1,252,953)		(256,226)	
Proceeds from borrowings under Credit Facilities	35,307,000		6,000,000	
Repayment of borrowings under Credit Facilities	(7,500,000)		(2,000,000)	
Financing fees paid in connection with borrowings	(406,859)		_	
Net cash flows provided by financing activities	29,922,736		4,743,063	
NET CHANGE IN CASH AND CASH EQUIVALENTS	(2,801,552)		(1,049,974)	
CASH AND CASH EQUIVALENTS, BEGINNING OF YEAR	7,554,388		2,184,259	
CASH AND CASH EQUIVALENTS, END OF PERIOD	\$ 4,752,836	\$	1,134,285	
SUPPLEMENTAL CASH FLOW INFORMATION:	1,102,000	_	_,,_00	
Interest paid	\$ 165,844	\$	104,764	
INDICASH FINANCING AND INVESTING ACTIVITIES:	Ψ 103,044	Ψ	104,704	
Property, plant and equipment purchases in accounts payable	\$ 7,662,437	\$	696,024	
r toperty, plant and equipment purchases in accounts payable	\$ 1,002,431	Ψ	090,024	

## NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(Unaudited)

### Note 1 - Nature of the Business and Summary of Significant Accounting Policies:

Nature of the Business – Freshpet, Inc. (hereafter referred to as "Freshpet" the "Company", "we" or "our"), a Delaware corporation, manufactures and markets natural fresh meals and treats for dogs and cats. The Company's products are distributed throughout the United States and other international markets, into major retail classes including Grocery (including online), Mass and Club, Pet Specialty, and Natural retail.

Basis of Presentation — The accompanying unaudited consolidated financial statements have been prepared in accordance with accounting principles generally accepted in the U.S. ("U.S. GAAP"). The unaudited consolidated financial statements include the accounts of the Company as well as the Company's wholly-owned subsidiaries. All intercompany accounts and transactions have been eliminated in consolidation. The interim unaudited consolidated financial statements have been prepared on the same basis as the annual audited consolidated financial statements and in accordance with the rules and regulations of the United States Securities and Exchange Commission. In the opinion of management, the interim unaudited financial statements reflect all adjustments, which include only normal recurring adjustments, necessary for the fair statement of the Company's financial position as of June 30, 2019, the results of its operations and changes to stockholders' equity for the three and six months ended June 30, 2019 and 2018. The results for the three and six months ended June 2019 and 2018. The results for the three and six months ended June 30, 2019 are not necessarily indicative of results to be expected for the year ending December 31, 2019, or any other interim periods, or any future year or period.

These consolidated financial statements should be read in conjunction with the audited consolidated financial statements and accompanying notes in Item 8 of Part II, "Financial Statements and Supplementary Data," of our Annual Report on Form 10-K for the year ended December 31, 2018.

Estimates and Uncertainties – The preparation of our consolidated financial statements in conformity with U.S. GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the consolidated financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results, as determined at a later date, could differ from those estimates.

Restricted Stock Tax Withholdings – To meet payroll tax withholdings obligations arising from the vesting of restricted share units, the Company withheld 13,383 shares totaling \$.06 million, for the three month period ended June 30, 2019 and, withheld 31,734 shares totaling \$1.3 million, for the six month period ended June 30, 2019. Shares of common stock withheld for tax withholdings do not reduce the Company's total share repurchase authority.

Debt Issuance Cost - During the six months ended June 30, 2019 the Company incurred \$1.1 million of fees associated with the debt modification, of which \$0.7 million were paid to the creditor. Further, the Company noted that \$0.7 million of the fees were related to the Draw Term Loan with the remaining balance relating to the Revolving Loan Facility. The Company's policy is to record the debt issuance cost related to the Draw Term Loan, net of debt, for the portion of the Draw Term Loan that is outstanding, with the remaining amount recorded within assets. As of June 30, 2019 there was less than \$0.1 million of debt issuance cost that were recorded as net of our debt outstanding. In addition, there was \$0.8 million of debt issuance cost that were recorded to other assets, and \$0.2 recorded to other current assets.

The Company amortizes debt issuance costs categorized as assets on a straight-line basis over the term of the loan, and amortizes the debt issuance costs that are categorized net of debt using the effective interest method, over the term of the loan.

### Note 2 - Recently Issued Accounting Standards:

In February 2016, the Financial Accounting Standards Board (FASB) issued Accounting Standards Update No. 2016-02 (Topic 842) "Leases." Topic 842 supersedes the lease requirements in Accounting Standards Codification (ASC) Topic

### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(Unaudited)

840, "Leases." Under Topic 842, lessees are required to recognize assets and liabilities on the balance sheet for most leases and provide enhanced disclosures. Leases will continue to be classified as either finance or operating. In July 2018, the FASB issued ASU 2018-11, "Leases (ASC 842): Targeted Improvements," which provides companies an optional adoption method to ASU 2016-02 whereby a company does not have to adjust comparative period financial statements for the new standard.

The reported results as of June 30, 2019 reflect the application of ASC 842, while the comparative information has not been restated and continues to be reported under the related lease accounting standards in effect for those periods. The adoption of this update represents a change in accounting principle and resulted in the recognition of right-of-use ("ROU") assets and operating lease liabilities. We elected the package of practical expedients, which permits us not to reassess our prior conclusions about lease identification, lease classification and initial direct costs incurred. We also elected the practical expedient to combine lease and non-lease components when determining the ROU asset and lease liability, as well as the practical expedient to exclude leases with an initial term of 12 months or less. The primary effect of adopting this standard relates to the recognition of our operating leases on our Consolidated Balance Sheets (Unaudited) and providing additional disclosures about our leasing activities.

The Company is a lessee in several noncancellable operating leases, primarily for office and warehouse space, as well as office equipment.

The Company determines if an arrangement is or contains a lease at contract inception. The Company recognizes a ROU asset and a lease liability at the lease commencement date. The operating lease liability is initially and subsequently measured at the present value of the unpaid lease payments at the lease commencement date.

Key estimates and judgments include how the Company determines (1) the discount rate it uses to discount the unpaid lease payments to present value, (2) lease term and (3) lease payments.

- ASC 842 requires a lessee to discount its unpaid lease payments using the interest rate implicit in the lease or, if that rate cannot be readily determined, its incremental borrowing rate. Generally, the Company cannot determine the interest rate implicit in the lease because it does not have access to the lessor's estimated residual value or the amount of the lessor's deferred initial direct costs. Therefore, at the date of adoption, we used our incremental borrowing rate based on the information available at January 1, 2019.
- The lease term for all of the Company's leases includes the noncancellable period of the lease plus any additional periods covered by either a Company option to extend (or not to terminate) the lease that the Company is reasonably certain to exercise.
- Lease payments included in the measurement of the lease liability comprise of fixed payments based on the terms of the lease. Certain arrangements have free rent periods or escalating rent payment provisions. We recognize rent expense on a straight-line basis over the lease term, including any periods of free rent.

ROU assets are initially and subsequently measured at cost, which comprises the initial amount of the lease liability adjusted for lease payments made at or before the lease commencement date, plus any initial direct

The ROU asset is subsequently measured throughout the lease term at the carrying amount of the lease liability, plus initial direct costs, plus (minus) any prepaid (accrued) lease payments, less the unamortized balance of lease incentives received. Lease expense for lease payments is recognized on a straight-line basis over the lease term.

ROU assets for operating leases are periodically reviewed for impairment losses. The Company uses the long-lived assets impairment guidance in "ASC Subtopic 360-10, Property, Plant, and Equipment – Overall", to determine whether a ROU asset is impaired, and if so, the amount of the impairment loss to recognize.

The Company monitors for events or changes in circumstances that require a reassessment of one of its leases. When a reassessment results in the remeasurement of a lease liability, a corresponding adjustment is made to the carrying

## NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(Unaudited)

amount of the corresponding ROU asset unless doing so would reduce the carrying amount of the ROU asset to an amount less than zero.

Operating lease ROU assets are presented as operating lease right of use assets on the consolidated balance sheet. The current portion of operating lease liabilities and the long-term portion are presented separately as current and long term operating lease liabilities on the consolidated balance sheet.

## Note 3 – Inventories:

	ıne 30, 2019	December 31, 2018
Raw Materials and Work in Process	\$ 3,492,941	\$ 2,784,233
Packaging Components Material	1,365,816	1,138,091
Finished Goods	8,471,453	5,442,338
	13,330,210	9,364,662
Reserve for Obsolete Inventory	(104,378)	(47,430)
	\$ 13,225,832	\$ 9,317,232

## Note 4 - Property, Plant and Equipment:

	June 30, 2019	December 31, 2018
Refrigeration Equipment	\$ 90,358,958	\$ 79,567,389
Machinery and Equipment	53,681,940	51,800,479
Building, Land, and Improvements	25,621,499	25,606,013
Furniture and Office Equipment	5,210,704	4,884,945
Leasehold Improvements	393,770	393,770
Automotive Equipment	319,490	319,496
Construction in Progress	19,510,25	4,769,268
	195,096,614	167,341,360
Less: Accumulated Depreciation	(72,701,670	(65,247,112)
	\$ 122.394.94	\$ 102.094.248

Depreciation expense related to property, plant and equipment totaled \$3,843,199 and \$7,486,692 for the three and six months ended June 30, 2019, respectively, of which \$1,588,692 and \$3,154,780 was recorded to cost of goods sold for the three and six months ended June 30, 2019, respectively, with the remainder of depreciation and amortization expense recorded to selling, general and administrative expense. Due to the continued growth of the Company's fresh pet food sales, the Company has undertaken a capital expansion project at its Freshpet Kitchens manufacturing facility to expand plant capacity and increase distribution. The Company invested \$11.2 million in the Freshpet Kitchens expansion during 2019.

Depreciation expense related to property, plant and equipment totaled \$3,389,814 and \$6,658,713 for the three and six ended June 30, 2018, respectively, of which \$1,498,334 and \$2,989,329 was recorded to cost of goods sold for the three and six months ended June 30, 2018, respectively, with the remainder of depreciation and amortization expense recorded to selling, general and administrative expense.

## NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(Unaudited)

### Note 5 - Accrued Expenses:

	 June 30, 2019	December 2018	
Accrued Compensation and Employee Related Costs	\$ 2,478,041	\$	5,276,552
Accrued Chiller Cost	1,312,623		1,401,762
Accrued Customer Consideration	1,887,361		650,567
Accrued Commission	204,800		114,901
Accrued Freight	653,192		405,733
Accrued Production Expenses	1,384,460		287,913
Accrued Marketing	596,884		551,681
Other Accrued Expenses	501,221		361,442
	\$ 9,018,582	\$	9,050,551

## Note 6 - Debt:

On May 15, 2019, the Company entered into the Fourth Amended and Restated Loan and Security Agreement (the "New Loan Agreement"), which amended and restated in full the Company's Third Amended and Restated Loan and Security Agreement, dated as of September 21, 2017. The New Loan Agreement provides for a \$90 million senior secured credit facility (the "New Credit Facility"), encompassing a \$55.0 million delayed draw term loan facility (the "Draw Facility") and a \$35.0 million revolving loan facility (the "Revolving Loan Facility"). The Company will have the ability to increase the New Credit Facility by up to an additional \$75.0 million, subject to certain conditions.

The New Credit Facility will mature on May 15, 2024 and borrowings thereunder will bear interest at variable rates depending on the Company's election, either at a base rate or at LIBOR, in each case, plus an applicable margin. Subject to the Company's leverage ratio, the applicable margin will vary between 0.50% and 1.00% for base rate loans and 1.50% and 2.00% for LIBOR loans. Upon closing the New Credit Facility, the Company borrowed \$15.0 million under the Revolving Loan Facility, which left \$20.0 million of availability. The Company has the option to borrow term loans under the Draw Facility ("Draw Term Loans") until May 15, 2021, subject to certain conditions. Commencing on June 30, 2021, the amount of any outstanding Draw Term Loans shall be repayable in equal consecutive quarterly installments equal to 1/28th of the outstanding Draw Term Loans and the remainder shall be due and payable on May 15, 2024.

Borrowings under the New Credit Facility are secured by substantially all of the Company's and certain of its subsidiaries' assets. The New Loan Agreement requires compliance with various covenants customary for agreements of this type, including financial covenants and negative covenants that limit, among other things, the Company's ability to incur additional debt, create or incur liens, engage in mergers or consolidations, sell, transfer or otherwise dispose of assets, make voluntary prepayments to subordinated debt, permit a change of control, pay dividends or distributions, make investments, and enter into certain transactions with affiliates. The New Loan Agreement also includes events of default customary for agreements of this type.

Net borrowing under our credit facilities totaled \$28.5 million at June 30, 2019, of which \$3.5 million was term loans and \$25.0 million was revolving loan. There was no debt outstanding as of December 31, 2018.

Interest expense and fees totaled \$0.3 million and \$0.4 for the three and six months ended June 30, 2019, respectively, and \$0.1 million and \$0.2 million for the three and six months ended June 30, 2018, respectively. There was \$0.1 million of accrued interest on the credit facilities as of June 30, 2019. In connection with the new credit facility, the Company accelerated the amortization of \$0.1 million of unamortized debt issuance costs related to the existing loan agreement. These costs are included in interest expense in the three and six months ended June 30, 2019. Also in connection with the new credit facility, the Company capitalized \$1.1 million of financing costs that are presented with other assets as of June 30, 2019.

## NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(Unaudited)

### Note 7- Leases:

We have various noncancellable lease agreements for office and warehouse space, as well as office equipment, with original remaining lease terms of two years to nine years, some of which include an option to extend the lease term for up to five years. Because the Company is not reasonably certain to exercise these renewal options, the options are not considered in determining the lease term, and associated potential option payments are excluded from lease payments. The Company's leases generally do not include termination options for either party to the lease or restrictive financial or other covenants.

Weighted-average remaining lease term (in years) and discount rate related to operating leases were as follows:

Weighted-average remaining lease term	6.69
Weighted-average discount rate	6.14%

As most of our leases do not provide an implicit rate, we use our incremental borrowing rate based on the information available at the date of adoption to determine the present value of lease payments.

Maturities of lease liabilities under noncancellable operating leases as of June 30, 2019 were as follows:

	Jun	e 30 ,2019
2019 (a)	\$	797,856
2020		1,723,951
2021		1,768,626
2022		1,763,787
2023		1,802,007
2024 and beyond		4,297,218
Total lease payments		12,153,445
Less: Imputed interest		(2,064,756)
Present value of lease liabilities	\$	10,088,689

Excluding the six months ended June 30, 2019. (a)

As of December 31, 2018 minimum lease payments under noncancellable operating leases by period were expected to be:

As of December 31, 2010 minimum lease payments under nonconcentable operating leases by	 December 31, 2018
2019	\$ 1,475,761
2020	1,626,179
2021	1,671,003
2022	1,664,510
2023	1,700,539
2024 and thereafter	4,246,015
	\$ 12,384,007

## NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(Unaudited)

A summary of rent expense for the years ended December 31, 2018 and 2017 was as follows:

	20:	18	2017
Rent expense	\$	515.565 \$	480.349

Supplemental cash flow information and non-cash activity relating to operating leases are as follows:

	For the Three	Months Ended June 30, 2019	For the Six Months Ended June 30, 2019		
Operating cash flow information:					
Cash paid for amounts included in the measurement of lease liabilities	\$	380,046	\$	626,781	

### Note 8 - Equity Incentive Plans:

Total compensation cost for share-based payments recognized for the three months ended June 30, 2019 and 2018 is \$1,480,882 and \$1,301,774, respectively. Total compensation cost for share-based payments recognized for the six months ended June 30, 2019 and 2018 \$2,741,009 and \$2,394,034, respectively.

2010 Stock Plan—The outstanding options are time-based (vest between two and four years). At June 30, 2019, there were zero shares available for grant as the 2010 Plan is frozen. The total number of unexercised options for the 2010 Plan is 294,955.

2014 Omnibus Incentive Plan—In November 2014, the Company approved the 2014 Omnibus Incentive Plan (the "2014 Plan"). Under the 2014 Plan 3,979,200 shares of common stock may be issued or used for reference purposes as awards granted. These awards may be in the form of stock options, stock appreciation rights, restricted stock, as well as other stock-based and cash-based awards. At June 30, 2019, the awards granted were either time-based, performance-based (vest when performance targets are met, as defined in the stock option grant agreement), or restricted stock units (employee RSUs vest over three years or cliff vest (as defined in the restricted stock agreement) and non-employee director RSUs vest over one year). The total number of unexercised options and RSUs for the 2014 Plan is 1,539,172.

At June 30, 2019, there were 1,674,876 shares of common stock available to be issued or used for reference purposes under the 2014 Plan.

NASDAQ Marketplace Rules Inducement Award—During fiscal year 2016, share-based awards were granted to the Company's Chief Executive Officer as an inducement under the NASDAQ Marketplace Rules, and therefore outside of any Plan. Under the terms of the agreement, the grant is governed as if issued under the 2014 Omnibus Plan. As of June 30, 2019, the awards granted were time-based (cliff vest over four years) and performance-based (vest when performance targets are met, as defined in the stock option grant agreement).

### Service Period Stock Options

The following table includes activity related to outstanding service period stock options during the six months ended June 30, 2019.

# NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

Service Period Stock Options	Shares	Weighted Average Exercise Price
Outstanding at December 31, 2018	1,793,981	\$ 9.84
Granted	21,101	42.29
Exercised	(407,245)	8.41
Forfeited	(1,191)	9.64
Outstanding at June 30, 2019	1,406,646	\$ 10.74

## Performance-Vested Stock Options

The following table includes activity related to outstanding performance-vested stock options during the six months ended June 30, 2019.

The following table includes deliving related to designating performance-vested stock opinions during the six months crided state so, 2015.		Weighte	ed Average Exercise
Performance Based Options	Shares		Price
Outstanding at December 31, 2018	1,275,746	\$	13.62
Granted	9,254	\$	42.29
Exercised	(35,447)		9.98
Forfeited	(39,253)		9.05
Outstanding at June 30, 2019	1,210,300	\$	14.09

As of June 30, 2019, 788,614 performance-vested stock options at a weighted average exercise price of \$11.20 have performance metrics that are probable of achievement. These shares are included in share-based compensation costs for the six months ended June 30, 2019.

## Restricted Stock Units

The following table includes activity related to outstanding restricted stock units during the six months ended June 30, 2019.

		Weighted-Av	erage Grant-Date
Restricted Stock Units	Shares	Fair Va	lue Per Unit
Outstanding at December 31, 2018	271,979	\$	19.07
Granted	75,413		41.71
Issued upon vesting	(122,392)		17.79
Forfeited	(7,819)		17.41
Outstanding at June 30, 2019	217,181	\$	27.71

### Note 9 - Loss Per Share:

Basic net loss per share of common stock is calculated by dividing net loss attributable to common stockholders by the weighted-average number of shares of common stock outstanding for the period. Diluted net loss per share of common stock is computed by giving effect to all potentially dilutive securities.

For the three and six months ended June 30, 2019 and 2018, there were no adjustments between net loss and net loss attributable to common stockholders.

# NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

The potentially dilutive securities are as follows:

	Three Month June 3		Six Months Ended June 30,		
	2019	2018	2019	2018	
Service Period Stock Options	1,519,431	2,070,511	1,602,279	2,048,563	
Restricted Stock Units	230,057	229,513	239,602	198,205	
Performance Stock Options	24,246	39,253	24,246	39,253	
Total	1,773,734	2,339,277	1,866,127	2,286,021	

For the three and six months ended June 30, 2019 and 2018, diluted net loss per share of common stock is the same as basic net loss per share of common stock, due to the fact that potentially dilutive securities would have an antidilutive effect as the Company incurred a net loss during such periods.

### Note 10 - Concentrations:

Concentration of Credit Risk—The Company maintains its cash balances in financial institutions that are insured by the Federal Deposit Insurance Corporation ("FDIC") up to \$250,000 each. At times, such balances may be in excess of the FDIC insurance limit.

Net Sales by Class of Retailer — The following table sets forth net sales by class of retailer:

	Three Mon June		ded		Six Months Ended June 30,				
	2019	019 2018			2019	2018			
Grocery (including Online), Mass and Club	\$ 49,920,536	\$	38,976,097	\$	95,628,433	\$	73,992,511		
Pet Specialty and Natural	10,131,643		8,648,859		19,215,948		16,802,046		
Net Sales	\$ 60,052,179	\$	47,624,956	\$	114,844,381	\$	90,794,557		

## Note 11 - Subsequent Events:

The Company evaluates events that have occurred after the balance sheet date but before the financial statements are issued.

Borrowings under the new credit facilities increased an additional \$5.4 million in July 2019.

Based upon further evaluation, the Company did not identify any additional recognized or unrecognized subsequent events that have required adjustment or disclosure in the financial statements.

## Item 2. Management's Discussion and Analysis of Financial Conditions and Results of Operations

The following discussion summarizes the significant factors affecting our consolidated operating results, financial condition, liquidity and cash flows as of and for the periods presented below. The following discussion and analysis should be read in conjunction with the accompanying unaudited consolidated financial statements and related notes in Item 1 and with the audited consolidated financial statements and the related notes included in our Annual Report on Form 10-K.

In addition to historical information, this discussion and analysis contains forward-looking statements based on current expectations that involve risks, uncertainties and assumptions, such as our plans, objectives, expectations, and intentions set forth under the sections entitled "Forward-Looking Statements" in this report and "Risk Factors" in our Annual Report on Form 10-K. Our actual results and the timing of events may differ materially from those anticipated in these forward-looking statements as a result of various factors, including those set forth in the section entitled "Risk Factors" in our Annual Report on Form 10-K.

We started Freshpet with a single-minded mission to bring the power of real, fresh food to our dogs and cats. We were inspired by the rapidly growing view among pet owners that their dogs and cats are a part of their family, leading them to demand healthier pet food choices. Since inception of the company in 2006, we have created a comprehensive business model to deliver wholesome pet food that pet parents can trust, and in the process we believe we have become one of the fastest growing pet food companies in North America. Our business model is difficult for others to replicate and we see significant opportunity for future growth by leveraging the unique elements of our business, including our brand, our product know-how, our Freshpet Kitchens, our refrigerated distribution, our Freshpet Fridge and our culture.

### Recent Developments

### Freshpet Kitchens Expansion

Due to the continued growth of the Company's fresh pet food sales, the Company has plans to continue expanding its manufacturing capacity. The Company converted three of its four manufacturing lines from five-day production to seven-day production, with plans to convert the fourth line to seven-day production within the next twelve months. The conversion to seven-day production has added more than 140 employees to the Freshpet team. Additionally the Company is in the process of building a 90,000 square-foot addition to our manufacturing location. The \$100 million strategic capital investment is expected to support Freshpet's growth in the United States, Canada and Europe by creating total capacity for approximately \$540 million in net sales from the facility. The facility "Freshpet Kitchens 2.0" will make greater use of automation to improve quality, safety and reduce costs. Production start-up is slated for the second half of 2020. During the first two quarters of 2019, we invested \$11.2 million of capital for the Freshpet Kitchens 2.0 project and have spent \$13.1 million on the project to date.

### **Components of our Operating Results**

### Not Calon

Our net sales are derived from the sale of pet food to our customers, who purchase either directly from us or through third-party distributors. Our products are sold to consumers through a fast-growing network of company-owned branded refrigerators, known as Freshpet Fridges, located in our customers' stores. We continue to roll out Freshpet Fridges at leading retailers across North America and parts of Europe and have installed Freshpet Fridges in over 20,400 retail stores as of June 30, 2019. Our products are sold under the Freshpet brand name with ingredients, packaging and labeling customized by class of retail. Sales are recorded net of discounts, slotting, returns and promotional allowances.

Our net sales growth is driven by the following key factors:

- Increasing sales velocity from the average Freshpet Fridge due to increasing awareness, trial and adoption of Freshpet products and innovation. Our investments in marketing and advertising help to drive awareness and trial at each point of sale.
- Increased penetration of Freshpet Fridge locations in major classes of retail, including Grocery, Mass, Club, Pet Specialty, and Natural. The impact of new Freshpet Fridge installations on our net sales varies by retail

- class and depends on numerous factors including store traffic, refrigerator size, placement within the store, and proximity to other stores that carry our products.
- Consumer trends including growing pet ownership, pet humanization and a focus on health and wellness.

We believe that as a result of the above key factors, we will continue to penetrate the pet food marketplace and increase our share of the pet food category.

### Cross Drofit

Our gross profit is net of costs of goods sold, which include the costs of product manufacturing, product ingredients, packaging materials and inbound freight.

Our gross profit margins are also impacted by the cost of ingredients, packaging materials, and labor and overhead and share based compensation related to direct labor and overhead. We expect to mitigate any adverse movement in input costs through a combination of cost management and price increases.

### Selling, General and Administrative Expenses

Our selling, general and administrative expenses consist of the following:

Outbound freight. We use a third-party logistics provider for outbound freight that ships directly to retailers as well as third-party distributors.

Marketing & advertising. Our marketing and advertising expenses primarily consist of national television media, digital marketing, social media and grass roots marketing to drive brand awareness. These expenses may vary from quarter to quarter depending on the timing of our marketing and advertising campaigns. Our Feed the Growth initiative will focus on growing the business through increased marketing investments.

Freshpet Fridge operating costs. Freshpet Fridge operating costs consist of repair costs and depreciation. The purchase and installation costs for new Freshpet Fridges are capitalized and depreciated over the estimated useful life. All new refrigerators are covered by a manufacturer warranty for three years. We subsequently incur maintenance and freight costs for repairs and refurbishments handled by third-party service providers.

Research & development. Research and development costs consist of expenses to develop and test new products. The costs are expensed as incurred.

Brokerage. We use third-party brokers to assist with monitoring our products at the point-of-sale as well as representing us at headquarters for various customers. These brokers visit our retail customers' store locations to ensure items are appropriately stocked and maintained.

Share based compensation. We account for all share-based compensation payments issued to employees, directors and non-employees using a fair value method. Accordingly, share-based compensation expense is measured based on the estimated fair value of the awards on the grant date. We recognize compensation expense for the portion of the award that is ultimately expected to vest over the period during which the recipient renders the required services to us using the straight-line single option method.

Other general & administrative costs. Other general and administrative costs include non-plant personnel salaries and benefits, as well as corporate general & administrative costs.

### Income Taxes

We had federal net operating loss ("NOL") carry forwards of approximately \$183.3 million as of December 31, 2018, of which, approximately \$175.0 million, generated in 2017 and prior, will expire between 2025 and 2037. The NOL generated in 2018, of approximately \$8.3 million, will have an indefinite carryforward. We may be subject to certain limitations in our annual utilization of NOL carry forwards to off-set future taxable income pursuant to Section 382 of the Internal Revenue Code, which could result in NOLs expiring unused. At December 31, 2018, we had approximately \$147.2 million of state NOLs, which expire between 2018 and 2038. At December 31, 2018, we had a full valuation allowance against our net deferred tax assets as the realization of such assets was not considered more likely than not.

### Consolidated Statements of Operations and Comprehensive Loss

		Three Months E	Ende	d June 30,				Six Months	End	led June 30,		
	2019		2018			2019				2018		
	 Amount	% of Net Sales		Amount	Amount % of Net Sales		Amount	Amount % of Net Sales		Amount	% of Net Sales	
	 	(Dollars in t	hous	sands)				(Dollars i	n th	_		
Net sales	\$ 60,052	100%	\$	47,625	100%	\$	114,844	100%	\$	90,795	100%	
Cost of goods sold	32,726	54		24,747	52		61,603	54		47,789	53	
Gross profit	 27,326	46		22,878	48		53,241	46		43,006	47	
Selling, general and administrative expenses	32,672	54		26,288	55		61,904	54		49,825	55	
Loss from operations	(5,346)	(9)		(3,410)	(7)		(8,663)	(8)		(6,819)	(8)	
Other income/(expenses), net	(21)	(0)		26	0		(3)	(0)		3	0	
Interest expense	(275)	(0)		(98)	(0)		(378)	(0)		(167)	(0)	
Loss before income taxes	(5,642)	(9)		(3,481)	(7)		(9,044)	(8)		(6,983)	(8)	
Income tax expense	 19	0		19	0		39	0		38	0	
Net Loss	\$ (5,661)	(9)%	\$	(3,501)	(7)%	\$	(9,083)	(8)%	\$	(7,021)	(8)%	

### Three Months Ended June 30, 2019 Compared to Three Months Ended June 30, 2018

The following table sets forth net sales by class of retailer:

	Three Months Ended June 30,									
		2019				2018				
		Amount	% of	Store Count		Amount	% of	Store Count		
		7 timount	Net Sales	Otoro oount		7	Net Sales			
				(Dollars in	thous	ands)				
Grocery (including Online), Mass and Club (1)	\$	49,921	83%	15,101	\$	38,976	82%	13,590		
Pet Specialty and Natural (2)		10,131	17	5,313		8,649	18	5,072		
Net Sales		60,052	100%	20,414	\$	47,625	100%	18,662		

- Stores at June 30, 2019 and June 30, 2018 consisted of 10,547 and 9,477 Grocery and 4,554 and 4,113 Mass and Club, respectively Stores at June 30, 2019 and June 30, 2018 consisted of 4,872 and 4,665 Pet Specialty and 441 and 407 Natural, respectively.

Net sales increased \$12.4 million, or 26.1%, to \$60.1 million for the three months ended June 30, 2019 as compared to the same period in the prior year. The \$12.4 million increase in net sales was driven by growth of \$10.9 million in our Grocery (including Online), Mass, and Club refrigerated channel and \$1.5 million in our Pet Specialty and Natural refrigerated channel. Our Freshpet Fridge store locations grew by 9.4% to 20,414 as of June 30, 2019 compared to 18,662 as of June 30, 2018.

Gross profit increased \$4.4 million, or 19.4%, to \$27.3 million for the three months ended June 30, 2019 as compared to the same period in the prior year. The increase in gross profit was driven by higher net sales.

Our gross profit margin of 45.5% for the three months ended June 30, 2019 decreased 250 basis points compared to the same period in the prior year, due to increased scrapping costs and incremental processing costs of 290 basis points, commodity inflation and in-bound freight costs of 120 basis points, partially offset by increases in sales price and shifting selling mix of 110 basis points, and leverage of depreciation expense of 50 hasis points.

Adjusted Gross Profit was \$29.1 million and \$24.5 million in the three months ended June 30, 2019 and 2018, respectively. Adjusted Gross Profit was 48.5% and 51.4% in the three months ended June 30, 2019 and 2018, respectively. Adjusted Gross Profit excludes \$1.6 million of depreciation expense and \$0.2 of share-based compensation expense in the three months ended June 30, 2019 and excludes \$1.5 million of depreciation expense and \$0.1 million of share-based compensation expense in the three months ended June 30, 2018. See "—Non-GAAP Financial Measures" for how we define Adjusted Gross Profit to Gro

### Selling, General and Administrative Expenses

Selling, general and administrative expenses increased \$6.4 million, or 24.3%, to \$32.7 million for the three months ended June 30, 2019 as compared to the same period in the prior year. Key components of the dollar increase include higher media spend of \$4.8 million, higher variable cost due to volume of \$1.1 million, higher depreciation and option expense of \$0.4 million and incremental operating expenses of \$0.3 million, offset by lower selling expense of \$0.2 million. The increased operating expenses were primarily due to new hires, and increased employee incentive and benefit costs.

As a percentage of net sales, selling, general and administrative expenses decreased to 54.4% for the three months ended June 30, 2019 from 55.2% for the three months ended June 30, 2018

Adjusted SG&A slightly increased as a percentage of net sales to 46.4% in the three months ended June 30, 2019 as compared to 46.2% of net sales in the three months ended June 30, 2018. The increase of 20 basis points in adjusted SG&A is a result of 400 basis point gain in SG&A leverage, offset by an increase of 420 basis points related to media ad spend increase. The media spend increase is due to the Company's Feed the Growth Initiative. Since the start of the initiative the Company has gained approximately 620 basis points of leverage on adjusted SG&A. Adjusted SG&A excludes \$2.3 million depreciation and amortization expense, \$1.2 million for non-cash share-based compensation, \$0.9 million launch expense, and \$0.3 secondary offering expense in the three months ended June 30, 2019 and \$2.0 million depreciation and amortization expense, \$1.2 million for share-based compensation expense, \$1.0 million launch expense and \$0.1 million litigation expense in the three months ended June 30, 2018. See "—Non-GAAP Financial Measures" for how we define Adjusted SG&A and a reconciliation of Adjusted SG&A to SG&A, the closest comparable U.S. GAAP measure.

### Loss from Operations

Loss from Operations increased \$1.9 million to \$5.3 million for the three months ended June 30, 2019 as compared to the same period in the prior year as a result of the factors discussed above.

### Interest Expense

Interest expense relating primarily to our credit facilities was \$0.3 million for the three months ended June 30, 2019 and less than \$0.1 million in the three months ended June 30, 2018. In connection with the new credit facility, the Company accelerated the amortization of \$0.1 million of unamortized debt issuance costs related to the existing loan agreement. These costs are included in interest expense in the three and six months ended June 30, 2019.

### Other Income/(Expenses), net

Other income (expenses), net decreased less than \$0.1 million for the three months ended June 30, 2019 compared to the same period in the prior year.

## Net Loss

Net Loss increased \$2.1 million to \$5.7 million for the three months ended June 30, 2019 as compared to loss of \$3.5 million for the same period in the prior year as a result of the factors discussed above.

### Six Months Ended June 30, 2019 Compared To Six Months Ended June 30, 2018

The following table sets forth net sales by class of retailer:

	_	Six Months Ended June 30,								
		2019				2018				
		Amount	% of Net Sales	Store Count	Amou	mount % of Net Sales		Store Count		
	<u> </u>			(Dollars in	thousands)					
Grocery (including Online), Mass and Club (1)	\$	95,628	83%	15,101	\$	73,993	81%	13,590		
Pet Specialty and Natural (2)		19,216	17	5,313		16,802	19	5,072		
Net Sales	\$	114,844	100%	20,414	\$	90,795	100%	18,662		

- Stores at June 30, 2019 and June 30, 2018 consisted of 10,547 and 9,477 Grocery and 4,554 and 4,113 Mass and Club, respectively Stores at June 30, 2019 and June 30, 2018 consisted of 4,872 and 4,665 Pet Specialty and 441 and 407 Natural, respectively.

Net sales increased \$24.0 million, or 26.5%, to \$114.8 million for the six months ended June 30, 2019 as compared to the same period in the prior year. The \$24.0 million increase in net sales was driven by growth of \$21.7 million in our Grocery (including Online), Mass, and Club refrigerated channel and \$2.3 million in our Pet Specialty and Natural refrigerated channel. Our Freshpet Fridge store locations grew by 9.4% to 20,414 as of June 30, 2019 compared to 18,662 as of June 30, 2018.

### **Gross Profit**

Gross profit increased \$10.2 million, or 23.8%, to \$53.2 million for the six months ended June 30, 2019 as compared to the same period in the prior year. The increase in gross profit was primarily driven by higher net sales

Our gross profit margin of 46.4% for the six months ended June 30, 2019 decreased 100 basis points compared to the same period in the prior year due to scrapping and incremental processing costs of 150 basis points, commodity inflation and in-bound freight costs of 120 basis points, unabsorbed labor in advance of a new seven day operation of 30 basis points, partially offset by increases in sales price and shifting selling mix of 120 basis points, and leverage of depreciation expense of 60 basis points, and increased production efficiencies of 20 basis points.

Adjusted Gross Profit was \$56.7 million and \$46.1 million in the six months ended June 30, 2019 and 2018, respectively. Adjusted Gross Profit Margin was 49.4% and 50.8% in the six months ended June 30, 2019 and 2018, respectively. Adjusted Gross Profit excludes \$3.2 million of depreciation expense and \$0.3 million non-cash share-based compensation expense in the six months ended June 30, 2019 and excludes \$3.0 million of depreciation expense and \$0.2 million of non-cash share-based compensation expense in the six months ended June 30, 2018. See "—Non-GAAP Financial Measures" for how we define Adjusted Gross Profit and a reconciliation of Adjusted Gross Profit to Gross Profit, the closest comparable U.S. GAAP measure.

### Selling, General and Administrative Expenses

Selling, general and administrative expenses increased \$12.1 million, or 24.2%, to \$61.9 million for the six months ended June 30, 2019 as compared to the same period in the prior year. Key components of the dollar increase include higher media spend of \$7.9 million, higher variable cost due to volume of \$2.4 million, higher depreciation and option expense of \$0.7 million and incremental operating expenses of \$1.7 million offset by lower selling expense of \$0.6 million. The increased operating expenses were primarily due to new hires, and increased employee incentive and benefit costs.

As a percentage of net sales, selling, general and administrative expenses slightly decreased to 53.9% for the six months ended June 30, 2019 from 54.9% for the six months ended June 30, 2018. Adjusted SG&A slightly decreased as a percentage of net sales to 45.9% in the six months ended June 30, 2019 as compared to 46.1% of net sales in the six months ended June 30, 2018. Adjusted SG&A excludes \$4.5 million of depreciation expense, \$2.3 million of non-cash share-based compensation expense, \$2.0 million launch expense and \$0.3 million of secondary offering expenses in the six months ended June 30, 2019 and excludes \$3.8 million of depreciation expense, \$2.2 million of share-based compensation expense, \$1.7 million of launch expense and \$0.2 million of litigation expense in the six months ended June 30, 2018. See "—Non-GAAP Financial Measures" for how we define Adjusted SG&A and a reconciliation of Adjusted SG&A, the closest comparable U.S. GAAP measure.

### Loss from Operations

Loss from Operations increased \$1.8 million to \$8.7 million for the six months ended June 30, 2019 as compared to the same period in the prior year as a result of the factors discussed above.

### Interest Expense

Interest expense relating primarily to our credit facilities was \$0.4 million \$0.2 million in the six months ended June 30, 2019 and 2018, respectively. In connection with the new credit facility, the Company accelerated the amortization of \$0.1 million of unamortized debt issuance costs related to the existing Loan Agreement. These costs are included in Interest Expense in the three and six months ended June 30, 2019.

### Other Income/(Expenses), net

Other income (expenses), net decreased less than \$0.1 million for the six months ended June 30, 2019 as compared to the same period in the prior year.

### Net Loss

Net Loss increased \$2.1 million to \$9.1 million for the six months ended June 30, 2019 as compared to net loss of \$7.0 million for the same period in the prior year as a result of the factors discussed above.

### Non-GAAP Financial Measures

Freshpet uses the following non-GAAP financial measures in its financial communications. These non-GAAP financial measures should be considered as supplements to the GAAP reported measures, should not be considered replacements for, or superior to, the GAAP measures and may not be comparable to similarly named measures used by other companies.

- · Adjusted Gross Profit
- Adjusted Gross Profit as a percentage of net sales
- Adjusted SG&A expenses
- Adjusted SG&A expenses as a percentage of net sales
- EBITDA
- Adjusted EBITDA
- Adjusted EBITDA as a percentage of net sales

Such financial measures are not financial measures prepared in accordance with U.S. GAAP. We define Adjusted Gross Profit as Gross Profit before non-cash depreciation expense and non-cash share-based compensation. We define Adjusted SG&A as SG&A expenses before depreciation and amortization expense, non-cash share-based compensation, launch expense, fees related to secondary offerings, and litigation expense. EBITDA represents net loss plus interest expense, income tax expense, and depreciation and amortization. Adjusted EBITDA represents EBITDA plus loss on disposal of equipment, non-cash share-based compensation, launch expenses, fees related to secondary offerings, and litigation expense.

We believe that each of these non-GAAP financial measures provide additional metrics to evaluate our operations and, when considered with both our U.S. GAAP results and the reconciliation to the closest comparable U.S. GAAP measures, provide a more complete understanding of our business than could be obtained absent this disclosure. We use the non-

GAAP financial measures, together with U.S. GAAP financial measures, such as net sales, gross profit margins and cash flow from operations, to assess our historical and prospective operating performance, to provide meaningful comparisons of operating performance across periods, to enhance our understanding of our operating performance and to compare our performance to that of our peers and competitors.

Adjusted EBITDA is also an important component of internal budgeting and setting management compensation.

The non-GAAP financial measures are presented here because we believe they are useful to investors in assessing the operating performance of our business without the effect of non-cash items, and other items as detailed below. The non-GAAP financial measures should not be considered in isolation or as alternatives to net loss, income from operations or any other measure of financial performance calculated and prescribed in accordance with U.S. GAAP. Neither EBITDA nor Adjusted EBITDA should be considered a measure of discretionary cash available to us to invest in the growth of our business. Our non-GAAP financial measures may not be comparable to similarly titled measures in other organizations because other organizations may not calculate non-GAAP financial measures in the same manner as we do.

Our presentation of the non-GAAP financial measures should not be construed as an inference that our future results will be unaffected by the expenses that are excluded from that term or by unusual or non-recurring items. We recognize that the non-GAAP financial measures have limitations as analytical financial measures. For example, the non-GAAP financial measures do not reflect:

- our capital expenditures or future requirements for capital expenditures;
- the interest expense, or the cash requirements necessary to service interest expense or principal payments, associated with indebtedness;
- depreciation and amortization, which are non-cash charges, although the assets being depreciated and amortized will likely have to be replaced in the future, nor any cash requirements for such replacements; and
- changes in or cash requirements for our working capital needs.

Additionally, Adjusted EBITDA excludes (i) non-cash share-based compensation expense, which is and will remain a key element of our overall long-term incentive compensation package, and (ii) certain costs essential to our sales growth and strategy, including an allowance for marketing expenses for each new store added to our network and non-capitalizable freight costs associated with Freshpet Fridge replacements. Adjusted EBITDA also excludes certain cash charges resulting from matters we consider not to be indicative of our ongoing operations. Other companies in our industry may calculate the non-GAAP financial measures differently than we do, limiting their usefulness as comparative measures.

The following table provides a reconciliation of EBITDA and Adjusted EBITDA to net loss, the most directly comparable financial measure presented in accordance with U.S. GAAP:

	Three Months Ended June 30,				Six Months Ended June 30,				
	 2019		2018		2019		2018		
	 (Dollars ir	thous	ands)		(Dollars	n thousar	nds)		
Net Loss	\$ (5,661)	\$	(3,501)	\$	(9,083)	\$	(7,021)		
Depreciation and amortization	3,923		3,462		7,643		6,796		
Interest expense	276		98		379		167		
Income tax expense	19		19		38		38		
EBITDA	\$ (1,443)	\$	78	\$	(1,023)	\$	(20)		
(Gain) loss on disposal of equipment	 (7)		48		1		76		
Non-cash share-based compensation	1,430		1,302		2,630		2,394		
Launch expense (a)	948		1,009		2,071		1,662		
Secondary offering expenses (b)	265		_		299		_		
Litigation expense (c)	_		93		_		228		
Adjusted EBITDA	\$ 1,193	\$	2,531	\$	3,978	\$	4,340		
Adjusted EBITDA as a % of Net Sales	 2.0%		5.3%		3.5%		4.8%		

- Represents new store marketing allowance of \$1,000 for each store added to our distribution network, as well as the non-capitalized freight costs associated with Freshpet Fridge replacements. The expense enhances the overall marketing spend to support our growing distribution network.

  Represents fees associated with secondary public offerings of our common stock. (a)
- (b) (c) Represents fees associated with two securities lawsuits.

The following table provides a reconciliation of Adjusted Gross Profit to Gross Profit, the most directly comparable financial measure presented in accordance with U.S. GAAP:

	Three Months Ended June 30,				Six Mo Ji	ed	
	 2019	2018		2019			2018
	 (Dollars i	ands)		(Dollars in thousands)			
Gross Profit	\$ 27,327	\$	22,878	\$	53,242	\$	43,006
Depreciation expense (a)	1,589		1,498		3,155		2,989
Non-cash share-based compensation (b)	186		89		334		153
Adjusted Gross Profit	\$ 29,102	\$	24,465	\$	56,731	\$	46,148
Adjusted Gross Profit as a % of Net Sales	48.5%		51.4%		49.4%		50.8%

- Represents depreciation and amortization expense included in cost of goods sold.
- Represents non-cash share-based compensation expense included in cost of goods sold.

The following table provides a reconciliation of Adjusted SG&A Expenses to SG&A Expenses, the most directly comparable financial measure presented in accordance with U.S. GAAP:

	Three Months Ended			Six Months Ended				
		June 30,			June 30,			
	2019		2018 20		2019 2018		2018	
		(Dollars in thousands)				(Dollars in thousands)		
SG&A expenses	\$	32,672	\$	26,288	\$	61,904	\$	49,825
Depreciation and amortization expense (a)		2,334		1,964	\$	4,486		3,807
Non-cash share-based compensation (b)		1,244		1,213	\$	2,296		2,241
Launch expense (c)		948		1,009	\$	2,071		1,662
Secondary offering expenses (d)		265		_	\$	299		_
Litigation expense (e)		_		93		_		228
Adjusted SG&A Expenses	\$	27,881	\$	22,009	\$	52,752	\$	41,887
Adjusted SG&A Expenses as a % of Net Sales		46.4%		46.2%		45.9%		46.1%

- Represents non-cash depreciation expense included in SG&A.
- (b) (c)
- Represents non-cash share-based compensation expense included in SG&A.

  Represents non-cash share-based compensation expense included in SG&A.

  Represents new store marketing allowance of \$1,000 for each store added to our distribution network, as well as the non-capitalized freight costs associated with Freshpet Fridge replacements. The expense enhances the overall marketing spend to support our growing distribution network.

  Represents fees associated with secondary public offerings of our common stock.

  Represents fees associated with two securities lawsuits.
- (d) (e)

### Liquidity and Capital Resources

Developing our business will require significant capital in the future. To meet our capital needs, we expect to rely on our current and future cash flow from operations and our current available borrowing capacity. Our ability to obtain additional funding will be subject to various factors, including general market conditions, our operating performance, the market's perception of our growth potential, lender sentiment and our ability to incur additional debt in compliance with other contractual restrictions, such as financial covenants under our debt agreements. Additionally, our ability to make payments on, and to refinance, any indebtedness under our Credit Facilities and to fund any necessary expenditures for our growth will depend on our ability to generate cash in the future. If our business does not achieve the levels of profitability or generate the amount of cash that we anticipate or if we expand faster than anticipated, we may need to seek additional debt or equity financing to operate and expand our business. Future third-party financing may not be available on favorable terms or at all.

Our primary cash needs are for ingredients, purchases and operating expenses, marketing expenses and capital expenditures to procure Freshpet Fridges and expand and improve our manufacturing plant to support our net sales growth. Over the next two years we also expect to invest approximately \$100 million in capital expenditures to expand our plant capacity and increase distribution. We believe that cash and cash equivalents, expected cash flow from operations and planned borrowing capacity are adequate to fund debt service requirements, operating lease obligations, capital expenditures and working capital obligations for the foreseeable future. However, our ability to continue to meet these requirements and obligations will depend on, among other things, our ability to achieve anticipated levels of revenue and cash flow from operations and our ability to manage costs and working capital successfully. Additionally, our cash flow generation ability is subject to general economic, financial, competitive, legislative and regulatory factors and other factors that are beyond our control. We cannot assure you that our business will generate cash flow from operations in an amount sufficient to enable us to fund our liquidity needs. Further, our capital requirements may vary materially from those currently planned if, for example, our revenues do not reach expected levels, or we have to incur unforeseen capital expenditures and make investments to maintain our competitive position. If this is the case, we may seek alternative financing, such as selling additional debt or equity securities, and we cannot assure you that we will be able to do so on favorable terms, if at all. Moreover, if we issue new debt securities, the debt holders would have rights senior to common stockholders to make claims on our assets, and the terms of any debt could restrict our operations, including our ability to pay dividends on our common stock. If we issue additional equity or convertible debt securities, existing stockholders may experienc

The following table sets forth, for the periods indicated, our working capital:

	June 30,	December 31,
	2019	2018
	(Dollars in thousa	nds)
Cash and cash equivalents	4,753	7,554
Accounts receivable, net of allowance for doubtful accounts	19,356	12,327
Inventories, net	13,226	9,317
Prepaid expenses	1,287	1,078
Other current assets	650	682
Accounts payable	(15,801)	(9,166)
Accrued expenses	(9,019)	(9,051)
Current operating lease liabilities	(1,068)	
Other current liabilities	(200)	_
Total Working Capital	\$ 13,184 \$	12,741

Working capital consists of current assets net of current liabilities. Working capital increased \$0.4 million to \$13.2 million at June 30, 2019 compared with working capital of \$12.7 million at December 31, 2018. The increase was a result of a decrease in cash, and an increase in accounts payable, offset by an increase in accounts receivable and inventory. The increase in accounts payable was mainly due capex spend of \$7.7 million that was within accounts payable as of June 30, 2019. The increase in account receivable and inventory is a result of the growth the Company has experienced during the year.

We normally carry three to four weeks of finished goods inventory. The average duration of our accounts receivable is approximately three weeks.

As of June 30, 2019, our capital resources consisted primarily of \$4.8 million cash on hand and \$61.5 million available under our credit facilities

We expect to fund our ongoing operations and obligations with cash and cash equivalents, cash flow from operations and available funds under our credit facilities.

The following table sets forth, for the periods indicated, our beginning balance of cash, net cash flows provided by (or used in) operating, investing and financing activities and our ending balance of cash.

	June 30,			
	 2019			
	 (Dollars in thousands)			
Cash at the beginning of period	\$ 7,554	\$	2,184	
Net cash provided by (used in) operating activities	(9,835)		3,140	
Net cash used in investing activities	(22,889)		(8,933)	
Net cash provided by financing activities	29,923		4,743	
Cash at the end of period	\$ 4,753	\$	1,134	

Six Months Ended

### Net Cash Used in Operating Activities

Cash used in operating activities consists primarily of net income adjusted for certain non-cash items (i.e. provision for loss on receivables, loss on disposal of equipment, change in reserve for inventory obsolescence, depreciation and amortization, amortization of deferred financing costs and loan discount, and share-based compensation).

For the six months ended June 30, 2019, net cash used in operating activities of \$9.8 million was primarily attributed to:

• \$11.4 million increase in working capital mainly due to \$11.4 million increase in assets, mainly a result of an increase in inventory, accounts receivable, prepaid expenses and other current assets.

This was partially offset by:

- \$1.4 million of net income, adjusted for reconciling non-cash items, which excludes \$10.4 million primarily related to \$2.6 million of share-based compensation and \$7.6 million of depreciation and amortization.
- \$0.2 million increase in other assets and liabilities.

For the six months ended June 30, 2018, net cash provided by operating activities of \$3.1 million was primarily attributed to:

• \$2.3 million net loss adjusted for reconciling non-cash items, which excludes \$9.4 million of non-cash items primarily relating to \$2.4 million of share-based compensation and \$6.8 million of depreciation and amortization.

This was partially offset by:

• \$0.8 million of change in operating assets and liabilities. The decrease in liabilities of \$0.2 million was mainly a result of result of timing of payments related to compensation that was partially offset by a \$1.0 million decrease in assets, which was the result of a decrease in accounts receivable, offset by an increase in inventory, prepaid expenses and other current assets.

### Net Cash Used in Investing Activities

Net cash used in investing activities of \$22.9 million for the six months ended June 30, 2019, was primarily related to:

- \$12.8 million in capital expenditures for Freshpet Kitchens, of which \$10.9 million relates to the Freshpet Kitchens expansion and \$1.9 million relates to recurring capital expenditures.
- \$10.1 million in capital expenditures relating to investment in fridges and other capital spend.

Net cash used in investing activities of \$8.9 million for the six months ended June 30, 2018, was primarily related to:

\$2.2 million in capital expenditures for Freshpet Kitchens.

\$6.7 million in capital expenditures related to investment in fridges and other capital spend.

### Net Cash Provided by Financing Activities

Net cash provided by financing activities was \$29.9 million for the six months ended June 30, 2019, attributable to:

- \$35.3 million of proceeds from borrowings under our Credit Facilities.
- \$3.8 million cash proceeds from the exercise of stock options.

### Primarily offset by:

- \$7.5 million repayment of borrowings under Credit Facilities.
- \$1.3 million for tax withholdings related to net share settlements of restricted stock units.
- \$0.4 million financing fees paid in connection with borrowings.

Net cash provided by financing activities was \$4.7 million for the six months ended June 30, 2018, attributable to

- \$6.0 million proceeds from borrowings under our Credit Facilities offset by repayments of \$2.0 million.
- \$1.0 million cash proceeds from the exercise of stock options offset by the repurchase of common stock related to employee tax withholdings upon issuance of restricted stock units of \$0.3 million.

## Indebtedness

For a discussion of our material indebtedness, see Note 6 to our consolidated financial statements included in this report.

### **Contractual Obligations**

There were no material changes to our commitments under contractual obligations, as disclosed in our latest Annual Report Form 10-K.

### Off Balance Sheet Arrangements

We have no off balance sheet arrangements or any holdings in variable interest entities.

### Critical Accounting Policies and Significant Estimates

Our management's discussion and analysis of financial condition and results of operations is based on our financial statements, which have been prepared in accordance with accounting principles generally accepted in the United States, or GAAP. The preparation of these financial statements requires us to make estimates and assumptions that affect the reported amounts of assets and liabilities and the disclosure of contingent assets and liabilities at the date of the financial statements, as well as the revenue and expenses incurred during the reported periods. On an ongoing basis, we evaluate our estimates and judgments, including those related to accrued expenses and share-based compensation. We base our estimates on historical experience and on various other factors that we believe are reasonable under the circumstances, the results of which form the basis for making judgments about the carrying value of assets and liabilities that are not apparent from other sources. Changes in estimates are reflected in reported results for the period in which they become known. Actual results may differ from these estimates under different assumptions or conditions.

With the exception of our newly adopted standards around lease accounting, there have been no material changes to our critical accounting policies and estimates as compared to the critical accounting policies and estimates described in our latest Annual Report Form 10-K.

### Recent Accounting Pronouncements

### Recently Adopted Standards:

### Leases:

For a discussion of the standard on leases, adopted in the first guarter of 2019, see Note 2 (Recently Issued Accounting Standards) and Note 7 (Leases) to our consolidated financial statements included in this report.

### Standards Effective in Future Years:

We consider the applicability and impact of all Accounting Standards Updates (ASUs) issued by the Financial Accounting Standards Board (FASB). ASUs not listed herein below were assessed and determined to be either not applicable or are expected to have minimal impact to our consolidated financial statements.

### Item 3. Quantitative and Qualitative Disclosures About Market Risks

### Interest Rate Risk

We are sometimes exposed to market risks from changes in interest rates on debt and changes in commodity prices. Our exposure to interest rate fluctuations is limited to our outstanding indebtedness under our credit agreements, which bears interest at variable rates. As of June 30, 2019, we had \$28.5 million outstanding under our credit facilities. A change in interest rates of 100 basis points would cause a \$0.3 million increase or decrease in annual interest expense.

### Commodity Price Risk

We purchase certain products that are affected by commodity prices and are, therefore, subject to price volatility caused by weather, market conditions and other factors which are not considered predictable or within our control. In many cases, we believe we will be able to address material commodity cost increases by either increasing prices or reducing operating expenses. However, increases in commodity prices, without adjustments to pricing or reduction to operating expenses, could increase our operating costs as a percentage of our net sales.

### Foreign Exchange Rates

Fluctuations in the currencies of countries where the Company operates outside the U.S. may have a significant impact on financial results. The Company is exposed to movements in the British pound sterling and Euro. The Statements of Financial Position of non-U.S. business units are translated into U.S. dollars using period-end exchange rates for assets and liabilities and weighted-average exchange rates for revenues and expenses. The percentage of consolidated revenue for both the three and six months ended June 30, 2019 recognized in Europe was approximately 1%.

The Company may, from time to time enter into forward exchange contracts to reduce the Company's exposure to foreign currency fluctuations of certain assets and liabilities denominated in foreign currencies. Historically, the foreign currency forward contracts have not been designated as hedges and, accordingly, any changes in their fair value are recognized on the Consolidated Statements of Operations and Comprehensive Loss in Other expenses, net, and carried at their fair value in the Consolidated Balance Sheet with gains reported in prepaid expenses and other current assets and losses reported in accrued expenses. As of June 30, 2019, there were no forward contracts outstanding.

### Item 4. Controls and Procedures

### **Evaluation of Disclosure Controls and Procedures**

Our management, with the participation of our Chief Executive Officer and Chief Financial Officer, has evaluated the effectiveness of our disclosure controls and procedures (as defined in Rules 13a-15(e) and 15d-15(e) under the Securities Exchange Act of 1934, as amended (the "Exchange Act")), as of the end of the period covered by this Quarterly Report on Form 10-Q. Based on such evaluation, our Chief Executive Officer and Chief Financial Officer have concluded that as of such date our disclosure controls and procedures were effective.

## Changes in Internal Control

There were no changes in our internal control over financial reporting identified in management's evaluation pursuant to Rules 13a-15(d) or 15d-15(d) of the Exchange Act during the period covered by this Quarterly Report on Form 10-Q that materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

### Limitations on Effectiveness of Controls and Procedures

Our management, including our Chief Executive Officer and Chief Financial Officer, believes that our disclosure controls and procedures and internal control over financial reporting are designed to provide reasonable assurance of achieving their objectives and are effective at the reasonable assurance level. However, our management does not expect that our disclosure controls and procedures or our internal control over financial reporting will prevent all errors and all fraud. A control system, no matter how well conceived and operated, can provide only reasonable, not absolute, assurance that the objectives of the control system are met. Further, the design of a control system must reflect the fact that there are resource constraints, and the benefits of controls must be considered relative to their costs. Because of the inherent limitations in all control systems, no evaluation of controls can provide absolute assurance that all control issues and instances of fraud, if any, have been detected. These inherent limitations include the realities that judgments in decision making can be faulty, and that breakdowns can occur because of a simple error or mistake. Additionally, controls can be circumvented by the individual acts of some persons, by collusion of two or more people or by management override of the controls. The design of any system of controls also is based in part upon certain assumptions about the likelihood of future events, and there can be no assurance that any design will succeed in achieving its stated goals under all potential future conditions. Over time, controls may become inadequate because of changes in conditions, or the degree of compliance with policies or procedures may deteriorate. Because of the inherent limitations in a cost-effective control system, misstatements due to error or fraud may occur and not be detected.

### PART II—OTHER INFORMATION

## Item 1. Legal Proceedings

A securities lawsuit, Curran v. Freshpet, Inc. et al, Docket No. 2:16-cv-02263, was instituted on April 21, 2016 in the United States District Court for the District of New Jersey against us and certain of our current and former executive officers and directors on behalf of certain purchasers of our common stock. We were served with a copy of the complaint in June 2016. The plaintiffs seek to recover damages for investors under the federal securities laws. Plaintiffs have filed a motion for class certification. Freshpet has filed an opposition to the plaintiffs' motion for class certification and requested an evidentiary hearing on the motion. Fact discovery is completed, but expert discovery and certain discovery related disputes are ongoing. The Company believes that the plaintiffs' allegations are without merit and intends to vigorously defend against the claims. Because the Company is in the early stages of litigation, the Company is unable to estimate a reasonably possible range of loss, if any, that may result from this matter.

A securities lawsuit, Meldon v. Freshpet, Inc. et al, Docket No. 2:18-cv-10166, was instituted on June 5, 2018 in the United States District Court for the District of New Jersey against us and certain of our current and

A securities lawsuit, Meldon v. Freshpet, Inc. et al, Docket No. 2:18-cv-10166, was instituted on June 5, 2018 in the United States District Court for the District of New Jersey against us and certain of our current and former executive officers and directors on behalf of certain holders of our common stock. We were served with a copy of the complaint in June 2018. The plaintiffs seek to recover damages for investors under the federal securities laws. On June 21, 2018, we were granted a motion to stay the Meldon case. On April 3, 2019, we were granted an extension of the stay pending (i) the close of expert discovery in the Curran action described in the preceding paragraph or (ii) the dismissal with prejudice of the Curran action. The Company believes that the plaintiffs' allegations are without merit and intends to vigorously defend against the claims. Because the Company is in the early stages of litigation, the Company is unable to estimate a reasonably possible range of loss, if any, that may result from this matter.

In addition, we are currently involved in various claims and legal actions that arise in the ordinary course of our business, including claims resulting from employment related matters. None of these claims or proceedings, most of which are covered by insurance, are expected to have a material adverse effect on our business, financial condition, results of operations or cash flows. However, a significant increase in the number of these claims or an increase in amounts owing under successful claims could materially and adversely affect our business, financial condition, results of operations or cash flows.

### Item 1A. Risk Factors

There have been no material changes from the risk factors previously disclosed in our Annual Report on Form 10-K for the year ended December 31, 2018.

Item 6. Exhibits  Exhibit No.	Description
10.1	Fourth Amended and Restated Loan and Security Agreement dated as of May 15, 2019, by and among Freshpet, Inc. as Borrower, City National Bank, a national banking association, as the arranger and administrative agent, and the lenders party thereto.
31.1	Certification of Chief Executive Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002
31.2	Certification of Chief Financial Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002
32.1	Certifications of Chief Executive Officer and Chief Financial Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002
EX-101.INS  EX-101.SCH EX-101.CAL EX-101.LAB EX-101.PRE EX-101.DEF	XBRL Instance Document – the instance document does not appear in the Interactive Date File because its XBRL tags are embedded within the inline XBRL document  XBRL Schema Documents  XBRL Calculation Linkbase Document  XBRL Labels Linkbase Document  XBRL Presentation Linkbase Document  XBRL Definition Linkbase Document

### SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

Date: August 6, 2019 FRES

FRESHPET, INC.

<u>/s/ William B. Cyr</u> William B. Cyr Chief Executive Officer (Principal Executive Officer)

Is/ Richard Kassar Richard Kassar Chief Financial Officer (Principal Financial and Accounting Officer)

# FOURTH AMENDED AND RESTATED LOAN AND SECURITY AGREEMENT by and among FRESHPET, INC., as Borrower,

## THE LENDERS THAT ARE SIGNATORIES HERETO

as the Lenders, and CITY NATIONAL BANK,

together with its successors and assigns as Lead Arranger and Administrative Agent  $\,$ 

Dated as of May 15, 2019

137020431v13

### **DEFINITIONS AND CONSTRUCTION.2** Definitions.2 1.1 Accounting Terms36 1.2 1.3 Code37 1.4 Construction37 Schedules and Exhibits37 1.5 2. **LOAN AND TERMS OF PAYMENT.38** Revolver Advances.38 Delayed Draw Term Loan.38 2.1 2.2 2.3 Borrowing Procedures.39 2.4 Payments.44 2.5 2.6 2.7 2.8 Promise to Pay; Promissory Notes.49 Interest Rates and Letter of Credit Fee: Rates, Payments, and Calculations50 Cash Management52 Crediting Payments53 Designated Account53 2.9 2.10 Maintenance of Loan Account; Statements of Obligations53 2.11 Fees54 2.12 Letters of Credit54 2.13 LIBOR Option.61 Capital Requirements.64 2.14 2.15 Incremental Commitment66 **CONDITIONS; TERM OF AGREEMENT.68** 3.1 Conditions Precedent to the Initial Extension of Credit68 Conditions Subsequent to the Initial Extension of Credit71 3.3 Conditions Precedent to all Extensions of Credit71 3.4 3.5 Term72 Effect of Termination72 Early Termination by Borrower73 3.6 **CREATION OF SECURITY INTEREST.73** 4.1 Grant of Security Interest73 Negotiable Collateral73 4.2 Collection of Accounts, General Intangibles, and Negotiable Collateral73 <u>Filing of Financing Statements; Commercial Tort Claims; Delivery of Additional Documentation Required.74</u> Power of Attorney75 4.3 4.4 4.5 Right to Inspect75 4.6 Control Agreements75 REPRESENTATIONS AND WARRANTIES 75 5.1 No Encumbrances 76 5.2 [Intentionally Omitted] 76 5.1 5.2 5.3 FP Foods76 5.4 Equipment76

1.

3.

4.

5.

1 137020431v13

5.5	Location of Inventory and Equipment76
5.6	Inventory Records76
5.7	State of Incorporation; Location of Chief Executive Office; Organizational Identification Number; Commercial Tort Claims, 70
5.8	Due Organization and Qualification; Subsidiaries.77
5.9	Due Authorization; No Conflict77
5.10	Litigation78
5.11	No Material Adverse Change78
5.12	Fraudulent Transfer79
5.13	Employee Benefits79
5.14	Environmental Condition79
5.15	Brokerage Fees79
5.16	Intellectual Property79
5.17	Leases79
5.18	Deposit Accounts and Securities Accounts80
5.19	Complete Disclosure80
5.20	Complete Discussified Indebtedness80
5.20 5.21	
5.21 5.22	Margin Stock80 Governmental Regulation80
5.22 5.23	OFAC80
5.24	Patriot Act81
5.25	No EEA Financial Institution.81
AFFIRM	ATIVE COVENANTS.81
6.1	Accounting System81
6.2	Collateral Reporting81
6.3	Financial Statements, Reports, Certificates81
6.4	Guarantor Reports83
6.5	[Intentionally Omitted]83
6.6	Maintenance of Properties83
6.7	Taxes83
6.8	Insurance84
6.9	Location of Inventory and Equipment85
6.10	Compliance with Laws85
6.11	Leases85
6.12	Existence85
6.13	Environmental85
6.14	Disclosure Updates86
6.15	Discussing Optimized Formation of Subsidiaries86
6.16	Canadian and Dutch Subsidiaries87
6.17	
0.17	Acquisition of Real Property Collateral87
	VE COVENANTS88
7.1	Indebtedness88
7.2	Liens89
7.3	Restrictions on Fundamental Changes.89
7.4	Disposal of Assets89
7. <del>4</del> 7.5	Change Name90
1.3	Change ryanieso

2 137020431v13

7.

6.

7.6	
	Nature of Business90
7.7	Prepayments and Amendments90
7.8	Change of Control90
7.9	Consignments90
7.10	Distributions90
7.11	Accounting Methods91
7.12	Investments91
7.13	Transactions with Affiliates91
7.14	Suspension91
7.15	Sale-Leaseback Transactions91
7.16	Use of Proceeds91
7.17	Restriction on FP Foods92
7.18	<u>Financial Covenants.92</u>
EVENTS	OF DEFAULT94
THE LEN	IDER GROUP'S RIGHTS AND REMEDIES96
9.1	Rights and Remedies96
9.2	Remedies Cumulative99
TAXES A	ND EXPENSES99
WAIVER	S; <u>INDEMNIFICATION99</u>
11.1	Demand; Protest; etc99
	The Lender Group's Liability for Borrower Collateral99
11.1	
11.1 11.2	The Lender Group's Liability for Borrower Collateral99 Indemnification100
11.1 11.2 11.3 NOTICES	The Lender Group's Liability for Borrower Collateral99 Indemnification100
11.1 11.2 11.3 NOTICES	The Lender Group's Liability for Borrower Collateral99 Indemnification100  S100  OF LAW AND VENUE; JURY TRIAL WAIVER; JUDICIAL REFERENCE PROVISION10
11.1 11.2 11.3 NOTICES	The Lender Group's Liability for Borrower Collateral99 Indemnification100  S100  OF LAW AND VENUE; JURY TRIAL WAIVER; JUDICIAL REFERENCE PROVISION10: MENTS AND PARTICIPATIONS; SUCCESSORS105
11.1 11.2 11.3 NOTICES CHOICE	The Lender Group's Liability for Borrower Collateral99 Indemnification100  S100  OF LAW AND VENUE; JURY TRIAL WAIVER; JUDICIAL REFERENCE PROVISION10
11.1 11.2 11.3 NOTICES CHOICE ASSIGNM 14.1 14.2	The Lender Group's Liability for Borrower Collateral99 Indemnification100  5100  OF LAW AND VENUE; JURY TRIAL WAIVER; JUDICIAL REFERENCE PROVISION103  MENTS AND PARTICIPATIONS; SUCCESSORS105  Assignments and Participations.105 Successors108
11.1 11.2 11.3 NOTICES CHOICE ASSIGNM 14.1 14.2	The Lender Group's Liability for Borrower Collateral99 Indemnification100  5100  OF LAW AND VENUE; JURY TRIAL WAIVER; JUDICIAL REFERENCE PROVISION103  MENTS AND PARTICIPATIONS; SUCCESSORS105     Assignments and Participations.105     Successors108  MENTS; WAIVERS108
11.1 11.2 11.3 NOTICES CHOICE - ASSIGNM 14.1 14.2 AMENDM 15.1	The Lender Group's Liability for Borrower Collateral99 Indemnification100  5100  OF LAW AND VENUE; JURY TRIAL WAIVER; JUDICIAL REFERENCE PROVISION10: MENTS AND PARTICIPATIONS; SUCCESSORS105     Assignments and Participations.105     Successors108  MENTS; WAIVERS108     Amendments and Waivers108
11.1 11.2 11.3 NOTICES CHOICE ASSIGNM 14.1 14.2 AMENDM	The Lender Group's Liability for Borrower Collateral99 Indemnification100  5100  OF LAW AND VENUE; JURY TRIAL WAIVER; JUDICIAL REFERENCE PROVISION103  MENTS AND PARTICIPATIONS; SUCCESSORS105     Assignments and Participations.105     Successors108  MENTS; WAIVERS108
11.1 11.2 11.3 NOTICES CHOICE ASSIGNM 14.1 14.2 AMENDM 15.1 15.2 15.3	The Lender Group's Liability for Borrower Collateral99 Indemnification100  5100  OF LAW AND VENUE; JURY TRIAL WAIVER; JUDICIAL REFERENCE PROVISION10:  MENTS AND PARTICIPATIONS; SUCCESSORS105 Assignments and Participations.105 Successors108  MENTS; WAIVERS108 Amendments and Waivers108 Replacement of Holdout Lender109
11.1 11.2 11.3 NOTICES CHOICE ASSIGNM 14.1 14.2 AMENDM 15.1 15.2 15.3	The Lender Group's Liability for Borrower Collateral99 Indemnification100  S100  OF LAW AND VENUE; JURY TRIAL WAIVER; JUDICIAL REFERENCE PROVISION10:  MENTS AND PARTICIPATIONS; SUCCESSORS105    Assignments and Participations.105    Successors108  MENTS; WAIVERS108    Amendments and Waivers108    Replacement of Holdout Lender109    No Waivers; Cumulative Remedies110  THE LENDER GROUP110    Appointment and Authorization of Agent110
11.1 11.2 11.3 NOTICES CHOICE ASSIGNN 14.1 14.2 AMENDN 15.1 15.2 15.3 AGENT;	The Lender Group's Liability for Borrower Collateral99 Indemnification100  5100  OF LAW AND VENUE; JURY TRIAL WAIVER; JUDICIAL REFERENCE PROVISION102  MENTS AND PARTICIPATIONS; SUCCESSORS105 Assignments and Participations.105 Successors108  MENTS; WAIVERS108 Amendments and Waivers108 Replacement of Holdout Lender109 No Waivers; Cumulative Remedies110  THE LENDER GROUP110 Appointment and Authorization of Agent110 Delegation of Duties111
11.1 11.2 11.3 NOTICES CHOICE ASSIGNM 14.1 14.2 AMENDM 15.1 15.2 15.3 AGENT, 16.1	The Lender Group's Liability for Borrower Collateral99 Indemnification100  S100  OF LAW AND VENUE; JURY TRIAL WAIVER; JUDICIAL REFERENCE PROVISION10:  MENTS AND PARTICIPATIONS; SUCCESSORS105    Assignments and Participations.105    Successors108  MENTS; WAIVERS108    Amendments and Waivers108    Replacement of Holdout Lender109    No Waivers; Cumulative Remedies110  THE LENDER GROUP110    Appointment and Authorization of Agent110
11.1 11.2 11.3 NOTICES CHOICE ASSIGNM 14.1 14.2 AMENDM 15.1 15.2 15.3 AGENT; 16.1 16.2	The Lender Group's Liability for Borrower Collateral99 Indemnification100  5100  OF LAW AND VENUE; JURY TRIAL WAIVER; JUDICIAL REFERENCE PROVISION102  MENTS AND PARTICIPATIONS; SUCCESSORS105 Assignments and Participations.105 Successors108  MENTS; WAIVERS108 Amendments and Waivers108 Replacement of Holdout Lender109 No Waivers; Cumulative Remedies110  THE LENDER GROUP110 Appointment and Authorization of Agent110 Delegation of Duties111
11.1 11.2 11.3 NOTICES CHOICE ASSIGNM 14.1 14.2 AMENDM 15.1 15.2 15.3 AGENT; 16.1 16.2 16.3	The Lender Group's Liability for Borrower Collateral99 Indemnification100  5100  OF LAW AND VENUE; JURY TRIAL WAIVER; JUDICIAL REFERENCE PROVISION103  MENTS AND PARTICIPATIONS; SUCCESSORS105  Assignments and Participations,105 Successors108  MENTS; WAIVERS108 Amendments and Waivers108 Replacement of Holdout Lender109 No Waivers; Cumulative Remedies110  THE LENDER GROUP110 Appointment and Authorization of Agent110 Delegation of Duties111 Liability of Agent111
11.1 11.2 11.3 NOTICES CHOICE ASSIGNN 14.1 14.2 AMENDN 15.1 15.2 15.3 AGENT; 16.1 16.2 16.3 16.4	The Lender Group's Liability for Borrower Collateral99 Indemnification100  S100  OF LAW AND VENUE; JURY TRIAL WAIVER; JUDICIAL REFERENCE PROVISION10: MENTS AND PARTICIPATIONS; SUCCESSORS105    Assignments and Participations.105    Successors108  MENTS; WAIVERS108    Amendments and Waivers108    Replacement of Holdout Lender109    No Waivers; Cumulative Remedies110  THE LENDER GROUP110    Appointment and Authorization of Agent110    Delegation of Duties111    Liability of Agent111    Reliance by Agent111

3 137020431v13

8. 9.

10. 11.

12. 13. 14.

15.

16.

16.8	Agent in Individual Capacity113
16.9	Successor Agent114
16.10	Lender in Individual Capacity114
16.11	Withholding Taxes.114
16.12	Collateral Matters.117
16.13	Restrictions on Actions by Lenders; Sharing of Payments118
16.14	Agency for Perfection119
16.15	Payments by Agent to the Lenders119
16.16	Concerning the Collateral and Related Loan Documents119
16.17	Field Examinations and Examination Reports; Confidentiality; Disclaimers
	by Lenders; Other Reports and Information 119
16.18	Several Obligations; No Liability120
16.19	Bank Product Providers121
16.20	Lead Arranger121
16.21	Legal Representation of Agent121
	N 0.1 11 11 11 11 11 11 11 11 11 11 11 11 1
	PROVISIONS121
17.1	Effectiveness121
17.2	Section Headings121
17.3	Interpretation121
17.4	Severability of Provisions121
17.5	Counterparts; Electronic Execution122
17.6	Revival and Reinstatement of Obligations; Certain Waivers.122
17.7	Confidentiality123
17.8	Integration123
17.9	Amendment and Restatement123
17.10	Keepwell124
17.11	Acknowledgement and Consent to Bail-In124
17.12	Patriot Act Notice125
17.13	No Advisory or Fiduciary Responsibility125

## Exhibits

17.

Exhibit A-1

Form of Assignment and Acceptance Exhibit B
Form of U.S. Tax Compliance Certificate Exhibit C
Form of Compliance Certificate

Exhibit L-1

Form of LIBOR Notice

Form of Notice of Borrowing

4 137020431v13

# Schedules

Agent's Account Commitments Schedule A-1 Schedule C-1 Schedule D-1 Designated Account Designated Account
Permitted Liens
Permitted Holders Schedule RReal Property Collateral Schedule
Cash Management Banks
Locations of Inventory and Equipment Schedule Schedule P-1 Schedule P-2 1 2.7(a) Schedule 5.5 5.7(a) Schedule 5.7(b) Schedule 5.7(c) States of Organization Chief Executive Offices Organizational Identification Numbers Schedule Commercial Tort Claims
Capitalization of Borrower's Subsidiaries
Subscriptions, Options, Warrants or Calls Relating to Shares of Borrower's Subsidiaries' Capital Stock
Litigation
Environmental Matters Schedule
Intellectual Property
Deposit Accounts and Securities Accounts Schedule
Permitted Indebtedness 5.7(d) Commercial Tort Claims Schedule 5.8(c) Schedule 5.8(d) Schedule 5.10 Schedule 5.14 5.16 Schedule 5.18 5.20

FOURTH AMENDED AND RESTATED LOAN AND SECURITY AGREEMENT
THIS FOURTH AMENDED AND RESTATED LOAN AND SECURITY
AGREEMENT (this "Agreement"), is entered into as of May 15, 2019, by and among the lenders identified on the signature pages hereof (such lenders, together with their respective successors and permitted assigns, are referred to hereinafter each individually as a "Lender" and collectively as the "Lenders"), CITY NATIONAL BANK, a national banking association ("CNB"), as the lead arranger and administrative agent for the Lenders (in such capacity as administrative agent, together with its successors and assigns in such capacity, "Agent"), and FRESHPET, INC., a Delaware corporation ("Borrower").

WHEREAS, Borrower and CNB, as the Agent and a Lender, are parties to that certain Third Amended and Restated Loan and Security Agreement, dated as of September 21, 2017 (as amended, restated, supplemented, or otherwise modified from time to time prior to the date hereof, the "Existing Loan Agreement") pursuant to which CNB, as sole initial Lender, made available to Borrower a revolving loan and letter of credit facility in the aggregate maximum principal amount of \$30,000,000, with an aggregate outstanding principal amount of \$15,000,000 as of the date hereof;

WHEREAS, FP Foods Realty PA, LLC, a Delaware limited liability company and a direct subsidiary of Borrower ("FP Foods"), guaranteed all of the Obligations (as defined in the Existing Loan Agreement), pursuant to that certain General Continuing Guaranty, dated as of October 5, 2016, delivered by FP Foods in favor of the Agent, and all of the Guarantied Obligations (as defined in such General Continuing Guaranty) are secured pursuant to (i) that certain Guarantor Security Agreement, dated as of October 5, 2016, between FP Foods and the Agent and (ii) the Mortgage relating to Real Property at address 146 N. Commerce Way, Bethlehem, PA 18017 ("Property A");

WHEREAS, Borrower has requested that the Existing Loan Agreement be amended and restated to (i) increase the revolving loan facility to \$35,000,000, (ii) add a delayed draw term loan facility in the aggregate maximum principal amount of \$55,000,000, (iii) permit FP Foods to be dissolved after all of its assets (including, without limitation, Property A) are transferred to Borrower, subject to Agent's continuing first priority Lien thereon, and (iii) make other changes to the Existing Loan Agreement as set forth herein; and

WHEREAS, the Agent and the Lenders have agreed to such requests, on the terms and subject to the conditions set forth herein.

NOW, THEREFORE, in consideration of the premises herein contained and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree to amend and restate the Existing Loan Agreement as follows, without constituting a novation:

137020431v13

### 1. DEFINITIONS AND CONSTRUCTION.

.1 Definitions

As used in this Agreement, the following terms shall have the following definitions:

"ACH Transactions" means any cash management or related services (including the Automated Clearing House processing of electronic fund transfers through the direct Federal Reserve Fedline system) provided by a Bank Product Provider for the account of Borrower or its Subsidiaries.

"Account" means an account (as that term is defined in the Code).

"Account Debtor" means any Person who is obligated on an Account, chattel paper, or a General Intangible.

"Accounting Changes" means changes in accounting principles required by the promulgation of any rule, regulation, pronouncement or opinion by the Financial Accounting Standards Board of the American Institute of Certified Public Accountants (or successor thereto or any agency with similar functions).

"Acquisition" means (a) the purchase or other acquisition by a Person or its Subsidiaries of all or substantially all of the assets of (or any division or business line of) any other Person, or (b) the purchase or other acquisition (whether by means of a merger, consolidation, or otherwise) by a Person or its Subsidiaries of all of the Stock of any other Person.

"Additional Documents" has the meaning set forth in Section 4.4(c).

"Adjusted EBITDA" means, with respect to any fiscal period, Borrower's and its Subsidiaries' (i) consolidated gross revenue; *minus* (ii) the cost of goods sold; *minus* (iii) selling, general, and administrative expenses; *minus* (iv) all non-cash gains; *plus* (v) (a) any costs associated with the installation of refrigerators in new locations and marketing expenses incurred in connection with the introduction of new locations (all such costs and expenses, collectively, not to exceed \$2,000 with respect to any location) and (b) non-recurring fees, charges and other one-time start-up costs in connection with any real property (including all buildings, fixtures, integrated equipment or other improvements located thereon) now, hereafter or heretofore owned, leased, operated or used by Borrower for the purpose of developing, manufacturing and marketing pet food; <u>provided</u> that the aggregate amount permitted to be added back pursuant to this clause (v) for the fiscal year of Borrower ending (a) December 31, 2019, shall not exceed \$6,000,000; *plus* (vi) all non-cash expenses or losses, including any depreciation and amortization expense; *plus* (vii) any costs or expenses incurred pursuant to any stock option plan or any other management or employee benefit plan, agreement or any stock subscription or stockholders agreement; *plus* (viii) other non-recurring fees, charges and other expenses that have been approved by the Required Lenders.

"Advances" has the meaning set forth in Section 2.1.

- 2 -137020431v13 "Affiliate" means, as applied to any Person, any other Person who, directly or indirectly through one or more intermediaries, controls, is controlled by, or is under common control with, such Person. For purposes of this definition, "control" means the possession, directly or indirectly through one or more intermediaries, of the power to direct the management and policies of a Person, whether through the ownership of Stock, by contract, or otherwise; provided, however, that, for purposes of Section 7.13 hereof: (a) any Person which owns directly or indirectly 10% or more of the Stock having ordinary voting power for the election of directors or other members of the governing body of a Person or 10% or more of the partnership or other ownership interests of a Person (other than as a limited partner of such Person) shall be deemed an Affiliate of such Person, and (c) each partnership or joint venture in which a Person is a partner or member of a joint venture shall be deemed an Affiliate of such Person.

"Agent" has the meaning set forth in the preamble to this Agreement. "Agent Advances" has the meaning set forth in Section 2.3(f)(i).

"Agent-Related Persons" means Agent, together with its Affiliates, officers, directors, employees, attorneys, and agents.

"Agent's Account" means the Deposit Account of Agent identified on

### Schedule A-1.

"Agent's Liens" means the Liens granted by Borrower or its Subsidiaries to Agent under this Agreement or the other Loan Documents.

"Agreement" has the meaning set forth in the preamble hereto. "Applicable Cross-Default Amount" means \$500,000.

"Applicable Margin" means, as of any date of determination and with respect to Base Rate Loans or LIBOR Rate Loans, as applicable, the applicable margin set forth in the following table that corresponds to the most recent Leverage Ratio calculation delivered to Agent pursuant to Section 6.3 of the Agreement (the "Leverage Ratio Calculation"); provided, that for the period from the Restatement Effective Date through and including the first day of the month following the date on which Agent receives the Leverage Ratio Calculation in respect of the testing period ending June 30, 2019, the Applicable Margin shall be set at the margin in the row styled "Level I"; provided further, that any time an Event of Default has occurred and is continuing, the Applicable Margin shall be set at the margin in the row styled "Level I":

Level     Leverage Ratio     Applicable Margin Relative to Base Rate Loans (the "Base Rate Margin")     Applicable Relative to Relative to LIBOR Rate Margin")	<u>to</u> <u>Loans</u> (the

- 3 -137020431v13

I	Greater than 2.50:1.00	1.00 percentage points	2.00 percentage points
II	Greater than 1.25:1.00 and less than or equal to 2.50:1.00	0.75 percentage points	1.75 percentage points
III	Less than or equal to 1.25:1.00	0.50 percentage points	1.50 percentage points

Except as set forth in the foregoing proviso, the Applicable Margin shall be based upon the most recent Leverage Ratio Calculation, which will be calculated as of the end of each fiscal quarter. Except as set forth in the foregoing proviso, the Applicable Margin shall be re-determined quarterly on the first day of the month following the date of delivery to Agent of the certification of the Leverage Ratio pursuant to Section 6.3 (each such date, a "Redetermination Date"); provided, however, that if Borrower fails to provide such certification when such certification is due, the Applicable Margin shall be set at the margin in the row styled "Level I" as of the first day of the month following the date on which the certification was required to be delivered until the date on which such certification is delivered (on which date (but not retroactively), without constituting a waiver of any Default or Event of Default occasioned by the failure to timely deliver such certification, the Applicable Margin shall be set at the margin based upon the calculations disclosed by such certification. In the event that the information regarding the Leverage Ratio contained in any certificate delivered pursuant to Section 6.3 of the Agreement is shown to be inaccurate, and such inaccuracy, if corrected, would have led to the application of a higher Applicable Margin for any period (an "Applicable Period") than the Applicable Margin shall be determined as if the correct Applicable Margin (as set forth in the table above) were applicable for such Applicable Period, which payment shall be promptly deliver to Agent full payment in respect of the accrued additional interest as a result of such increased Applicable Margin for such Applicable Period, which payment shall be promptly applied by Agent to the affected Obligations.

"Application Event" means the occurrence of (a) a failure by Borrower to repay all of the Revolver Obligations in full on the Revolver Maturity Date, (b) a failure by Borrower to repay all of the Delayed Draw Term Loan Obligations in full on the Delayed Draw Term Loan Maturity Date, or (c) an Event of Default and the election by Agent or the Required Lenders to require that payments and proceeds of Collateral be applied pursuant to Section 2.4(b) of this Agreement.

"Approved Fund" means, with respect to any Lender, any Person (other than a natural Person) that (a) (i) is or will be engaged in making, purchasing, holding or otherwise investing in commercial loans and similar extensions of credit in the ordinary course of business or (ii) temporarily warehouses loans for any Lender or any Person described in <a href="clause(i)">clause(i)</a> above and (b) is advised or managed by (i) such Lender, (ii) any Affiliate of such Lender or (iii) any

Person (other than a natural Person) or any Affiliate of any Person (other than a natural Person) that administers or manages such Lender.

"Assignee" has the meaning set forth in Section 14.1(a).

"Assignment and Acceptance" means an Assignment and Acceptance Agreement substantially in the form of Exhibit A-1.

"Authorized Person" means any officer or employee of Borrower.

"Availability." means, as of any date of determination, the amount that Borrower is entitled to borrow as Advances or Delayed Draw Term Loan, as applicable, hereunder (after giving effect to all then outstanding Obligations (other than Bank Product Obligations) and all sublimits and reserves then applicable hereunder).

"Bail-In Action" means the exercise of any Write-down and Conversion Powers by the applicable EEA Resolution Authority in respect of any liability of an EEA Financial Institution.

"Bail-In Legislation" means, with respect to any EEA Member Country implementing Article 55 of Directive 2014/59/EU of the European Parliament and of the Council of the European Union, the implementing law for such EEA Member Country from time to time which is described in the EU Bail-In Legislation Schedule.

"Bank Product" means any financial accommodation extended to Borrower or its Subsidiaries by a Bank Product Provider (other than pursuant to this Agreement) including:

(a) credit cards, (b) credit card processing services, (c) debit cards, (d) purchase cards, (e) ACH Transactions, (f) cash management, including controlled disbursement, accounts or services, or (g) transactions under Hedge Agreements.

"Bank Product Agreements" means those agreements entered into from time to time by Borrower or its Subsidiaries with a Bank Product Provider in connection with the obtaining of any of the Bank Products.

"Bank Product Collateralization" means providing cash collateral (pursuant to documentation reasonably satisfactory to Agent) to be held by Agent for the benefit of the Bank Product Providers in an amount determined by Agent as sufficient to satisfy the reasonably estimated credit exposure with respect to the then existing Bank Product Obligations.

"Bank Product Obligations" means all obligations, liabilities, contingent reimbursement obligations, fees, and expenses owing by Borrower or its Subsidiaries to any Bank Product Provider pursuant to or evidenced by the Bank Product Agreements and irrespective of whether for the payment of money, whether direct or indirect, absolute or contingent, due or to become due, now existing or hereafter arising, and including all such amounts that Borrower or its Subsidiaries are obligated to reimburse to Agent or any member of the Lender Group as a result of Agent or such member of the Lender Group purchasing participations from, or executing indemnities or reimbursement obligations to, a Bank Product

Provider with respect to the Bank Products provided by such Bank Product Provider to Borrower or its Subsidiaries.

"Bank Product Provider" means any Person that at the time it enters into a Bank Product Agreement, is a Lender or an Affiliate of a Lender, in its capacity as a party to such Bank Product Agreement.

"Bankruptcy Code" means title 11 of the United States Code, as in effect from

time to time.

"Base LIBOR Rate" means a rate per annum determined by Agent (which determination shall be conclusive and binding absent manifest error), rounded upwards, if necessary, to the nearest 1/16 of 1.00%, equal to the rate of interest which is identified and normally published by Bloomberg on the applicable page as the offered rate for loans in United States dollars for the applicable Interest Period; provided that, if the Base LIBOR Rate shall be less than zero, such rate shall be deemed zero for purposes of this Agreement. The rate is set by the ICE Benchmark Administration or any successor determining administrator as of 11:00 a.m. (London time) on the second Business Day preceding the first day of each Interest Period. If Bloomberg (or another nationallyrecognized rate reporting source acceptable to the Agent) no longer reports such rate or the Agent determines in good faith that the rate so reported no longer accurately reflects the rate available to the Agent in the London Interbank Market or if such index no longer exists or if the applicable Bloomberg page no longer exists or accurately reflects the rate available to the Agent in the London Interbank Market, "Base LIBOR Rate" shall be determined by reference to such other comparable publicly available service for displaying eurodollar rates as may be selected by the Agent in its discretion.

"Base Rate" means, for any day, a fluctuating rate per annum equal to the greatest of (a) the Federal Funds Rate plus 0.50%, (b) the LIBOR Rate (which rate shall be calculated based upon an Interest Period of 1 month and shall be determined on a daily basis), plus 1 percentage point, and (c) the Prime Rate. The Base Rate is a reference rate and does not necessarily represent the lowest or best rate actually charged to any customer. The Agent and the Lenders may make commercial loans or other loans at rates of interest at, above or below the Base Rate. If, for any reason, the Agent shall have determined (which determination shall be conclusive absent manifest error) that it is unable to ascertain the Federal Funds Rate for any reason, including the inability or failure of the Agent to obtain sufficient quotations in accordance with the terms hereof, the Base Rate shall be determined without regard to clause (a) of the first sentence of this definition until the circumstances giving rise to such inability no longer exist. Any change in the Base Rate due to a change in the Federal Funds Rate or the "Prime Rate", as the case may be, shall be effective on the effective date of such change in the Federal Funds Rate or the "Prime Rate", as applicable.

"Base Rate Loan" means the portion of an Advance, a Delayed Draw Term Loan or any Incremental Term Loan that bears interest at a rate determined by reference to the Base Rate.

"Beneficial Ownership Certification" means a certification regarding beneficial ownership required by the Beneficial Ownership Regulation.

"Beneficial Ownership Regulation" means 31 C.F.R. § 1010.230.

"Benefit Plan" means a "defined benefit plan" (as defined in Section 3(35) of ERISA) for which Borrower or any Subsidiary or ERISA Affiliate of Borrower has been an "employer" (as defined in Section 3(5) of ERISA) within the past six years.

"Board of Directors" means the board of directors (or comparable managers) of Borrower or any committee thereof duly authorized to act on behalf of the board of directors (or comparable managers).

"Books" means all of Borrower's and its Subsidiaries' now owned or hereafter acquired books and records (including all of their Records indicating, summarizing, or evidencing their assets (including the Collateral) or liabilities, all of Borrower's and its Subsidiaries' Records relating to their business operations or financial condition, and all of their goods or General Intangibles related to such information).

"Borrower" has the meaning set forth in the preamble to this Agreement. "Borrower Collateral" means all of Borrower's now owned or hereafter acquired

right, title, and interest in and to each of the following:

(b)

(a)	all of its Accounts,

- (c) all of its commercial tort claims described on Schedule 5.7(d),
- (d) all of its Deposit Accounts,

all of its Books,

- (e) all of its Equipment,
- (f) all of its General Intangibles,
- (g) all of its Inventory,
- (h) all of its Investment Property (including all of its securities and Securities

Accounts),

(i) all of its Negotiable Collateral,

(j) all of its Supporting Obligations,

(k) money or other assets of Borrower that now or hereafter come into the

possession, custody, or control of any member of the Lender Group, and

(l) the proceeds and products, whether tangible or intangible, of any of the foregoing, including proceeds of insurance covering any or all of the foregoing, and any and all Accounts, Books, Deposit Accounts, Equipment, General Intangibles, Inventory, Investment Property, Negotiable Collateral, Real Property, Supporting Obligations, money, or other tangible

- 7 -137020431v13 or intangible property resulting from the sale, lease, license, exchange, collection, or other disposition of any of the foregoing, or any portion thereof or interest therein, and the proceeds thereof; <u>provided</u> that "Borrower Collateral" shall not include the Excluded Property; <u>provided</u>, <u>further</u>, that if and when any property shall cease to be Excluded Property, "Borrower Collateral" shall include such property and a Lien on and security interest in such property shall be deemed granted therein.

"Borrowing" means a borrowing hereunder consisting of Advances (or term loans, in the case of the Delayed Draw Term Loan) made on the same day by the Lenders (or Agent on behalf thereof), or by Agent in the case of an Agent Advance.

"Business Day." means any day that is not a Saturday, Sunday, or other day on which banks are authorized or required to close in the state of California, except that, if a determination of a Business Day shall relate to a LIBOR Rate Loan, the term "Business Day" also shall exclude any day on which banks are closed for dealings in Dollar deposits in the London interbank market.

"Canadian Subsidiary" means Professor Connors Canada, Inc., a company organized under the laws of the province of Ontario.

"Canadian Subsidiary Dissolution" means any of the consolidation, combination or merger of the Canadian Subsidiary with and into Borrower or the liquidation, wind up, dissolution or other similar transaction reasonably approved by the Agent, of the Canadian Subsidiary (in each case, including any similar transaction under local law governing of such Subsidiary); provided, that if the aggregate Net Cash Proceeds received from all such transactions are in an amount greater than \$1,000,000, the remaining assets of the Canadian Subsidiary (if any) and any proceeds of any of the foregoing shall be transferred to Borrower.

"Capital Expenditures" means, with respect to any Person for any period, the aggregate of all expenditures by such Person and its Subsidiaries during such period that are capital expenditures as determined in accordance with GAAP, whether such expenditures are paid in cash or financed excluding (a) interest capitalized during such period, (b) any expenditure described above to the extent such expenditure is part of the aggregate amounts payable as consideration for any Permitted Acquisition consummated during or prior to such period, (c) to the extent permitted by this Agreement, a reinvestment of the Net Cash Proceeds of any Disposition by Borrower or any of its Subsidiaries in accordance with Section 2.4(e)(ii).

(d) expenditures of proceeds of insurance settlements, condemnation awards and other settlements in respect of lost, destroyed, damaged or condemned assets, equipment or other property to the extent such expenditures are made to replace or repair such lost, destroyed, damaged or condemned assets, equipment or other property or otherwise to acquire, maintain, develop, construct, improve, upgrade or repair assets or properties useful in the business of Borrower and the Subsidiaries, (e) expenditures that are accounted for as capital expenditures of such person and that actually are paid for by, or for which Borrower or any Subsidiary receives reimbursement in cash from, a third party and for which none of Borrower or any Subsidiary has provided or is required to provide or incur, directly or indirectly, any consideration or obligation to such third party or any other person (whether before, during or after such period), and (f) other

expenditures that Agent determines in its discretion to exclude from this definition of "Capital Expenditures".

"Capitalized Lease Obligation" means that portion of the obligations under a Capital Lease that is required to be capitalized in accordance with GAAP.

"Capital Lease" means a lease that is required to be capitalized for financial reporting purposes in accordance with GAAP. Notwithstanding anything else set forth herein, any lease that was or would have been treated as an operating lease under GAAP as in effect on the Restatement Effective Date that would be treated as a capital lease solely as a result of a change in GAAP after the Restatement Effective Date shall always be treated as an operating lease for all purposes and at all times under this Agreement.

"Cash Equivalents" means (a) marketable direct obligations issued by, or unconditionally guaranteed by, the United States or issued by any agency thereof and backed by the full faith and credit of the United States, in each case maturing within 1 year from the date of acquisition thereof, (b) marketable direct obligations issued by any state of the United States or any political subdivision of any such state or any public instrumentality thereof maturing within 1 year from the date of acquisition thereof and, at the time of acquisition, having one of the two highest ratings obtainable from either Standard & Poor's Rating Group ("S&P") or Moody's Investor Service, Inc. ("Moody's"), (c) commercial paper maturing no more than 270 days from the date of creation thereof and, at the time of acquisition, having a rating of at least A-1 from S&P or at least P-1 from Moody's, (d) certificates of deposit or bankers' acceptances maturing within 1 year from the date of acquisition thereof issued by any bank organized under the laws of the United States or any state thereof having at the date of acquisition thereof combined capital and surplus of not less than \$250,000,000, (e) Deposit Accounts maintained with (i) any bank that satisfies the criteria described in clause (d) above, or (ii) any other bank organized under the laws of the United States or any state thereof so long as the amount maintained with any such other bank is less than or equal to \$100,000 and is insured by the Federal Deposit Insurance Corporation, and (f) Investments in money market funds substantially all of whose assets are invested in the types of assets described in clauses (a) through (e) above.

"Cash Management Account" has the meaning set forth in Section 2.7(a).

"Cash Management Agreements" means those certain cash management agreements, in form and substance reasonably satisfactory to Agent, each of which is among Borrower or one of its Subsidiaries, Agent, and one of the Cash Management Banks.

"Cash Management Bank" has the meaning set forth in Section 2.7(a).

"CFC" means any Subsidiary of a Loan Party that is a controlled foreign corporation (as that term is defined in Section 957 of the IRC).

"CFC Holdco" means any Subsidiary of a Loan Party substantially all of the assets of which consist of the Stock (including, for this purpose, any debt or other instrument treated as equity for U.S. federal income tax purposes) or Stock and indebtedness of any CFC or any other CFC Holdco.

- 9 -

### "Change of Control" means:

- (a) any Person or two or more Persons acting in concert (other than Permitted Holders), shall have acquired beneficial ownership, directly or indirectly, of Stock of Borrower (or other securities convertible into such Stock) representing 35% or more of the combined voting power of all Stock of Borrower entitled (without regard to the occurrence of any contingency) to vote for the election of members of the Board of Directors of Borrower: or
- (b) any Person or two or more Persons acting in concert (other than Permitted Holders), shall have acquired by contract or otherwise, or shall have entered into a contract or arrangement that, upon consummation thereof, will result in its or their acquisition of the power to exercise, directly or inclinedly, a controlling influence over the management or policies of Borrower or control over the Stock of such Person entitled to vote formembers of the Board of Directors of Borrower on a fully-diluted basis (and taking into account all such Stock that such Person or group has the right to acquire pursuant to any option right) representing 35% or more of the combined voting power of such Stock; or
- (c) Borrower shall cease to hold 100% of the Stock and voting power of each of its Subsidiaries that it a Guarantor (other than in connection with any transaction permitted pursuant to Section 7.3).

"Change in Law" means the occurrence after the date of the Agreement of:

- (a) the adoption or effectiveness of any law, rule, regulation, judicial ruling, judgment or treaty,
- (b) any change in any law, rule, regulation, judicial ruling, judgment or treaty or in the administration, interpretation, implementation or application by any Governmental Authority of any law, rule, regulation, guideline or treaty, or (c) the making or issuance by any Governmental Authority of any request, rule, guideline or directive, whether or not having the force of law; provided that notwithstanding anything in the Agreement to the contrary, (i) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, guidelines or directives thereunder or issued in connection therewith and (ii) all requests, rules, guidelines or directives concerning capital adequacy promulgated by the Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities shall, in each case, be deemed to be a "Change in Law," regardless of the date enacted, adopted or issued.

"Chillers" means a refrigerated unit out of which Borrower's products are sold. "Code" means the California Uniform Commercial Code, as in effect from time to

time; <u>provided</u>, <u>however</u>, , that in the event that, by reason of mandatory provisions of law, any or all of the attachment, perfection, priority, or remedies with respect to Agent's Lien on any Collateral is governed by the Uniform Commercial Code as enacted and in effect in a jurisdiction other than the State of California, the term "Code" shall mean the Uniform Commercial Code as enacted and in effect in such other jurisdiction solely for purposes of the provisions thereof relating to such attachment, perfection, priority, or remedies

"Collateral" means all assets and interests in assets and proceeds thereof now owned or hereafter acquired by Borrower or its Subsidiaries in or upon which a Lien is granted under any of the Loan Documents.

"Collateral Access Agreement" means a landlord waiver, bailee letter, or acknowledgement agreement of any lessor, warehouseman, processor, consignee, or other Person in possession of, having a Lien upon, or having rights or interests in Borrower's or its Subsidiaries' Books, Equipment, or Inventory, in each case, in form and substance reasonably satisfactory to Agent.

"Collections" means all cash, checks, notes, instruments, and other items of payment (including insurance proceeds, proceeds of cash sales, rental proceeds, and tax refunds).

"Commercial Tort Claim Assignment" has the meaning set forth in

Section 4.4(b).

"Commitment" means, with respect to each Lender, its Revolver Commitment or

its Delayed Draw Term Loan Commitment, as the context requires, and, with respect to all Lenders, their Revolver Commitments or their Delayed Draw Term Loan Commitments, as the context requires, in each case as such Dollar amounts are set forth beside such Lender's name under the applicable heading on Schedule C-1 or in the Assignment and Acceptance pursuant to which such Lender became a Lender hereunder, as such amounts may be reduced or increased from time to time pursuant to assignments made in accordance with the provisions of Section 14.1.

"Commitment Increase Notice" has the meaning set forth in Section 2.15(a).

"Commodity Exchange Act" means the Commodity Exchange Act (7 U.S.C. § 1 et seq.) as amended from time to time, and any successor statute.

"Compliance Certificate" means a certificate substantially in the form of Exhibit C-1 delivered by the chief financial officer of Borrower to Agent.

"Consolidated Funded Indebtedness" means, with respect to any Person at any date, all Indebtedness for borrowed money of such Person, determined on a consolidated basis in accordance with GAAP (other than Subordinated Debt and Permitted Preferred Stock), including, in any event, but without duplication, with respect to Borrower and its Subsidiaries, the Advances, Purchase Money Indebtedness, and the amount of their Capitalized Lease Obligations, in each case exclusive of Indebtedness owed by one Loan Party to another Loan Party and any Indebtedness in respect of any of the foregoing.

"Control Agreement" means a control agreement, in form and substance reasonably satisfactory to Agent, executed and delivered by Borrower or one of its Subsidiaries, Agent, and the applicable securities intermediary (with respect to a Securities Account) or bank (with respect to a Deposit Account), including without limitation, that certain Deposit Account Control Agreement, dated as of May 18, 2015, among Borrower, the Agent and Bank of America, N.A, in each case, as the same may be amended, restated, supplemented, or otherwise modified from time to time in accordance to the terms thereof and hereof.

"Daily Balance" means, as of any date of determination and with respect to any Obligation, the amount of such Obligation owed at the end of such day.

"<u>Daily Delayed Draw Term Loan Unused Line Fee</u>" means, as of any date, the product of (a) the aggregate amount of remaining Delayed Draw Term Loan Commitments on such date *times* (b) 0.30% per annum (calculated in accordance with the provisions of <u>Section 2.11(c)</u>).

"<u>Daily Revolver Unused Line Fee</u>" means, as of any date, the product of (a) the difference between (i) the aggregate amount of Revolver Commitments on such date *minus* 

(ii) the Revolver Usage on such date times (b) 0.30% per annum (calculated in accordance with the provisions of Section 2.11(b)).

"Default" means an event, condition, or default that, with the giving of notice, the passage of time, or both, would be an Event of Default.

"Defaulting Lender" means any Lender that (a) has failed to fund any amounts required to be funded by it under the Agreement on the date that it is required to do so under the Agreement (including the failure to make a required payment in connection with a Letter of Credit Disbursement), (b) notified Borrower, Agent, or any Lender in writing that it does not intend to comply with all or any portion of its funding obligations under the Agreement, (c) has made a public statement to the effect that it does not intend to comply with its funding obligations under the Agreements generally (as reasonably determined by Agent) under which it has committed to extend credit, (d) failed, within 1 Business Day after written request by Agent, to confirm that it will comply with the terms of the Agreement relating to its obligations to fund any amounts required to be funded by it under the Agreement,

(e) otherwise failed to pay over to Agent or any other Lender any other amount required to be paid by it under the Agreement on the date that it is required to do so under the Agreement, or

(f) (i) becomes or is insolvent or has a parent company that has become or is insolvent or

(ii) becomes the subject of a bankruptcy or insolvency proceeding, or has had a receiver, conservator, trustee, or custodian or appointed for it, or has taken any action in furtherance of, or indicating its consent to, approval of or acquiescence in any such proceeding or appointment or has a parent company that has become the subject of a bankruptcy or insolvency proceeding, or has had a receiver, conservator, trustee, or custodian appointed for it, or has taken any action in furtherance of, or indicating its consent to, approval of or acquiescence in any such proceeding or appointment.

"Defaulting Lender Rate" means (a) for the first 3 days from and after the date the relevant payment is due, the Base Rate, and (b) thereafter, the interest rate then applicable to Advances that are Base Rate Loans (inclusive of the Base Rate Margin applicable thereto).

"Delayed Draw Term Loan" has the meaning set forth in Section 2.2(a).

"Delayed Draw Term Loan Amount" means \$55,000,000, as such amount may be reduced from time to time in accordance with the provisions hereof.

"<u>Delayed Draw Term Loan Commitment</u>" means, with respect to each Lender, its Delayed Draw Term Loan Commitment, and, with respect to all Lenders, their Delayed Draw

Term Loan Commitments, in each case as such Dollar amounts are set forth beside such Lender's name under the applicable heading on <u>Schedule C-1</u> or in the Assignment and Acceptance pursuant to which such Lender became a Lender hereunder, as such amounts may be reduced or increased from time to time pursuant to assignments made in accordance with the provisions of <u>Section 14.1</u>.

"<u>Delayed Draw Term Loan Commitment Expiration Date</u>" means the earlier to occur of (a) May 15, 2021, and (b) the date when the Delayed Draw Term Loan Commitments have been reduced to \$0.

"Delayed Draw Term Loan Maturity Date" means May 15, 2024.

"Delayed Draw Term Loan Obligations" means all Obligations in respect of or relating to the Delayed Draw Term Loan (including principal, interest, premiums, if any, fees, costs, and expenses (including Lender Group Expenses) in respect thereof).

"Delayed Draw Term Loan Unused Line Fee" has the meaning set forth in Section 2.11(c).

"Deposit Account" means any deposit account (as that term is defined in the

Code)

"<u>Designated Account</u>" means the Deposit Account of Borrower identified on

Schedule D-1, as such Schedule may be updated by Borrower with the consent of the Agent.

"<u>Disbursement Letter</u>" means an instructional letter executed and delivered by Borrower to Agent regarding the extensions of credit to be made on the Restatement Effective Date, the form and substance of which is reasonably satisfactory to Agent.

"<u>Disposition</u>" means any transaction, or series of related transactions, pursuant to which any Person or any of its Subsidiaries sells, assigns, transfers or otherwise disposes of any property or assets (whether now owned or hereafter acquired) to any other Person, in each case, whether or not the consideration therefor consists of cash, securities or other assets owned by the acquiring Person, whether voluntary or involuntary and including any casualty losses or condemnations or other loss or destruction of any property of Borrower or any of its Subsidiaries.

"Distribution" has the meaning set forth in Section 7.10.

"Division/Series Transaction" means, with respect to the Loan Parties and their Subsidiaries, that any such Person (a) divides into two or more Persons (whether or not the original Loan Party or Subsidiary thereof survives such division) or (b) creates, or reorganizes into, one or more series, in each case as contemplated under the laws of any jurisdiction.

"Dollars" or "\$" means United States dollars.

"Dutch Subsidiary" means Freshpet NE B.V., a company organized under the laws of Netherlands.

"EEA Financial Institution" means (a) any credit institution or investment firm established in any EEA Member Country which is subject to the supervision of an EEA Resolution Authority, (b) any entity established in an EEA Member Country which is a parent of an institution described in clause (a) of this definition, or (c) any financial institution established in an EEA Member Country which is a Subsidiary of an institution described in clauses (a) or (b) of this definition and is subject to consolidated supervision with its parent.

"EEA Member Country" means any of the member states of the European Union, Iceland, Liechtenstein, and Norway.

"EEA Resolution Authority" means any public administrative authority or any Person entrusted with public administrative authority of any EEA Member Country (including any delegee) having responsibility for the resolution of any EEA Financial Institution

"Eligible Equipment" means (a) the purchase of refrigerators, and (b) the purchase of Equipment to be used at new or existing production facilities of

"Environmental Actions" means any complaint, summons, citation, notice, directive, order, claim, litigation, investigation, judicial or administrative proceeding, judgment, letter, or other communication from any Governmental Authority, or any third party involving violations of Environmental Laws or releases of Hazardous Materials from (a) any assets, properties, or businesses of Borrower, its Subsidiaries, or any of their predecessors in interest,

(b) from adjoining properties or businesses, or (c) from or onto any facilities which received Hazardous Materials generated by Borrower, its Subsidiaries, or any of their predecessors in interest.

"Environmental Indemnity." means, with respect to any Real Property, an environmental indemnity agreement (whether provided as a separate document or included in another document required to be delivered to the Agent pursuant to this Agreement or any other Loan Document in connection with such Real Property), executed by the Loan Parties, in form and substance reasonably satisfactory to the Agent.

"Environmental Law" means any applicable federal, state, provincial, foreign or local statute, law, rule, regulation, ordinance, code, binding and enforceable guideline, binding and enforceable written policy, or rule of common law now or hereafter in effect and in each case as amended, or any judicial or administrative interpretation thereof, including any judicial or administrative order, consent decree or judgment, in each case, to the extent binding on Borrower or its Subsidiaries, relating to the environment, the effect of the environment on employee health, or Hazardous Materials, including the Comprehensive Environmental Response Compensation and Liability Act, 42 USC §9601 et seq.; the Resource Conservation and Recovery Act, 42 USC §6901 et seq.; the Federal Water Pollution Control Act, 33 USC §1251 et seq.; the Toxic Substances Control Act, 15 USC §2601 et seq.; the Clean Air Act, 42 USC §7401 et seq.; the Safe Drinking Water Act, 42 USC §3803 et seq.; the Oil Pollution Act of 1990, 33 USC §2701 et seq.; the Emergency Planning and the Community Right-to-Know Act of 1986, 42 USC §11001 et seq.; the Hazardous Material Transportation Act, 49 USC §1801 et seq.; and the Occupational Safety and Health Act, 29 USC §651 et seq. (to the

Borrower.

extent it regulates occupational exposure to Hazardous Materials); any state and local or foreign counterparts or equivalents, in each case as amended from time to time.

"Environmental Liabilities and Costs" means all liabilities, monetary obligations, losses, damages, punitive damages, consequential damages, treble damages, costs and expenses (including all reasonable fees, disbursements and expenses of counsel, experts, or consultants, and costs of investigation and feasibility studies), fines, penalties, sanctions, and interest incurred as a result of any claim or demand, or Remedial Action required, by any Governmental Authority or any third party, and which relate to any Environmental Action.

"Environmental Lien" means any Lien in favor of any Governmental Authority for Environmental Liabilities and Costs.

"Equipment" means equipment (as that term is defined in the Code) and includes machinery, machine tools, motors, furniture, furnishings, fixtures, vehicles (including motor vehicles), computer hardware, tools, parts, and goods (other than consumer goods, farm products, or Inventory), wherever located, including all attachments, accessories, accessions, replacements, substitutions, additions, and improvements to any of the foregoing.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended, and any successor statute thereto.

"ERISA Affiliate" means (a) any Person subject to ERISA whose employees are treated as employed by the same employer as the employees of Borrower or its Subsidiaries under IRC Section 414(b), (b) any trade or business subject to ERISA whose employees are treated as employed by the same employer as the employees of Borrower or its Subsidiaries under IRC Section 414(c), (c) solely for purposes of Section 302 of ERISA and Section 412 of the IRC, any organization subject to ERISA that is a member of an affiliated service group of which Borrower or any of its Subsidiaries is a member under IRC Section 414(m), or (d) solely for purposes of Section 302 of ERISA and Section 412 of the IRC, any Person subject to ERISA that is a party to an arrangement with Borrower or any of its Subsidiaries and whose employees are aggregated with the employees of Borrower or its Subsidiaries under IRC Section 414(o).

time to time.

"EU Bail-In Legislation Schedule" means the document described as such and published by the Loan Market Association (or any successor Person) from

"Event of Default" has the meaning set forth in Section 8.

"Excess Cash Flow" means, with respect to any fiscal period and with respect to Borrower determined on a consolidated basis in accordance with GAAP

the result of:

- (a) TTM EBITDA, minus
- (b) the sum of
- (i) the cash portion of Interest Expense paid during such fiscal period,
- (ii) the cash portion of income taxes paid during such period,

- 15 -137020431v13 (iii) all scheduled principal payments made in respect of the Delayed Draw Term Loan during such period, (iv) the cash portion of unfinanced Capital Expenditures made during such

for such period.

period, and

(v) all other cash amounts added back in the determination of TTM EBITDA

"Exchange Act" means the Securities Exchange Act of 1934, as in effect from

time to time.

"Excluded Deposit Account" means a deposit account used solely for (i) 401(k) or

other employee benefit plans, (ii) tax deposits, (iii) trust or escrow or fiduciary accounts, (iv) petty cash accounts or (v) zero balance accounts.

"Excluded Hedge Obligations" means, with respect to any Guarantor, any Hedge Obligation if, and to the extent that, all or a portion of the Guaranty of such Guarantor of, or the grant by such Guarantor of a security interest pursuant to the Loan Documents to secure, such Hedge Obligation (or any guarantee thereof) is or would otherwise have become illegal or unlawful under the Commodity Exchange Act or any rule, regulation or order of the Commodity Futures Trading Commission (or the application or official interpretation of any thereof) by virtue of such Guarantor's failure for any reason to constitute an "eligible contract participant" as defined in the Commodity Exchange Act and the regulations thereunder (determined after giving effect to Section 17.10 hereof and any other "keepwell, support or other agreement" for the benefit of such Loan Party and any and all guarantees of such Loan Party's Hedge Obligations by other Loan Parties) at the time the Guaranty of such Guarantor or the grant of such security interest would otherwise have become effective with respect to such related Hedge Obligation. If a Hedge Obligation arises under a master agreement governing more than one swap, such exclusion shall apply only to the portion of such Hedge Obligation that is attributable to swaps for which such Guaranty or security interest is or becomes excluded in accordance with the first sentence of this definition.

"Excluded Property." means (i) any Stock in or assets of any Excluded Subsidiary (other than 65% of the outstanding voting Stock (and 100% of the outstanding non-voting Stock) of any CFC or CFC Holdco that is a direct Subsidiary of a Loan Party), (ii) any rights or interest in any contract, lease, permit, license, or license agreement covering real or personal property of Borrower if under the terms of such contract, lease, permit, license, or license agreement, or applicable law with respect thereto, the grant of a security interest or Lien therein is prohibited as a matter of law or under the terms of such contract, lease, permit, license, or license agreement and such prohibition or restriction has not been waived or the consent of the other party to such contract, lease, permit, license, or license agreement has not been obtained (provided, that, (A) the foregoing exclusions of this clause (ii) shall in no way be construed (1) to apply to the extent that any described prohibition or restriction is unenforceable under Section 9-406, 9-407, 9-408, or 9-409 of the Code or other applicable law, or (2) to apply to the extent that any consent or waiver has been obtained that would permit Agent's security interest or Lien notwithstanding the prohibition or restriction on the pledge of such contract, lease, permit, license, or license

- 16 -137020431v13 agreement and (B) the foregoing exclusions of clauses (i) and (ii) shall in no way be construed to limit, impair, or otherwise affect any of Agent's, any other member of the Lender Group's or any Bank Product Provider's continuing security interests in and Liens upon any rights or interests of Borrower in or to (1) monies due or to become due under or in connection with any described contract, lease, permit, license, license agreement, or Stock (including any Accounts or Stock), or (2) any proceeds from the sale, license, lease, or other dispositions of any such contract, lease, permit, license, license agreement, or Stock), (iii) leasehold interests in any real property with fair market value lower than \$10,000,000, for any such property individually; provided that in no event shall the foregoing be construed to modify the requirements in Section 6.9 with respect to any such real property and (iv) any United States intent-to-use trademark applications to the extent that, and solely during the period in which, the grant of a security interest therein would impair the validity or enforceability of such intent-to-use trademark applications under applicable federal law, provided that upon submission and acceptance by the United States Patent and Trademark Office of an amendment to allege use pursuant to 15 U.S.C. Section 1060(a) (or any successor provision), such intent-to-use trademark application shall not be considered Excluded Property.

"Excluded Subsidiary" means any Subsidiary of a Loan Party that is a CFC, a CFC Holdco or a direct or indirect Subsidiary of a CFC or CFC Holdco.

"Excluded Tax" means any of the following Taxes imposed on or with respect to a Recipient or required to be withheld or deducted from a payment to a Recipient, (a) Taxes imposed on or measured by net income (however denominated), franchise Taxes, and branch profits Taxes, in each case, (i) imposed as a result of such Recipient being organized under the laws of, or having its principal office or, in the case of any Lender, its applicable lending office located in, the jurisdiction imposing such Tax (or any political subdivision thereof) or (ii) that are Other Connection Taxes, (b) in the case of a Lender, U.S. federal withholding Taxes imposed on amounts payable to or for the account of such Lender with respect to an applicable interest in a loan under this Agreement or Commitment pursuant to a law in effect on the date on which (i) such Lender acquires such interest in such loan or Commitment (other than pursuant to an assignment request by the Borrower under Section 2.3(d)) or (ii) such Lender changes its lending office, except in each case to the extent that, pursuant to Section 16.11, amounts with respect to such Taxes were payable either to such Lender's assignor immediately before such Lender became a party hereto or to such Lender immediately before it changed its lending office, (c) Taxes attributable to such Recipient's failure to comply with Section 16.11 and (d) any withholding Taxes imposed under FATCA.

"Existing Loan Agreement" has the meaning set forth in the recitals to this

Agreement.

"Fee Letter" means that certain Amended and Restated Fee Letter, dated as of

even date herewith, between Borrower and Agent, as the same may be amended, restated, supplemented, or otherwise modified from time to time in accordance to the terms thereof and hereof

"FATCA" means Sections 1471 through 1474 of the IRC, as of the date of this Agreement (or any amended or successor version that is substantively comparable and not

- 17 -137020431v13 materially more onerous to comply with) and any current or future regulations or official interpretations thereof, any applicable agreements entered into pursuant to Section 1471(b)(1) of the IRC, any applicable intergovernmental agreements with respect to the implementation of such Sections of the IRC, and any fiscal or regulatory legislation, rules or official administrative practices adopted pursuant to any intergovernmental agreement, treaty or convention among Governmental Authorities entered into in connection with the implementation of the foregoing.

"Federal Funds Rate" means, for any period, a fluctuating interest rate per annum equal to, for each day during such period, the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers, as published on the next succeeding Business Day by the Federal Reserve Bank of New York, or, if such rate is not so published for any day which is a Business Day, the average of the quotations for such day on such transactions received by Agent from three Federal funds brokers of recognized standing selected by it.

"Fixed Charges" means with respect to Borrower and its Subsidiaries for any period, the sum, without duplication, of (a) Interest Expense and (b) principal payments required to be paid during such period in respect of the Indebtedness.

"Fixed Charge Coverage Ratio" means, with respect to Borrower and its Subsidiaries on a consolidated basis, for any period, the ratio of (i) TTM EBITDA for such period *minus* the sum of (A) Maintenance Capital Expenditures made (to the extent not already incurred in a prior period) or incurred during such period *plus* (B) all federal, state, and local income taxes paid in cash during such period, *plus* (C) all Distributions made during such period, to (ii) Fixed Charges for such period.

"Flood Insurance Laws" means, collectively, (i) the National Flood Insurance Act of 1968 as now or hereafter in effect or any successor statute thereto, (ii) the Flood Disaster Protection Act of 1973 as now or hereafter in effect or any successor statute thereto, (iii) the National Flood Insurance Reform Act of 1994 as now or hereafter in effect or any successor statute thereto, (iv) the Flood Insurance Reform Act of 2004 as now or hereafter in effect or any successor statute thereto and (v) the Biggert-Waters Flood Insurance Reform Act of 2012 as now or hereafter in effect or any successor statute thereto.

"FP Foods" has the meaning set forth in the recitals to this Agreement.

"FP Foods Dissolution" means the dissolution of FP Foods on or prior to December 31, 2019 (or such later date as the Agent may agree); provided, that, on or prior to Restatement Effective Date any remaining assets of FP Foods (including, without limitation, Property A) shall have been transferred to Borrower, subject to Agent's continuing first priority Lien thereon.

"Funding Date" means the date on which a Borrowing occurs. "Funding Losses" has the meaning set forth in Section 2.13(b)(ii).

"GAAP" means generally accepted accounting principles as in effect from time to time in the United States, consistently applied.

"General Intangibles" means general intangibles (as that term is defined in the Code), including payment intangibles, contract rights, rights to payment, rights arising under common law, statutes, or regulations, choses or things in action, goodwill, patents, trade names, trade secrets, trademarks, servicemarks, copyrights, blueprints, drawings, purchase orders, customer lists, monies due or recoverable from pension funds, route lists, rights to payment and other rights under any royalty or licensing agreements, infringement claims, computer programs, information contained on computer disks or tapes, software, literature, reports, catalogs, insurance premium rebates, tax refunds, and tax refund claims, and any other personal property other than Accounts, Deposit Accounts, goods, Investment Property, and Negotiable Collateral.

"Governing Documents" means, with respect to any Person, the certificate or articles of incorporation, by-laws, or other organizational documents of such Person.

"Governmental Authority." means any federal, state, local, or other governmental or administrative body, instrumentality, board, department, or agency or any court, tribunal, administrative hearing body, arbitration panel, commission, or other similar dispute-resolving panel or body.

"Guarantors" means each Subsidiary of Borrower that executes, or otherwise becomes a party to, a Guaranty, including pursuant to the provisions of Section 6.15 of this Agreement, and "Guarantor" means any one of them. For the avoidance of doubt, no Excluded Subsidiary shall be a Guarantor.

"Guarantor Security Agreement" means one or more security agreements executed and delivered by each Guarantor in favor of Agent, in each case, in form and substance reasonably satisfactory to Agent.

"Guaranty" means that certain general continuing guaranty executed and delivered by each Guarantor in favor of Agent, in form and substance reasonably satisfactory to Agent.

"<u>Hazardous Materials</u>" means (a) substances that are defined or listed in, or otherwise classified pursuant to, any applicable laws or regulations as "hazardous substances," "hazardous materials," "hazardous wastes," "toxic substances," or any other formulation intended to define, list, or classify substances by reason of deleterious properties such as ignitability, corrosivity, reactivity, carcinogenicity, reproductive toxicity, or "EP toxicity",

(b) oil, petroleum, or petroleum derived substances, natural gas, natural gas liquids, synthetic gas, drilling fluids, produced waters, and other wastes associated with the exploration, development, or production of crude oil, natural gas, or geothermal resources, (c) any flammable substances or explosives or any radioactive materials, and (d) asbestos in any form or electrical equipment that contains any oil or dielectric fluid containing levels of polychlorinated biphenyls in excess of 50 parts per million.

"Hedge Agreement" means any and all agreements or documents now existing or hereafter entered into by Borrower or any of its Subsidiaries that provide for an interest rate, credit, commodity or equity swap, cap, floor, collar, forward foreign exchange transaction, currency swap, cross currency rate swap, currency option, or any combination of, or option with

respect to, these or similar transactions, for the purpose of hedging Borrower's or any of its Subsidiaries' exposure to fluctuations in interest or exchange rates, loan, credit exchange, security, or currency valuations or commodity prices.

"Hedge Obligations" means any and all obligations or liabilities, whether absolute or contingent, due or to become due, now existing or hereafter arising, of Borrower or its Subsidiaries arising under, owing pursuant to, or existing in respect of any Hedge Agreements.

"Holdout Lender" has the meaning set forth in <u>Section 15.2(a)</u>. "Increased Amount <u>Date</u>" has the meaning set forth in <u>Section 2.15(b)</u>.

"Incremental Facility Supplement" has the meaning set forth in Section 2.15(d). "Incremental Term Loan" has the meaning set forth in Section 2.15(a).

"Incremental Term Loan Commitment"

has the meaning set forth in Section

2.15(a).

"Incremental Term Loan Obligations" all Obligations in respect of or relating to

any Incremental Term Loans (including principal, interest, premiums, if any, fees, costs, and expenses (including Lender Group Expenses) in respect thereof).

"Incremental Term Note" has the meaning set forth in Section 2.15. "Indebtedness" means (a) all obligations for borrowed money (and Prohibited

Preferred Stock), (b) all obligations evidenced by bonds, debentures, notes, or other similar instruments and all reimbursement or other obligations in respect of letters of credit, bankers acceptances, interest rate swaps, or other financial products, (c) all obligations as a lessee under Capital Leases, (d) all obligations or liabilities of others secured by a Lien on any asset of a Person or its Subsidiaries, irrespective of whether such obligation or liability is assumed, (e) all obligations to pay the deferred purchase price of assets (other than trade payables, deferred rent, taxes or compensation incurred in the ordinary course of business and repayable in accordance with customary trade practices), (f) all Hedge Obligations, and (g) any obligation guaranteeing or intended to guarantee (whether directly or indirectly guaranteed, endorsed, co-made, discounted, or sold with recourse) any obligation of any other Person that constitutes Indebtedness under any of clauses (a) through (f) above.

"Indemnified Liabilities" has the meaning set forth in Section 11.3. "Indemnified Person" has the meaning set forth in Section 11.3.

"Indemnified Taxes" means (a) Taxes, other than Excluded Taxes, imposed on or with respect to any payment made by or on account of any obligation of the Borrower under any Loan Document and (b) to the extent not otherwise described in clause (a), Other Taxes.

"Insolvency Proceeding" means any proceeding commenced by or against any Person under any provision of the Bankruptcy Code or under any other

state or federal

- 20 -137020431v13 bankruptcy or insolvency law, assignments for the benefit of creditors, formal or informal moratoria, compositions, extensions generally with creditors, or proceedings seeking reorganization, arrangement, or other similar relief.

"Installed Store" means a store or other place of business not owned, leased or operated by a Loan Party or any of its Affiliates in which Borrower's products are maintained and sold.

"Intercompany Subordination Agreement" means that certain Amended and Restated Intercompany Subordination Agreement, dated as of November 13, 2014, executed and delivered by Borrower, the Canadian Subsidiary and Agent, as supplemented by (i) that certain Joinder Agreement, dated as of October 5, 2016, by and among FP F Foods, Borrower and the Agent, (ii) that certain Joinder Agreement, dated as of September 21, 2017, by and among Freshpet Europe LTD, a United Kingdom private limited company. Borrower and the Agent and

(iii) that certain Joinder Agreement, dated as of May 15, 2019, by and among the Dutch Subsidiary, Borrower and the Agent, and as the same may be further amended, restated, supplemented, or otherwise modified from time to time in accordance to the terms thereof and hereof.

"Interest Expense" means, for any period, the aggregate of the cash interest expense of Borrower and its Subsidiaries for such period, determined on a consolidated basis in accordance with GAAP.

"Interest Period" means, with respect to each LIBOR Rate Loan, a period commencing on the date of the making of such LIBOR Rate Loan (or the continuation of a LIBOR Rate Loan or the conversion of a Base Rate Loan to a LIBOR Rate Loan) and ending 1, 2, 3 or 6 months thereafter; provided, however, that (a) if any Interest Period would end on a day that is not a Business Day, such Interest Period shall be extended (subject to clauses (c)-(e) below) to the next succeeding Business Day, (b) interest shall accrue at the applicable rate based upon the LIBOR Rate from and including the first day of each Interest Period to, but excluding, the day on which any Interest Period expires, (c) any Interest Period that would end on a day that is not a Business Day shall be extended to the next succeeding Business Day unless such Business Day falls in another calendar month, in which case such Interest Period shall end on the next preceding Business Day, (d) with respect to an Interest Period that begins on the last Business Day of a calendar month (or on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period), the Interest Period shall end on the last Business Day of the calendar month that is 1, 2, or 3 months after the date on which the Interest Period began, as applicable, (e) Borrower may not elect an Interest Period with respect to any portion of the Delayed Draw Term Loan which will end after the Delayed Draw Term Loan Maturity Date and (g) Borrower may not elect an Interest Period with respect to any portion of an Incremental Term Loan which will end after the maturity date of such Incremental Term Loan.

"Inventory" means inventory (as that term is defined in the Code).

"Investment" means, with respect to any Person, any investment by such Person in any other Person (including Affiliates) in the form of loans, guarantees, advances, or capital contributions (excluding (a) commission, travel, and similar advances to officers and employees of such Person made in the ordinary course of business, and (b) bona fide Accounts arising in the ordinary course of business consistent with past practice), purchases or other acquisitions of Indebtedness, Stock, or all or substantially all of the assets of such other Person (or of any division or business line of such other Person), and any other items that are or would be classified as investments on a balance sheet prepared in accordance with GAAP.

"Investment Property" means investment property (as that term is defined in the

Code). "IRC" means the Internal Revenue Code of 1986, as in effect from time to time. "Issuing Lender" means CNB or any other Lender that, at the request of

and with the consent of Agent, agrees, in such Lender's sole discretion, to become an Issuing Lender for the purpose of issuing Letters of Credit pursuant to Section 2.12.

"L/C Disbursement" means a payment made by the Issuing Lender pursuant to a Letter of Credit.

"<u>Lender</u>" and "<u>Lenders</u>" have the respective meanings set forth in the preamble to this Agreement, and shall include any other Person made a party to this Agreement in accordance with the provisions of <u>Section 14.1</u> or <u>Section 2.15(c)</u>.

"Lender Group" means, individually and collectively, each of the Lenders (including the Issuing Lender) and Agent.

"Lender Group Expenses" means all (a) documented costs or expenses (including taxes, and insurance premiums) required to be paid by Borrower or its Subsidiaries under any of the Loan Documents that are paid, advanced, or incurred by the Lender Group, (b) fees or charges paid or incurred by Agent in connection with the Lender Group's transactions with Borrower or its Subsidiaries, including, the contact of the department of motor vehicles), filing, recording, publication, appraisal (including periodic collateral appraisals or business valuations to the extent of the fees and charges (and up to the amount of any limitation) contained in this Agreement), real estate surveys, real estate title policies and endorsements, and environmental examinations, (c) costs and expenses incurred by Agent in the disbursement of funds to Borrower or other members of the Lender Group (by wire transfer or otherwise), (d) charges paid or incurred by Agent resulting from the dishonor of checks, (e) documented reasonable costs and expenses paid or incurred by the Lender Group to correct any default or enforce any provision of the Loan Documents, or in gaining possession of, maintaining, handling, preserving, storing, shipping, selling, preparing for sale, or advertising to sell the Collateral, or any portion thereof, irrespective of whether a sale is consummated,

(f) documented audit fees and expenses of Agent related to audit examinations of the Books to the extent of the fees and charges (and up to the amount of any limitation)

Agreement, (g) documented reasonable costs and expenses of third party claims or any other suit paid or incurred by the Lender Group in enforcing or defending the Loan Documents or in connection with the transactions contemplated by the Loan Documents or the Lender Group's relationship with Borrower or any its Subsidiaries, (h) Agent's and each Lender's documented reasonable costs and expenses (including attorneys' fees of one primary counsel and one local counsel in each relevant jurisdiction) incurred in advising, structuring, drafting, reviewing, administering, syndicating, or amending the Loan Documents, and (i) Agent's and each Lender's documented reasonable costs and expenses (including attorneys', accountants', consultants', and other advisors' fees and expenses incurred in terminating, enforcing (including attorneys', accountants', consultants', and other advisors' fees and expenses incurred in connection with a "workout," a "restructuring," or an Insolvency Proceeding concerning Borrower or its Subsidiaries or in exercising rights or remedies under the Loan Documents), or defending the Loan Documents, irrespective of whether suit is brought, or in taking any Remedial Action concerning the Collateral.

"Lender-Related Person" means, with respect to any Lender, such Lender, together with such Lender's Affiliates, officers, directors, employees, attorneys,

and agents.

"Letter of Credit" has the meaning set forth in Section 2.12(a).

"Letter of Credit Collateralization" means either (a) providing cash collateral (pursuant to documentation reasonably satisfactory to Agent, including provisions that specify that the Letter of Credit Fees and all commissions, fees, charges and expenses provided for in Section 2.12(d) of this Agreement (including any fronting fees) will continue to accrue while the Letters of Credit are outstanding) to be held by Agent for the benefit of the Lenders holding a Revolver Commitment in an amount equal to 105% of the then existing Letter of Credit Usage,

105% of the then existing Letter of Credit Usage,
(b) delivering to Agent documentation executed by all beneficiaries under the Letters of Credit, in form and substance reasonably satisfactory to Agent and Issuing Lender, terminating all of such beneficiaries' rights under the Letters of Credit, or (c) providing Agent with a standby letter of credit, in form and substance reasonably satisfactory to Agent, from a commercial bank acceptable to Agent (in its sole discretion) in an amount equal to 105% of the then existing Letter of Credit Usage (it being understood that the Letter of Credit Fee and all fronting fees set forth in the Agreement will continue to accrue while the Letters of Credit are outstanding and that any such standby letter of credit).

"<u>Letter of Credit Exposure</u>" means, as of any date of determination with respect to any Lender, such Lender's Pro Rata Share of the Letter of Credit Usage on such date.

"Letter of Credit Usage" means, as of any date of determination, the aggregate undrawn amount of all outstanding Letters of Credit.

"Leverage Ratio" means, as of any date of determination, the ratio of

(a) Consolidated Funded Indebtedness of Borrower and its Subsidiaries as of such date of determination to (b) TTM EBITDA of Borrower and its Subsidiaries as of such date of determination.

- 23 -137020431v13 "LIBOR Deadline" has the meaning set forth in Section 2.13(b)(i). "LIBOR Notice" means a written notice in the form of

Exhibit L-1. "LIBOR Option" has the meaning set forth in Section 2.13(a).

"LIBOR Rate" means, for each Interest Period for each LIBOR Rate Loan, the rate per annum determined by Agent (rounded upwards, if necessary, to the next 1/16th of one percent) by dividing (a) the Base LIBOR Rate for such Interest Period, by (b) 100% minus the Reserve Percentage. The LIBOR Rate shall be adjusted on and as of the effective day of any change in the Reserve Percentage.

"LIBOR Rate Loan" means each portion of an Advance, a Delayed Draw Term Loan or any Incremental Term Loan that bears interest at a rate determined by reference to the LIBOR Rate.

"Lien" means any interest in an asset securing an obligation owed to, or a claim by, any Person other than the owner of the asset, irrespective of whether (a) such interest is based on the common law, statute, or contract, (b) such interest is recorded or perfected, and (c) such interest is contingent upon the occurrence of some future event or events or the existence of some future circumstance or circumstances. Without limiting the generality of the foregoing, the term "Lien" includes the lien or security interest arising from a mortgage, deed of trust, encumbrance, pledge, hypothecation, assignment, deposit arrangement, security agreement, conditional sale or trust receipt, or from a lease, consignment, or bailment for security purposes and also includes reservations, exceptions, encroachments, easements, rights-of-way, covenants, conditions, restrictions, leases, and other title exceptions and encumbrances affecting Real Property. For the avoidance of doubt, the term "Lien" shall not be deemed to include any nonexclusive intellectual property license.

"Loan" an Advance, a Delayed Draw Term Loan or an Incremental Term Loan, as applicable; and "Loans" means the aggregate of all Advances, all Delayed Draw Term Loans and all Incremental Term Loans, as applicable, outstanding at any given time.

"Loan Account" has the meaning set forth in Section 2.10.

"Loan Documents" means this Agreement, the Master Reaffirmation Agreement, the Cash Management Agreements, the Control Agreements, the Fee Letter, any Guarantor Security Agreement, each Guaranty, the Intercompany Subordination Agreement, the Mortgages, any Environmental Indemnity, the Stock Pledge Agreement, the Trademark Security Agreement, any Incremental Facility Supplement, any note or notes executed by Borrower in connection with this Agreement and payable to a Lender, and any other agreement entered into, now or in the future, by Borrower and Agent or any Lender in connection with either this Agreement or the Existing Loan Agreement.

"Loan Parties" means Borrower and each Guarantor, and "Loan Party" means any

one of them.

- 24 -137020431v13 "Maintenance Capital Expenditures" means, for any period, the sum of (without duplication) (a) Capital Expenditures made during such period on account of the maintenance of the Equipment of Borrower and its Subsidiaries, and (b) the higher of (i) actual amount spent on Capital Expenditures made during such period on account of maintenance of the Chillers and (ii) Capital Expenditures made during such period on account of maintenance of the Chillers, calculated as the product of (x) the total number of Installed Stores as of the end of such period multiplied by (y) \$16.

"Margin Stock" means "margin stock" as defined in Regulation U of the Board of Governors of the Federal Reserve System of the United States (or any successor) as in effect from time to time.

"Master Reaffirmation Agreement" means that certain Master Reaffirmation Agreement, dated as of even date herewith, executed and delivered by Borrower, each of its Subsidiaries, and Agent, as the same may be amended, restated, supplemented, or otherwise modified from time to time in accordance to the terms thereof and hereof

"Material Adverse Change" means (a) a material adverse change in the business, operations, results of operations, assets, liabilities or condition (financial or otherwise) of Borrower and its Subsidiaries, taken as a whole, (b) a material impairment of Borrower's and its Subsidiaries' ability to perform their respective obligations under the Loan Documents to which they are parties or of the Lender Group's ability to enforce the Obligations or realize upon the Collateral, or (c) a material impairment of the enforceability or priority of the Agent's Liens with respect to the Collateral as a result of an action or failure to act on the part of Borrower or its Subsidiaries.

"Material Contract" means, with respect to any Person, (a) each contract or agreement to which such Person or any of its Subsidiaries is a party involving aggregate consideration payable to or by such Person or such Subsidiary of \$2,500,000 or more (other than purchase orders in the ordinary course of the business of such Person or such Subsidiary and other than contracts that by their terms may be terminated by such Person or Subsidiary in the ordinary course of its business upon less than 60 days' notice without penalty or premium), and

all other contracts or agreements, the loss or termination of which could reasonably be expected to result in a Material Adverse Change.

"Maximum Commitment Amount" has the meaning set forth in Section 2.15(a). "Maximum Revolver Amount" means \$35,000,000, as such amount may be reduced from time to time in accordance with the provisions hereof.

"Mortgages" means, individually and collectively, (i) that certain Open-End Mortgage, Assignment of Leases and Rents, Security Agreement and Fixture Filing, dated as of October 5, 2016, made by and from FP Foods to Agent, and recorded with the Recorder of Deeds, Northampton County, Pennsylvania, in Book 2016-1, Starting Page 231021, relating to Property A, as the same may be amended, restated, supplemented, or otherwise modified from time to time in accordance to the terms thereof and hereof, (ii) that certain Open-End Mortgage, Assignment of Leases and Rents, Security Agreement and Fixture Filing, dated as of October 5,

2016, made by and from Borrower to Agent, and recorded with the Recorder of Deeds, Northampton County, Pennsylvania, in Book 2016-1, Starting Page 231055, in each case, relating to Property B, as the same may be amended, restated, supplemented, or otherwise modified from time to time in accordance to the terms thereof and hereof, and (iii) any other mortgages, deeds of trust, or deeds to secure debt, executed and delivered by Borrower or its Subsidiaries in favor of Agent, in form and substance reasonably satisfactory to Agent, that encumber the Real Property Collateral.

"Negotiable Collateral" means letters of credit, letter of credit rights, instruments, promissory notes, drafts, documents, and chattel paper (including electronic chattel paper and tangible chattel paper).

"Net Cash Proceeds" means, (i) with respect to any Disposition by any Person or any of its Subsidiaries, the amount of cash received (directly or indirectly) from time to time (whether as initial consideration or through the payment or disposition of deferred consideration) by or on behalf of such Person or such Subsidiary, in connection therewith after deducting therefrom only (A) the amount of any Indebtedness secured by any Permitted Lien on any asset (other than Indebtedness assumed by the purchaser of such asset) which is required to be, and is, repaid in connection with such Disposition (other than Indebtedness under this Agreement),

(B) reasonable expenses related thereto incurred by such Person or such Subsidiary in connection therewith, (C) transfer taxes paid to any taxing authorities by such Person or such Subsidiary in connection therewith, and (D) net income taxes paid or reasonably estimated to be payable in connection with such Disposition and (ii) with respect to the issuance or incurrence of any Indebtedness by any Person or any of its Subsidiaries, or the sale or issuance by any Person or any of its Subsidiaries of any shares of its Stock, the aggregate amount of cash received (directly or indirectly) from time to time (whether as initial consideration or through the payment or disposition of deferred consideration) by or on behalf of such Person or such Subsidiary in connection therewith, after deducting therefrom only (A) reasonable expenses related thereto incurred by such Person or such Subsidiary in connection therewith, (B) transfer taxes paid by such Person or such Subsidiary in connection therewith, in each case of clause (i) and (ii) to the extent, but only to the extent, that the amounts so deducted are properly attributable to such transaction or to the asset that is the subject thereof.

"New Lender" has the meaning set forth in Section 2.15(c).

"Non-Defaulting Lender" means, at any time, a Lender that at such time is not a Defaulting Lender.

"Obligations" means (a) all loans (including Delayed Draw Term Loan and, if the Incremental Term Loan Facility shall be activated, Incremental Term Loans), Advances, debts, principal, interest (including any interest that accrues after the commencement of an Insolvency Proceeding, regardless of whether allowed or allowable in whole or in part as a claim in any such Insolvency Proceeding), contingent reimbursement obligations with respect to outstanding Letters of Credit, premiums, liabilities (including all amounts charged to Borrower's Loan Account pursuant hereto), obligations (including indemnification obligations), fees (including the fees provided for in the Fee Letter), charges, costs, Lender Group Expenses (including any

fees or expenses that accrue after the commencement of an Insolvency Proceeding, regardless of whether allowed or allowable in whole or in part as a claim in any such Insolvency Proceeding), lease payments, guaranties, covenants, and duties of any kind and description owing by Borrower to the Lender Group pursuant to or evidenced by the Loan Documents and irrespective of whether for the payment of money, whether direct or indirect, absolute or contingent, due or to become due, now existing or hereafter arising, and including all interest not paid when due and all Lender Group Expenses that Borrower is required to pay or reimburse by the Loan Documents, by law, or otherwise, and (b) all Bank Product Obligations; provided that, notwithstanding the foregoing, the Obligations shall exclude any Excluded Hedge Obligations. Any reference in this Agreement or in the Loan Documents to the Obligations shall include all extensions, modifications, renewals, or alterations thereof, both prior and subsequent to any Insolvency Proceeding.

"Originating Lender" has the meaning set forth in Section 14.1(f).

"Other Connection Taxes" means, with respect to any Recipient, Taxes imposed as a result of a present or former connection between such Recipient and the jurisdiction imposing such Tax (other than connections arising from such Recipient having executed, delivered, become a party to, performed its obligations under, received payments under, received or perfected a security interest under, engaged in any other transaction pursuant to or enforced any Loan Document, or sold or assigned an interest in any Loan Document).

"Other Taxes" means all present or future stamp, court or documentary, intangible, recording, filing or similar Taxes that arise from any payment made under, from the execution, delivery, performance, enforcement or registration of, from the receipt or perfection of a security interest under, or otherwise with respect to, any Loan Document, except any such Taxes that are Other Connection Taxes imposed with respect to an assignment (other than an assignment made at the request of a Loan Party).

- "Overadvance" has the meaning set forth in Section 2.4(e)(i). "Participant" has the meaning set forth in Section 14.1(f).
- "Participant Register" has the meaning set forth in Section 14.1(f). "Patriot Act" has the meaning set forth in Section 5.24.

"Permitted Acquisition" means any Acquisition so long as:

(a) no Default or Event of Default shall have occurred and be continuing or would result from the consummation of the proposed Acquisition and the proposed Acquisition is

consensual

(b) Borrower has provided Agent with written confirmation, supported by reasonably detailed calculations, that on a proform a basis (including proform a adjustments arising out of events which are directly attributable to such proposed Acquisition, are factually supportable, and are expected to have a continuing impact, in each case, determined as if the combination had been accomplished at the beginning of the relevant period; such eliminations

- 27 -137020431v13 and inclusions to be mutually and reasonably agreed upon by Borrower and Agent) created by adding the historical combined financial statements of Borrower (including the combined financial statements of any other Person or assets that were the subject of a prior Permitted Acquisition during the relevant period) to the historical consolidated financial statements of the Person to be acquired (or the historical financial statements related to the assets to be acquired) pursuant to the proposed Acquisition, Borrower and its Subsidiaries would have been in compliance with the financial covenants in Section 7.18 of the Agreement for the 4 fiscal quarter period ended immediately prior to the proposed date of consummation of such proposed Acquisition,

- (c) Bonower has provided Agent with its due diligence package relative to the proposed Acquisition, consisting of, to the extent reasonably available, forecasted balance sheets, profit and loss statements, and cash flow statements of the Person or assets to be acquired, together with appropriate supporting details and a statement of underlying assumptions for the 1 year period following the date of the proposed Acquisition, on a quarter by quarter basis),
  - (d) [Intentionally Omitted],
- (e) Borrower has provided Agent with written notice of the proposed Acquisition at least 5 Business Days prior to the anticipated closing date of the proposed Acquisition, copies of the acquisition agreement and other material documents relative to the proposed Acquisition, which agreement and documents must be reasonably acceptable to Agent,
- (f) the assets being acquired (other than a de minimis amount of assets in relation to Bonower's and its Subsidiaries' total assets), or the Person whose Stock is being acquired, are useful in or engaged in, as applicable, the business of Bonower and its Subsidiaries or a business reasonably related thereto,
- (g) the assets being acquired (other than a de minimis amount of assets in relation to the assets being acquired) are located within the United States or Canada, or the Person whose Stock is being acquired is organized in a jurisdiction located within the United States or Canada,
- (h) the subject assets or Stock, as applicable, are being acquired directly by Borrower or a Subsidiary, and, in connection therewith, such Borrower or Subsidiary shall have provided such documents and instruments as requested are required by <u>Section 6.15</u> of the Agreement), and
- (i) the consideration payable in respect of the proposed Acquisition shall be composed solely of (i) common Stock of Borrower; (ii) proceeds of equity contributions made to Borrower by a Permitted Holder for the purpose of funding, in whole or in part, a proposed Acquisition, or (iii) Subordinated Debt permitted under this Agreement.

"Permitted Discretion" means a determination made in the exercise of reasonable (from the perspective of a secured lender) business judgment.

"Permitted Dispositions" means (a) sales or other dispositions of Equipment that is substantially worn, damaged, or obsolete in the ordinary course of business or no longer used or useful, (b) sales of Inventory to buyers in the ordinary course of business, (c) the use or transfer of money or Cash Equivalents in a manner that is not prohibited by the terms of this Agreement or the other Loan Documents, (d) the licensing, on a non-exclusive basis, of patents, trademarks, copyrights, and other intellectual property rights in the ordinary course of business, or the lapse of intellectual property that is immaterial or no longer used in or necessary to its business, (e) any Dispositions of any property by any Subsidiary to Borrower to the extent

(i) any resulting Investment constitutes a Permitted Investment, (ii) the foregoing constitutes a distribution permitted pursuant to Section 7.10, or (iii) the foregoing constitutes a transaction permitted by Section 7.3. (f) any Disposition or issuance by Borrower of its own equity interests to the extent that any such issuance does not result in a Change of Control, (g) any other Disposition of property of Borrower; provided that the aggregate consideration received during any fiscal year of Borrower for all such Dispositions shall not exceed \$1.000.000.

(h) Dispositions of property to the extent that (i) such property is exchanged for credit against the purchase price of similar property or (ii) the proceeds of such Disposition are promptly applied to the purchase price of such similar property, (i) Dispositions of accounts receivables in connection with the collection or compromise thereof in the ordinary course of business, (j) the granting of a Permitted Lien, and (k) Dispositions resulting from property loss events or takings and transfers of property that has suffered a property loss event or a taking (constituting a total loss or constructive total loss of such property) upon receipt of the Net Cash Proceeds of such property loss event or taking; provided however that in no event shall sales or other dispositions of Real Property Collateral be considered a Permitted Disposition.

"<u>Permitted Holders</u>" means the Persons identified on <u>Schedule P-2</u> hereto. "<u>Permitted Investments</u>" means (a) Investments in cash and Cash Equivalents,

- (b) Investments in negotiable instruments for collection, (c) advances made in connection with purchases of goods or services in the ordinary course of business, (d) loans and advances to officers, managers, directors and employees of any Loan Party in the ordinary course of business for reasonable and customary business-related travel, entertainment, relocation and analogous ordinary business purposes (including employee payroll advances), (e) Investments resulting from entering into any Hedge Agreements, (f) Investments received in settlement of amounts due to Borrower or any of its Subsidiaries effected in the ordinary course of business or owing to Borrower or any of its Subsidiaries as a result of Insolvency Proceedings involving an Account Debtor or upon the foreclosure or enforcement of any Lien in favor of Borrower or its Subsidiaries, (g) Permitted Acquisitions, (h) Investments by a Loan Party in another Loan Party,
- (i) other Investments in an aggregate outstanding amount not to exceed \$1,000,000 at any time,
- (j) any Investment owned by a Person at the time such Person is acquired and becomes a Subsidiary pursuant to a Permitted Acquisition; <u>provided</u> that such Investment was not made in connection with or in contemplation of such Permitted Acquisition, (k) any Investments made from casualty insurance proceeds in connection with the replacement, substitution, restoration or repair of assets on account of an insurable event to the extent such assets were covered by casualty insurance maintained by Borrower or a Subsidiary, (l) Investments consisting of purchases and acquisitions of inventory, supplies, material or equipment, in each such case in the ordinary course of business, and (m) any other Investments (other than an Investment that constitutes an Acquisition) made with the proceeds of equity interests issued by Borrower to a

Permitted Holder. For purposes of covenant compliance, the amount of any Permitted Investment shall be the amount actually invested less cash returns on such Investment.

"Permitted Liens" means (a) Liens held by Agent, (b) Liens for unpaid taxes that either (i) are not yet delinquent, or (ii) do not constitute an Event of Default hereunder and are the subject of Permitted Protests, (c) Liens set forth on Schedule P-1, (d) the interests of lessors under operating leases, (e) purchase money Liens or the interests of lessors under Capital Leases to the extent that such Liens or interests secure Permitted Purchase Money Indebtedness and so long as such Lien attaches only to the asset purchased or acquired and the proceeds thereof,

(f) Liens arising by operation of law in favor of warehousemen, landlords, carriers, mechanics, materialmen, laborers, or suppliers, incurred in the ordinary course of business and not in connection with the borrowing of money, and which Liens either (i) are for sums not yet delinquent by more than 30 days, or (ii) are the subject of Permitted Protests, (g) Liens on amounts deposited in connection with obtaining worker's compensation or other unemployment insurance, (h) Liens on amounts deposited in connection with the making or entering into of bids, tenders, or leases in the ordinary course of business and not in connection with the borrowing of money, (i) Liens on amounts deposited as security for surety or appeal bonds in connection with obtaining such bonds in the ordinary course of business, (j) Liens resulting from any judgment or award that is not an Event of Default hereunder, (k) with respect to any Real Property, easements, rights of way, and zoning restrictions that do not materially interfere with or impair the use or operation thereof and (l) Liens securing Subordinated Debt.

"Permitted Preferred Stock" means and refers to any Preferred Stock issued by Borrower (and not by one or more of its Subsidiaries) that is not Prohibited Preferred Stock.

"Permitted Protest" means the right of Borrower or any of its Subsidiaries to protest any Lien (other than any Lien that secures the Obligations), taxes (other than payroll taxes or taxes that are the subject of a United States federal tax lien), or rental payment, provided that (a) a reserve with respect to such obligation is established on the Books in such amount as is required under GAAP, (b) any such protest is instituted promptly and prosecuted diligently by Borrower or any of its Subsidiaries, as applicable, in good faith, and (c) Agent is satisfied that, while any such protest is pending, there will be no impairment of the enforceability, validity, or priority of any of the Agent's Liens.

"Permitted Purchase Money Indebtedness" means, as of any date of determination, Purchase Money Indebtedness that does not constitute Subordinated Debt and that is incurred after the Restatement Effective Date in an aggregate amount outstanding at any one time not in excess of \$2,000,000, plus any Purchase Money Indebtedness used to finance the purchase of Chillers.

"Preferred Stock" means, as applied to the Stock of any Person, the Stock of any class or classes (however designated) that is preferred with respect to the payment of dividends, or as to the distribution of assets upon any voluntary or involuntary liquidation or dissolution of such Person, over shares of Stock of any other class of such Person.

"Prime Rate" means, as of any time of determination, a fluctuating per annum rate of interest equal at all times to the rate of interest in effect for such day as announced from time

- 30 -137020431v13 to time by CNB as its "prime rate". Any change in the prime rate announced by CNB shall take effect at the opening of business on the day specified in the public announcement of such change.

"Prohibited Preferred Stock" means any Preferred Stock that by its terms is mandatorily redeemable or subject to any other payment obligation (including any obligation to pay dividends, other than dividends of shares of Preferred Stock of the same class and series payable in kind or dividends of shares of common stock) on or before a date that is less than 6 months after the later of the Revolver Maturity Date, the Delayed Draw Term Loan Maturity Date and the maturity date of any Incremental Term Loan, or, on or before the date that is less than 6 months after the later of then extant Revolver Maturity Date, the Delayed Draw Term Loan Maturity Date and the maturity date of any Incremental Term Loan, is redeemable at the option of the holder thereof for cash or assets or securities (other than distributions in kind of shares of Preferred Stock of the same class and series or of shares of common stock).

"Property A" has the meaning set forth in the recitals to this Agreement. "Property B" means Real Property at address 176 N. Commerce Way, Bethlehem,

PA 18017.

"Person" means natural persons, corporations, limited liability companies, limited

partnerships, general partnerships, limited liability partnerships, joint ventures, trusts, land trusts, business trusts, or other organizations, irrespective of whether they are legal entities, and governments and agencies and political subdivisions thereof.

"Projections" means Borrower's forecasted (a) balance sheets, (b) profit and loss statements, (c) cash flow statements, and (d) statement of projected capital expenditures, all prepared on a basis consistent with Borrower's historical financial statements, together with appropriate supporting details and a statement of underlying assumptions.

"Pro Rata Share" means, as of any date of determination:

- (a) with respect to a Lender's obligation to make Advances and receive payments of principal, interest, fees, costs, and expenses with respect thereto, (i) prior to the Revolver Commitments being terminated or reduced to zero, the percentage obtained by dividing
- (y) such Lender's Revolver Commitment, by (2) the aggregate Revolver Commitments of all Lenders, and (ii) from and after the time that the Revolver Commitments have been terminated or reduced to zero, the percentage obtained by dividing (y) the aggregate outstanding principal amount of such Lender's Advances by (2) the aggregate outstanding principal amount of such Lender's Advances by (2) the aggregate outstanding principal amount of such Lender's Advances by (2) the aggregate outstanding principal amount of such Lender's Advances by (2) the aggregate outstanding principal amount of such Lender's Advances by (2) the aggregate outstanding principal amount of such Lender's Advances by (2) the aggregate outstanding principal amount of such Lender's Advances by (2) the aggregate outstanding principal amount of such Lender's Advances by (2) the aggregate outstanding principal amount of such Lender's Advances by (2) the aggregate outstanding principal amount of such Lender's Advances by (2) the aggregate outstanding principal amount of such Lender's Advances by (2) the aggregate outstanding principal amount of such Lender's Advances by (2) the aggregate outstanding principal amount of such Lender's Advances by (2) the aggregate outstanding principal amount of such Lender's Advances by (2) the aggregate outstanding principal amount of such Lender's Advances by (2) the aggregate outstanding principal amount of such Lender's Advances by (3) the aggregate outstanding principal amount of such Lender's Advances by (3) the aggregate outstanding principal amount of such Lender's Advances by (3) the aggregate outstanding principal amount of such Lender's Advances by (3) the aggregate outstanding principal amount of such Lender's Advances by (3) the aggregate outstanding principal amount of such Lender's Advances by (3) the aggregate outstanding principal amount of such Lender's Advances by (3) the aggregate outstanding principal amount of such Lender's Advances by (3) the aggregate outstanding principal amount of such Lender
- (b) with respect to a Lender's obligation to participate in Letters of Credit, to reimburse the Issuing Lender, and to receive payments of fees with respect thereto, (i) prior to the Revolver Commitments being terminated or reduced to zero, the percentage obtained by dividing (y) such Lender's Revolver Commitment, by (z) the aggregate Revolver Commitments of all Lenders, and (ii) from and after the time that the Revolver Commitments bave been terminated or reduced to zero, the percentage obtained by dividing (y) the aggregate outstanding

principal amount of such Lender's Advances by (z) the aggregate outstanding principal amount of all Advances,

## (c) [Intentionally Omitted],

- (d) with respect to a Lender's obligation to make a Delayed Draw Term Loan and receive payments of interest, fees, and principal with respect thereto, (i) prior to the Delayed Draw Term Loan Commitment, and (B) the outstanding principal balance of such Lender's Delayed Draw Term Loans, by (2) the sum of (A) the aggregate amount of all Lenders' remaining Delayed Draw Term Loan Commitment, and (B) the aggregate outstanding principal balance of all Delayed Draw Term Loans, and (ii) from and after the Delayed Draw Term Loan Commitment Expiration Date, the percentage obtained by dividing (y) the outstanding principal balance of such Lender's Delayed Draw Term Loans by (z) the aggregate outstanding principal balance of such Lender's Delayed Draw Term Loans by (z) the aggregate outstanding principal balance of all Delayed Draw Term Loans by (z) the aggregate outstanding principal balance of all Delayed Draw Term Loans by (z) the aggregate outstanding principal balance of all Delayed Draw Term Loans by (z) the aggregate outstanding principal balance of all Delayed Draw Term Loans by (z) the aggregate outstanding principal balance of all Delayed Draw Term Loans by (z) the aggregate outstanding principal balance of all Delayed Draw Term Loans by (z) the aggregate outstanding principal balance of all Delayed Draw Term Loans by (z) the aggregate outstanding principal balance of all Delayed Draw Term Loans by (z) the aggregate outstanding principal balance of all Delayed Draw Term Loans by (z) the aggregate outstanding principal balance of all Delayed Draw Term Loans by (z) the aggregate outstanding principal balance of all Delayed Draw Term Loans by (z) the aggregate outstanding principal balance of all Delayed Draw Term Loans by (z) the aggregate outstanding principal balance of all Delayed Draw Term Loans by (z) the aggregate outstanding principal balance of all Delayed Draw Term Loans by (z) the aggregate outstanding principal balance of all Delayed Draw Term Loans by (z) the aggregate outstanding principal balance of a
- (e) with respect to all other matters as to a particular Lender (including the indemnification obligations arising under Section 16.7), the percentage obtained by dividing (i) the sum of (A)(x) prior to the termination or the reduction to zero of the Revolver Commitments, such Lender's Revolver Commitment, or (y) from and after the termination or the reduction to zero of the Revolver Commitments, the sum of the outstanding principal amount of such Lender's Advances plus such Lender's ratable portion of the Risk Participation Liability with respect to outstanding Letters of Credit, plus (B) (x) prior to the Delayed Draw Term Loan Commitment Expiration Date, the sum of (f) such Lender's Delayed Draw Term Loan Commitment after the Delayed Draw Term Loan Commitment Expiration Date, the outstanding principal balance of such Lender's Delayed Draw Term Loans, by (ii) the sum of (A) (x) prior to the termination or the reduction to zero of the Revolver Commitments, the aggregate amount of Revolver Commitments of all Lenders, or (y) from and after the termination or the reduction to zero of the Revolver Commitments, the sum of outstanding principal amount of all Advances plus the aggregate amount of the Risk Participation Liability with respect to all outstanding Letters of Credit, plus (B) (x) prior to the Delayed Draw Term Loan Commitment Expiration Date, the aggregate amount of all Lenders' remaining Delayed Draw Term Loan Commitment Expiration Date, the aggregate outstanding principal balance of all Delayed Draw Term Loans, or (y) from and after the Delayed Draw Term Loan Commitment Expiration Date, the aggregate outstanding principal balance of all Delayed Draw Term Loans.

"Purchase Money Indebtedness" means Indebtedness (other than the Obligations, but including Capitalized Lease Obligations), incurred at the time of, or within 90 days after, the acquisition of any fixed assets for the purpose of financing all or any part of the acquisition cost thereof.

"Qualified ECP Guarantor" shall mean, in respect of any Hedge Obligation, each Guarantor with total assets exceeding \$10,000,000 at the time the relevant Guaranty or grant of the relevant security interest becomes effective with respect to such Hedge Obligation or such other person as constitutes an "eligible contract participant" under the Commodity Exchange Act or any regulations promulgated thereunder and can cause another person to qualify as an

"eligible contract participant" at such time by entering into a keepwell under Section 1a(18)(A)(v)(II) of the Commodity Exchange Act.

"Real Property" means any estates or interests in real property now owned or hereafter acquired by Borrower or any of its Subsidiaries and the improvements thereto.

"Real Property Collateral" means the Real Property identified on Schedule R-1 and any Real Property hereafter acquired by Borrower or any of its Subsidiaries; provided that Real Property Collateral shall not include Excluded Property, so long as such Real Property Collateral constitutes Excluded Property.

"Recipient" means (a) the Agent, (b) any Lender or (c) any Issuing Lender, as

applicable.

"Record" means information that is inscribed on a tangible medium or which is stored in an electronic or other medium and is retrievable in perceivable form. "Register" has the meaning set forth in Section

<u>14.1(c)</u>.

"Remedial Action" means all actions taken to (a) clean up, remove, remediate, contain, treat, monitor, assess, evaluate, or in any way address Hazardous Materials in the indoor or outdoor environment, (b) prevent or minimize a release or threatened release of Hazardous Materials so they do not migrate or endanger or threaten to endanger public health or welfare or the indoor or outdoor environment, (c) restore or reclaim natural resources or the environment, (d) perform any pre-remedial studies, investigations, or post-remedial operation and maintenance activities, or (e) conduct any other actions with respect to Hazardous Materials

authorized by Environmental Laws.

"Replacement Lender" has the meaning set forth in Section 15.2(a). "Report" has the meaning set forth in Section 16.17.

"Required Lenders" means, at any time, (a) if there are more than two Lenders who are not Affiliates of one another, at least two Lenders who are not Affiliates of one another, at least two Lenders who are not Affiliates of one another whose aggregate Pro Rata Shares (calculated under clause (e) of the definition of Pro Rata Shares) exceed 50%, and (b) otherwise, all of the Lenders.

"Reserve Percentage" means, on any day, for any Lender, the maximum percentage prescribed by the Board of Governors of the Federal Reserve System (or any successor Governmental Authority) for determining the reserve requirements (including any basic, supplemental, marginal, or emergency reserves) that are in effect on such date with respect to eurocurrency funding (currently referred to as "eurocurrency liabilities") of that Lender, but so long as such Lender is not required or directed under applicable regulations to maintain such reserves, the Reserve Percentage shall be zero.

"Resolution Authority," means any body which has authority to exercise any Write-down and Conversion Powers.

"Restatement Effective Date" means May 15, 2019.

"Revolver Commitment" means, with respect to each Lender, its Revolver Commitment, and, with respect to all Lenders, their Revolver Commitments, in each case as such Dollar amounts are set forth beside such Lender's name under the applicable heading on Schedule C-1 or in the Assignment and Acceptance pursuant to which such Lender became a Lender hereunder, as such amounts may be reduced or increased from time to time pursuant to assignments made in accordance with the provisions of Section 14.1.

"Revolver Maturity Date" means May 15, 2024.

"Revolver Obligations" all Obligations in respect of or relating to the Advances, Letters of Credit and Bank Product Obligations (including principal, interest, premiums, if any, fees, costs, and expenses (including Lender Group Expenses) in respect thereof).

"Revolver Usage" means, as of any date of determination, the sum of (a) the amount of outstanding Advances, plus (b) the amount of the Letter of Credit

Usage.

"Revolving Loan Exposure" means, with respect to any Lender, as of any date of determination (a) prior to the termination of the Revolver Commitments, the amount of such Lender's Revolver Commitment, and (b) after the termination of the Revolver Commitments, the aggregate outstanding principal amount of the Advances of such Lender.

"Risk Participation Liability." means, as to each Letter of Credit, all reimbursement obligations of Borrower to the Issuing Lender with respect to such Letter of Credit, consisting of (a) the amount available to be drawn or which may become available to be drawn, (b) all amounts that have been paid by the Issuing Lender with respect thereto to the extent not reimbursed by Borrower, whether by the making of an Advance or otherwise, and (c) all accrued and unpaid interest, fees, and expenses payable with respect thereto.

"Sanctioned Entity" means (a) a country or territory or a government of a country or territory, (b) an agency of the government of a country or territory, (c) an organization directly or indirectly controlled by a country or territory or its government, or (d) a Person resident in or determined to be resident in a country or territory, in each case of clauses (a) through (d) that is a target of a country or territory sanctions program, including a target of any country or territory sanctions program administered and enforced by OFAC.

"Sanctioned Person" means, at any time (a) any Person named on the list of Specially Designated Nationals and Blocked Persons maintained by OFAC or any other sanctions-related list maintained by any United States Governmental Authority, (b) a Person or legal entity that is a target of sanctions, (c) any Person operating, organized or resident in a Sanctioned Entity, or (d) any Person directly or indirectly owned or controlled (individually or in the aggregate) by or acting on behalf of any such Person or Persons described in clauses (a) through (c) above.

"SEC" means the United States Securities and Exchange Commission and any successor thereto.

"Securities Account" means a securities account (as that term is defined in the

Code).

"Solvent" means, with respect to any Person on a particular date, that, at fair

valuations, the sum of such Person's assets is greater than all of such Person's debts.

"Specified Loan Party" means any Loan Party that is not then a Qualified ECP

Guarantor.

"Stock" means all shares, options, warrants, interests, participations, or other equivalents (regardless of how designated) of or in a Person, whether voting or nonvoting, including common stock, preferred stock, or any other "equity security" (as such term is defined in Rule 3a11-1 of the General Rules and Regulations promulgated by the SEC under the Exchange Act).

"Stock Pledge Agreement" means that certain Amended and Restated Stock Pledge Agreement, dated as of November 13, 2014, executed and delivered by Borrower to Agent with respect to the pledge of the Stock owned by Borrower, as supplemented by (i) that certain Joinder Agreement, dated as of October 5, 2016, by and among FP Foods, Borrower and the Agent, (ii) that certain Pledged Interests Addendum with respect to FP Foods, dated as of October 5, 2016, by Borrower, (iii) that certain Pledged Interests Addendum with respect to Freshpet Europe LTD, a United Kingdom private limited company, dated as of March 8, 2017, by Borrower and (iv) that certain Pledged Interests Addendum with respect to the Dutch Subsidiary, dated as of May 15, 2019, by Borrower, and as the same may be otherwise amended, restated, supplemented, or otherwise modified from time to time in accordance to the terms thereof and hereof.

"Subordinated Debt" means Indebtedness of Borrower that is on terms and conditions (including payment terms, interest rates, covenants, remedies, defaults and other material terms) satisfactory to the Agent and which (a) has been expressly subordinated in right of payment to all Obligations by the execution and delivery of a subordination agreement, in form and substance satisfactory to Agent and (b) if such Indebtedness is secured by a Lien, such Lien is expressly subordinated to the Liens granted to Lender by the execution and delivery of a subordination agreement, in form and substance satisfactory to Agent.

"Subsidiary." of a Person means a corporation, partnership, limited liability company, or other entity in which that Person directly or indirectly owns or controls the shares of Stock having ordinary voting power to elect a majority of the board of directors (or appoint other comparable managers) of such corporation, partnership, limited liability company, or other entity.

"Supporting Obligation" means a letter-of-credit right or secondary obligation that supports the payment or performance of an Account, chattel paper, document, General Intangible, instrument, or Investment Property.

"<u>Taxes</u>" means, any taxes, levies, imposts, duties, fees, assessments or other charges of whatever nature now or hereafter imposed by any jurisdiction or by any political

- 35 -137020431v13 subdivision or taxing authority thereof or therein with respect to such payments and all interest, penalties or similar liabilities with respect thereto.

"Tax Group" has the meaning set forth in Section 7.10.

"<u>Trademark Security Agreement</u>" means that certain Third Amended and Restated Trademark Security Agreement, dated as of May 15, 2019, executed and delivered by Borrower and Agent, and any other trademark security agreement from time to time executed by Borrower and Agent, in form and substance reasonably satisfactory to Agent, in each case, as the same may be amended, restated, supplemented, or otherwise modified from time to time in accordance to the terms thereof and hereof.

"TTM EBITDA" means, as of any date of determination, Adjusted EBITDA of Borrower determined on a consolidated basis in accordance with GAAP, for the 12 month period most recently ended.

"United States" means the United States of America.

"U.S. Person" means any Person that is a "United States person" as defined in Section 7701(a)(30) of the Code.

"Voidable Transfer" has the meaning set forth in Section 17.6.

"Write-down and Conversion Powers" means, with respect to any EEA Resolution Authority, the write-down and conversion powers of such EEA Resolution Authority from time to time under the Bail-In Legislation for the applicable EEA Member Country, which write-down and conversion powers are described in the EU Bail-In Legislation Schedule.

1.2 Accounting Terms. All accounting terms not specifically defined herein shall be construed in accordance with GAAP; provided, that if Borrower notifies Agent that Borrower requests an amendment to any provision hereof to eliminate the effect of any Accounting Change occurring after the Restatement Effective Date or in the application thereof on the operation of such provision (or if Agent notifies Borrower that the Required Lenders request an amendment to any provision hereof for such purpose), regardless of whether any such notice is given before or after such Accounting Change or in the application thereof, then Agent and Borrower agree that they will negotiate in good faith amendments to the provisions of this Agreement that are directly affected by such Accounting Change with the intent of having the respective positions as of the date of this Agreement and, until any such amendments have been agreed upon and agreed to by the Required Lenders, the provisions in this Agreement shall be calculated as if no such Accounting Change had occurred. When used herein, the term "financial statements" shall include the notes and schedules thereto. Whenever the term "Borrower" is used in respect of a financial covenant or a related definition, it shall be understood to mean Borrower and its Subsidiaries on a consolidated basis unless the context clearly requires otherwise. Notwithstanding anything to the contrary contained herein, (a) all financial statements delivered hereunder shall be prepared, and all financial covenants contained herein shall be calculated, without giving effect to any election under the Statement of Financial Accounting Standards No. 159 (or any similar accounting principle) permitting a Person to value

its financial liabilities or Indebtedness at the fair value thereof, and (b) the term "unqualified opinion" as used herein to refer to opinions or reports provided by accountants shall mean an opinion or report that is (i) unqualified, and (ii) does not include any explanation, supplemental comment, or other comment concerning the ability of the applicable Person to continue as a going concern or concerning the scope of the audit.

- 1.3 <u>Code</u>. Any terms used in this Agreement that are defined in the Code shall be construed and defined as set forth in the Code unless otherwise defined herein; <u>provided</u>, <u>however</u>, that to the extent that the Code is used to define any term herein and such term is defined differently in different Articles of the Code, the definition of such term contained in Article 9 shall govern.
- 1.4 Construction. Unless the context of this Agreement or any other Loan Document clearly requires otherwise, references to the plural include the singular, references to the singular include the plural, the terms "includes" and "including" are not limiting, and the term "or" has, except where otherwise indicated, the inclusive meaning represented by the phrase "and/or." The words "hereof," "herein," "hereby," "hereunder," and similar terms in this Agreement or any other Loan Document refer to this Agreement or such other Loan Document, as the case may be, as a whole and not to any particular provision of this Agreement or such other Loan Document, as the case may be, as a whole and not to any particular provision of this Agreement or such other Loan Document, as the case may be. Section, subsection, clause, schedule, and exhibit references herein are to this Agreement unless otherwise specified. The words "asset" and "property" shall be construed to have the same meaning and effect and to refer to any and all tangible and intangible assets and properties. Any reference in this Agreement or in the other Loan Documents to any agreement, instrument, or document shall include all alterations, amendments, changes, extensions, modifications, renewals, replacements, substitutions, joinders, and supplements set forth herein). Any reference herein to the satisfaction or repayment in full of the Obligations shall mean the repayment in full in cash (or cash collateralization in accordance with the terms hereof) of all Obligations other than contingent indemnification Obligations as to which no claim as been asserted and other than any Bank Product Obligations that, at such time, are allowed by the applicable Bank Product Provider to remain outstanding and are not required to be repaid or cash collateralized pursuant to the provisions of this Agreement, and the termination of all Commitments of the Lenders. Any reference herein to any Person shall be construed to include such Person's successors and assigns. Any requireme

5 Schedules and Exhibits. All of the schedules and exhibits attached to this Agreement shall be deemed incorporated herein by reference.

### 2. LOAN AND TERMS OF PAYMENT.

### 2.1 Revolver Advances.

- (a) Effective as of the Restatement Effective Date, all "Advances" (as defined in the Existing Loan Agreement) and any "Letters of Credit" (as defined in the Existing Loan Agreement) and any "Letters of Credit" (as defined in the Existing Loan Agreement) that are "Advances" (as defined in the Existing Loan Agreement)
- (b) Subject to the terms and conditions of this Agreement, and cluring the term of this Agreement, each Lender with a Revolver Commitment agrees (severally, not jointly or jointly and severally) to make additional advances ("Advances") to Bouwer in an amount at any one time outstanding not to exceed such Lender's Pro Rata Share of an amount equal to the Maximum Revolver Amount less the Letter of Gredit Usage.
- (c) The Lendes with Revolver Commitments shall have no obligation to make additional Advances have under to the extent such additional Advances would cause the Revolver Usage to exceed the Maximum Revolver Amount.
  - (d) Amounts bourowed pursuant to this <u>Section 2.1</u> may be repaid and, subject to the terms and conditions of this Agreement, rebourowed at any time during the term of this Agreement.

# 2.2 <u>Delayed Draw Term Loan</u>.

(a) <u>Delyed Draw Term Loan.</u> Subject to the terms and conditions of this Agreement, each Lender with a Delayed Draw Term Loan Commitment agrees (severally, not jointly or jointly and severally) to make term loans (each a "<u>Delayed Draw Term Loans</u>" and collectively, the "<u>Delayed Draw Term Loans</u>") to Borrower from time to time from the Restatement Effective Date until the Delayed Draw Term Loan Commitment Expiration Date, or until the eatien reduction of its Delayed Draw Term Loan Commitment The aggregate principal amount not to exceed the unused portion of such Loader's Delayed Draw Term Loan Commitment. The aggregate principal amount of the Delayed Draw Term Loan Commitment of each Lender shall (x) automatically and permanently be reduced to be extent that such Lender makes a Delayed Draw Term Loan Do Browse; and (y) automatically and permanently be reduced to zero on the Delayed Draw Term Loan Commitment Expiration Date. Each Delayed Draw Term Loan Commitment Expiration Date. Each Delayed Draw Term Loan Commitment Delayed Draw Term Loan Commitment Delayed Draw Term Loan Commitment Draw Term Loan Commitment Delayed Draw Term Loan Commitment Draw Term Loan Commitment Delayed Draw Term Loan Commitment Draw Term Loan Commitment Draw Term Loan Commitment Delayed Draw Term Loan Commitment Delayed Draw Term Loan Commitment Dr

\$2,500,000, and in integral multiples of \$500,000 in excess thereof. Any principal amount of the Delayed Draw Term Loans that is repaid or prepaid may not be reborrowed.

- 38 -137020431v13 (b) <u>Repayment.</u> On the Delayed Draw Term Loan Commitment Expiration Date, the aggregate principal amount of all outstanding Delayed Draw Term Loans shall be convented to a single term loan (the "<u>Combined Delayed Draw Term Loan</u>") and shall the reafter be repaid in 12 equal consecutive quarterly installments, each in an amount equal to 1/28 th of the Combined Delayed Draw Term Loan, which installment payments shall be due and payable on the last day of each fiscal quarter, commencing on June 30, 2021. The outstanding unpaid principal balance and all accounted and unpaid interest under the Delayed Draw Term Loan shall be due and payable on the Delayed Draw Term Loan Shall be due and payable on the Delayed Draw Term Loan Shall be due and payable on the Delayed Draw Term Loan Shall be due and payable on the Delayed Draw Term Loan Shall be due and payable on the Delayed Draw Term Loan Shall constitute Obligations.

### 2.3 Borrowing Procedures.

(a) Procedure for Borrowing. Each Borrowing shall be made by an inevocable written request in substantially the form of Exhibit N-1 hereto (a "Notice of Borrowing") by an Authorized Person delivered to Agent. In the case of an equest for a Base. Rate Loan, such notice must be received by Agent no later than 10:00 a.m. (California time) on the Business Day prior to the date that is the requested Funding Date, in the case of a request for a LIBOOR Rate Loan, no later than 10:00 a.m. (California time) at least 38 business Days prior to the date that is the requested Funding Date, in the case of a request for a LIBOOR Rate Loan, no later than 10:00 a.m. (California time) at least 38 business Days prior to the date that is the requested Funding Date, in the case of a request for a LIBOOR Rate Loan, no later than 10:00 a.m. (California time) and in the case of a request Date of the requested Funding Date, which shall be a Business Days. At Agent's election, in lieu of delivering the above-described written request, any Authorized Person may give Agent telephonic notice, but the failure to provide such written confirmation shall not affect the validity of the request.

(b) [Intentionally Omitted.]

(c) Making of Loans.

(i) Promptly after receipt of a request for a Borrowing pursuant to Section 23(a), Agent shall notify the Lenders, not later than 1:00 p.m. (California time) on the Business Day immediately preceding the Funding Date applicable thereto, by telecopy, telephone, or other similar form of transmission, of the requested Borrowing. Each Lender shall make the amount of such Lender's Pro Rata Share of the requested Borrowing available to Agent in immediately available funds, to Agent's Account, not later than 1:000 a.m. (California time) on the Funding Date applicable thereto. After Agent's receipt of the proceeds of such Advances (or the Delayed Draw Term Loan), Agent shall make the proceeds thereof available to Borrower's Designated Account; provided however, that, subject to the provisions of Section 23(i), Agent shall not request any Lender to make, and no Lender shall have the obligation to make, any Advance (or its portion of the Delayed Draw Term Loan) if Agent shall have a actual knowledge that (1) one or more of the applicable conditions precedent set for in Section 3 will not be satisfied on the requested Funding Date for the applicable

- 39 -137020431v13 Borrowing unless such condition has been waived, or (2) the requested Borrowing would exceed the Availability on such Funding Date.

(ii) Unless Agent receives notice from a Lender on or prior to the Restatement Effective Date or, with respect to any Borrowing after the Restatement Effective Date, prior to 9:00 am (California time) on the date of such Borrowing, that such Lender will not make available as and when required hereunder to Agent for the account of Borrower the amount of that Lender's Pro Rata Share of the Borrowing, Agent may assume that each Lender has made or will make such amount available to Agent in immediately available innos on the Funding Date and Agent may (but shall not be so required), in reliance upon such assumption, make available to Borrower on such date a conesponding amount. If and to the extent any Lender's hall not have made its full amount available to Agent in immediately available funds and Agent in such circumstances have made available to Borrowing and Funding Date make such amount available to Agent, together with interest at the Defaulting Lender's Rate for each day during such period. A notice submitted by Agent to any Lender with interest and amounts owing under this subsection shall be conclusive, absent manifest error. If such amount is somed available, such payment to Agent shall constitute such Lender's Advance (or portion of the Delayed Draw Term Loan) on the date of Borrowing for all purposes of this Agreement. If such amount is not made available to Agent on the Business Day following the Funding Date, Agent will notify Borrower of such failure to fund and, upon demand by Agent, Borrower shall pay such amount to Agent for Agent's account, together with interest the each day elapsed since the date of such Borrowing, at anate per annume qual to the interest rate applicable at the time to the Advances (or portion of the Delayed Draw Term Loan) on such Funding Date shall not lender of any obligation hereunder to make any Advance (or portion of the Delayed Draw Term Loan) to the Delayed Draw Term Loan) to be made by such other Lender on any Funding Date.

### (d) Defaulting Lenders

(i) Notwithstanding the provisions of Section 24(b). Agent shall not be obligated to transfer to a Defaulting Lender any payments made by Borrower to Agent for the Defaulting Lender's benefit or any proceeds of Collateral that would otherwise be remitted hereunder to the Defaulting Lender, and, in the absence of such transfer to the Defaulting Lender, Agent shall transfer any such payments (A) first, to Issuing Lender, to the extent of the portion of a Letter of Credit Disbursement that was required to be, but was not, paid by the Defaulting Lender, (B) second, to each Non-Defaulting Lender ratably in accordance with their Commitments (but, in each case, only to the extent that such Defaulting Lender's portion of an Advance or Delayed Draw Term Loan (or other funding obligation) was furnled by such other Non-Defaulting Lender), (C) third, to as usperse account maintained by Agent, the proceeds of which shall be retained by Agent and may be made available to be re-advanced to or for the benefit of Borrower (upon the request of Borrower and subject to the conditions set forth in Section 3.2) as if such Defaulting Lender in accordance with

- 40 -1370<u>2</u>0431v13 tier (J) of Section 2.4(b)(iii). Subject to the foregoing, Agent may hold and, in its Permitted Discretion, re-lend to Borrower for the account of such Defaulting Lender the amount of all such payments received and retained by Agent for the account of such Defaulting Lender. Solely for the purposes of voting or consenting to matters with respect to the Loan Documents (including the calculation of Pro Rata Share in connection therewith) and for the purpose of calculating the fee payable under Section 2.11(b), such Defaulting Lender shall be deemed not to be a "Lender" and such Lender's Commitment shall be deemed to be zero. This Section shall remain effective with respect to such Lender until the earlier of (y) the date on which all of the Non-Defaulting Lenders, Agent, Issuing Lender, and Borrower shall have waived, in writing, the application of this Section 2.3(d) to such Defaulting Lender, or (z) the date on which such Defaulting Lender makes payment of all amounts that it was obligated to fund hereunder, pays to Agent all amounts owing by Defaulting Lender in respect of the amounts that it was obligated to fund hereunder, and, if requested by Agent, provides adequate assurance of its ability to perform its future obligations hereunder (on which earlier date, so long as no Event of Default has occurred and is continuing, any remaining cash collateral held by Agent pursuant to Section 2.3(d) (ii) shall be released to Borrower). The operation of this Section shall not be construed to increase or otherwise affect the Commitment of any Lender, to relieve or excuse the performance by such Defaulting Lender or any other Lender of its duties and obligations hereunder, or to relieve or excuse the performance by Borrower of its duties and obligations hereunder to Agent or to the Lenders other than such Defaulting Lender. Any failure by a Defaulting Lender to fund amounts that it was obligated to fund hereunder shall constitute a material breach by such Defaulting Lender of this Agreement and shall entitle Borrower, at its option, upon written notice to Agent, to arrange for a substitute Lender to assume the Commitment of such Defaulting Lender, such substitute Lender to be reasonably acceptable to Agent. In connection with the arrangement of such a substitute Lender, the Defaulting Lender shall have no right to refuse to be replaced hereunder, and agrees to execute and deliver a completed form of Assignment and Acceptance in favor of the substitute Lender (and agrees that it shall be deemed to have executed and delivered such document if it fails to do so) subject only to being repaid its share of the outstanding Obligations (other than Bank Product Obligations, but including (1) all interest, fees, and other amounts that may be due and payable in respect thereof, and (2) an assumption of its Pro Rata Share of the Risk Participation Liability) without any premium or penalty of any kind whatsoever; provided, however, that any such assumption of the Commitment of such Defaulting Lender shall not be deemed to constitute a waiver of any of the Lender Groups' or Borrower's rights or remedies against any such Defaulting Lender arising out of or in relation to such failure to fund. In the event of a direct conflict between the priority provisions of this Section 2.3(d) and any other provision contained in this Agreement or any other Loan Document, it is the intention of the parties hereto that such provisions be read together and construed, to the fullest extent possible, to be in concert with each other. In the event of any actual, irreconcilable conflict that cannot be resolved as aforesaid, the terms and provisions of this Section 2.3(d) shall control and

(ii) If any Letter of Credit is outstanding at the time that a Lender becomes a Defaulting Lender then:

(A) such Defaulting Lender's Letter of Credit Exposure shall be reallocated among the Non-Defaulting Lenders in accordance with their respective Pro Rata Shares but only to the extent the sum of all Non-Defaulting Lenders' Revolving Loan Exposures

- 41 -137020431v13 plus such Defaulting Lender's Letter of Credit Exposure does not exceed the total of all Non- Defaulting Lenders' Revolver Commitments;

- if the reallocation described in clause (A) above cannot, or can only partially, be effected, Borrower shall within one Business Day following notice by the Agent (or a longer period as Agent may reasonably agree) cash collateralize such Defaulting Lender's Letter of Credit Exposure (after giving effect to any partial reallocation pursuant to clause (A) above), pursuant to a cash collateral agreement to be entered into in form and substance reasonably satisfactory to the Agent, for so long as such Letter of Credit Exposure is outstanding; provided, that Borrower shall not be obligated to cash collateralize any Defaulting Lender's Letter of Credit Exposure if such Defaulting Lender is also the Issuing Lender;
- (C) if Borrower cash collateralizes any portion of such Defaulting Lender's Letter of Credit Exposure pursuant to this Section 2.3(d)(ii), Borrower shall not be required to pay any Letter of Credit Fees to Agent for the account of such Defaulting Lender pursuant to Section 2.6(b) with respect to such cash collateralized portion of such Defaulting Lender's Letter of Credit Exposure during the period such Letter of Credit Exposure is cash collateralized;
- (D) to the extent the Letter of Credit Exposure of the Non-Defaulting Lenders is reallocated pursuant to this Section 2.3(d)(ii), then the Letter of Credit Fees payable to the Non-Defaulting Lenders pursuant to Section 2.6(b) shall be adjusted in accordance with such Non-Defaulting Lenders' Letter of Credit Exposure:
- (E) to the extent any Defaulting Lender's Letter of Credit Exposure is neither cash collateralized nor reallocated pursuant to this Section 2.3(d)(ii), then, without prejudice to any rights or remedies of the Issuing Lender or any Lender hereunder, all Letter of Credit Fees that would have otherwise been payable to such Defaulting Lender under Section 2.6(b) with respect to such portion of such Letter of Credit Exposure shall instead be payable to the Issuing Lender until such portion of such Defaulting Lender's Letter of Credit Exposure is cash collateralized or reallocated;
- (F) so long as any Lender is a Defaulting Lender the Issuing Lender shall not be required to issue, amend, or increase any Letter of Credit, in each case, to the extent (x) the Defaulting Lender's Pro Rata Share of such Letter of Credit cannot be reallocated pursuant to this Section 2.3(d)(ii) or (y) the Issuing Lender has not otherwise entered into arrangements reasonably satisfactory to Issuing Lender and Borrower to eliminate or Issuing Lender's risk with respect to the Defaulting Lender's participation in Letters of Credit; and
- (G) Agent may release any cash collateral provided by Borrower pursuant to this Section 2.3(d)(ii) to the Issuing Lender and the Issuing Lender may apply any such cash collateral to the payment of such Defaulting Lender's Pro Rata Share of any Letter of Credit Disbursement that is not reimbursed by Borrower pursuant to Section 2.12(a).
  - (e) [Intentionally Omitted.]
  - (f) Agent Advances.

- 42 -137020431v13

(i)	Agent hereby is authorized by Borrower and the Lenders, from time to time in Agent's sole discretion, (1) after the occurrence and during the
continuance of a Default or an Event of Default, or (2) at any time that any of the other applicable	e conditions precedent set forth in <u>Section 3</u> have not been satisfied, to make Advances to Bonower on behalf of the Lenders that Agent, in its Permitted
Discretion deems necessary or desirable (A) to preserve or protect the Collateral, or any portion t	hereof, (B) to enhance the likelihood of repayment of the Obligations (other than the Bank Product Obligations), or (C) to pay any other amount
chargeable to Borrower pursuant to the terms of this Agreement, including Lender Group Expe	enses and the costs, fees, and expenses described in <u>Section 10</u> (any of the Advances described in this <u>Section 23</u> (f) shall be referred to as " <u>Agent</u>
Advances"), Each Agent Advance shall be deemed to be an Advance hereunder, except that no	o such Agent Advance shall be eligible to be a LIBOR Rate Loan and all payments thereon shall be payable to Agent solely for its own account.

(ii) The Agent Advances shall be repayable on demand, secured by the Agent's Liens granted to Agent under the Loan Documents, constitute Obligations hereunder, and bear interest at the rate applicable from time to time to Advances that are Base Rate Loans.

(iii) Each Lender with a Revolver Commitment shall be obligated to settle with Agent for the amount of such Lender's Pro Rata Share of any Agent Advances made as permitted under this <u>Section 2.3</u>(f).

# (g) [Intentionally Omitted.]

- (h) **Notation.** In accordance with <u>Section 14.1(c)</u>, the Agent, as anon-fiduciary agent for Borrower, shall maintain a Registershowing the principal amount of the Advances, any portion of the Delayed Draw Term Loan and any portion of the Incremental Term Loans, owing to each Lender, including the Agent Advances, and the interess therein of each Lender, from time to time and such Registershall, absent manifest error; conclusively be presumed to be consect and accurate.
- (i) Lenders' Failure to Perform. All Advances (other than Agent Advances) and Delayed Draw Term Loans shall be made by the Lender scontemporaneously and in accordance with their Pro Rata Shares. It is understood that (i) no Lendershall be responsible for any failure by any other Lender to perform its obligations have index any Advance or Delayed Draw Term Loans (or other extension of credit) have index nor shall any Commitment of any Lender be increased or decreased as a result of any failure by any other Lender to perform its obligations have under; and (ii) no failure by any Lender to perform its obligations have under.
- (i) Optional Overadvances. Any contrary provision of this Agreement not with standing, the Lendes hereby authorize Agent and Agent may, but is not obligated to, knowingly and intentionally, continue to make Advances to Borrowernot with standing that an Overadvance exists or thereby would be overled, so long as after giving effect to such Advances, the outstanding Revolver Usage (except for and excluding amounts charged to the Lond Account for interest, fees, or Lender Group Expenses) does not exceed the Maximum Revolver Amount. The foregoing provisions are for the exclusive benefit of Agent and the Lendes and are not

intended to benefit Borrower in any way. The Advances that are made pursuant to this <u>Section 2.3(i)</u> shall be subject to the same terms and conditions as any other Advance, except that they shall not be eligible for the LIBOR Option and the rate of interest applicable thereto shall be the rate applicable to Advances that are Base Rate Loans under <u>Section 2.6(c)</u> hereof without regard to the presence or absence of a Default or Event of Default.

- (i) In the event Agent obtains actual knowledge that the Revolver Usage exceeds the amounts permitted by the preceding paragraph, regardless of the amount of, or reason for, such excess, Agent shall notify the Lenders as soon as practicable (and prior to making any (or any additional) intentional Overadvances (except for and excluding amounts charged to the Loan Account for interest, fees, or Lender Group Expenses) unless Agent determines that prior notice would result in imminent harm to the Collateral or its value), and the Lenders with Revolver Commitments thereupon shall, together with Agent, jointly determine the terms of arrangements that shall be implemented with Borrower intended to reduce, within a reasonable time, the outstanding principal amount of the Advances to Borrower to an amount permitted by the preceding paragraph. In the event Agent or any Lender disagrees over the terms of reduction or repayment of any Overadvance, the terms of reduction or repayment thereof shall be implemented accounting to the determination of the Required Lenders.
- (ii) Each Lender with a Revolver Commitment shall be obligated to settle with Agent for the amount of such Lender's Pro Rata Share of any unintentional Overactvances by Agent reported to such Lender, any intentional Overactvances made as permitted under this Section 23(i), and any Overactvances resulting from the charging to the Loan Account of interest, fees, or Lender Group Expenses.

### 2.4 Payments.

# (a) Payments by Borrower.

- (i) Except as otherwise expressly provided herein, all payments by Borrower shall be made to Agent's Account for the account of the Lender Group and shall be made in immediately available funds, no later than 11:00 am. (California time) on the date specified herein. Any payment received by Agent later than 11:00 am. (California time) shall be deemed to have been received (unless Agent, in its sole discretion, elects to credit it on the date received) on the following Business Day.
- (ii) Unless Agentreceives notice from Borrower prior to the date on which any payment is due to the Lenders that Borrower will not make such payment in full to Agent on such date in immediately available funds and Agent may (but shall not be so required), in reliance upon such assumption, distribute to each Lender on such due date an amount equal to the amount then due such Lender. If and to the extent Borrower does not make such payment in full to Agent on the date when due, each Lender severally shall repay to Agent on demand such amount distributed to such Lender, together with interest thereon at the Defaulting Lender Rate for each day from the date such amount is distributed to such Lender until the date repaid.

Apportionment and Application. (b)  $So long as no Application \ Event has occurred and is continuing and except as otherwise provided herein with respect to Defaulting \ Lenders, all \ Lenders, all \ Lenders in the lenders of the lende$ principal and interest payments received by Agent shall be apportioned ratably among the Lenders (according to the unpaid principal balance of the Obligations to which such payments relate held by each Lender) and all payments of fees and expenses received by Agent (other than fees or expenses that are for Agent's separate account or for the separate account of fissuing Lender) shall be apportioned ratably among the Lenders having a Pro Rata Share of the type of Commitment or Obligation to which a particular fee or expense relates.  $Subject to \underline{Section 22(b)}, \underline{Section 24(b)(v)}, \underline{Section 24(d)}, \underline{Section 24(d$ be remitted to Agent and all such payments, and all proceeds of Collateral received by Agent, shall be applied, so long as no Application Event has occurred and is continuing and except as otherwise provided herein with respect to Defaulting Lenders, to reduce the balance of the Advances outstanding and, thereafter, to Borrower (to be wired to the Designated Account) or such other Person entitled thereto under applicable law. At any time that an Application Event has occurred and is continuing and except as otherwise provided herein with respect to Defaulting Lenders, all payments remitted to Agent and all proceeds of Collateral received by Agent shall be applied as follows:  $\ensuremath{(A)}$  in each case, then due to Agent under the Loan Documents, until paid in full; first, to pay any Lender Group Expenses (including cost or expense reimbursements) or indemnities, second, to pay any fees or premiums then due to Agent under the Loan Documents until paid in full; (B) (C) third, to pay interest due in respect of all Agent Advances until paid in full; in full; (D) fourth, to pay the principal of all Agent Advances until paid (E) fifth, ratably, to pay any Lender Group Expenses (including cost or expense reimbursements) or indemnities, in each case, then due to any of the Lenders under the Loan Documents, until paid in full;

sixth, ratably, to pay any fees or premiums then due to any of the Lenders under the Loan Documents

ninth, ratably, to pay interest accrued in respect of the Advances (other than Agent Advances) and the

Delayed Draw Term Loan until paid in full;

until paid in full;

- 45 -137020431v13 (F)

(G)

(H) tenth, ratably (i) to pay the principal of all Advances until paid in full, (ii) to Agent, to be held by Agent, for the benefit of Issuing Lender (and for the ratable benefit of each of the Lenders that have an obligation to pay to Agent, for the account of Issuing Lender, a share of each Letter of Credit Disbursement), as cash collateral in an amount up to 105% of the Letter of Credit Usage (to the extent permitted by applicable law, such cash collateral shall be applied to the reimbursement of any Letter of Credit Disbursement as and when such disbursement occurs and, if a Letter of Credit expires undrawn, the cash collateral held by Agent in respect of such Letter of Credit shall, to the extent permitted by applicable law, be reapplied pursuant to this Section 2.4(b)(iii), beginning with tier (A) hereofy, (iii) ratably, to the Bank Product Providers based upon amounts then certified by the applicable Bank Product Provider to Agent (in form and substance satisfactory to Agent) to be due and payable to such Bank Product Providers on account of Bank Product Obligations and (iv) to pay the outstanding principal balance of the Delayed Draw Term Loan (in the inverse order of the maturity of the installments due thereunder) until the Delayed Draw Term Loan is paid in full;

(I) <u>eleventh</u>, to pay any other Obligations other than Obligations owed to Defaulting Lenders;

(J) <u>twelfth</u>, ratably to pay any Obligations owed to Defaulting

Lenders; and

(K) <u>thirteenth</u>, to Borrower (to be wired to the Designated

Account) or such other Person entitled thereto under applicable law.

Excluded Hedge Obligations with respect to any Guarantor shall not be paid with amounts received from such Guarantor or its assets, but appropriate adjustments shall be made with respect to payments from other Loan Parties to preserve the allocation to Obligations otherwise set forth above in this Section.

(iv) Agent promptly shall distribute to each Lender, pursuant to the applicable wire instructions received from each Lender in writing, such funds as it may be entitled to receive.

 $(v) In each instance, so long as no Application Event has occurred and is continuing, \underline{Section 24(b)(iii)} shall not apply to any payment made by Borrower to Agent and specified by Borrower to be for the payment of specific Obligations then due and payable (or prepayable) under any provision of this Agreement or any other Loan Document. \\$ 

(vi) For purposes of <u>Section 2.4(b)(iii)</u>, "paid in full" of a type of Obligation means payment in cash or immediately available funds of all amounts owing on account of such type of Obligation, including interest accound after the commencement of any Insolvency Proceeding, default interest, interest on interest, and expense reimbursements, in espective of whether any of the foregoing would be cris allowed or disallowed in whole or in part in any Insolvency Proceeding.

(vii) In the event of a direct conflict between the priority provisions of this <u>Section 24</u> and any other provision contained in this Agreement or any other Loan Document, it is the intention of the parties hereto that such provisions be read together and

- 46 -137020431v13 construed, to the fullest extent possible, to be in concert with each other. In the event of any actual, irreconcilable conflict that cannot be resolved as aforesaid, if the conflict relates to the provisions of  $\underbrace{Section\ 2.3(d)}$  and this  $\underbrace{Section\ 2.4}$ , then the provisions of  $\underbrace{Section\ 2.3(d)}$  shall control and govern, and if otherwise, then the terms and provisions of this  $\underbrace{Section\ 2.4}$  shall control and govern.

#### (c) Reduction of Commitments

(i) Revolver Commitments. The Revolver Commitments shall terminate on the Revolver Maturity Date. Borrower may reduce the Revolver Commitments, without premium or penalty, to an amount (which may be zero) not less than the sum of (A) the Revolver Usage as of such date, plus (B) the principal amount of all Advances not yet made as to which arequest has been given by Borrower pursuant to Section 2.12(a). Each such reduction shall be in an amount which is not less than \$1,000,000 (unless the Revolver Commitments are being reduced to zero and the amount of the Revolver Commitments in effect immediately prior to such reduction are less than \$1,000,000, shall be made by providing not less than 10 Business Days prior written notice to Agent, and shall be intervocable. Once reduced, the Revolver Commitments may not be increased. Each such reduction of the Revolver Commitments shall reduce the Revolver Commitments of each Lender proportionately in accordance with its stable share thereof

(ii) Delayed Draw Term Loan Commitments. The Delayed Draw Term Loan Commitments (A) shall be reduced to the extent that the Delayed Draw Term Loan Commitment Expiration Date.

#### (d) Optional Prepayments.

Advances Bonowermay prepay the principal of any Advance at any time in whole or in part, without premium or penalty.

(ii) **Delayed Draw Term Loan.** Bonower may, upon at least 10 Business Days prior written notice to Agent, prepay the principal of the Delayed Draw Term Loan, in whole or in part. Each prepayment made pursuant to this <u>Section 24 (d)(ii</u>) shall be accompanied by the payment of accound interest to the date of such payment on the amount prepaid. Each such prepayment shall be applied against the remaining installments of principal due on the Delayed Draw Term Loan in the inverse order of maturity (for the avoidance of doubt, any amount that is due and payable on the Delayed Draw Term Loan Maturity Date shall constitute an installment).

### (e) Mandatory Prepayments.

(i) If, at any time or for any reason, the amount of Obligations owed by Borrower to Lender pursuant to <u>Sections 2.1</u> is greater than the Dollar limitations set forth in <u>Sections 2.1</u>, (an "<u>Overadvance</u>"), Borrower immediately shall pay to Agent, in cash, the amount of such excess, which amount shall be used by Agent to reduce the Obligations in accordance with the priorities set forth in <u>Section 2.4(b)</u>. In addition, Borrower hereby promises to pay the

- 47 -137020431v13

Obligations (including principal, interest, fees, costs, and expenses) in Dollars in full to Agent as and when due and payable under the terms of this Agreement and the other Loa
Documents.

(ii) Immediately upon the receipt by Borrower or any of its Subsidiaries of the proceeds of any Disposition by Borrower or any of its Subsidiaries of property or assets (excluding sales or dispositions which qualify as Permitted Dispositions under clauses (a) through (f), (h), or (i) of the definition of Permitted Dispositions), Borrower shall prepay the outstanding principal amount of the Obligations in accordance with Section 2.4(f) in an amount equal to 100% of the Net Cash Proceeds (including insurance proceeds and condemnation awards and payments in lieuthereof) received by such Person in connection with such sales or dispositions, provided that, so long as (A) no Default or Event of Defaults shall have occurred and is continuing (B) the Net Cash Proceeds of such Disposition are held in a cash collateral account in which Agent has a perfected flust-priority security interest, and (C) Borrower or its Subsidiaries, as applicable, complete such replacement, purchase, or construction within 180 days after the initial accept of such monies, Borrower and its soliciaries is shall have the option to apply such monies to the costs of replacement of the property or assets that are the subject of such sale or disposition unless and to the extent that such applicable period shall have expired without such replacement, purchase or construction being made or completed, in which case, such monies shall be paid to Agent and applied in accordance with Section 2.4(f). Nothing contained in this Section 2.4(e)(ii) shall permit Borrower or any of its Subsidiaries to sell or otherwise dispose of any property or assets other than in accordance with Section 7.4.

(iii) Promptly upon the issuance or incurrence by Bonower or any of its Subsidianies of any Indebtedness (except for Indebtedness permitted under Section 7.1) Bonower shall prepay the outstanding principal amount of the Obligations in accordance with Section 2.4(f) in an amount equal to 100% of the Net Cash Proceeds received by such Person in connection with such issuance or incurrence. The provisions of this Section 2.4(e) (iii) shall not be deemed to be implied consent to any such issuance or incurrence otherwise prohibited by the terms and conditions of this Agreement.

(iv) Promptly upon the issuance by Borrower or any of its Subsidiaries of any Stock (except for (A) the issuance of Stock by Borrower to any Permitted Holder, (B) the issuance of Stock of Borrower to directors, officers and employees of Borrower and its Subsidiaries pursuant to employee stock option plans (or other employee incentive plans or other compensation amangements) approved by the Board of Directors) Borrower shall prepay the outstanding principal amount of the Obligations in accordance with Section 2.4(f) in an amount equal to 50% of the Net Cash Proceeds received by such Person in connection with such issuance, provided however, that if the Leverage Ratio of Borrower and its Subsidiaries as of the end of the fiscal quarter most recently ended prior to the date of the issuance of such Stock as to which financial statements were required to be delivered pursuant to this Agreement was equal to or less than 20.1.0, then no prepayment in respect of such issuance of Stock shall be required. The provisions of this Section 2.4(e)(iv) shall not be deemed to be implied consent to any such issuance otherwise prohibited by the terms and conditions of this Agreement.

(v) Within 10 days of delivery to Agent of audited annual financial statements pursuant to <u>Section 6.3(b)</u>, commencing with the delivery to Agent of the financial statements for Bonower's fiscal year ended December 31, 2021 or, if such financial statements

- 48 -137020431v13 are not delivered to Agent on the date such statements are required to be delivered pursuant to Section 6.3(b), within 10 days after the date such statements were required to be delivered to Agent pursuant to Section 6.3(b), Borrower shall (A) if such financial statements demonstrate that the Leverage Ratio of Borrower and its Subsidiaries as of the end of such fiscal year was 1.50:1.00 or greater, prepay the outstanding principal amount of the Obligations in accordance with Section 2.4(f) in an amount equal to 50% of the Excess Cash Flow of Borrower and its Subsidiaries for such fiscal year, and (B) if such financial statements demonstrate that the Leverage Ratio of Borrower and its Subsidiaries as of the end of such fiscal year was less than 1.50:1.0, then no prepayment shall be required. The foregoing to the contrary notwithstanding, at Agent's election, in its sole and absolute discretion, if the remaining amount of the capital expenditures projected to be made during such fiscal year exceeds the remaining amount of the Delayed Draw Term Loan Commitments at such time, then Agent may permit Borrower to reduce the amount of the mandatory prepayment that would otherwise be due and payable pursuant to this Section 2.4(e)(v) to the extent of such excess (but not to an amount that is less than \$0).

### (f) Application of Payments.

(i) Each prepayment pursuant to Section 24(e)(i) shall, (A) so long as no Application Event shall have occurred and be continuing, be applied, first, to the outstanding principal amount of the Advances until paid in full, and second, to cash collateralize the Letters of Credit in an amount equal to 105% of the then outstanding Letter of Credit Usage, and (B) if an Application Event shall have occurred and be continuing, be applied in the manner set forth in Section 24(t)(iii).

(ii) Each prepayment pursuant to Section 24(e)(ii), 24(e)(iii), 24(e)(iy), or 24(e)(v) shall (A) so long as no Application Event shall have occurred and be continuing be applied, first, ratably to the outstanding principal amount of the Delayed Draw Term Loan until paid in full, and second, to the outstanding principal amount of the Advances (without a conesponding permanent reduction in the Maximum Revolver Amount unless such prepayment is made pursuant to Section 24(e)(iii), until paid in full, and third, to cash collateralize the Letters of Credit in an amount equal to 105% of the then outstanding Letter of Oredit Usage (without a conesponding permanent reduction in the Maximum Revolver Amount), and (B) if an Application Event shall have occurred and be continuing be applied in the manner set forth in Section 24(b)(iii). Each such prepayment of the Delayed Draw Term Loan in the inverse order of maturity (for the avoidance of doubt, any amount that is due and payable on the Delayed Draw Term Loan Maturity Dates shall constitute an installment).

### 2.5 Promise to Pay; Promissory Notes.

(a) Borrower agrees to pay the Lender Group Expenses on the earlier of

(i) the first day of the month following the date on which the applicable Lender Group Expenses were first incurred or (ii) the date on which demand therefor is made by Agent (it being acknowledged and agreed that any charging of such costs, expenses or Lender Group Expenses to the Loan Account pursuant to the provisions of Section 2.6(d) shall be deemed to constitute a demand for payment thereof for the purposes of this subclause (ii)). Borrower promises to pay

- 49 -137020431v13 all of the Revolver Obligations (including principal, interest, premiums, if any, fees, costs, and expenses (including Lender Group Expenses) in respect thereof) in full on the Revolver Maturity Date or, if earlier, on the date on which the Revolver Obligations (other than the Bank Product Obligations) become due and payable pursuant to the terms of this Agreement. Borrower promises to pay all of the Delayed Draw Term Loan Obligations (including principal, interest, premiums, if any, fees, costs, and expenses (including Lender Group Expenses) in respect thereof) in full on the Delayed Draw Term Loan Maturity Date or, if earlier, on the date on which the Delayed Draw Term Loan Obligations become due and payable pursuant to the terms of this Agreement. Borrower promises to pay all of the Incremental Term Loan Obligations (including principal, interest, premiums, if any, fees, costs, and expenses (including Lender Group Expenses) in respect thereof) in full on the maturity date of any Incremental Term Loan or, if earlier, on the date on which the Incremental Term Loan Obligations become due and payable pursuant to the terms of this Agreement. Borrower agrees that its obligations contained in the first sentence of this Section 2.5(a) shall survive payment or satisfaction in full of all other Obligations.

(b) Any Lendermay request that any portion of its Commitments or the Loans made by it be evidenced by one or more promissory notes. In such event, Borrower shall execute and deliver to such Lender and its registered assigns in a form furnished by Agent and reasonably satisfactory to Borrower. The reafler, the portion of the Commitments and Loans evidenced by such promissory notes and interest the econ shall at all times be represented by one or more promissory notes in such form payable to the payeen amed therein.

### 2.6 <u>Interest Rates and Letter of Credit Fee: Rates, Payments, and Calculations</u>

- (a) Interest Rates Except as provided in clause (c) below, all Obligations (except for undrawn Letters of Credit and except for Bank Product Obligations) that have been charged to the Loan Account pursuant to the terms hereof shall bear interest on the Daily Balance thereof as follows (i) if the relevant Obligation is a LiBOR Rate Loan, at a per annum rate equal to the LiBOR Rate plus the LiBOR Rate Margin, and (ii) if the relevant Obligation is a Base Rate Loan, at a per annum rate equal to the Base Rate plus the Base Rate Margin.
- (b) Letter of Credit Fee. Borrowershall pay Agent (for the ratable benefit of the Lenders with a Revolver Commitment), a Letter of Credit Fee '1) (which fee shall be in addition to the fronting fees and commissions, other fees, charges and expresses set for thin <u>Section 2.12(di</u>) that shall accure at a *per arrum* mate equal to the LIBOR Rate Margin times the Daily Balance of the undawn amount of all outstanding Letters of Credit.
  - (c) Default Rate. Upon the occurrence and during the continuation of an Event of Default (and at the election of Agent or the Required Lenders),
- (i) all Obligations (except for undrawn Letters of Gredit and except for Bank Product Obligations) that have been charged to the Loan Account pursuant to the terms hereof shall bear interest on the Daily Balance thereof at a per annum rate equal to 200 percentage points above the per annum rate otherwise applicable hereunder, and

- 50 -137020431v13 (ii) the Letter of Credit fee provided for above shall be increased to 2.00 percentage points above the per annum rate otherwise applicable hereunder.

- (d) Payment Except to the extent provided to the contrary in Section 2.11, Section 2.12(d), or Section 2.13(a), (i) all interest, Letter of Credit fees, and all other fees payable has under shall be due and payable, in an east, on the first day of each quater at any time that Obligations or Commitments are outstanding and (ii) all costs and expenses payable has under or under any of the other Loan Documents, and all Lender Group Expenses shall be due and payable on the earlier of (e) the first day of the month following the date on which the applicable costs, expenses or Lender Group Expenses were first invoiced to Borrower or (v) the claim which demand therefor is made by Agant (their go advowledged and agreed that any charging of such costs, expenses or Lender Group Expenses to the Loan Account pursuant to the provisions of the following sentence shall be deemed to constitute admand for payment thereof for the purposes of this subclass (v)). To the extert that such amounts are not of any when the Loan Documents (as and when payable has under or under the other Loan Documents) by debiting the Designated Account in an amount equal to the amount the eof. Borrower hereby authorizes Agant, from time to time without prior or the total committee of the amount the eof. Borrower hereby authorizes Agant, from time to time without prior or the ender of the amount the eof. Borrower hereby authorizes Agant, from time to time without prior or the ender of the amount the eof. Borrower hereby authorizes Agant, from time to time without prior or the ender of the amount the eof. Borrower hereby authorizes Agant, from time to time without prior or the ender of the ender
- (A) as and when due and payable all other payment obligations payable under any Loan Document or any Bank Product Agreement (including any amounts due and payable to the Bank Product Providers in respect of Bank Products). All amounts (including interest, fees, costs, expenses, Lender Group Expenses, or other amounts payable hereunder or under any other Loan Document or under any Bank Product Agreement) charged to the Loan Account shall thereupon constitute Advances hereunder, shall constitute Obligations hereunder, and shall initially accrue interest at the rate then applicable to Advances that are Base Rate Loans (unless and until converted into LIBOR Rate Loans in accordance with the terms of this Agreement).
- (e) Computation. All interest and fees chargeable under the Loan Documents shall be computed on the basis of a 360 day year for the actual number of days elapsed (except in the case of Base Rate Loans, which shall be 365 days). In the event the Base Rate is changed from time to time hereafter, the rates of interest hereunder based upon the Base Rate automatically and immediately shall be increased or decreased by an amount equal to such darage in the Base Rate.
- (f) Intent to Limit Charges to Maximum Lawful Rate. In no event shall the interestrate on a tespayable under this Agreement, plus any other amounts paid in connection herewith, exceed the highest rate permissible under any law that account of competent

jurisdiction shall, in a final determination, deem applicable. Borrower and the Lender Group, in executing and delivering this Agreement, intend legally to agree upon the rate or rates of interest and manner of payment stated within it; <a href="mailto:provided">provided</a>, <a href="mailto:however">however</a>, that, anything contained herein to the contrary notwithstanding, if said rate or rates of interest or manner of payment exceeds the maximum allowable under applicable law, then, <a href="mailto:ipso facto">ipso facto</a>, as of the date of this Agreement, Borrower is and shall be liable only for the payment of such maximum as allowed by law, and payment received from Borrower in excess of such legal maximum, whenever received, shall be applied to reduce the principal balance of the Obligations to the extent of such excess.

#### 2.7 Cash Management.

- (a) Bonowershill and shall cause each of its Subsidiaries to (i) establish and maintain cash management services of a type and on terms reasonably satisfactory to Agent at one or more of the banks set forthors. Schedule 27(a) (each, a "Cash Management Bank"), and shall request in writing and otherwise take such reasonable steps to ensure that all of its and its Subsidiaries' Account Debtors for was observed by them directly to such Cash Management Bank, and (ii) deposit or cause to be deposited promptly, and many event no later than the first. Business Day after the date of receipt thee of, all of their Collections (including those sent directly by their Account Debtors to Bonower or one of its Subsidiaries) into a bank account in Agent's name (a "Cash Management Account") at one of the Cash Management Banks, provided that (i) Bonower may keep up to \$25,000 in the aggregate in accounts that are not subject to the foregoing and (ii) the requirements in this Section 27 or Section 47 shall not apply to Excluded Deposit Accounts.
- (b) Each Cash Management Bank shall establish and maintain Cash Management Agreements with Agent and Borower, in form and substance reasonably acceptable to Agent. Each such Cash Management Agreement shall provide, among other things, that (i) the Cash Management Bank will comply with any instructions originated by Agent directing the disposition of the funds in such Cash Management Account without further consent by Borrower or its Subsidiaries, as applicable, (iii) the Cash Management Bank has notights of set off on ecoupment or any other claim against the applicable Cash Management Account other than for payment of its service fees and other charges directly related to the administration of such Cash Management Account and for returned deeds on other charges and initiation of Agent (an "Activation Instruction"), the Cash Management Bank will forward by daily sweep all amounts in the applicable Cash Management Account of the Agent's Account. Agent agrees not to issue an Activation Instruction is used.
- (c) Solong-as no Defaultor Event of Defaulth-as occurred and is continuing. Borrowermay ament <u>Schedule 27</u>(a) to add one place a Cash Management Bank or Cash Management Account, <u>provided however</u>, that (i) such prospective Cash Management Bank shall be reasonably satisfactory to Agent, and (ii) prior to the time of the opening of such Cash Management Account, Borrower (crits Subsidiary, as applicable) and such prospective Cash Management Bank shall have executed and delivered to Agent a Cash Management. Borrower (crits Subsidiaries, as applicable) shall dose any of its Cash Management Accounts (and establish replacement cash management accounts in accordance with the foregoing

sentence) promptly and in any event within 30 days of notice from Agent (or such longer period as the Agent may reasonably agree; but in no event later than 60 days after such notice is given) that the creditworthiness of any Cash Management Bank is no longer acceptable in Agent's reasonable judgment, or as promptly as practicable and in any event within 60 days of notice from Agent (or such longer period as the Agent may reasonably agree; but in no event later than 90 days after such notice is given) that the operating performance, funds transfer, or availability procedures or performance of the Cash Management Bank with respect to Cash Management Accounts or Agent's liability under any Cash Management Agreement with such Cash Management Bank is no longer acceptable in Agent's reasonable judgment

### (d) The Cash Management Accounts shall be cash collateral accounts subject to Control Agreements.

- 2.8 Crediting Payments. The receipt of any payment item by Agent (whether from transfers to Agent by the Cash Management Banks pursuant to the Cash Management Agreements or otherwise) shall not be considered a payment on account unless such payment item is a wire transfer of immediately available federal funds made to the Agent's Account or unless and until such payment item is honored when presented for payment, then Borrower shall be deemed not to have made such payment and interest shall be calculated accordingly. Anything to the contrary contained herein notwithstanding, any payment item shall be deemed received by Agent only if it is received into the Agent's Account on a Business Day on or before 11:00 a.m. (California time). If any payment item is received into the Agent's Account on a non-Business Day or after 11:00 a.m. (California time) on a Business Day, it shall be deemed to have been received by Agent as of the opening of business on the immediately following Business Day.
- 2.9 <u>Designated Account</u>. Agent is authorized to make the Advances, the Delayed Draw Term Loan and any Incremental Term Loans, and Issuing Lender is authorized to issue the Letters of Credit, under this Agreement based upon telephonic or other instructions received from anyone purporting to be an Authorized Person or, without instructions, if pursuant to <u>Section 2.6(d</u>). Borrower agrees to establish and maintain the Designated Account with CNB for the purpose of receiving the proceeds of the Advances, the Delayed Draw Term Loan and the Incremental Term Loans requested by Borrower and made by Agent or the Lenders hereunder. Unless otherwise agreed by Agent and Borrower, any Advance, Agent Advance, Delayed Draw Term Loan or Incremental Term Loan requested by Borrower and made by Agent or the Lenders hereunder shall be made to the Designated Account.
- 2.10 Maintenance of Loan Account; Statements of Obligations. Agent shall maintain an account on its books in the name of Borrower (the "Loan Account") on which Borrower will be charged with the Incremental Term Loans, the Delayed Draw Term Loan, all Advances (including Agent Advances) made by Agent or the Lenders to Borrower or for Borrower's account, the Letters of Credit issued by Issuing Lender for Borrower's account, and with all other payment Obligations hereunder or under the other Loan Documents (except for Bank Product Obligations), including, accrued interest, fees and expenses, and Lender Group Expenses. In accordance with Section 2.8, the Loan Account will be credited with all payments received by Agent from Borrower or for Borrower's account, including all amounts received in the Agent's Account from any Cash Management Bank. Agent shall render statements regarding

the Loan Account to Borrower, including principal, interest, fees, and including an itemization of all charges and expenses constituting Lender Group Expenses owing, and such statements, absent manifest error, shall be conclusively presumed to be correct and accurate and constitute an account stated between Borrower and the Lender Group unless, within 30 days after receipt thereof by Borrower, Borrower shall deliver to Agent written objection thereto describing the error or errors contained in any such statements.

- 2.11 Fees. Borrower shall pay to Agent the following fees and charges, which fees and charges shall be non-refundable when paid (irrespective of whether this Agreement is terminated thereafter) and shall be apportioned among the Lenders in accordance with the terms of agreements between Agent and individual Lenders:
  - (a) Fee Letter Fees. As and when due and payable under the terms of the Fee Letter; the fees set forth in the Fee Letter;
- (b) **Revolver Unused Line Fee.** Borrowershall pay to Agent, for the natible account of the Lendes holding a Revolver Commitment, a revolver unused line fee (the "<u>Revolver Unused Line Fee</u>"), quaterly in areas, on the first day of each quater from and after the Residement Effective Date up to the first day of the quater prior to the date on which the Revolver Obligations are paid in full and the Revolver Obligations are paid in full and the Revolver Obligations are terminated or are reduced to zero, in an amount equal to the result of the sum of the Daily Revolver Unused Line Fee for each day during such fixed quarter.
- (c) **Dekyed Draw Term Loan Unused Line Fee.** Borowershall pay to Agent, for the ratable account of the Lendas holding a Dekyed Draw Term Loan Commitment, adekyed draw term loan unused line fee (the "<u>Dekyed Draw Term Loan Unused Line Fee</u>"), quarterly in an easy on the first day of each quarter from and after the Restaument Effective Date up to the first day of the quarter prior to the Dekyed Draw Term Loan Commitment Expiration Date and on the Dekyed Draw Term Loan Commitment Expiration Date, in an amount equal to the result of the sum of the Daily Dekyed Draw Term Loan Unused Line Fee for each day during such fiscal quarter.
- (d) Audit, Appraisal, and Valuation Charges. Audit, appraisal, and valuation fees and charges shall consist of the actual charges paid or incurred by Agent fit elects to employ the services of one or more third Persons to perform financial audits of Borrower or its Subsidiaries, to establish electronic collateral reporting systems, to appraise the Collateral, or any portion thereof, or to assess Borrower's or its Subsidiaries' business valuation; provided, however, that so long as no Event of Default shall have occurred and becoming Borrower shall not be obligated to reimburse. Agent for more than 2 audits, appraisals or valuations cluring any calendar year.

### 2.12 Letters of Credit

(a) Subject to the terms and conditions of this Agreement, upon the request of Borrower made in accordance herewith, and prior to the Revolver Maturity Date, the Issuing Lender agrees to issue letters of oredit for the account of Borrower (each, a "Letter of Oredit").

By submitting a request to Issuing Lender for the issuance of a Letter of Credit, Borrower shall be deemed to have requested that Issuing Lender issue the requested Letter of Credit. Each request for the issuance of a Letter of Credit, or the amendment, renewal, or extension of any outstanding Letter of Credit, shall be irrevocable and shall be made in writing by an Authorized Person and delivered to Issuing Lender via telefacsimile or other electronic method of transmission reasonably acceptable to Issuing Lender and reasonably in advance of the requested date of issuance, amendment, renewal, or extension. Each such request shall be in form and substance satisfactory to the Issuing Lender in its Permitted Discretion and shall specify (i) the amount of such Letter of Credit, (ii) the date of issuance, amendment, renewal, or extension of such Letter of Credit, (iii) the proposed expiration date of such Letter of Credit, (iv) the name and address of the beneficiary thereof, and (v) such other information (including, in the case of an amendment, renewal, or extension, identification of the outstanding Letter of Credit to be so amended, renewed, or extended) as shall be necessary to prepare, amend, renew, or extend such Letter of Credit. The Issuing Lender shall have no obligation to issue a Letter of Credit if any of the following would result after giving effect to the issuance of such requested Letter of Credit:

the Letter of Credit Usage would exceed \$10,000,000, or

(ii) the Letter of Credit Usage would exceed the Maximum Revolver Amount less the outstanding amount Advances.

Each Letter of Credit shall be in form and substance acceptable to the Issuing Lender (in the exercise of its Permitted Discretion), including the requirement that the amounts payable thereunder must be payable in Dollars. If Issuing Lender is obligated to advance funds under a Letter of Credit, Borrower promptly shall reimburse such L/C Disbursement to Issuing Lender by paying to Agent an amount equal to such L/C Disbursement not later than 11:00 a.m., California time, on the date that such L/C Disbursement is made, if Borrower shall have received written or telephonic notice of such L/C Disbursement prior to 10:00 a.m., California time, on such date, or, if such notice has not been received by Borrower prior to such time on such date, then not later than 11:00 a.m., California time, on the Business Day that Borrower receives such notice, if such notice is received prior to 10:00 a.m., California time, on the date of receipt, and, in the absence of such reimbursement, the L/C Disbursement immediately and automatically shall be deemed to be an Advance hereunder and, thereafter, shall bear interest at the rate then applicable to Advances that are Base Rate Loans under Section 2.6. To the extent an L/C Disbursement is deemed to be an Advance hereunder, Borrower's obligation to reimburse such L/C Disbursement shall be discharged and replaced by the resulting Advance. Promptly following receipt by Agent of any payment from Borrower pursuant to this paragraph, Agent shall distribute such payment to the Issuing Lender or, to the extent that Lenders have made payments pursuant to Section 2.12(c) to reimburse the Issuing Lender, then to such Lenders and the Issuing Lender as their interests may appear.

(b) Promptly following receipt of anotice of LCD isbursement pursuant to Section 2.12(a), each Lender with a Revolver Commitment agrees to fund its Pro Rata Share of any Advance deemed made pursuant to the foregoing subsection on the same terms and conditions as if Borrower had requested such Advance and Agents hall promptly pay to Issuing Lender the amounts so received by it from the Lenders. By the issuance of a Letter of Credit (or an amendment to a Letter of Credit increasing the amount the ecol and without any further action

- 55 -137020431v13 on the part of the Issuing Lender or the Lenders with Revolver Commitments, the Issuing Lender shall be deemed to have granted to each Lender with a Revolver Commitment, and each Lender with a Revolver Commitment shall be deemed to have purchased, a participation in each Letter of Credit, in an amount equal to its Pro Rata Share of the Risk Participation Liability of such Letter of Credit, and each such Lender agrees to pay to Agent, for the account of the Issuing Lender, such Lender's Pro Rata Share of any payments made by the Issuing Lender under such Letter of Credit. In consideration and in furtherance of the foregoing, each Lender with a Revolver Commitment hereby absolutely and unconditionally agrees to pay to Agent, for the account of the Issuing Lender, such Lender's Pro Rata Share of each L/C Disbursement made by the Issuing Lender and not reimbursed by Borrower on the date due as provided in clause (a) of this Section, or of any reimbursement payment required to be refunded to Borrower for any reason. Each Lender with a Revolver Commitment acknowledges and agrees that its obligation to deliver to Agent, for the account of the Issuing Lender, an amount equal to its respective Pro Rata Share of each L/C Disbursement made by the Issuing Lender pursuant to this Section 2.12(b) shall be absolute and unconditional and such remittance shall be made notwithstanding the occurrence or continuation of an Event of Default or Default or the failure to satisfy any condition set forth in Section 3 hereof. If any such Lender fails to make available to Agent the amount of such Lender's Pro Rata Share of each L/C Disbursement made by the Issuing Lender in respect of such Letter of Credit as provided in this Section, such Lender shall be deemed to be a Defaulting Lender and Agent (for the account of the Issuing Lender) shall be entitled to recover such amount on demand from such Lender together with interest thereon at the Defaulting Lender Rate until paid in full. In the event there is a Defaulting Lender as of the date of any request for the issuance of a Letter of Credit, the Issuing Lender shall not be required to issue or arrange for such Letter of Credit to the extent (i) the Defaulting Lender's Letter of Credit Exposure with respect to such Letter of Credit may not be reallocated pursuant to Section 2.3(d)(ii), or (ii) the Issuing Lender has not otherwise entered into arrangements reasonably satisfactory to it and Borrower to eliminate the Issuing Lender's risk with respect to the participation in such Letter of Credit of the Defaulting Lender, which arrangements may include Borrower cash collateralizing such Defaulting Lender's Letter of Credit Exposure in accordance with Section 2.3(d)(ii). Additionally, Issuing Lender shall have no obligation to issue a Letter of Credit if (A) any order, judgment, or decree of any Governmental Authority or arbitrator shall, by its terms, purport to enjoin or restrain Issuing Lender from issuing such Letter of Credit, or any law applicable to Issuing Lender or any request or directive (whether or not having the force of law) from any Governmental Authority with jurisdiction over Issuing Lender shall prohibit or request that Issuing Lender refrain from the issuance of letters of credit generally or such Letter of Credit in particular, (B) the issuance of such Letter of Credit would violate one or more policies of Issuing Lender applicable to letters of credit generally, or (C) if amounts demanded to be paid under any Letter of Credit will or may not be in United States Dollars.

(c) Borrowerheeby agresto indemnify, save, defend, and hold harmlesseed member of the Lender Group (including Issuing Lender and its branches, Affiliates, and consepondents) and each such Person's respective directors, offices, employees, attorneys and agents (each, including Issuing Lender, a "Letter of Credit Related Person') (to the full est extent permitted by law) from and against any and all claims, demands, suits, actions, investigations, proceedings, liabilities, fines, costs, penalties, and damages, and all resonable fees and disbursements of attorneys, experts, or consultants and all other costs and expenses actually incurred in connection therewith or in connection with the enforcement of this indemnification

- 56 -137020431v13

(as and when they are incurred and	i iiiespeciive c	in whether suit is brought), (the <u>letter of Credit indefinition Costs</u> ), and which arise out of the Collection with, or as a result of
	(i)	any Letter of Credit or any pre-advice of its issuance;
connection with any Letter of Credit;	(ii)	any  transfer, sale,  delivery, sumender or endorsement of any  Drawing  Document  at any  time (s)  held  by   any  such  Letter  of  Credit  Related  Person  in  Drawing  Dr
,	(iii) rrestrainany preser	any action or proceeding arising out of, or in connection with, any Letter of Credit (whether administrative, judicial or in connection with arbitration), tation or payment under any Letter of Credit, or for the wrongful dishonor of, or honoring a presentation under, any Letter of Credit;
Letter of Credit;	(iv)	any independent undertakings issued by the beneficiary of any
connection with any Letter of Crec	(v) lit or requested	anyunauthorized instruction or request made to Issuing Lender in I Letter of Credit or error in computer or electronic transmission;
	(vi)	an adviser; confirmer or other nominated person seeking to be reimbursed, indemnified or compensated;
of an instrument or document:	(vii)	any third party seeking to enforce the rights of an applicant, beneficiary, nominated person, transferee, assignee of Letter of Credit proceeds or holder
	(viii)	the fiaud, forgery or illegal action of parties other than the Letter of Credit Related Person;
	(ix)	Issuing Lender's performance of the obligations of a confirming institution or entity that wrongfully dishonors a confirmation; or

in each case, including that resulting from the Letter of Credit Related Person's own negligence; provided, however, that such indemnity shall not be available to any Letter of Credit Related Person claiming indemnification under clauses (i) through (x) above to the extent that such Letter of Credit Indemnified Costs may be finally determined in a final, non-appealable judgment of a court of competent jurisdiction to have resulted directly from the gross negligence or willful misconduct of the Letter of Credit Related Person claiming indemnity. Borrower hereby agrees to pay the Letter of Credit Related Person claiming indemnity on demand from time to time all amounts owing under this Section 2.12(c). If and to the extent that the obligations of Borrower under this Section 2.12(c) are unenforceable for any reason, Borrower agrees to make the maximum contribution to the Letter of Credit Indemnified Costs permissible

the acts or omissions, whether rightful or wrongful, of any present or future dejure or defactogovernmental or regulatory authority or cause or event

(x)

beyond the control of the Letter of Credit Related Person;

under applicable law. Borrower hereby acknowledges and agrees that neither the Lender Group nor the Issuing Lender shall be responsible for delays, errors, or omissions resulting from the malfunction of equipment in connection with any Letter of Credit. This indemnification provision shall survive termination of this Agreement and all Letters of

- (d) Any and all charges, commissions, fees, and costs incurred by the Issuing Lender relating to Letters of Credit shall be Lender Group Expenses for purposes of this Agreement and promptly shall be reimbursable by Borrowerto Agent for the account of the Issuing Lender; it being advrowledged and agreed by Borrowertot, as of the Restatement Effective Date, the issuance charge imposed by Issuing Lender is 0.125% per annum times the face amount of each Letter of Credit, that such issuance charge may be charged from time to time, and that the Issuing Lender also imposes as shedule of charges for amendments, extensions, drawings, and renewals.
- (e) The liability of Issuing Lender (or any other Letter of Cedit Related Person) under, in connection with or aising out of any Letter of Cedit (or pre-advice), regardless of the form or legal grounds of the action or proceeding, shall be limited to direct damages suffered by Borrower that are cassed directly by Issuing Lender's gross negligence or will full misconduct in (i) honoring a presentation under a Letter of Cedit (iii) failing to homor a presentation under a Letter of Cedit that sair thy complies with the terms and conditions of such Letter of Cedit (iii) tetaining dawing documents presented under a Letter of Cedit. Issuing Lender shall be deemed to have eated with due diligence and tress or able care if Issuing Lender should is in accordance with this Agreement. Borrower's aggregate mendies against Issuing Lender and any Letter of Cedit or wrongfully honoring a presentation under and y Letter of Cedit or wrongfully retaining honored Drawing Documents shall innove entexceed the aggregate amount paid by Borrower to Issuing Lender and any Letter of Cedit and inconnection with such Letter of Cedit runder Section 21.2. Borrower shall take action to avoid and mitigate the amount of any damages claimed against Issuing Lender or any other Letter of Cedit Related Person, including by enforcing its sights against the beneficianies of the Letters of Cedit. Any claim by Borrower and enrich or alleged wrongful conduct complained of, and (y) the amount (if any) of the loss that would have been avoided had Borrower taken all resonable steps to mitigate any loss, and in case of action of wrongful dishonor, by specifically and timely authorizing Issuing Lender to effect a one
- (f) Borower is responsible for preparing or approving the final text of the Letter of Credit as issued by Issuing Lender; in respective of any assistance Issuing Lendermay provide such as drafting or recommending text or by Issuing Lender's use on refusal to use text submitted by Borrower is solely responsible for the suitability of the Letter of Credit for Borrower's purposes. With respect to any Letter of Credit containing an "automatic amendment" to extend the expiration date of such Letter of Credit, Issuing Lender, in its sole and absolute discretion, may give notice of nonrenewal of such Letter of Credit and, if Borrower does not at any time want such Letter of Credit to be renewed, Borrower will so notify Agent and Issuing Lender at least 15 calendar days before Issuing Lender is required to notify the

beneficiary of such Letter of Credit or any advising bank of such nonrenewal pursuant to the terms of such Letter of Credit.

- (g) Borrower's reimbusement and payment obligations under this <u>Section 2.12</u> are absolute, unconditional and inevocable and shall be performed strictly in accordance with the terms of this Agreement under any and all circumstances what soever; including
  - any lack of validity, enforceability or legal effect of any Letter of Credit or this Agreement or any term or provision therein;
- (ii) payment against presentation of any draft, demand or daim for payment under any drawing document that does not comply in whole or in part with the terms of the applicable Letter of Credit or which proves to be fraudulent, forged or invalid in any respect or any statement therein being untrue or inaccurate in any respect, or which is signed, issued or presented by a Person or a transferee of such Person purporting to be a successor or transferee of the beneficiary of such Letter of Credit;
  - (iii) Issuing Lender or any of its branches or Affiliates being the beneficiary of any Letter of Credit;
- (iv) Issuing Lender or any correspondent honoring a drawing against a drawing document up to the amount available under any Letter of Credit even if such drawing document drains an amount in excess of the amount available under the Letter of Credit;
- (v) the existence of any claim, set-off, defense or other night that Borrower or any other Person may have at any time against any beneficiary, any assignee of proceeds, Issuing Lender or any other Person;
- (vi) any other event, circumstance or conduct what soever, whether or not similar to any of the foregoing that might, but for this Section 2.12(g), constitute a legal or equitable defense to or discharge of, or provide a right of set-off against, Borrower's reimbursement and other payment obligations and liabilities, arising under; or in connection with, any Letter of Credit, whether against Issuing Lender, the beneficiary or any other Person; or
  - (vii) the fact that any Default or Event of Default shall have occurred

and be continuing;

provided, however, that subject to Section 2.12(e) above, the foregoing shall not release Issuing Lender from such liability to Borrower as may be finally determined in a final, non-appealable judgment of a court of competent jurisdiction against Issuing Lender following reimbursement or payment of the obligations and liabilities, including reimbursement and other payment obligations, of Borrower to Issuing Lender arising under, or in connection with, this Section 2.12 or any Letter of Credit.

(h) Without limiting any other provision of this Agreement, Issuing Lender and each other Letter of Credit Related Person (if applicable) shall not be responsible to

- 59 -137020431v13

Borrower for, and Issuing Lender's rights and remedies against Borrow shall not be impaired by:	wer and the obligation of Borrower to reimburse Issuing Lender for each drawing under each Letter of Credi
(i) the Letter of Credit requires strict compliance by the beneficiary;	honor of a presentation under any Letter of Credit that on its face substantially complies with the terms and conditions of such Letter of Credit, even if the conditions of such Letter of Credit, even if the conditions of such Letter of Credit, even if the conditions of such Letter of Credit, even if the conditions of such Letter of Credit, even if the conditions of such Letter of Credit, even if the conditions of such Letter of Credit that on its face substantially complies with the terms and conditions of such Letter of Credit, even if the conditions of such Letter of Credit that on its face substantially complies with the terms and conditions of such Letter of Credit, even if the conditions of such Letter of Credit, even if the conditions of such Letter of Credit, even if the conditions of such Letter of Credit, even if the conditions of such Letter of Credit, even if the conditions of such Letter of Credit, even if the conditions of such Letter of Credit, even if the conditions of such Letter of Credit, even if the conditions of such Letter of Credit, even if the conditions of such Letter of Credit (even if the conditions of such Letter of Credit (even if the conditions of such Letter of Credit (even if the conditions of even if the condition is even if the conditions of even if the conditio
(ii) nansferee of any beneficiary or other Person required to sign, present or issue such drawing docum (B) under a new name of the beneficiary;	honor of a presentation of any drawing document that appears on its face to have been signed, presented or issued (A) by any purported successor or ment or
(iii) draft or not withstanding any requirement that such draft, demand or request bear any or adequate	acceptance as a diaft of any written or electronic demand or request for payment under a Letter of Credit, even if nonnegotiable or not in the form of reference to the Letter of Credit;
(iv) document (other than Issuing Lender's determination that such drawing document appears on its	the identity or authority of any presenter or signer of any drawing document or the form, accuracy, genuineness or legal effect of any drawing face substantially to comply with the terms and conditions of the Letter of Credit);
(v) givenby a Person authorized to give such instruction or request;	acting upon any instruction or request relative to a Letter of Credit or requested Letter of Credit that Issuing Lender in good faith believes to have been described by the contract of the
(vi) or for enors in interpretation of technical terms or in translation or any delay in giving or failing to g	any errors, omissions, interruptions or delays in transmission or delivery of any message, advice or document (regardless of how sent or transmitted give notice to Borrower;
(vii) between the beneficiary and Borrower or any of the parties to the underlying transaction to which	any acts, omissions or fraud by, or the insolvency of, any beneficiary, any nominated person or entity or any other Person or any breach of contract rathe Letter of Credit relates;
(viii) drawing document be presented to it at a particular hour or place;	$assertion\ or\ waiver\ of\ any\ provision\ of\ the\ ISP\ or\ UCP\ that\ primarily\ benefits\ an issuer\ of\ a letter\ of\ credit,\ including\ any\ requirement\ that\ any\ any\ benefits\ and\ any\ and\ any\ and\ any\ any\ any\ any\ any\ any\ any\ any$

(ix) payment to any paying or negotiating bank (designated or permitted by the terms of the applicable Letter of Credit) claiming that it nightfully honored or is entitled to reimbursement or indemnity under Standard Letter of Credit Practice applicable to it;

(x) advised or negotiated such Letter of Credit, as the case may be; acting or failing to act as required or permitted under standard letter of credit practice applicable to where Issuing Lender has issued, confirmed,

- 60 -137020431v13

dishonored by Issuing Lender if subsequ	(xi) pently Issuing Lenderorans	honor of a presentation after the expiration date of any Letter of Credit not with standing that a presentation was made prior to such expiration date an count or other finder of fact determines such presentation should have been honored;
caractery acting a care in career	<u> </u>	- Compared to the compared to
	(xii)	dishonor of any presentation that does not strictly comply or that is fraudulent, forged or otherwise not entitled to honor; or
restrictions on the transaction of business	(xiii) swith certain prohibited Per	honor of a presentation that is subsequently determined by Issuing Lender to have been made in violation of international, federal, state or local sons.
(i) whether having the force of law) of any G	Governmental Authority or m	If by nesson of (x) any Change in Law, or (y) compliance by Issuing Lender or any other member of the Lender Group with any direction, request, or requirement (in espective of coretary authority including Regulation Dof the Board of Governors as from time to time in effect (and any successor thereto):
hereunder or hereby, or	(1)	$any \textit{reserve}, deposit, or similar \textit{requirement} is \textit{or shall} \textit{be imposed or modified in respect of any Letter of Credit is sued or caused to \textit{be} i$
	(ii)	there shall be imposed on Issuing Lender or any other member of the Lender Group any other condition regarding any Letter of Credit,
received is reduced, notify Borrower, an cost or reduced receipt, together with inte	d Bonowershall pay within erest on such amount from t	and the result of the foregoing is to increase, directly or indirectly, the cost to Issuing Lender or any other member of the Lender Group of issuing buce the amount receivable in respect thereof, then, and in any such case, Agent may, at any time within a reasonable period after the additional cost is incurred or the amount 130 days after demand therefor, such anounts as Agent may specify to be necessary to compensate Issuing Lender or any other member of the Lender Group for such additional ne date of such demand until payment in full thereof at the rate then applicable to Base Rate Loans hereufer. The determination by Agent of any amount due pursuant to this thereof in treasonable detail, shall, in the absence of manifest or demonstrable error, be final and conclusive and binding on all of the parties hereto.

# 2.13 <u>LIBOR Option</u>.

(a) Interest and Interest Payment Dates. In lieu of having interest charged at the rate based upon the Base Rate, Borrower shall have the option, subject to Section 2.13(b) below (the "LIBOR Option") to have interest on all or a portion of the Advances, the Delayed Daw Term Loan or any Incerneral Term Loans be charged (whether at the time when made (unless otherwise provided herein), upon conversion from a Base Rate Loan to a LIBOR Rate Loan, or upon continuation of a LIBOR Rate Loans a LIBOR Rate Loan) at a rate of interest based upon the LIBOR Rate. Interest on LIBOR Rate Loans shall be payable on the earliest of (i) the last day of the Interest Period applicable thereto (provided, however, that, subject to the

- 61 -137020431v13 following clauses (ii) and (iii), in the case of any Interest Period greater than 3 months in duration, interest shall be payable at 3 month intervals after the commencement of the applicable Interest Period and on the last day of such Interest Period), (ii) the date on which all or any portion of the Obligations are accelerated pursuant to the terms hereof, or (iii) termination of this Agreement pursuant to the terms hereof. On the last day of each applicable Interest Period, unless Borrower properly has exercised the LIBOR Option with respect thereto, the interest rate applicable to such LIBOR Rate Loan automatically shall convert to the rate of interest then applicable to Base Rate Loans of the same type hereunder. At any time that an Event of Default has occurred and is continuing, Borrower no longer shall have the option to request that Advances, the Delayed Draw Term Loan or any Incremental Term Loans bear interest at a rate based upon the LIBOR Rate and Agent shall have the right to convert the interest rate on all outstanding LIBOR Rate Loans to the rate then applicable to Base Rate Loans hereunder.

#### (b) LIBOR Election

(i) Borrowermay, at any time and from time to time, so long as no Event of Defaulthas occurred and is continuing, elect to exercise the LIBOR Option by notifying Agent prior to 10:00 a.m. (California time) at least 3 Business Days prior to the commencement of the proposed Interest Period (the "<u>LIBOR Deadline</u>"). Notice of Borrower's election of the LIBOR Option for a permitted portion of the Advances, the Delayed Draw Term Loan or any Incremental Term Loans and an Interest Period pursuant to this Section shall be made by delivery to Agent of a LIBOR Notice received by Agent before the LIBOR Deadline (to be confirmed by delivery to Agent of a LIBOR Notice received by Agent prior to 5:00 p.m. (California time) on the same day). Promptly upon its receipt of each such LIBOR Notice, Agent shall provide a copy thereof to each of the Lendes having a Revolver Commitment.

(ii) Each LIBOR Noticeshall be intervocable and binding on Borrower. In connection with each LIBOR Rate Loan, Borrowershall indemnify, defend, and hold Agent and the Lendershamiless against any loss, cost, or expense incurred by Agent or any Lender as a result of (a) the payment of any principal of any LIBOR Rate Loan other than on the last day of an Interest Period applicable thereto, or (c) the failure to borrow, convert, continue or prepay any LIBOR Rate Loan on the date specified in any LIBOR Notice delivered pusuant hereto (such losses, costs, and expenses, collectively, "Furting Losses"). Funding Losses shall, with respect to Agent or any Lender, be deemed to equal the amount determined by Agent or such Lender to be the excess, if any, of

(i) the amount of interest that would have accrued on the principal amount of such LIBOR Rate Loan had such event not occurred, at the LIBOR Rate that would have been applicable thereto, for the period from the date of such event to the last day of the then current Interest Period therefor (or, in the case of a failure to borrow, convert, or continue, for the period that would have been the Interest Period therefor), *minus* (ii) the amount of interest that would accrue on such principal amount for such period at the interest rate which Agent or such Lender would be offered were it to be offered, at the commencement of such period, Dollar deposits of a comparable amount and period in the London interbank market. A certificate of Agent or a Lender delivered to Borrower setting forth any amount or amounts that Agent or such Lender is entitled to receive pursuant to this Section 2.13 shall be conclusive absent manifest error.

(iii) Borrower shall have not more than 5 LIBOR Rate Loans in effect at any given time. Borrower only may exercise the LIBOR Option for LIBOR Rate Loans of at least \$100,000 and integral multiples of \$100,000 in excess thereof.

Prepayments Bonowermay prepay LIBOR Rate Loans at any time; provided however; that in the event that LIBOR Rate Loans are prepaid on any date that is not the last day of the Interest Period applicable theeto, including as a result of any automatic prepayment through the required application by Agent of proceeds of Borrower's and its Subsidiaries' Collections in accordance with Section 24(b) or for any other neason, including early termination of the terms of this Agreement or acceleration of all or any portion of the Obligations pursuant to the terms hereof, Borrower shall indemnify, defend, and hold Agent and the Lenders and their Participants harmless against any and all Funding Losses in accordance with dause (b)(ii) above.

### Special Provisions Applicable to LIBOR Rate.

(i) The LIBOR Rate may be adjusted by Agent with respect to any Lender on a prospective basis to take into account any additional or increased costs to such Lender of maintaining or obtaining any eurodollar deposits or increased costs, in each case, due to changes in applicable law occurring subsequent to the commencement of the then applicable Interest Period, including any Changes in Law (including any changes in tax laws (except changes of general applicability in corporate income tax laws)) and changes in the reserve requirements imposed by the Board of Governors of the Federal Reserve System (or any successor), excluding the Reserve Percentage, which additional or increased costs would increase the cost of funding or maintaining loans bearing interest at the LIBOR Raie. In any such event, the affected Lender shall give Bonower and Agent notice of such a determination and adjustment and Agent promptly shall transmit the notice to each other Lender and, upon its receipt of the notice from the affected Lender; Borrower may, by notice to such affected Lender (A) require such Lender to furnish to Borrower a statement setting forth in reasonable detail the basis for adjusting such LIBOR Rate and the method for determining the amount of such adjustment, or (B) repay the LIBOR Rate Loans of such Lender with respect to which such adjustment is made (together with any amounts due under Section 2.13(b)(ii)).

In the event that any change in market conditions or any Change in Law shall at any time after the date hereof, in the reasonable opinion of any Lender, make it unlawfull or impractical for such Lender to fund or maintain LIBOR Rate Loans or to continue such funding or maintaining, or to determine or charge interestrates at the LIBOR Rate, such Lendershall give notice of such changed circumstances to Agent and Borrower and Agent promptly shall transmit the notice to each other Lender and (y) in the case of any LIBOR Rate Loans of such Lender that are outstanding, the date specified in such Lender's notice shall be deemed to be the last day of the Interest Period of such LIBOR Rate Loans, and interest upon the LIBOR Rate Loans of such Lender thereafter shall accuve interest at the rate then applicable to Base Rate Loans, and (2) Borrower shall not be entitled to elect  $the LIBOR\ Option\ until such\ Lender\ determines\ that it\ would no\ longer\ be\ unlawful\ or impractical\ to\ do\ so.$ 

If at any time the Agent determines (which determination shall be conclusive absent manifest error) that either (i) the circumstances set forth in

Section 2.13(d)(ii)

- 63 -137020431v13

have arisen and such circumstances are unlikely to be temporary or (ii) the circumstances set forth in Section 2.13(d)(ii) have not arisen but the supervisor for the administrator of LIBOR or a Governmental Authority having jurisdiction over the Agent has made a public statement identifying a specific date after which LIBOR shall no longer be used for determining interest rates for loans (in the case of either such clause (i) or (ii), an "Alternative Interest Rate Election Event"), the Agent shall designate an alternative rate of interest with the consent of Borrower, and shall enter into an amendment to this Agreement to reflect such alternate rate of interest and such other related changes to this Agreement as may be applicable. Notwithstanding anything to the contrary in Section 15.1. such amendment shall become effective without any further action or consent of any other party to this Agreement so long as the Agent shall not have received, within five (5) Business Days after the date notice of such alternate rate of interest is provided to the Lenders, a written notice from the Required Lenders stating that they object to such amendment (which amendment shall not be effective prior to the end of such five (5) Business Day notice period). To the extent an alternate rate of interest is adopted as contemplated hereby, the approved rate shall be applied in a manner consistent with prevailing market convention; provided that, to the extent such prevailing market convention; provided that, to the extent such prevailing market conventions of this paragraph, (x) any LIBOR Notice that requests the conversion of any Base Rate Loan to, or continuation of a LIBOR Rate Loan as, a LIBOR Rate Loan shall be ineffective, and (y) if any Notice of Borrowing requests a LIBOR Rate Loan, such Borrowing shall be made as a Base Rate Loan; provided that, to the extent such Alternative Interest Rate Election Event is as a result of clause (ii) above in this Section 2.13(c), then clauses (x) and (y) of this sentence shall apply durin

(e) No Requirement of Matched Funding. Anything to the contrary contained herein not with standing, neither Agent, nor any Lender, nor any Ortheir Participants, is required actually to acquire eurodollar deposits to fund or otherwise match funded any Obligation as to which interest is accurring at the LIBOR Rate Dya Rate Dya

#### 2.14 Capital Requirements

(a) If, after the date hereof, Issuing Lender or any Lender determines that

(i) any Change in Law regarding capital or reserve requirements for banks or bank holding companies, or (ii) compliance by Issuing Lender or such Lender, or their respective parent bank holding companies, with any guideline, request or directive of any Governmental Authority regarding capital adequacy (whether or not having the force of law), has the effect of reducing the return on Issuing Lender's, such Lender's, or such holding companies' capital as a consequence of Issuing Lender's or such Lender's commitments hereunder to a level below that

- 64 -137020431v13 which Issuing Lender, such Lender, or such holding companies could have achieved but for such Change in Law or compliance (taking into consideration Issuing Lender's, such Lender's, or such holding companies' then existing policies with respect to capital adequacy and assuming the full utilization of such entity's capital) by any amount deemed by Issuing Lender or such Lender to be material, then Issuing Lender or such Lender may notify Borrower and Agent in writing thereof. Following receipt of such written notice, Borrower agrees to pay Issuing Lender or such Lender on demand the amount of such reduction of return of capital as and when such reduction is determined, payable within 60 days after presentation by Issuing Lender or such Lender of a statement in the amount and setting forth in reasonable detail Issuing Lender's or such Lender's calculation thereof and the assumptions upon which such calculation was based (which statement shall be deemed true and correct absent manifest error). In determining such amount, Issuing Lender or such Lender may use any reasonable averaging and attribution methods. Failure or delay on the part of Issuing Lender to demand compensation pursuant to this Section shall not constitute a waiver of Issuing Lender's or such Lender's right to demand such compensation.

(b) If Issuing Lenderorany Lenderrequests additional orinnessed costs referred to in Section 2.13(d)(i) or amounts under Section 2.14(a) or sends anotice under Section 2.13(d)(ii) relative to changed circumstances or if the Borower's required to pay any Indemnified Taxes or additional amounts to any Lenderor any Governmental Authority for the account of any Lender pursuant to Section 16.11 (such Issuing Lenderor Lender, an' 'Affected Lenders'), then such Affected Lenders' all use measurable efforts to promptly designate addifinent one of its lending offices or to assign its injection of a sign in the reasonable independent and ner of its offices or branches, if (i) in the reasonable judgment of such Affected Lenders, such designation or assignment would not subject its one you applicable, or would eliminate the illegality or impractically officing or maintaining LIBOR Rate Loans and (iii) in the reasonable judgment of such Affected Lenders, such designation or assignment would not subject its one you are present would not always be materially disadvantage outs to it. Borrower agrees to pay all treasonable out-of-pode tooss and expenses incured by such Affected Lender incorrection with any such designation or assignment. If, after such reasonable out-of-pode tooss and expenses incured by such Affected Lender incorrection with any such designation or assignment. If, after such reasonable efforts, such Affected Lender or substitute and a such additional amounts to such Affected Lender pursuant to Section 2.12(b), Section 2.13(d)(i), Section 2.13(d)(i), Section 2.13(d)(i), Section 2.13(d)(i), Section 2.13(d)(i), Section 2.13(d)(i), Section 2.14(a) or Section 16.11, as applicable) may, unless prior to the effective date of any such assignment the Affected Lender vision in the additional amounts under Section 2.12(i), Section 2.13(d)(i), Section 2.14(a) or Section 16.11, as applicable, or include a vision such assignment and a vision such additional amounts under Section 2.12(i), Section 2.13(d)(i), Section 2.14(a) o

- 65 -137020431v13 Agreement and such Affected Lender shall cease to be "Issuing Lender" or a "Lender" (as the case may be) for purposes of this Agreement.

(c) Notwithstanding anything herein to the contrary, the protection of Sections 2.12(1), 2.13(d), 2.14 and 16.11 shall be available to Issuing Lender and each Lender (as applicable) regardless of any possible contention of the invalidity or inapplicability of the law, rule, regulation, judicial ruling judgment, guideline, treaty or other change or condition which shall have occurred or been imposed, so long as it shall be outstorn any for issuing banks or landers affected thereby to comply the ewith.

### 2.15 <u>Incremental Commitment</u>.

(a) Borowermay, by written notice to the Agent (each, a "Commitment Increase Notice"), at any time after the Delayed Draw Term Loan Commitment Expiration Date until six months prior to the Revolver Maturity Date and the Delayed Draw Term Loan Maturity Date, request the establishment of an ewterm loan facility (each, an "Incremental Term Loan Commitment") and collectively, the "Incremental Term Loan Commitments" and any such term loans, an "Incremental Term Loan" and collectively the "Incremental Term Loans"); purovided that:

the Agent has consented to the proposed Incremental Term Loans,

in writing;

(i)

(ii)

the aggregate principal amount of Incremental Term Loan

Commitments shall not exceed \$75,000,000 (the "Maximum Commitment Amount");

(iii)

(iv)

Borrower may exercise such increase request option up to two times after the Delayed Draw Term Loan Commitment Expiration Date during the term of this Agreement;

each exercise of the increase request option shall be in a minimum principal amount of not less than \$25,000,000 and integral multiples of \$1,000,000 in excess thereof;

(v) after giving effect to any applicable transaction permitted pursuant to Section 7.16(c) to be financed with such Incremental Term Loans, calculated on a pro forma basis (including the funding of such Incremental Term Loans and all other sources and uses of funds to be applied in consummation of such transaction), (A) no Default or Event of Default shall have occurred and be continuing or would result from the consummation of such transaction, (B) the Leverage Ratio of Borrower and its Subsidiaries as of the end of the fiscal quarter most recently ended as to which financial statements were required to be delivered pursuant to this Agreement is at least 0.25 less than the maximum Leverage Ratio permitted pursuant to Section 7.18(a)(ii) for such fiscal quarter and (C) Borrower and its Subsidiaries shall be in compliance on a pro forma basis with Sections 7.18(a)(ii), 7.18(a)(iii) and 7.18(b) for the 4 fiscal quarters (on a quarter-by-quarter basis) immediately following the incurrence of such Incremental Term Loans and the consummation of such transaction;

- 66 -137020431v13

the Agent shall have received an amendment to this Agreement, in form and substance acceptable to the Agent, to incorporate any changes Agent reasonably deems necessary with regard to such term loan facility;	(vii)
the Borrower shall have satisfied the applicable requirements specified in $\underbrace{\text{Section } 6.17}$ , in accordance therewith; and	(viii)
the Agent shall have received a certificate from a senior officer of the Borrower, in form acceptable to the	(ix)

(vi)

Incremental Term Loans shall be used by Borrower solely as permitted pursuant to Section 7.16(c);

Agent, as to the conditions set forth in clause (v) above.

- (b) Each such Commitment Incresse Notice shall specify (i) the date (the "<u>Incressed Amount Date</u>") on which Borrower proposes that the new Incremental Term Loan Commitment shall be effective, which shall be additionable so that is the Agent partie of the Agent part
- (c) UponieceptofaCommiment Incresse Notice, the Agent shall forward acopy thereof to each Lender, and each Lender shall have the right, but not the obligation, to participate in such incresse in an amount equal to its provides a sharp of control of the provided in the sharp of the provided incressed of the provided increased of

reasonably satisfactory to the Agent. Subject to the foregoing, final allocations with respect to any Incremental Term Loan Commitment shall be determined solely by Borrower and the Agent.

- (d) Any Incremental Term Loan Commitment shall be effected pursuant to as upplement to this Agreement executed by Borower, the Agent and the Loan Gommitment, in form and substance acceptable to the Agent (an "<u>Incremental Facility Supplement</u>"), which supplement shall set forth the terms and conditions relating to such Incremental Term Loan Commitment (including as to interest ate, amortization and any upfront or commitment fee), <u>provided</u> that:
  - (i) the final maturity date of any Incremental Term Loan Commitment shall be no earlier than the Revolver Maturity Date and the Delayed Draw Term Loan Maturity Date; and
  - (ii) except with respect to maturity as set forth in clause (i) above, any Incernental Term Loan Commitment shall be on terms consistent with the Delayed Draw Term Loans (including, without limitation, as to interest rate, amortization and any upfront or commitment fee).

Each Incremental Facility Supplement may, without the consent of any other Lenders, effect such amendments to this Agreement and the other Loan Documents as may be necessary, in the opinion of Agent, to effect the provisions of this Section 2.15. From and after the effectiveness of any Incremental Term Loan Commitment the Loans and Commitments established pursuant to this Section 2.15 shall constitute Loans and Commitments under, and shall be entitled to all the benefits afforded by, this Agreement and the other Loan Documents, and shall, without limiting the foregoing, benefit equally and ratably from any guarantees and the security interests created hereunder and the other applicable Loan Documents. Borrower shall take any actions reasonably required by Agent to ensure and demonstrate that the Liens and security interests granted by the Loan Documents continue to be perfected under the Code or otherwise after giving effect to the establishment of any such new Incremental Term Loan Commitments and Maximum Commitment Amount. Any Incremental Term Loans made on an Increased Amount Date shall be designated a separate series of term loans for all purposes of this Agreement. If so requested by any Lender having an Incremental Term Loan Commitment by written notice to Borrower (with a copy to the Agent), Borrower shall execute and deliver to such Lender a promissory note (an "Incremental Term Note") in form acceptable to the Agent to evidence such Lender's Incremental Term Loan Commitment.

### 3. CONDITIONS; TERM OF AGREEMENT.

- 3.1 <u>Conditions Precedent to the Initial Extension of Credit.</u> The obligation of each Lender to make its initial extension of credit provided for hereunder, is subject to the prior fulfillment, to the satisfaction of Agent and each Lender (the making of such initial extension of credit by a Lender being conclusively deemed to be its satisfaction or waiver of the following), of each of the following conditions precedent:
  - (a) the Restatement Effective Date shall occur on or before May 15, 2019;

- 68 -137020431v13

<b>(b)</b>	Ag	entshall have receive	edeach of the following documents, in form and substance satisfactory to Agent, duly executed, and each such documents hall be in full force and effect:
	(i)	a Notice of Bono	wing
	(ii)	the Disbursement	Letter;
	(iii)	the Fee Letter;	
(iv)	)		promissory notes evidencing  all  Commitments  and  Loans  of the  Lenders, as  of the  Restatement  Effective  Date;
	(v)	the Master Reaffin	mation Agreement,
	(vi)	Tiademark Securi	ity Agreement,
(vii necessary or desirable to evidence the transfer of all Rea		lownedbyFPFoo	an amended, restated and consolidated mortgage with respect to the existing Mortgages, together with any other documentation Agent deems als to Borrower, subject to Agent's continuing first priority Lien; and
(viii	i)		with respect to all Real Property Collateral owned by Borrower, or any Subsidiary thereof, on the Restatement Effective Date:
not less than \$30,000,000,		(A)	a lenders' policy of title insurance with an insured amount
or such Subsidiary expect to initiate on	any Real Prop		construction plans and budget for all projects that Borrower
\$60,000,000, in the aggregate, for all su	ch Real Prope		an appraisal evidencing a minimum appraised value of
relating thereto,	(D)		a Phase I environmental site assessment report, along with a reliance letter in favor of the Agent
	ent, such flood ir	ısurance in such ı	evidence as to whether any such Real Property Collateral is located in an area identified by the Federal hazard area" (or a similar designation) and, if it is, evidence that Borrower or such Subsidiary has obtained, with insurance reasonable total amount as the Agent may reasonably require, and otherwise sufficient to comply with all applicable rules and ble to the Agent,
		(F)	a survey with respect to Property A,
		(G)	a zoning opinion with respect to Property A,

- 69 -137020431v13

(H) an Environmental Indemnity, and (I) a legal opinion with respect to each existing Mortgage and the documents referred to in  $\underline{Section\ 3.1(b)}$ (vii) above, delivered by Borrower's or such Subsidiary's counsel in the State in which the Real Property Collateral is located, including, without limitation, Permsylvania; [Intentionally Omitted]; (ix) Joinder Agreement with respect to the Intercompany Subordination Agreement, by the Dutch Subsidiary and acknowledged by the Borrower and (x) the Agent; (xi) Pledged Interests Addendum with respect to the Dutch Subsidiary, by the Borrower; Agent shall have received a certificate from the Secretary of Borrower and each other Loan Party (i) attesting to the resolutions of such Person's Board of Directors authorizing its execution, delivery, and performance of this Agreement and the other Loan Documents to which such Person is a party, (iii) authorizing specific offices of such Person to execute the same, and (iii) attesting to the incumber cy and signatures of such specific Agent shall have received copies of Borrower's each other Loan Party's and each of their respective Subsidiaries' Governing Documents, as amended, modified, or supplemented tothe Restatement Effective Date, certified by the Secretary of such Person; Agent shall have received a certificate of status with respect to Borrower, each other Loan Party and each of their respective Subsidiaries, each dated within 10 days of the Restatement and the statement of their respective Subsidiaries, each dated within 10 days of the Restatement and their respective Subsidiaries, each dated within 10 days of the Restatement and their respective Subsidiaries, each dated within 10 days of the Restatement and their respective Subsidiaries, each dated within 10 days of the Restatement and their respective Subsidiaries, each dated within 10 days of the Restatement and their respective Subsidiaries, each dated within 10 days of the Restatement and their respective Subsidiaries, each dated within 10 days of the Restatement and their respective Subsidiaries, each dated within 10 days of the Restatement and their respective Subsidiaries, each dated within 10 days of the Restatement and their respective Subsidiaries, each dated within 10 days of the Restatement and their respective Subsidiaries, each dated within 10 days of the Restatement and their respective Subsidiaries, each dated within 10 days of the Restatement and their respective Subsidiaries and the Restatement and their respective Subsidiaries and the Restatement and the RestaEffective Date, such certificate to be issued by the appropriate officer of the jurisdiction of organization of such Person, which certificate shall indicate the such Person is in good standing in such jurisdiction; Agent shall have received certificates of status with respect to Borrower, each other Loan Party and each of their respective Subsidiaries, each dated within 30 days of the Resiatement Effective Date, such certificates to be issued by the appropriate officer of the jurisdictions (other than the jurisdiction of organization of such Person) in which such Person's failure to be duly qualified or licensed would constitute a Material Adverse Change, which certificates shall indicate that such Person is in good standing in such jurisdictions;  $Agent shall have received certificates of insurance and endorsements as are required by \underline{Section 68}, the formand substance of which shall be reasonably satisfactory to Agent;$ (g) (h) Agent shall have received an opinion of Borrower's counsel in form and substance reasonably satisfactory to Agent; Agentshall have received satisfactory evidence (including acertificate of the chief financial officer of Borrower) that all tax returns required to be filed by Borrower and (i)

- 70 -137020431v13

upon Borrower and its Subsidiaries or their properties, assets, income, and franchises (including (a) Real Property taxes and gate, reasonably be expected to result in a liability to Borrower or any of its Subsidiaries in excess of \$10,000, and (b) payroll ch taxes that are the subject of a Permitted Protest;
nrowershall have paid all Lender Group Expenses incurred in connection with the transactions evidenced by this Agreement and invoiced at least 2 Business Days prior to the
gentshall have received a Beneficial Ownership Certification in relation to any Loan Party, or Subsidiary thereof, requested by the Agent or any Lender at least 2 Business Days prior
nower and each of its Subsidianies shall have received all licenses, approvals or evidence of other actions required by any Governmental Authority in connection with the execution three consummation of the transactions contemplated thereby.
to the Initial Extension of Credit. The obligation of the Lender Group (or any member thereof) to continue to make Advances or to make der) is subject to the fulfillment, on or before the date applicable thereto, of each of the conditions subsequent set forth below (the failure by an Event of Default):
t later than December 31, 2019 (cr such later date as the Agent may agree) the Bourower shall deliver to the Agent evidence of completion of the FP Foods Dissolution in accordance
onowershall use commercially reasonable efforts to provide to Agant within 60 days of the Restatement Effective Date (or such later date as the Agant may agree) a Collateral athead Road, Bethlehem, PA 18020, (ii) 787 California Road, Qualvertown, PA 18951 and (iii) 303 West Marquette Avenue, Oak Greek, WI 53154; and

3.3 Conditions Precedent to all Extensions of Credit. The obligation of the Lender Group (or any member thereof) to make any Advances or Delayed Draw Term Loans hereunder at any time (or to extend any other credit hereunder) shall be subject to the following conditions precedent:

- (a) the representations and warranties of Borrower or its Subsidiaries contained in this Agreement or in the other Loan Documents shall be true and correct in all material respects (except that such material in qualifiers hall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof) or and as of the date of such extension of credit, as though made on and as of such date (except to the extent that such representations and warranties relate solely to an earlier date, in which case such representations and warranties that already are qualified or modified by materiality in the text thereof) as of such earlier date).
  - (b) no Default ar Event of Default shall have occurred and be continuing on the date of such extension of a edit, norshall either result from the making thereof;
- (c) no injunction, writ, restaining order, or other order of any nature restricting or prohibiting, directly or indirectly, the extending of such ore dirichall have been issued and remain in force by any Governmental Authority against Borrower, Agent, any Lender, or any of their Affiliates,
  - (d) no Material Adverse Change shall have occurred;
- (e) if any portion of the proceeds of such Advance or Delayed Draw Term Loan (or any other extension of credit hereunder) will be used to acquire any Real Property Collateral, all actions and conditions specified in Section 6.17 shall have been taken or satisfied, as applicable, with respect to such Real Property; and
- (f) solely with respect to Delayed Draw Term Loans, after giving proforma effect to any such Delayed Draw Term Loans, the Leverage Ratio of Bourower and its Subsidiaries shall not exceed the Leverage Ratio equived by Section 7.18(a)(iii) minus 0.25, and Bourower shall have delivered to the Agenta Compliance Certificate demonstrating compliance with such conclusion.
- 3.4 Term. This Agreement shall continue in full force and effect for a term ending on the later of the Revolver Maturity Date, the Delayed Draw Term Loan Maturity Date and the maturity date of any Incremental Term Loan. The foregoing notwithstanding, the Lender Group, upon the election of the Required Lenders, shall have the right to terminate its obligations under this Agreement immediately and without notice upon the occurrence and during the continuation of an Event of Default.
- 3.5 Effect of Termination. On the date of termination of this Agreement, pursuant to the provisions hereof, all commitments of the Lender Group to provide additional credit hereunder shall automatically be terminated and, except to the extent due and payable prior to such date pursuant to the provisions hereof, all Obligations (including contingent reimbursement obligations of Borrower with respect to outstanding Letters of Credit and including all Bank Product Obligations) immediately shall become due and payable without notice or demand (including (a) providing Letter of Credit Collateralization, and (b) providing Bank Product Collateralization). No termination of this Agreement, however, shall relieve or discharge

Borrower or its Subsidiaries of their duties, Obligations, or covenants hereunder or under any other Loan Document and the Agent's Liens in the Collateral shall remain in effect until all Obligations have been paid in full and the Lender Group's obligations to provide additional credit hereunder have been terminated. When this Agreement has been terminated and all of the Obligations have been paid in full and the Lender Group's obligations to provide additional credit under the Loan Documents have been terminated irrevocably, Agent will, at Borrower's sole expense, execute and deliver any termination statements, lien releases, mortgage releases, re-assignments of trademarks, discharges of security interests, and other similar discharge or release documents (and, if applicable, in recordable form) as are reasonably necessary to release, as of record, the Agent's Liens and all notices of security interests and liens previously filed by Agent with respect to the Obligations.

3.6 <u>Early Termination by Borrower</u>. Borrower has the option, at any time upon 3 Business Days prior written notice to Agent, to terminate this Agreement by paying to Agent, in cash, the Obligations (including (a) providing Letter of Credit Collateralization, and (b) providing Bank Product Collateralization), in full. If Borrower has sent a notice of termination pursuant to the provisions of this Section, then the Commitments shall terminate and Borrower shall be obligated to repay the Obligations (including (a) providing Letter of Credit Collateralization, and (b) providing Bank Product Collateralization), in full.

## 4. CREATION OF SECURITY INTEREST.

- 4.1 Grant of Security Interest. Borrower hereby grants to Agent, for the benefit of the Lender Group and the Bank Product Providers, a continuing security interest in all of its right, title, and interest in all currently existing and hereafter acquired or arising Borrower Collateral in order to secure prompt repayment of any and all of the Obligations in accordance with the terms and conditions of the Loan Documents and in order to secure prompt performance by Borrower of each of its covenants and duties under the Loan Documents. The Agent's Liens in and to the Borrower Collateral shall attach to all Borrower Collateral without further act on the part of Agent or Borrower. Anything contained in this Agreement or any other Loan Document to the contrary notwithstanding, except for Permitted Dispositions permitted pursuant to Section 7.4, Borrower and its Subsidiaries have no authority, express or implied, to dispose of any item or portion of the Collateral.
- 4.2 Negotiable Collateral. In the event that any Borrower Collateral, including proceeds, is evidenced by or consists of Negotiable Collateral, and if and to the extent that Agent determines that perfection or priority of Agent's security interest is dependent on or enhanced by possession, Borrower, promptly upon the request of Agent, shall endorse and deliver physical possession of such Negotiable Collateral to Agent.
- 4.3 <u>Collection of Accounts, General Intangibles, and Negotiable Collateral</u>. At any time after the occurrence and during the continuation of an Event of Default, Agent or Agent's designee may (a) notify Account Debtors of Borrower that Borrower's Accounts, chattel paper, or General Intangibles have been assigned to Agent or that Agent has a security interest therein, or (b) collect Borrower's Accounts, chattel paper, or General Intangibles directly and charge the collection costs and expenses to the Loan Account. Borrower agrees that it will hold in trust for the Lender Group, as the Lender Group's trustee, any of its or its Subsidiaries'

- 73 -137020431v13 Collections that it receives and immediately will deliver such Collections to Agent or a Cash Management Bank in their original form as received by Borrower or its Subsidiaries.

## 4.4 Filing of Financing Statements; Commercial Tort Claims; Delivery of Additional Documentation Required.

- (a) Borrower authorizes Agent at any time and from time to file, transmit, or communicate, as applicable, financing statements and amendments (i) describing the Borrower Collateral as being of equal or lessers cope or with greater detail, or (iii) that contain any information required by part 5 of Auticle 9 of the Code for the sufficiency or filing office acceptance. Borrower also heeby ratifies any and all financing statements or amendments previously filed by Agent in any jurisdiction.
- (b) If Borower or its Subsidiaries acquire any commercial tort dairns with a value in excess of \$250,000 after the date hereof, Borrower's hall promptly (but in any event within 3 Business Days after such acquisition) deliver to Agent, a written description of such commercial tort dairn and shall deliver a written agreement, in form and substance reasonably satisfactory to Agent, pursuant to which Borrower or its Subsidiary, as applicable, shall grant a perfected security interest in all of its right, title and interest in and to such commercial tort dairn to Agent, as security for the Obligations (a "Commercial Tort Claim Assignment").
- (c) At any time upon the request of Agent, Borrowershall execute or deliver to Agent, and shall cause its Subsidiaries to execute or deliver to Agent, any and all financing statements, original financing statements in lieu of continuation statements, amendments to financing statements, instrue filings, security agreements, pledges, assignments, and continue perfected on to better perfect the Agent's Liens in the assessor Borrowers (collectively, the "Additional Documents") that Agent may request in its Permitted Discretion, informands obstance reasonably satisfactory to Agent, to create perfect, and continue perfected on to better perfect the Agent's Liens in the assessor Borrower and its Subsidiaries (whether mow owned or hereafter arising or acquired, tangible orintangible, real or personal), to create end perfect. Liens in favor of Agent in any Real Property Collateral acquired after the Restatement Effective Date, and in order to fully consummate all of the transactions contemplated the edy and under the other Loan Documents. To the maximum extent permitted by applicable law, Borrower authorizes Agent to execute any such Additional Documents in Borrower is name and authorizes Agent to files uche executed Additional Documents in any appropriate filing office. In addition, on such periodic basis as Agent shall require, Borrower shall (i) provide Agent with a report of all new U.S. federal patent, applications, patents, registered copyrights, copyright applications, registered to Agent acquired, and delivered to Agent applicable Loan Documents to identify such patents, copyrights, and trademarks as being subject to the security interests created the eurobe; provided, however, that neither Borrower morany of its Subsidiaries shall register with the U.S. Copyright Office any unregistered copyrights and patent points and accurate the statement of Effective Date or the reafter acquired, aising or developed) unless within 20 days of the filing of an application for such registration, the applicable Pason executes a

supplemental schedules to any existing copyright security agreement, or such other documentation as Agent reasonably deems necessary in order to perfect and continue perfected Agent's Liens on such copyrights following such registration.

- 4.5 **Power of Attorney.** Borrower hereby invevocably makes, constitutes, and appoints Agent (and any of Agent's officers, employees, or agents designated by Agent) as Borrower's true and lawful attorney, with power to (a) if Borrower refuses to, or fails timely to execute and deliver any of the documents described in Section 4.4, sign the name of Borrower on any of the documents described in Section 4.4 (b) at any time that an Event of Default has occurred and is continuing, sign Borrower's name on any invoice or bill of lading relating to the Borrower Collateral, drafts against Account Debtors, or notices to Account Debtors, (c) send requests for verification of Borrower's or its Subsidiaries' Accounts, (d) endorse Borrower's name on any of its payment items (including all of its Collections) that may come into the Lender Group's possession, (e) at any time that an Event of Default has occurred and is continuing, make, settle, and adjust all claims under Borrower's policies of insurance and make all determinations and decisions with respect to such policies of insurance, and (f) at any time that an Event of Default has occurred and is continuing, settle and adjust disputes and claims respecting Borrower's or its Subsidiaries' Accounts, chattel paper, or General Intangibles directly with Account Debtors, for amounts and upon terms that Agent determines to be reasonable, and Agent may cause to be executed and delivered any documents and releases that Agent determines to be necessary. The appointment of Agent as Borrower's attorney, and each and every one of its rights and powers, being coupled with an interest, is invevocable until all of the Obligations have been fully and finally repaid and performed and the Lender Group's obligations to extend credit hereunder are terminated.
- 4.6 Right to Inspect. Agent and each Lender (through any of their respective officers, employees, or agents) shall have the right, from time to time hereafter to inspect the Books and make copies or abstracts thereof and to check, test, and appraise the Collateral, or any portion thereof, in order to verify Borrower's and its Subsidiaries' financial condition or the amount, quality, value, condition of, or any other matter relating to, the Collateral.
- 4.7 <u>Control Agreements.</u> Borrower agrees that it will and will cause its Subsidiaries to take commercially reasonable steps in order for Agent to obtain control in accordance with Sections 8-106, 9-104, 9-105, 9-106, and 9-107 of the Code with respect to (subject to the proviso contained in <u>Section 7.12</u>) all of its or their Securities Accounts, Deposit Accounts, electronic chattel paper, Investment Property, and letter-of-credit rights; <u>provided</u> that (i) Borrower may keep up to \$25,000 in the aggregate in accounts that are not subject to the foregoing and (ii) the requirements in this Section 4.7 or Section 2.7 shall not apply to Excluded Deposit Accounts. Upon the occurrence and during the continuance of an Event of Default, Agent may notify any bank or securities intermediary to liquidate the applicable Deposit Account or Securities Account or any related Investment Property maintained or held thereby and remit the proceeds thereof to the Agent's Account.

## 5. REPRESENTATIONS AND WARRANTIES.

In order to induce the Lender Group to enter into this Agreement, Borrower makes the following representations and warranties to the Lender Group which shall be true,

- 75 -137020431v13 correct, and complete, in all material respects (except that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof), as of the Restatement Effective Date, and shall be true, correct, and complete, in all material respects (except that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof), as of the date of the making of each Advance or Delayed Draw Term Loan (or other extension of credit) made thereafter, as though made on and as of the date of such Advance or Delayed Draw Term Loan (or other extension of credit) (except to the extent that such representations and warranties relate solely to an earlier date, in which case such representations and warranties shall be true and correct in all material respects (except that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof) as of such earlier date) and such representations and warranties shall survive the execution and delivery of this Agreement:

- 5.1 No Encumbrances. Borrower and its Subsidiaries have good and indefeasible title to, or a valid leasehold interest in, their personal property assets and good and marketable title to, or a valid leasehold interest in, their Real Property, in each case, free and clear of Liens except for Permitted Liens.
  - 5.2 [Intentionally Omitted].
  - 5.3 FP Foods. As of the Restatement Effective Date, FP Foods does not own any assets or conduct any business.
- 5.4 Equipment. All of the Equipment of Borrower and its Subsidiaries that is necessary or material to conduct their business is used or held for use in their business and is fit for such purposes, except for any failure to comply with the foregoing with respect to Equipment with a market value less than or equal to \$250,000.
- 5.5 <u>Location of Inventory and Equipment.</u> The Inventory and Equipment of Borrower and its Subsidiaries (other than (w) Equipment out for repair, (x) Equipment in transit between such locations, (y) Chillers located at one or more of the Installed Stores and (z) other Equipment with an after market value not in excess of \$500,000 in the aggregate) are not stored with a bailee, warehouseman, or similar party and (other than (w) Equipment out for repair, (x) Equipment in transit between such locations, (y) Chillers located at one or more of the Installed Stores and (z) other Equipment with an after market value not in excess of \$500,000 in the aggregate) are located only at, or in-transit between, the locations identified on <u>Schedule 5.5</u> (as such Schedule may be updated pursuant to <u>Section 6.9</u>).
- 5.6 **Inventory\_Records.** Borrower keeps correct and accurate records itemizing and describing the type, quality, and quantity of its and its Subsidiaries' Inventory and the book value thereof in all material respects.
  - 5.7 State of Incorporation; Location of Chief Executive Office; Organizational Identification Number; Commercial Tort Claims.

- 76 -137020431v13

to Agent.	(a)	$The jurisdiction of aganization of Bourower and each of its Subsidiaries is set for thon \underline{Schedule 5.7(a)}, as such Schedule may be amended upon no less than 30 days prior written notice$
	(b)	$The chief executive of fice of Bourover and each of its Subsidiaries is located at the address indicated on \underline{Schedule 5.7(b)} (as such Schedule may be updated pursuant to Section 6.9).$
5.7(c), as such Schedule may be	(c) eamended upon no less than 30 days pi	Bourower's and each of its Subsidiaries' organizational identification numbers in their jurisdiction of organization and employer identification number; if any, are identified on Schedule ior written notice to Agent.
upon no less than 30 days prior	(d) written notice to Agent.	$As of the \textit{Restatement Effective Date}, \textit{Borrower and its Subsidiaries} do not hold any commercial tort dains, except as set for thon \underline{Schedule 5.7(d)}, as such Schedule may be amended any commercial tort dains, except as set for thou \underline{Schedule 5.7(d)}, as such Schedule may be amended any commercial tort dains, except as set for thou \underline{Schedule 5.7(d)}, as such Schedule may be amended any commercial tort dains, except as set for thou \underline{Schedule 5.7(d)}, as such Schedule may be amended any commercial tort dains, except as set for thou \underline{Schedule 5.7(d)}, as such Schedule may be amended any commercial tort dains, except as set for thou \underline{Schedule 5.7(d)}, as such Schedule may be amended as a such Schedule may be a set for thou \underline{Schedule 5.7(d)}, as such Schedule may be a set for thou \underline{Schedule 5.7(d)}, as such Schedule may be a set for thou \underline{Schedule 5.7(d)}, as such Schedule may be a set for thou \underline{Schedule 5.7(d)}, as such Schedule may be a set for thou \underline{Schedule 5.7(d)}, as such Schedule may be a set for thou \underline{Schedule 5.7(d)}, as such Schedule may be a set for thou \underline{Schedule 5.7(d)}, as such Schedule may be a set for \underline{Schedule 5.7(d)}, as such Schedule may be a set for \underline{Schedule 5.7(d)}, as such Schedule may be a set for \underline{Schedule 5.7(d)}, as such Schedule may be a set for \underline{Schedule 5.7(d)}, as such Schedule may be a set for \underline{Schedule 5.7(d)}, as such Schedule may be a set for \underline{Schedule 5.7(d)}, as such Schedule may be a set for \underline{Schedule 5.7(d)}, as such Schedule may be a set for \underline{Schedule 5.7(d)}, as such Schedule may be a set for \underline{Schedule 5.7(d)}, as such Schedule may be a set for \underline{Schedule 5.7(d)}, as such Schedule may be a set for \underline{Schedule 5.7(d)}, as such Schedule may be a set for \underline{Schedule 5.7(d)}, as such Schedule may be a set for \underline{Schedule 5.7(d)}, as such Schedule may be a set for \underline{Schedule 5.7(d)}, as such Schedule may be a set for \underline{Schedule 5.7(d)}, as such Schedule may be a set for \underline{Schedule 5.7(d)}, as such Schedule may be a set for \underline{Schedule 5.7(d)}, as such Sched$
5.8	Due Organization and O	Qualification; Subsidiaries.
qualified ressonably could be ex	(a) spected to result in a Material Adverse C	Borrower is duly organized and existing and in good standing under the laws of the jurisdiction of its organization and qualified to do business in any state where the failure to be so hange.
	(b) [Intentional	y Omitted].
		Set forthon <u>Scheckle 5.9(c)</u> , as such Scheckle may be amanded upon no less than 30 days prior written notice to Agent, is a complete and accurate list of Bourower's direct and inclined in her of shares of each dass of common and preferred Stock authorized for each of such Subsidiaries, and (iii) the number and the percentage of the outstanding shares of each such dass Stock of each such Subsidiary has been validly issued and is fully paid and in the case of a corporation, non-assessable.
		Except asset forth on <u>Scheck lie 58(d)</u> , as such Schedule may be amended upon no less than 30 days prior written notice to Agent there are no subscriptions, options, warrants, or calls gany right of conversion or exchange under any outstanding security or other instrument. Neither Borrower nor any of its Subsidiaries is subject to any obligation (contingent or otherwise) beidiaries' capital Stock or any security convertible into or exchangeable for any such capital Stock.
5.9	<u>Due Authorization; No</u>	Conflict.
of Bonower:	(a)	The execution, delivery, and performance by Borrower of this Agreement and the Loan Documents to which it is a party have been duly authorized by all necessary action on the part of the execution, delivery, and performance by Borrower of this Agreement and the Loan Documents to which it is a party have been duly authorized by all necessary action on the part of the execution, delivery, and performance by Borrower of this Agreement and the Loan Documents to which it is a party have been duly authorized by all necessary action on the part of the execution of th
- 77 - 137020431v13		

- (b) The execution, delivery, and performance by Borrower of this Agreement and the other Loan Documents to which it is a party do not and will not (i) violate any provision of federal, state, or local law or regulation applicable to Borrower, (ii) conflict with, result in a breach of, or constitute (with due notice or lapse of time or both) a default under any contractual obligation of Borrower, (iii) result in or require the creation or imposition of any nature what sever upon any properties or assets of Borrower, other than Permitted Liens, or (iv) require any approval of Borrower's interest holdes or any approval or consent of any Person under any contractual obligation of Borrower, other than consents or approvals that have been obtained and that are still in force and effect except in the case of clauses (ii) or (iv) where the failure to do so could not reasonably be expected to result in a Material Adverse Change.
- (c) Other than the filing of financing statements and the recordation of the Mortgages, the execution, delivery, and performance by Borower of this Agreement and the other Loan Documents to which Borower is a party do not and will not require any registration with, consent, or approval of, or notice to, or other action with or by, any Governmental Authority or any other Person, other than consents or approvals that have been obtained and that a restill inforce and effect.
- (d) This Agreement and the other Loan Documents to which Bourower is a party, and all other documents contemplated hereby, when executed and delivered by Bourower will be the legally valid and binding obligations of Bourower, enforceable against Bourower in accordance with their respective terms, except as enforcement may be limited by equitable principles or by bankruptcy, insolvency, recognization, moratorium, or similar laws relating to or limiting aceditors' rights generally.
- (e) The Agent's Liens are validly orated, perfected, and first priority Liens, subject only to Permitted Liens (with respect to copyrights, trademarks, and patents, if and to the extent that perfection can be achieved by UCC filings or filings with the U.S. Patent and Tiademark Office or U.S. Copyright Office, as applicable).
- 5.10 Litigation. Other than those matters described on Schedule 5.10 and other than matters arising after the Restatement Effective Date that reasonably could not be expected to result in a Material Adverse Change, there are no actions, suits, investigations or proceedings pending or, to the best knowledge of Borrower, threatened against Borrower or any of its Subsidiaries.
- 5.11 No Material Adverse Change. All financial statements relating to Borrower and its Subsidiaries that have been delivered by Borrower to Agent have been prepared in accordance with GAAP (except, in the case of unaudited financial statements, for the lack of footnotes and being subject to year-end audit adjustments and reclassifications) and present fairly in all material respects, Borrower's and its Subsidiaries' financial condition as of the date thereof and results of operations for the period then ended. There has not been a Material Adverse Change with respect to Borrower and its Subsidiaries since the date of the latest financial statements submitted to Agent on or before the Restatement Effective Date.

- 78 -137020431v13

### 5.12 Fraudulent Transfer.

- (a) The Loan Parties on a consolidated basis are Solvent.
- (b) Notarsferof property is being made by Bourower or its Subsidiaries and morbligation is being incurred by Bourower or its Subsidiaries in connection with the transactions contemplated by this Agreement or the other Loan Documents with the intent to hinder, delay, or definant either present or future creditors of Bourower or its Subsidiaries.
  - 5.13 Employee Benefits. None of Borrower, any of its Subsidiaries, or any of their ERISA Affiliates maintains or contributes to any Benefit Plan.
- 5.14 Environmental Condition. Except as set forth on Schedule 5.14, (a) to Borrower's knowledge, none of Borrower's or its Subsidiaries' properties or assets has ever been used by Borrower, its Subsidiaries, or by previous owners or operators in the disposal of, or to produce, store, handle, treat, release, or transport, any Hazardous Materials, where such use, production, storage, handling, treatment, release or transport was in violation, in any material respect, of any applicable Environmental Law, (b) to Borrower's knowledge, none of Borrower's or its Subsidiaries' properties or assets has ever been designated or identified in any manner pursuant to any environmental protection statute as a Hazardous Materials disposal site, (c) neither Borrower nor any of its Subsidiaries has received notice that a Lien arising under any Environmental Law has attached to any revenues or to any Real Property owned or operated by Borrower or its Subsidiaries, and (d) neither Borrower nor its Subsidiaries has received a summons, citation, notice, or directive from the United States Environmental Protection Agency or any other federal or state governmental agency concerning any action or omission by Borrower or its Subsidiaries resulting in the releasing or disposing of Hazardous Materials into the environment that individually or in the aggregate could reasonably be expected to result in a Material Adverse Change.
- 5.15 <u>Brokerage Fees.</u> Neither Borrower nor any of its Subsidiaries has utilized the services of any broker or finder in connection with Borrower's obtaining financing from the Lender Group under this Agreement and no brokerage commission or finder's fee is payable by Borrower or its Subsidiaries in connection herewith.
- 5.16 Intellectual Property. Borrower and its Subsidiaries own, hold licenses in or otherwise have a right to use all trademarks, trade names, copyrights, patents, patent rights, and licenses that are necessary to the conduct of its business as currently conducted, and attached hereto as Schedule 5.16 (as updated from time to time) is a true, correct, and complete listing of all material patents, patent applications, registered trademarks, trademark applications, copyright registrations and copyright applications as to which Borrower or one of its Subsidiaries is the owner or is an exclusive licensee.
- 5.17 Leases. Borrower and its Subsidiaries enjoy peaceful and undisturbed possession under all leases material to their business and to which they are parties or under which they are operating, and all of such leases are valid and subsisting and no material default by Borrower or its Subsidiaries exists under any of them.

- 79 -137020431v13

5.18	Deposit Accounts and Securities Accounts. Set forth on Schedule 5.18 (as updated from time to time) is a listing of all of Borrower's and its Subsidiaries' Deposit
Accounts and Securities Accoun	ts, including, with respect to each bank or securities intermediary (a) the name and address of such Person, and (b) the account numbers of the Deposit Accounts or Securitie
Accounts maintained with such 1	Person.

- Complete Disclosure. All factual information (taken as a whole) furnished by or on behalf of Borrower or its Subsidiaries in writing to Agent or any Lender (including all information contained in the Schedules hereto or in the other Loan Documents) for purposes of or in connection with this Agreement, the other Loan Documents, or any transaction contemplated herein or therein is, and all other such factual information (taken as a whole) hereafter furnished by or on behalf of Borrower or its Subsidiaries in writing to Agent or any Lender will be, true and accurate, in all material respects, on the date as of which such information is dated or certified and not incomplete by omitting to state any fact necessary to make such information (taken as a whole) not misleading in any material respect at such time in light of the circumstances under which such information was provided. On the Restatement Effective Date, the Restatement Effective Date Projections represent, and as of the date on which any other Projections are delivered to Agent, such additional Projections represent Borrower's good faith estimate of its and its Subsidiaries' future performance for the periods covered thereby (it being understood that actual results may vary from such forecasts and that such variances may be material).
- 5.20 Indebtedness. Set forth on Schedule 5.20 is a true and complete list of all Indebtedness of Borrower and its Subsidiaries outstanding immediately prior to the Restatement Effective Date that is to remain outstanding after the Restatement Effective Date and such Schedule accurately reflects the aggregate principal amount of such Indebtedness and describes the principal terms thereof.
- 5.21 Margin Stock. No Loan Party nor any of its Subsidiaries is engaged principally, or as one of its important activities, in the business of extending credit for the purpose of purchasing or carrying any Margin Stock. No part of the proceeds of the loans made to Borrower will be used to purchase or carry any Margin Stock or to extend credit to others for the purpose of purchasing or carrying any Margin Stock or for any purpose that violates the provisions of Regulation T, U or X of the Board of Governors.
- 5.22 Governmental Regulation. No Loan Party nor any of its Subsidiaries is subject to regulation under the Federal Power Act or the Investment Company Act of 1940 or under any other federal or state statute or regulation which may limit its ability to incur Indebtedness or which may otherwise render all or any portion of the Obligations unenforceable. No Loan Party nor any of its Subsidiaries is a "registered investment company" or a company "controlled" by a "registered investment company" or a "principal underwriter" of a "registered investment company" as such terms are defined in the Investment Company Act of 1940.
- 5.23 OFAC. No Loan Party nor any of its Subsidiaries is in violation of any of the country or list based economic and trade sanctions administered and enforced by OFAC. No Loan Party nor any of its Subsidiaries (a) is a Sanctioned Person or a Sanctioned Entity, (b) has its assets located in Sanctioned Entities, or (c) derives revenues from investments in, or

- 80 -137020431v13 transactions with Sanctioned Persons or Sanctioned Entities. No proceeds of any loan made hereunder will be used to fund any operations in, finance any investments or activities in, or make any payments to, a Sanctioned Person or a Sanctioned Entity.

- 5.24 Patriot Act, Anti-Corruption Laws and Beneficial Ownership Regulation. To the extent applicable, each Loan Party is in compliance, in all material respects, with the (a) Trading with the Enemy Act, as amended, and each of the foreign assets control regulations of the United States Treasury Department (31 CFR, Subtitle B, Chapter V, as amended) and any other enabling legislation or executive order relating thereto, (b) Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism (USA Patriot Act of 2001) (the "Patriot Act") and (c) the FCPA and any other applicable anti- corruption laws. No part of the proceeds of the loans made hereunder will be used by any Loan Party or any of their Affiliates, directly or indirectly, for any payments to any governmental official or employee, political party, official of a political party, candidate for political office, or anyone else acting in an official capacity, in order to obtain, retain or direct business or obtain any improper advantage, in violation of the United States Foreign Corrupt Practices Act of 1977, as amended, or the rules and regulations thereunder (collectively, the "FCPA"), or any other applicable anti-corruption law. As of the Restatement Effective Date, the information included in the Beneficial Ownership Certification, if applicable, is true and correct in all respects.
  - 5.25 No EEA Financial Institution. No Loan Party is an EEA Financial Institution.

### 6. AFFIRMATIVE COVENANTS.

Borrower covenants and agrees that, until termination of all of the Commitments and payment in full of the Obligations, Borrower shall and shall cause each of its Subsidiaries to do all of the following:

- 6.1 Accounting System. Maintain a system of accounting that enables Borrower to produce financial statements in accordance with GAAP and maintain records pertaining to the Collateral that contain information as from time to time reasonably may be requested by Agent. Borrower also shall keep a reporting system that shows all additions, sales, claims, returns, and allowances with respect to its and its Subsidiaries' sales.
- 6.2 <u>Collateral Reporting</u>. Provide Agent (and if so requested by Agent, with copies for each Lender) with the following documents upon reasonable request by the Agent, in form reasonably satisfactory to Agent:
  - (a) adetailed list of Borrower's and its Subsidiaries' customers, including number of Installed Store locations, and
  - (b) such other reports as to the Collateral or the financial condition of Bourower and its Subsidiaries, as Agent may request.
  - 6.3 **Financial Statements, Reports, Certificates.** Deliver to Agent, with copies to each Lender:

(a)	wit	nin 45 days after the end of each of the first three quarters during each of Borrower's fiscal years,
(i) such period and for the three month period then ended,		an unaudited consolidated balance sheet, income statement, and statement of cash flow covering Borrower's and its Subsidiaries' operations during
(ii) and any construction updates for such period, and		management discussion and analysis report of Borrower and its Subsidiaries, describing in reasonable detail their operations and financial condition
	(iii)	a Compliance Certificate;
provided that the foregoing clauses (i) a	nd (ii) shall	the satisfied by filing of Borrower's 10-0 with the SEC including such financial statements and reports (and the public filin

provided, that the foregoing clauses (i) and (ii) shall be satisfied by filing of Borrower's 10-Q with the SEC including such financial statements and reports (and the public filing of such report with the SEC shall constitute delivery under this Section 6.3(a));

(b) within 120 days after the end of each of Borrower's fiscal years (for the avoidance of doubt, including Borrower's fiscal year ended December 31, 2018),

consolidated financial statements of Borrower and its Subsidiaries for each such fiscal year, audited by independent certified public accountants reasonably acceptable to Agent and certified, without any qualifications (including any (A) "going concern" or like equalification or exception (except to the extent that such qualification or exception is cheesely to the fact that the Revolver Maturity Date or the Delayed Draw Term Loan Maturity Date at the time of such audit its scheduled to occur within when on the sort of such fiscal year), (B) qualification or exception as to the scope of such audit, or (C) qualification which relates to the tenter the processor of such audit its scheduled to occur within the end of such fiscal year), (B) qualification or exception as to the scope of such audit on the tente the tenter and to such a such as a condition to the removal of such qualification, would require an adjustment to such tiem, the effect of which would be to cause any noncompliance with the provisions of <u>Section 7.18</u>), by such accountants to have been prepared in accordance with GAAP (such audited financial statements to include a balance sheet, income statement, and statement of cash flow and, if prepared, such accountants' letter to management) and

(ii) a Compliance Certificate;

<u>provided</u>, that the foregoing clause (i) shall be satisfied by filing of Borrower's 10-K with the SEC including such financial statements (and the public filing of such report with the SEC shall constitute delivery under this Section 6.3(b));

(c) within 60 days after the start of each of Borrower's fiscal years, copies of Borrower's Projections, in form (including as to scope) satisfactory to Agent, in its Permitted Discretion, for the forthcoming fiscal year, month by month, certified by the chief financial officer of Borrower as being such officer's good faith estimate of the financial performance of Borrower during the period covered thereby,

(d) if and when filed by Borrower;

- 82 -137020431v13 Form 10-Q quarterly reports, Form 10-K annual reports, and Form 8-K current reports, and

- (i)
  - (ii) any other filings made by Borrower with the SEC;
- (e) promptly, but in any event within 5 days after a senior officer of Bonowerhas knowledge of any event or condition that constitutes a Default or an Event of Default, notice thereof and a statement of the curative action that Bonower proposes to take with respect thereto (if any);
- (f) promptly after the commencement thereof, but in any event within 5 days after the service of process with respect thereto on Borrower or any of its Subsidiaries, notice of all actions, suits, or proceedings brought by or against Borrower or any of its Subsidiaries before any Governmental Authority which reasonably could be expected to result in a Material Adverse Change;
- (g) promptly following any request therefor; provide information and documentation reasonably requested by the Agent or any Lender for purposes of compliance with applicable "knowyour customer" and anti-money-kundering rules and regulations, including without limitation, the PATRIOT Act and the Beneficial Ownership Regulation; and
  - (h) upon the request of Agent, any other information reasonably requested relating to the financial condition of Borrower or its Subsidiaries.

In addition, Borrower agrees that no Subsidiary of Borrower will have a fiscal year different from that of Borrower. Borrower also agrees to cooperate with Agent to allow Agent to consult with its independent certified public accountants if Agent reasonably requests the right to do so and that, in such connection, its independent certified public accountants are authorized to communicate with Agent and to release to Agent whatever financial information concerning Borrower or its Subsidiaries that Agent reasonably may request.

6.4 <u>Guarantor Reports.</u> Cause each Guarantor to deliver its annual financial statements at the time when Borrower provides its audited financial statements to Agent, but only to the extent such Guarantor's financial statements are not consolidated with Borrower's financial statements.

#### 6.5 [Intentionally Omitted]

- 6.6 Maintenance of Properties. Maintain and preserve all of its properties which are necessary or material to the proper conduct to its business in good working order and condition, ordinary wear and tear excepted, and comply at all times with the provisions of all material leases to which it is a party as lessee, so as to prevent any loss or forfeiture thereof or thereunder, except for any failure to comply with the foregoing with respect to properties with a market value less than or equal to \$250,000.
- 6.7 Taxes. Cause all assessments and taxes, whether real, personal, or otherwise, due or payable by, or imposed, levied, or assessed against Borrower, its Subsidiaries, or any of their respective assets to be paid in full, before delinquency or before the expiration of any extension

- 83 -137020431v13 period, except to the extent that the validity of such assessment or tax shall be the subject of a Permitted Protest. Borrower will and will cause its Subsidiaries to make timely payment or deposit of all tax payments and withholding taxes required of it and them by applicable laws, including those laws concerning F.I.C.A., F.U.T.A., state disability, and local, state, and federal income taxes, and will, upon request, furnish Agent with proof satisfactory to Agent indicating that Borrower and its Subsidiaries have made such payments or deposits.

## 6.8 <u>Insurance</u>.

- (a) At Borower's expense, maintain insurance respecting its and its Subsidiaries' assets where vertocated, covering loss or damage by fire, theft, explosion, and all other hezards and its sacdinarily are insured against by other Persons engaged in the same or similar businesses. Borrower also shall maintain business interruption, public liability, and product liability insurance, as well as insurance against larceny, embezzlement, and oin intal misppropriation. All such policies of insurance shall be insuch an ounts and with such insurance companies as are reasonably satisfactory to Agent (being understood and agreed that the insurance company that provides such insurance company that provides a section of the policy of company that provides such insurance company that provides a section of the policy of company that provides such insurance company that provides a section of the policy of company that provides such insurance company that provides a section of the policy of company that provides such as the provides of the provides such as the provides such as the provides such as the provides such as the provides of the provides such as the provi
- (b) Borrower shall give Agent prompt notice of any loss in excess of \$250,000 covered by such insurance. Agent shall have the exclusive right to adjust any losses claimed under any such insurance policies after the occurrence and during the continuation of an Event of Default, without any liability to Borrower whatsoever in respect of such adjustments. Any monies received as payment for any loss under any insurance policy mentioned above (other than liability insurance policies) or as payment of any award or compensation for condemnation or taking by eminent domain shall be applied as set forth in, subject to Section 2.4(c).
- (c) Borrower will not suffer or permit its Subsidiaries to take out separate insurance concurrent in form or contributing in the event of loss with that required to be maintained under this <u>Section 68</u>, unless Agent is included thereon as an additional insured or loss payee under a lender's loss payable endorsement. Borrower promptly shall notify Agent whenever such separate insurance is taken out, specifying the insurenthere under and full particulars as to the policies evidencing the same, and copies of such policies promptly shall be provided to Agent.
- (d) With respect to any Real Property Collateral located in an area identified by the Federal Emergency Management Agency (or any successor agency) as a "special flood hazard area" (or a similar designation), Borrowershall, and shall cause its Subsidiaries to, provide to Agent evidence that Borrower or such Subsidiary maintains, with insurance companies as are reasonably satisfactory to Agent, such flood insurance in such reasonable total amount and informand substance as the Agent may from time to time reasonably require, and

otherwise sufficient to comply with the Flood Insurance Laws and all other applicable rules and regulations relating to flood insurance. At the request of the Agent, Borrower shall deliver copies of all such policies to the Agent, including evidence of annual renewals of such insurance.

- Location of Inventory and Equipment. Keep Borrower's and its Subsidiaries' Inventory and Equipment (other than (w) Equipment out for repair, (x) Equipment in transit between such locations, (y) Chillers located at, or in transit to, one or more of the Installed Stores and (z) other Equipment with an after market value not in excess of \$500,000 in the aggregate) only at the locations identified on Schedule 5.5 and their chief executive offices only at the locations identified on Schedule 5.7(b); provided, however, that Borrower may amend Schedule 5.5 and Schedule 5.7(b) so long as such amendment occurs by written notice to Agent not less than 30 after the date on which such Inventory or Equipment is moved to such new location or such chief executive office is relocated, so long as such new location is within the continental United States, and so long as, at the time of such written notification, Borrower uses commercially reasonable efforts to provide to Agent a Collateral Access Agreement with respect thereto (other than with respect to such locations that are subject to a Mortgage) (for the avoidance of doubt, a Collateral Access Agreement will not be required with respect to any locations described in clauses (w) through (z) in the parenthetical above)
- 6.10 <u>Compliance with Laws</u>. Comply with the requirements of all applicable laws, rules, regulations, and orders of any Governmental Authority, other than laws, rules, regulations, and orders the non-compliance with which, individually or in the aggregate, could not reasonably be expected to result in a Material Adverse Change.
- 6.11 Leases. Pay when due all rents and other amounts, in each case to the extent that such amounts that are not paid when due exceed \$250,000 in the aggregate at any one time, payable under any material leases to which Borrower or any of its Subsidiaries is a party or by which Borrower's or any such Subsidiaries' properties and assets are bound, unless such payments are the subject of a Permitted Protest.
- 6.12 Existence. At all times preserve and keep in full force and effect Borrower's and its Subsidiaries' existence and good standing and any rights and franchises material to their businesses; provided, however, that neither Borrower or any of its Subsidiaries shall be required to preserve any such right or franchise if such Person's board of directors (or similar governing body) shall determine that the preservation thereof is no longer desirable in the conduct of the business of such Person, and that the loss thereof is not disadvantageous in any material respect to such Person or to the Lenders; provided further that this Section 6.12 shall not prohibit or otherwise limit the Canadian Subsidiary Dissolution or the FP Foods Dissolution, in each case, conducted in accordance with the terms hereof.
- 6.13 Environmental. (a) Keep any property either owned or operated by Borrower or its Subsidiaries free of any Environmental Liens which could, individually or in the aggregate, reasonably be expected to secure a liability to Borrower or any of its Subsidiaries in excess of \$500,000, or post bonds or other financial assurances sufficient to satisfy the obligations or liability evidenced by such Environmental Liens, (b) comply with Environmental Laws where any such failure to comply could, individually or in the aggregate, reasonably be expected to result in a liability to Borrower or any of its Subsidiaries in excess of \$500,000 and provide to

Agent documentation of such compliance which Agent reasonably requests, (c) promptly notify Agent of any release of a Hazardous Material in any reportable quantity from or onto property owned or operated by Borrower or its Subsidiaries which could, individually or in the aggregate, reasonably be expected to result in a liability to Borrower or any of its Subsidiaries in excess of

\$500,000 and take any Remedial Actions required to abate said release or otherwise to come into compliance with applicable Environmental Law, and (d) promptly, but in any event within 5 days of its receipt thereof, provide Agent with written notice of any of the following: (i) notice that an Environmental Lien which could, individually or in the aggregate, reasonably be expected to secure a liability to Borrower or any of its Subsidiaries in excess of \$500,000 has been filed against any of the real or personal property of Borrower or its Subsidiaries,

- (ii) commencement of any Environmental Action or notice that an Environmental Action will be filed against Borrower or its Subsidiaries, and (iii) notice of a violation, citation, or other administrative order which reasonably could be expected to result in a Material Adverse Change.
- 6.14 **Disclosure Updates.** Promptly and in no event later than 5 Business Days after obtaining knowledge thereof, notify Agent if any written information, exhibit, or report furnished to Agent contained, at the time it was furnished, any untrue statement of a material fact or omitted to state any material fact necessary to make the statements contained therein not misleading in light of the circumstances in which made. The foregoing to the contrary notwithstanding, any notification pursuant to the foregoing provision will not cure or remedy the effect of the prior untrue statement of a material fact or omission of any material fact nor shall any such notification have the effect of amending or, modifying this Agreement or any of the Schedules hereto.
- 6.15 **Formation of Subsidiaries.** At the time that Borrower or any Guarantor forms any direct or indirect Subsidiary or acquires any direct or indirect Subsidiary after the Restatement Effective Date, Borrower or such Guarantor shall (a) cause such new Subsidiary to provide to Agent a Guaranty, Guarantor Security Agreement, and Intercompany Subordination Agreement (or joinder thereto), together with such other security documents (including delivery of Mortgages, and satisfaction of each other requirement specified in Section 6.17 in the time periods set forth therein, with respect to any Real Property Collateral of such new Subsidiary), as well as appropriate financing statements (and with respect to all property subject to a Mortgage, fixture filings), all in form and substance satisfactory to Agent (including being sufficient to grant Agent a first priority Lien (subject to Permitted Liens) in and to the assets of such newly formed or acquired Subsidiary other than any Excluded Property); provided that the Guaranty, Guarantor Security Agreement, and such other security documents shall not be required to be provided to Agent with respect to any direct or indirect Subsidiary of a Loan Party that is an Excluded Subsidiary, (b) within 10 days of such formation or acquisition (or such later date as permitted by Agent in its sole discretion) provide to Agent a pledge agreement and appropriate certificates and powers or financing statements, hypothecating all of the direct or beneficial ownership interest in such new Subsidiary, in form and substance satisfactory to Agent; provided that no such pledge shall be made with respect to any Excluded Subsidiary (other than 65% of the total outstanding voting Stock (and 100% of the total outstanding non-voting Stock) of any CFC or CFC Holdco that is a direct Subsidiary of a Loan Party), and (c) within 10 days of such formation or acquisition (or such later date as permitted by Agent) provide to Agent all other documentation, including one or more opinions of counsel satisfa

referred to above (including p	oolicies of title insurance or	other documentation	with respect to al	l property subje	ect to a Mortgage).	Any document,	agreement,	or instrument
executed or issued pursuant to t	this Section 6.15 shall be a I	oan Document						

- 6.16 <u>Canadian and Dutch Subsidiaries</u>. (a) Within 30 days (or such longer period as reasonably agreed to by Agent) after the written request therefor by Agent, Borrower shall deliver an additional stock pledge agreement (in addition to the Stock Pledge Agreement) which is governed by Ontario law and which is otherwise in form and substance reasonably satisfactory to Agent, pursuant to which Borrower shall pledge 65% of the total outstanding voting Stock (and 100% of the total outstanding non-voting Stock) of the Canadian Subsidiary (and none of any Subsidiary of such Canadian Subsidiary).
- (b) Within 30 days (or such longer period as reasonably agreed to by Agent) after the written request therefor by Agent, Borrower shall deliver an additional stock pledge agreement (in addition to the Stock Pledge Agreement) which is governed by Dutch law and which is otherwise in form and substance reasonably satisfactory to Agent, pursuant to which Borrower shall pledge 65% of the total outstanding voting Stock (and 100% of the total outstanding non-voting Stock) of the Dutch Subsidiary (and none of any Subsidiary of such Dutch Subsidiary).
- 6.17 Acquisition of Real Property Collateral. Not less than 30 days (or such shorter period as reasonably agreed to by Agent) prior to the acquisition thereof, Borrower shall (i) notify the Agent in writing of any Real Property Collateral to be acquired by Borrower or any Subsidiary, on or after the Restatement Effective Date and (ii) cause to be delivered to the Agent with respect to such Real Property Collateral (in each case, in form and substance reasonably satisfactory to the Agent):
  - (a) an appraisal,
  - (b) a Phase I environmental site assessment report, along with a reliance letter in favor of the Agent relating thereto,
- (c) evidence as to whether any such Real Property Collateral is located in an area identified by the Federal Emergency Management Agency (or any successor agency) as a "special flood hazard area" (or a similar designation) and, if it is, evidence that Borrower or such Subsidiary has obtained (or, concurrently with the acquisition thereof, will obtain), with insurance companies as are reasonably satisfactory to Agent, such flood insurance in such reasonable total amount as the Agent may reasonably require, and otherwise sufficient to comply with all applicable rules and regulations relating to flood insurance, in form and substance reasonably acceptable to the Agent,
  - (d) a survey,
  - (e) a zoning report, and
- (f) to the extent then available, construction plans and budget for all projects Borrower expects to initiate on such Real Property

Collateral.

- 87 -137020431v13

		is reasonably agreed to by Agent) after such acquisition of any Real Property Collateral, the Borrower shall cause to be delivered to the y Collateral (in each case, in form and substance reasonably satisfactory to the Agent):
	(a)	a duly executed Mortgage, along with a fixture filing,
	(b)	a lenders' policy of title insurance with respect thereto, in form and substance, and with an insured amount, reasonably satisfactory
to the Agent,		
	(c)	an Environmental Indemnity,
	(d)	to the extent not previously delivered, construction plans and budget for all projects Borrower expects to initiate on such Real
Property Collateral, and		
	(e)	a legal opinion with respect to the documents referred to in clause (a) above, delivered by Borrower's counsel in the State in
which such Real Propert	y Collateral is loc	cated.

## 7. NEGATIVE COVENANTS.

Borrower covenants and agrees that, until termination of all of the Commitments and payment in full of the Obligations, Borrower will not and will not permit any of its Subsidiaries to do any of the following:

7.1 Indebtedness, except:	Indebtedness.	Create, incur; assume, suffer to exist, guarantee, or otherwise become or remain, directly or indirectly, liable with respect to any
(a	a) Indebtedness evidenced by	ythis Agreement and the other Loan Documents,
(t	b) Indebtedness set forth on <u>S</u>	ichedule 520,
(0	c) Permitted Purchase Mone	y Indebtechess,

(d) refinancings, nerwals, or extensions of Indebtechess permitted under clauses (b) and (c) above (and continuance on nerwal of any Permitted Liens associated therewith) so long as (b) the terms and conditions of such refinancings, nerwals, or extensions do not, in Agent's reasonable judgment, materially impaint propects of repayment of the Obligations by Borrower or materially impair Borrower's credit worthiness, (ii) such refinancings, nerwals, or extensions do not result in an increase in the principal amount of, or interest are with respect to, the Indebtechess so refinanced, nerwed, or extended, (iii) such refinancings, nerwals, or extensions do not result in astroner in a strategic weighted maturity of the Indebtechess so refinanced, nerwed, or extended was subordinated in night of payment to the Obligations, then the terms and conditions of the refinancing nerwal, or extension Indebtechess must include subordination terms and conditions that are at least as favorable to the

Lenders as those that were applicable to the refinanced, renewed, or extended Indebtedness, and
(v) the Indebtedness that is refinanced, renewed, or extended is not recourse to any Person that is liable on account of the Obligations other than those Persons which were obligated with respect to the Indebtedness that was refinanced, renewed, or extended,

- (e) endorsement of instruments or other payment items for deposit,
- (f) Indebtedness composing Permitted Investments,
- (g) Subordinated Debt in an aggregate amount not to exceed \$5,000,000 at any time outstanding.
  - (h) Hedge Obligations, and
- (i) other unsecured Indebtedness in an aggregate amount not to exceed

\$2,500,000 at any time outstanding.

7.2 **Liens.** Create, incur, assume, or suffer to exist, directly or indirectly, any Lien on or with respect to any of its assets, of any kind, whether now owned or hereafter acquired, or any income or profits therefrom, except for Permitted Liens (including Liens that are replacements of Permitted Liens to the extent that the original Indebtedness is refinanced, renewed, or extended under <u>Section 7.1(d</u>) and so long as the replacement Liens only encumber those assets that secured the refinanced, renewed, or extended Indebtedness).

## 7.3 <u>Restrictions on Fundamental Changes.</u>

- (a) Enterinto any mega; consolidation, reorganization, or reclassify its Stock other than (i) to consummate a Permitted Acquisition and (ii) megas between Bourower and any of its Subsidiaries, provided that Bourower is the surviving entity of such megar and remains an entity organized under the laws of a state within the United States.
  - (b) Consummate a Division Series Transaction without the prior written consent of the Agent.
- (c) Liquidate, windrup, or dissolve itself (or suffer any liquidation or dissolution); <u>provided</u> that this <u>Section 7.3(</u>c) shall not prohibit or otherwise limit the Canadian Subsidiary Dissolution or the FPF Foods Dissolution, in each case, conducted in accordance with the terms hereof.
  - (d) Convey, sell, lesse, license, assign, transfer, or otherwise dispose of, in one transaction or a senies of transactions, all or any substantial part of its assets.
- 7.4 <u>Disposal of Assets.</u> Other than Permitted Dispositions, convey, sell, lease, license, assign, transfer, or otherwise dispose of any of Borrower's or its Subsidiaries assets; <u>provided</u> that this <u>Section 7.4</u> shall not prohibit or otherwise limit the Canadian Subsidiary Dissolution, conducted in accordance with the terms hereof; <u>provided further</u> that (i) none of the foregoing shall be deemed to permit a Division/Series Transaction and (ii) in no event shall

- 89 -137020431v13 Borrower sell or otherwise dispose of, or permit any of its Subsidiaries to sell or otherwise dispose of, any Real Property Collateral without the prior written consent of the Agent.

- 7.5 Change Name. Change Borrower's or any of its Subsidiaries' names, organizational identification number, state of organization or organizational identity; provided, however, that Borrower or any of its Subsidiaries may change their names upon at least 30 days prior written notice to Agent of such change and so long as, at the time of such written notification, Borrower or its Subsidiary provides any financing statements necessary to perfect and continue perfected the Agent's Liens.
  - 7.6 Nature of Business. Make any change in the principal nature of its or their business.
  - 7.7 **Prepayments and Amendments**. Except in connection with a refinancing permitted by Section 7.1(e),
  - (a) optionally prepay, redeem, defesse, purchase, or otherwise acquire any Subordinated Debt of Borrower or its Subsidiaries, other than the Obligations in accordance with this Agreement,
- $(b) {\it directly or inclined by, amend, modify, alter, increase, or change any of the terms or conditions of any agreement, instrument, document, indenture, or other writing evidencing or concerning Indebtedness permitted under <math>\underline{Section 7.1(b)}$  to the extent that such amendment, modification, or change could, individually or in the aggregate, tessonably be expected to be materially adverse to the interests of the Lender Group.
  - 7.8 Change of Control. Cause, permit, or suffer, directly or indirectly, any Change of Control.
  - 7.9 Consignments. Consign any of its or their Inventory or sell any of its or their Inventory on bill and hold, sale or return, sale on approval, or other conditional terms of sale.
- 7.10 <u>Distributions.</u> Make any distribution or declare or pay any dividends (in cash or other property, other than common Stock) on, or purchase, acquire, redeem, or retire any of Borrower's Stock, of any class, whether now or hereafter outstanding (each, a "<u>Distribution</u>"), other than (a) so long as no Default or Event of Default has occurred and is continuing or would result therefrom, Borrower may make Distributions to present or former employees, officers, or directors of Borrower (or any spouses, ex-spouses, or estates of any of the foregoing) on account of redemptions of Stock of Borrower held by such Persons, provided that the aggregate amount of such redemptions made by Borrower during any fiscal year does not exceed \$1,000,000, (b) dividends and distributions declared and paid on the Stock of any Subsidiary of Borrower ratably to the holders of such Stock, (c) cashless repurchases of Stock of Borrower deemed to occur upon exercise of stock options or warrants if such Stock represent a portion of the exercise price of such options or warrants, (d) Distributions within 60 days after the date of declaration thereof if at such date of declaration such Distribution would have been permitted under this Society.

7.10 so long as no Default or Event of Default shall have occurred and be continuing or would result therefrom, (e) for any taxable year in which Borrower or any of its Subsidiaries is a member of a consolidated, combined. affiliated, unitary, or similar income tax group of which

Borrower or a direct or indirect parent of Borrower is the common parent (a "<u>Tax Group</u>"), Borrower or any Subsidiary may make distributions to its direct or indirect parent entity necessary to permit such parent entity to pay any income taxes of such Tax Group that are due and payable in respect of taxable income attributable to Borrower and/or its Subsidiaries and (f) other Distributions if after giving *pro forma* effect thereto, Borrower is in compliance with the Leverage Ratio required by Section 7.18(a)(ii) and so long as no Default or Event of Default shall have occurred and be continuing or would result therefrom. Notwithstanding anything herein to the contrary, nothing in this Section 7.10 shall be deemed to permit a Division/Series Transaction.

- 7.11 Accounting Methods. Modify or change its fiscal year or its method of accounting (other than as may be required to conform to GAAP) or enter into, modify, or terminate any agreement currently existing, or at any time hereafter entered into with any third party accounting firm or service bureau for the preparation or storage of Borrower's or its Subsidiaries' accounting records without said accounting firm or service bureau agreeing to provide Agent information regarding Borrower's and its Subsidiaries' financial condition.
- 7.12 Investments. Except for Permitted Investments, directly or indirectly, make or acquire any Investment or incur any liabilities (including contingent obligations) for or in connection with any Investment; provided, however, that Borrower and its Subsidiaries shall not have Permitted Investments (other than in the Cash Management Accounts) in Deposit Accounts or Securities Accounts in an aggregate amount in excess of \$25,000 at any one time unless Borrower or its Subsidiary, as applicable, and the applicable securities intermediary or bank have entered into Control Agreements governing such Permitted Investments in order to perfect (and further establish) the Agent's Liens in such Permitted Investments. Subject to the foregoing proviso, Borrower shall not permit its Subsidiaries to establish or maintain any Deposit Account or Securities Account unless Agent shall have received a Control Agreement in respect of such Deposit Account or Securities Account. Notwithstanding anything herein to the contrary, nothing in this Section 7.12 shall be deemed to permit a Division/Series Transaction.
- 7.13 Transactions with Affiliates. Directly or indirectly enter into or permit to exist any transaction with any Affiliate of Borrower except for transactions that (i) are in the ordinary course of Borrower's business, (ii) are upon fair and reasonable terms, and (iii) are no less favorable to Borrower or its Subsidiaries, as applicable, than would be obtained in an arm's length transaction with a non-Affiliate; provided, that none of the foregoing shall be deemed to permit a Division/Series Transaction.
  - 7.14 <u>Suspension</u>. Suspend or go out of a substantial portion of its or their business.
- 7.15 Sale-Leaseback Transactions. Borrower shall not, and shall not permit any Subsidiary to, sell, assign or otherwise transfer any of its properties, rights or assets (whether now owned or hereafter acquired) to any Person and thereafter directly or indirectly lease back the same or similar property.
- 7.16 Use of Proceeds. Use the proceeds of (a) the Advances for any purpose other than (i) on the Restatement Effective Date, to repay the Indebtedness evidenced by the Existing Loan Agreement and to pay transactional fees, costs, and expenses incurred in connection with

this Agreement, the other Loan Documents, and the transactions contemplated hereby and thereby, and (ii) thereafter, to purchase Eligible Equipment, finance Capital Expenditures, finance general corporate expenses and to use in any other manner consistent with the terms and conditions hereof, for its lawful and permitted purposes, (b) the Delayed Draw Term Loan for any purpose other than to purchase Eligible Equipment, finance tenant improvements and finance development of its manufacturing facilities (including financing the acquisition of real estate in connection with any such development), or (c) the Incremental Term Loans for any purpose other than the acquisition and development of Real Property Collateral necessary or material for the conduct of the business of Borrower and its Subsidiaries; provided, that in no event shall any proceeds of the Advances, the Delayed Draw Term Loans or the Incremental Term Loans be used to refinance or repay any Subordinated Debt, any Indebtedness owed to any Permitted Holder or to any Affiliate of Borrower or any Permitted Preferred Stock.

7.17 Restriction on FP Foods. After the Restatement Effective Date FP Foods shall not own or hold any assets and shall not conduct any business except to the extent necessary to complete its dissolution.

## 7.18 Financial Covenants.

(a) Fail to maintain or achieve:

(ii)

(i) Adjusted EBITDA. TTM EBITDA for the 12 month period ending June 30, 2019, and for each 12 month period ending on the last day of any fiscal quarter thereafter; that is greater than or equal to the amount set forth in the following table for the applicable period:

12 Month Period Ending	Minimum TTM EBITDA
June 30, 2019	\$16,020,000
September 30, 2019	\$19,620,000
December 31, 2019	\$23,640,000
March 31, 2020	\$28,938,000
June 30, 2020	\$33,741,000
September 30, 2020	\$39,088,000
December 31, 2020	\$45,150,000

; provided that if Borrower delivers to the Agent a certificate of occupancy with respect to Property A, in form and substance satisfactory to the Agent, the testing of the financial covenant set forth in this  $\underline{Section 7.18(a)(i)}$  shall cease.

endingon the

Leverage Ratio. A Leverage Ratio, measured on a quarter-end basis, for the 12 month period ending. June 30, 2019, and for each 12 month period ending. June 30, 2019, and 2019, and

- 92 -137020431v13 last day of any fiscal quarter thereafter, that is less than or equal to the amount set forth in the following table for the applicable period:

12 Month Period Ending	Maximum Leverage Ratio
June 30, 2019	3.00:1.00
September 30, 2019	3.00:1.00
December 31, 2019	3.00:1.00
March 31, 2020	3.00:1.00
June 30, 2020	3.00:1.00
September 30, 2020	3.00:1.00
December 31, 2020	3.00:1.00
March 31, 2021	3.00:1.00
June 30, 2021	3.00:1.00
September 30, 2021	3.00:1.00
December 31, 2021	2.75:1.00
March 31, 2022	2.75:1.00
June 30, 2022	2.75:1.00
September 30, 2022	2.75:1.00
December 31, 2022 and each fiscal quarter thereafter	2.50:1.00

(iii) Fixed Charge Coverage Ratio. A Fixed Charge Coverage Ratio, measured on a quarter-end basis, for the 12 month period ending June 30, 2019, and for each 12 month period ending on the last day of any fiscal quarter thereafter, of at least 125:1.00.

(b) Make:

(i)

 $\textbf{\textit{Capital Expenditures.} Capital Expenditures in any fiscal year in excess of the amount set forth in the following table for the applicable fiscal year:} \\$ 

Fiscal Year	Maximum Capital Expenditures
2019	\$81,000,000
2020	\$69,000,000
2021	\$13,000,000

Fiscal Year	Maximum Capital Expenditures
2022	\$13,000,000
2023	\$23,000,000
2024	\$25,000,000

provided, that if the amount of the Capital Expenditures permitted to be made in any fiscal year as set forth in the above table is greater than the actual amount of the Capital Expenditures (excluding the amount, if any, of Capital Expenditures made with Net Cash Proceeds reinvested pursuant to the proviso in Section 2.4(e)(ii)) actually made in such fiscal year (the amount by which such permitted Capital Expenditures for such fiscal year exceeds the actual amount of Capital Expenditures for such fiscal year, the "Excess Amount"), then 100% of such Excess Amount (the "Carry-Over Amount") may be carried forward to the next succeeding Fiscal Year (the "Succeeding Fiscal Year"), so long as no Default or Event of Default shall have occurred and be continuing or would result therefrom; provided further that the Carry-Over Amount applicable to a particular Succeeding Fiscal Year may not be used in that Fiscal Year may not be used in full and the Carry-Over Amount applicable to a particular Succeeding Fiscal Year may not be carried forward to another fiscal year.

### 8. EVENTS OF DEFAULT.

Any one or more of the following events shall constitute an event of default (each, an "Event of Default") under this Agreement:

8.1 If Borrower fails to pay when due and payable, or when declared due and payable,

(a) all or any portion of the Obligations consisting of the reimbursement of Lender Group Expenses (other than any portion thereof constituting principal, any amount payable to Issuing Lender in reimbursement of any drawing under a Letter of Credit, fees or interest) constituting Obligations (including any portion thereof that accrues after the commencement of an Insolvency Proceeding, regardless of whether allowed or allowable in whole or in part as a claim in any such Insolvency Proceeding), and such failure continues for a period of 10 days, (b) all or any portion of the Obligations consisting of principal or any amount payable to Issuing Lender in reimbursement of any drawing under a Letter of Credit, or (c) any interest or other amounts that do not constitute Lender Group Expenses (including any portion thereof that accrues after the commencement of an Insolvency Proceeding, regardless of whether allowed or allowable in whole or in part as a claim in any such Insolvency Proceeding), principal or any amount payable to Issuing Lender in reimbursement of any drawing under a Letter of Credit and such failure continues for a period of five days;

8.2 If Borrower or any of its Subsidiaries:

(a) failstoperform or observe any covenant or other agreement contained in any of (i) <u>Sections 62, 63, 68, 612, 615, 616</u> or <u>617</u> of this Agreement, or (ii) Sections <u>7.1</u> through <u>7.18</u> of this

Agreement; or

- 94 -137020431v13

- (b) fails to perform or observe any covenant or other agreement contained in this Agreement, or in any of the other Loan Documents, in each case, other than any such covenant or agreement that is the subject of another provision of this <u>Section 8</u> (in which event such other provision of this <u>Section 8</u> shall goven), and such failure continues for a period of 30 days after the earlier of (i) the date on which such failure shall first become known to any officer of Borrower or (ii) the date on which written notice thereof is given to Borrower by Agent;
  - 8.3 If any material portion of Borrower's or any of its Subsidiaries' assets is attached, seized, subjected to a writ or distress warrant or levied upon;
  - 8.4 If an Insolvency Proceeding is commenced by Borrower or any of its Subsidiaries;
- 8.5 If an Insolvency Proceeding is commenced against Borrower, or any of its Subsidiaries, and any of the following events occur: (a) Borrower or such Subsidiary consents to the institution of such Insolvency Proceeding against it, (b) the petition commencing the Insolvency Proceeding is not timely controverted, (c) the petition commencing the Insolvency Proceeding is not dismissed within 45 calendar days of the date of the filing thereof, (d) an interim trustee is appointed to take possession of all or any substantial portion of the properties or assets of, or to operate all or any substantial portion of the business of, Borrower or any of its Subsidiaries, or (e) an order for relief shall have been entered therein;
  - 8.6 If Borrower or any of its Subsidiaries is enjoined, restrained, or in any way prevented by court order from continuing to conduct all or any material part of its business affairs;
- 8.7 If a notice of Lien, levy, or assessment is filed of record involving an amount of the Applicable Cross-Default Amount or more with respect to any of Borrower's or any of its Subsidiaries' assets by the United States, or any department, agency, or instrumentality thereof, or by any state, county, municipal, or governmental agency, or if any taxes or debts owing at any time hereafter to any one or more of such entities becomes a Lien, whether choate or otherwise, upon any of Borrower's or any of its Subsidiaries' assets and the same is not paid before such payment is delinquent;
- 8.8 If a judgment or other claim becomes a Lien or encumbrance involving an amount of the Applicable Cross-Default Amount or more upon any material portion of Borrower's or any of its Subsidiaries' assets;
- 8.9 If there is a default in any material agreement of Indebtedness or any Material Contract to which Borrower or any of its Subsidiaries is a party involving an amount of the Applicable Cross-Default Amount or more and such default (a) occurs at the final maturity of the obligations thereunder, or (b) results in a right by the other party thereto, irrespective of whether exercised, to accelerate the maturity of Borrower's or its Subsidiaries' obligations thereunder or to terminate such agreement;
- 8.10 If Borrower or any of its Subsidiaries makes any payment on account of Indebtedness that has been contractually subordinated in right of payment to the payment of the Obligations (other than reimbursement of a *de minimis* amount of expenses), except to the extent

such payment is permitted by the terms of the subordination provisions applicable to such Indebtedness;

- 8.11 If any warranty, representation, certificate, statement, or Record made herein or in any other Loan Document or delivered in writing to Agent or any Lender in connection with this Agreement or any other Loan Document proves to be untrue in any material respect (except that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof) as of the date of issuance or making or deemed making thereof;
  - 8.12 If the obligation of any Guarantor under any Guaranty is limited or terminated by operation of law or by any such Guarantor thereunder;
- 8.13 If this Agreement or any other Loan Document that purports to create a Lien, shall, for any reason, fail or cease to create a valid and perfected and, except to the extent permitted by the terms hereof or thereof, first priority Lien on or security interest in the Collateral covered hereby or thereby (with respect to intellectual property, limited to perfection in the United States), except as a result of a disposition of the applicable Collateral in a transaction permitted under this Agreement; or
- 8.14 Any provision of any Loan Document shall at any time for any reason be declared to be null and void, or the validity or enforceability thereof shall be contrested by Borrower or its Subsidiaries, or a proceeding shall be commenced by Borrower or its Subsidiaries, or by any Governmental Authority having jurisdiction over Borrower or its Subsidiaries seeking to establish the invalidity or unenforceability thereof, or Borrower or its Subsidiaries, shall deny that Borrower or its Subsidiaries has any liability or obligation purported to be created under any Loan Document.

### 9. THE LENDER GROUP'S RIGHTS AND REMEDIES.

- 9.1 **Rights and Remedies.** Upon the occurrence, and during the continuation, of an Event of Default, the Required Lenders (at their election but without notice of their election and without demand) may authorize and instruct Agent to do any one or more of the following on behalf of the Lender Group (and Agent, acting upon the instructions of the Required Lenders, shall do the same on behalf of the Lender Group), in addition to any other rights or remedies provided for hereunder or under any other Loan Document or by applicable law, all of which are authorized by Borrower:
- (a) (i) Declare the principal of, and any and all accound and unpaid interest and fees in respect of, the Loans and all other Obligations (other than the Bank Product Obligations), whether evidenced by this Agreement or by any of the other Loan Documents to be immediately due and psychle and power shall be obligated to repay all of such Obligations in full, without presentment, demand, protest, or further motice or other requirements of any kind, all of which are hereby expressly waived by Borrower, and (ii) direct Borrower agrees that upon receipt of such motice it will provide). Letter of Credit Collateralization to Agent

to be held as security for Borrower's reimbursemen	t obligations for drawings that may subsequently occur under issued and outstanding Letters of Credit;
(b) Barrower and the Lender Group;	Cesse advancing money or extending credition or for the benefit of Bourower under this Agreement, under any of the Loan Documents, crunder any other agreement between
(c) Obligation of Issuing Lender to issue Letters of Credit and (iii) the obligation	Declare the Commitments terminated, whereupon the Commitments shall immediately be terminated together with (1) any obligation of any Lendertomake Advances, (ii) the not any Lendertomake Delayed Draw Term Loans;
	Settle or adjust disputes and claims directly with Bourower's Account Debtors for amounts and upon terms which Agent considers advisable, and in such cases, Agent will credit in payment of such disputed Accounts after deducting all Lender Group Expenses incurred or expended in connection therewith;
(e)	Cause Bourower to hold all of its returned Inventory in trust for the Lender Group and segregate all such Inventory from all other assets of Bourower or in Bourower's possession;
Borrower agrees to assemble the Collateral if Agent so requires, and to mail Collateral is located, to take and maintain possession of the Collateral, or any	Without notice to ordemand upon Bourower, make such payments and do such acts as Agent considers necessary or reasonable to protect its security interests in the Collateral. Identine Collateral available to Agent at a place that Agent may designate which is reasonably convenient to both parties. Bourower authorizes Agent to enter the premises where the y part of it, and to pay, purchase, contest, or compromise any Lien that in Agent's determination appears to conflict with the priority of Agent's Liens in and to the Collateral and to pay all n Account therefor. With respect to any of Bourower's owned or lessed premises, Bourower hereby grants Agent a license to enter into possession of such premises and to occupy the some medies provided herein, at law, in equity, or otherwise;
	Without notice to Borrower (such notice being expressly waived), and without constituting an acceptance of any collateral in full or partial satisfaction of an obligation (within the alances and deposits of Borrower held by the Lender Group (including any amounts received in the Cash Management Accounts), or (ii) Indebtedness at any time owing to or for the
(h) final repayment of all of the Obligations;	Hold, as cash collateral, any and all balances and deposits of Borrowerheld by the Lender Group, and any amounts received in the Cash Management Accounts, to secure the full and

Ship, reclaim, recover, store, finish, maintain, repair; prepare for sale, advertise for sale, and sell (in the manner provided for herein) the Borrower Collateral.

(i)

Borrower hereby grants to Agent a license or other right to use, without charge, Borrower's labels, patents, copyrights, trade secrets, trade names, trademarks, service marks, and advertising matter, or any property of a similar nature, as it pertains to the Borrower Collateral, in completing production of, advertising for sale, and selling any Borrower Collateral and Borrower's rights under all licenses and all franchise agreements shall inure to the Lender Group's benefit;

- (j) Sell the Borrower Collateral at either a public or private sale, or both, by way of one or more contracts or transactions, for cash or on terms, in such manner and at such places (including Borrower's premises) as Agent determines is commercially reasonable. It is not necessary that the Borrower Collateral be present at any such sale;
  - (k) Except in those circumstances where no notice is required under the Code, Agent shall give notice of the disposition of the Borrower Collateral as follows:
- (i) Agent shall give Borrower a notice in writing of the time and place of public sale, or, if the sale is a private sale or some other disposition other than a public sale is to be made of the Borrower Collateral, the time on or after which the private sale or other disposition is to be made; and
- (ii) The notice shall be personally delivered or mailed, postage prepaid, to Borrower as provided in <u>Section 12</u>, at least 10 days before the earliest time of disposition set forth in the notice; no notice needs to be given prior to the disposition of any portion of the Borrower Collateral that is perishable or threatens to decline speedily in value or that is of a type customarily sold on a recognized market;
  - (1) Agent, on behalf of the Lender Group, may credit bid and purchase at any

public sale;

(m) Agent may seek the appointment of a receiver or keeper to take possession

of all or any portion of the Borrower Collateral or to operate same and, to the maximum extent permitted by law, may seek the appointment of such a receiver without the requirement of prior notice or a hearing; and

(n) The Lender Groupshall have all other nights and nemedies available at law or in equity or pursuant to any other Loan Document.

The foregoing to the contrary notwithstanding, upon the occurrence of any Event of Default described in Section 8.4 or Section 8.5, in addition to the remedies set forth above, without any notice to Borrower or any other Person or any act by the Lender Group, the Commitments shall automatically terminate and the Obligations (other than the Bank Product Obligations), inclusive of the principal of, and any and all accrued and unpaid interest and fees in respect of, the Loans and all other Obligations (other than the Bank Product Obligations), whether evidenced by this Agreement or by any of the other Loan Documents, shall automatically become and be immediately due and payable and Borrower shall automatically be obligated to repay all of such Obligations in full (including Borrower being obligated to provide (and Borrower agrees that it will provide) (1) Letter of Credit Collateralization to Agent to be held as security for Borrower's reimbursement obligations in respect of drawings that may subsequently occur under issued and

outstanding Letters of Credit, and (2) Bank Product Collateralization to be held as security for Borrower's or its Subsidiaries' obligations in respect of outstanding Bank Products), without presentment, demand, protest, or notice or other requirements of any kind, all of which are expressly waived by Borrower.

9.2 Remedies Cumulative. The rights and remedies of the Lender Group under this Agreement, the other Loan Documents, and all other agreements shall be cumulative. The Lender Group shall have all other rights and remedies not inconsistent herewith as provided under the Code, by law, or in equity. No exercise by the Lender Group of one right or remedy shall be deemed an election, and no waiver by the Lender Group of any Event of Default shall be deemed a continuing waiver. No delay by the Lender Group shall constitute a waiver, election, or acquiescence by it.

#### 10. TAXES AND EXPENSES.

If Borrower fails to pay any monies (whether taxes, assessments, insurance premiums, or, in the case of leased properties or assets, rents or other amounts payable under such leases) due to third Persons, or fails to make any deposits or furnish any required proof of payment or deposit, all as required under the terms of this Agreement, then, Agent, in its sole discretion and without prior notice to Borrower, may do any or all of the following: (a) make payment of the same or any part thereof, (b) set up such reserves against the Maximum Revolver Amount as Agent deems necessary to protect the Lender Group from the exposure created by such failure, or (c) in the case of the failure to comply with Section 6.8 hereof, obtain and maintain insurance policies of the type described in Section 6.8 and take any action with respect to such policies as Agent deems prudent. Any such amounts paid by Agent shall constitute Lender Group Expenses and any such payments shall not constitute an agreement by the Lender Group to make similar payments in the future or a waiver by the Lender Group of any Event of Default under this Agreement. Agent need not inquire as to, or contest the validity of, any such expense, tax, or Lien and the receipt of the usual official notice for the payment thereof shall be conclusive evidence that the same was validly due and owing.

### 11. WAIVERS; INDEMNIFICATION.

- 11.1 <u>Demand; Protest; etc.</u> Borrower waives demand, protest, notice of protest, notice of default or dishonor, notice of payment and nonpayment, nonpayment at maturity, release, compromise, settlement, extension, or renewal of documents, instruments, chattel paper, and guarantees at any time held by the Lender Group on which Borrower may in any way be liable.
- 11.2 The Lender Group's Liability for Borrower Collateral. Borrower hereby agrees that: (a) so long as Agent complies with its obligations, if any, under the Code, the Lender Group shall not in any way or manner be liable or responsible for: (i) the safekeeping of the Borrower Collateral, (ii) any loss or damage thereto occurring or arising in any manner or fashion from any cause, (iii) any diminution in the value thereof, or (iv) any act or default of any carrier, warehouseman, bailee, forwarding agency, or other Person, and (b) all risk of loss, damage, or destruction of the Borrower Collateral shall be bome by Borrower.

- 99 -137020431v13 11.3 Indemnification. Borrower shall pay, indemnify, defend, and hold the Agent-Related Persons, the Lender-Related Persons, and each Participant (each, an "Indemnified Person") hamless (to the fullest extent permitted by law) from and against any and all claims, demands, suits, actions, investigations, proceedings, and damages, and all reasonable attorneys' fees and disbursements and other costs and expenses actually incurred in connection therewith or in connection with the enforcement of this indemnification (as and when they are incurred and irrespective of whether suit is brought), at any time asserted against, imposed upon, or incurred by any of them (a) in connection with or as a result of or related to the execution, delivery, enforcement, performance, or administration (including any restructuring or workout with respect hereto) of this Agreement, any of the other Loan Documents, or the transactions contemplated hereby or thereby or the monitoring of Borrower's and its Subsidiaries' compliance with the terms of the Loan Documents, and (b) with respect to any investigation, litigation, or proceeding related to this Agreement, any other Loan Document, or the use of the proceeds of the credit provided hereunder (irrespective of whether any Indemnified Person is a party thereto), or any act, omission, event, or circumstance in any manner related thereto (all the foregoing, collectively, the "Indemnified Liabilities"). The foregoing to the contrary notwithstanding, Borrower shall have no obligation to any Indemnified Person under this Section 11.3 with respect to any Indemnified Liabilities at the repayment of the Obligations. If any Indemnified Person makes any payment to any other Indemnified Person with respect to an Indemnified Liability as to which Borrower was required to indemnify the Indemnified Person receiving such payment, the Indemnified Person making such payment is entitled to be indemnified and reimbursed by Borrower with respect thereto. This Section 11.3 shall not apply with respect to Taxe

### 12. NOTICES.

Unless otherwise provided in this Agreement, all notices or demands by Borrower or Agent to the other relating to this Agreement or any other Loan Document shall be in writing and (except for financial statements and other informational documents which may be sent by first-class mail, postage prepaid) shall be personally delivered or sent by registered or certified mail (postage prepaid, return receipt requested), overnight courier, electronic mail (at such email addresses as Borrower or Agent, as applicable, may designate to each other in accordance herewith), or telefacsimile to Borrower or Agent, as the case may be, at its address set forth below:

If to Borrower:

FRESHPET, INC. 400 Plaza Drive FL1 Secaucus, NJ 07094 Attn: Richard Kassar Fax No. (201) 866-2018

- 100 -137020431v13 with copies to:

#### KIRKLAND & ELLIS LLP

601 Lexington Avenue New York, NY 10022 Attn: Jason Kanner, Esq. Fax No. (212) 446-

If to Agent:

CITY NATIONAL BANK

555 S. Flower Street 21st Floor Los Angeles, California 90071 Attn: Garen Papazyan

Fax No.: (213) 673-9801

with copies to:

CITY NATIONAL BANK

agencyservices@cnb.com and KATTEN MUCHIN ROSENMAN LLP

515 South Flower Street Suite 1000 Los Angeles, California 90071 Attention: Jan Harris Cate,

Esq. Facsimile: (213) 947-1151

Any party hereto may change the address at which they are to receive notices hereunder, by notice in writing in the foregoing manner given to the other party. All notices or demands sent in accordance with this Section 12, shall be deemed received on the earlier of the date of actual receipt or 3 Business Days after the deposit thereof in the mail; provided, that

notices sent by overnight courier service shall be deemed to have been given when received,

notices by facsimile shall be deemed to have been given when sent (except that, if not given during normal business hours for the recipient, shall be deemed to have been given at the opening of business on the next Business Day for the recipient) and (c) notices by electronic mail shall be deemed received upon the sender's receipt of an acknowledgment from the intended recipient (such as by the "return receipt requested" function, as available, return email or other written acknowledgment).

# CHOICE OF LAW AND VENUE; JURY TRIAL WAIVER; JUDICIAL REFERENCE PROVISION.

(a) THE VALIDITY OF THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS (UNLESS EXPRESSLY PROVIDED TO THE CONTRARY IN ANOTHER LOAN DOCUMENT IN RESPECT OF SUCH OTHER LOAN DOCUMENT), THE CONSTRUCTION, INTERPRETATION, AND ENFORCEMENT HEREOF AND THEREOF, THE RIGHTS OF THE PARTIES HERETO AND THERETO WITH RESPECT TO ALL MATTERS ARISING HEREUNDER OR

- 101 -137020431v13 THEREUNDER OR RELATED HERETO OR THERETO, AND ANY CLAIMS, CONTROVERSIES OR DISPUTES ARISING HEREUNDER OR THEREUNDER OR RELATED HERETO OR THERETO SHALL BE DETERMINED UNDER, GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF CALIFORNIA.

- (b) THE PARTIES AGREE THAT ALL ACTIONS OR PROCEEDINGS ARISING IN CONNECTION WITH THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS SHALL BE TRIED AND LITIGATED ONLY IN THE STATE AND, TO THE EXTENT PERMITTED BY APPLICABLE LAW, FEDERAL COURIS LOCATED IN THE COUNTY OF LOS ANGELES, STATE OF CALIFORNIA; PROVIDED, THAT ANY SUIT SEEKING ENFORCEMENT AGAINST ANY COLLATERAL OR OTHER PROPERTY MAY BE BROUGHT; AT AGENT'S OPTION, IN THE COURTS OF ANY JURISDICTION WHERE AGENT ELECTS TO BRING SUCH ACTION OR WHERE SUCH COLLATERAL OR OTHER PROPERTY MAY BE FOUND. BORROWER AND EACH MEMBER OF THE LENDER GROUP WAIVE, TO THE EXTENT PERMITTED UNDER APPLICABLE LAW, ANY RIGHT EACH MAY HAVE TO ASSERT THE DOCTRINE OF FORUM NON CONVENIENS OR TO OBJECT TO VENUE TO THE EXTENT ANY PROCEEDING IS BROUGHT IN ACCORDANCE WITH THIS SECTION 13(b).
- (c) TO THE MAXIMUM EXTENT PERMITTED BY APPLICABLE LAW, BORROWER AND EACH MEMBER OF THE LENDER GROUP HEREBY WAIVE THEIR RESPECTIVE RIGHTS, IF ANY, TO A JURY TRIAL OF ANY CLAIM, CONTROVERSY, DISPUTE OR CAUSE OF ACTION DIRECTLY OR INDIRECTLY BASED UPON OR ARISING OUT OF ANY OF THE LOAN DOCUMENTS OR ANY OF THE TRANSACTIONS CONTEMPLATED THEREIN, INCLUDING CONTRACT CLAIMS, TORT CLAIMS, BREACH OF DUTY CLAIMS, AND ALL OTHER COMMON LAW OR STATUTORY CLAIMS (EACH A "CLAIM"), BORROWER AND EACH MEMBER OF THE LENDER GROUP REPRESENT THAT EACH HAS REVIEWED THIS WAIVER AND EACH KNOWINGLY AND VOLUNTARILY WAIVES ITS JURY TRIAL RIGHTS FOLLOWING CONSULTATION WITH LEGAL COUNSEL. IN THE EVENT OF LITIGATION, A COPY OF THIS AGREEMENT MAY BE FILED AS A WRITTEN CONSENT TO A TRIAL BY THE COURT.

(d) BORROWER HEREBY IRREVOCABLY
AND UNCONDITIONALLY SUBMITS TO THE EXCLUSIVE JURISDICTION OF THE STATE AND FEDERAL
COURTS LOCATED IN THE COUNTY OF LOS ANGELES AND THE STATE OF CALIFORNIA, IN ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO ANY LOAN
DOCUMENTS, OR FOR RECOGNITION OR ENFORCEMENT OF ANY JUDGMENT. FACH OF THE PARTIES HEREITO AGREES THAT A FINAL JUDGMENT IN ANY SUCH ACTION OR
PROCEEDING SHALL BE CONCLUSIVE AND MAY BE ENFORCED IN OTHER JURISDICTIONS BY SUIT ON THE JUDGMENT OR IN ANY OTHER MANNER PROVIDED BY LAW.
NOTHING IN THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT SHALL AFFECT ANY RIGHT THAT AGENT MAY OTHERWISE HAVE TO BRING ANY ACTION OR
PROCEEDING RELATING TO THIS AGREEMENT OR ANY OTHER

- 102 -137020431v

### LOAN DOCUMENT AGAINST ANY LOAN PARTY OR ITS PROPERTIES IN THE COURTS OF ANY JURISDICTION.

- (e) NO CLAIMMAY BE MADE BY ANY LOAN PARTY AGAINST THE AGENT, ISSUING LENDER, ANY OTHER LENDER OR ANY AFFILIAIE, DIRECTOR, OFFICER, EMPLOYEE, COUNSEL, REPRESENTATIVE, AGENT, OR ATTORNEY-IN-FACT OF ANY OF THEM FOR ANY SPECIAL, INDIRECT, CONSEQUENTIAL, PUNTITIVE OR EXEMPLARY DAMAGES OR LOSSES IN RESPECTOF ANY CLAIM FOR BREACH OF CONTRACT OR ANY OTHER THEORY OF LIABILITY ARISING OUT OF OR RELATED TO THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT, OR ANY ACT, OMISSION, OR EVENT OCCURRING IN CONNECTION THEREWITH, AND EACH LOAN PARTY HEREBY WAIVES, RELEASES, AND AGREES NOT TO SUE UPON ANY CLAIM FOR SUCH DAMAGES, WHETHER OR NOT ACCRUED AND WHETHER OR NOT KNOWN OR SUSPECTED TO EXIST IN TIS FAVOR.
- (f) INTHE EVENT ANY LEGAL PROCEEDING IS FILED IN A COURT OF THE STATE OF CALIFORNIA (THE "COURT") BY OR AGAINST ANY PARTY HERETO IN CONNECTION WITH ANY CLAIM AND THE WAIVER SET FORTH IN CLAUSE (C) ABOVE IS NOT ENFORCEABLE IN SUCH PROCEEDING, THE PARTIES HERETO AGREE AS FOLLOWS:
- (I) WITH THE EXCEPTION OF THE MATTERS SPECIFIED IN SUBCLAUSE (II) BELOW, ANY CLAIM SHALL BE DETERMINED BY A GENERAL REFERENCE PROCEEDING IN ACCORDANCE WITH THE PROVISIONS OF CALLFORNIA CODE OF CIVIL PROCEDURE SECTIONS 688 THROUGH 645.1 THE PARTIES INTENDITHIS GENERAL REFERENCE AGREEMENT TO BE SPECIFICALLY ENFORCEABLE. VENUE FOR THE REFERENCE PROCEEDING SHALL BE IN THE COUNTY OF LOS ANG 5F IS. CALIFORNIA.
- (i) THEFOLLOWING MATTERS SHALL NOT BESUBJECT TO A GENERAL REFERENCE PROCEEDING: (A) NON-JUDICIAL FORECLOSURE OF ANY SECURITY INTERESTS IN REAL OR PERSONAL PROPERTY, (B) EXERCISE OF SELF-HELP REMEDIES (INCLUDING SET OFF OR RECOUPMENT), (C) APPOINTMENT OF A RECEIVER, AND (D) TEMPORARY, PROVISIONAL, OR ANCILLARY REMEDIES (INCLUDING WRITS OF ATTACHMENT, WRITS OF POSSESSION, TEMPORARY RESTRAINING ORDERS, OR PRELIMINARY INJUNCTIONS). THIS AGREEMENT DOES NOT LIMIT THE RIGHT OF ANY PARTY TO EXERCISE OR OPPOSE ANY OF THE RIGHTS AND REMEDIES DESCRIBED IN CLAUSES (A) (D) AND ANY SUCH EXERCISE OR OPPOSITION DOES NOT WAIVE THE RIGHT OF ANY PARTY TO PARTICIPATE IN A REFERENCE PROCEEDING PURSUANT TO THIS AGREEMENT WITH RESPECT TO ANY OTHER MATTER.

- 103 -137020431v13 UPONTHE WRITTENREQUEST OF ANY PARTY, THE PARTIES SHALL SELECT ASINGLE REFEREE, WHO SHALL BE ARETIRED JUDGE OR JUSTICE. IF THE PARTIES DO NOT AGREE UPON A REFEREE WITHIN 10 DAYS OF SUCH WRITTENREQUEST, THEN, ANY PARTY SHALL HAVE THE RIGHT TO REQUEST THE COURT TO APPOINT A REFEREE PURSUANT TO CALIFORNIA CODE OF CIVIL PROCEDURE SECTION 640(B). THE REFEREE SHALL BE APPOINTED TO SIT WITH ALL OF THE POWERS PROVIDED BY LAW. PENDING APPOINTMENT OF THE REFEREE, THE COURT SHALL HAVE THE POWER TO ISSUE TEMPORARY OR PROVISIONAL REMEDIES.

- (iv) EXCEPT AS EXPRESSLY SET FORTH IN THIS AGREEMENT, THE REFEREE SHALL DETERMINE THE MANNER IN WHICH THE REFERENCE PROCEEDING IS CONDUCTED INCLUDING THE TIME AND PLACE OF HEARINGS, THE ORDER OF PRESENTATION OF EVIDENCE, AND ALL OTHER QUESTIONS THAT ARISE WITH RESPECT TO THE COURSE OF THE REFERENCE PROCEEDING, ALL PROCEEDINGS AND HEARINGS CONDUCTED BEFORE THE REFEREE, EXCEPT FOR TRIAL, SHALL BE CONDUCTED WITHOUT A COURT REPORTER, EXCEPT WHEN ANY PARTY SO REQUESTS A COURT REPORTER AND A TRANSCRIPT IS ORDERED, A COURT REPORTER SHALL BE USED AND THE REFEREE SHALL BE PROVIDED A COURTESY COPY OF THE TRANSCRIPT. THE PARTY MAKING SUCH REQUEST SHALL HAVE THE OBLIGATION TO ARRANGE FOR AND PAY THE COSTS OF THE COURT REPORTER, PROVIDED THAT SUCH COSTS, ALONG WITH THE REFEREE'S FEES, SHALL ULTIMATELY BE BORNE BY THE PARTY WHO DOES NOT PREVAIL. AS DETERMINED BY THE REFEREE.
- (v) THEREFERE MAY REQUIRE ONE OR MORE PREHEARING CONFERENCES. THE PARTIES HERE TO SHALL BEENTITLED TO DISCOVERY, AND THE REFERE SHALL OVERSEE DISCOVERY IN ACCORDANCE WITH THE RULES OF DISCOVERY, AND SHALL ENFORCE ALL DISCOVERY ORDERS IN THE SAME MANNER AS ANY TRIAL COURT JUDGE IN PROCEEDINGS AT LAW IN THE STATE OF CALIFORNIA.
- (vi) THEREFERESHALL APPLYTHE RULES OF EVIDENCE APPLICABLE TO PROCEEDINGS AT LAW IN THE STATE OF CALIFORNIA AND SHALL DETERMINE ALL ISSUES IN ACCORDANCE WITH CALIFORNIA SUBSTANTIVE AND PROCEDURAL LAW. THEREFEREE SHALL BE EMPOWERED TO ENTER EQUITABLE AS WELL AS LEGAL RELIEF AND RULE ON ANY MOTION WHICH WOULD BE AUTHORIZED IN A TRIAL, INCLUDING MOTIONS FOR DEFAULT JUDGMENT OR SUMMARY JUDGMENT. THEREFEES HALL REPORTHIS ON HER DECISION, WHICH REPORT SHALL ALSO INCLUDE FINDINGS OF FACT AND CONCLUSIONS OF LAW. THE REFEREE SHALL ISSUE ADECISION AND PURSUANT TO CALIFORNIA CODE OF CIVIL PROCEDURE, SECTION 644, THE REFEREE'S DECISION SHALL BE ENTERED BY THE COURT AS A JUDGMENT IN THE SAME MANNER AS IF THE ACTION HAD BEEN TRIED BY THE COURT. THE FINAL JUDGMENT OR ORDER

- 104 -137020431v13 FROM ANY APPEALABLE DECISION OR ORDER ENTERED BY THE REFEREE SHALL BE FULLY APPEALABLE AS IF IT HAS BEEN ENTERED BY THE

(g) THE PARTIES RECOGNIZE AND AGREE THAT ALL CLAIMS RESOLVED IN A GENERAL REFERENCE PROCEEDING PURSUANT HERETO WILL BE DECIDED BY A REFEREE AND NOT BY A JURY. AFTER CONSULTING (OR HAVING HAD THE OPPORTUNITY TO CONSULT) WITH COUNSEL OF THEIR OWN CHOICE, EACH PARTY HERETO KNOWINGLY AND VOLUNTARILY AND FOR THEIR MUTUAL BENEFIT AGREES THAT THIS REFERENCE PROVISION SHALL APPLY TO ANY DISPUTE BETWEEN THEM THAT ARISES OUT OF OR IS RELATED TO THIS AGREEMENT OR THE OTHER LOAN DOCUMENTS.

### 14. ASSIGNMENTS AND PARTICIPATIONS; SUCCESSORS.

## 14.1 Assignments and Participations.

(a) Any Lendermay assign and delegate to one or more assigness (each an "Assignes") all, or any natable part of all, of the Obligations, the Commitments and the other nights and obligations of such Lender here under and under the other Loan Documents, in a minimum amount of \$5,000,000 (except that in the case of an assignment of the entire remaining amount of the assigning Lender's Commitment and Loans at the time owing to it under the Loan Documents, no minimum amount need be assigned, and in connection with simultaneous assignments to two or more related. Approved Funck shall be treated as one assignee for purposes of determining compliance with the minimum assignment amount referred be assigned, by nowided, however, that (w) a Lendermay not assign all or any portion of the Obligations or the Commitments to Borrower, any Permitted Holder or any Affiliate thereof, (x) the prior written consent of Agent (not to be unessorably withheld, delayed or conditioned) shall be required prior to any such assignment by a Lender to another Person that is not a Lender, and Affiliate of a Lender or and Approved Funck provided that Borrowershall be deemed to have consented to any such assignment unless its hall object thereto by written notice to Agent within 5 Business Days after having received notice thereof, and (z) Borrower and Agent may continue to deal solely and directly with such Lender in connection with the intens to assignee have been given to Borrower and Agent by such Lender or Assignee, (i) such Lender and its Assignee have delivered to Borrower and Agent and Assignment is in connection with any meng; consolidation, sale, transfer, or other disposition of all or any substantial portion of the business of ban portfolio of the assigning Lender.

- 105 -137020431v13

(b) From and after the date that Agent notifies the assigning Lender (with a copy to Bourower) that it has received an executed Assignment and Acceptance and payment of the above-
referenced processing fee, (1) the Assignment neutral beaparty hereto and, to the extent that rights and obligations hereunder have been assigned to it pursuant to such Assignment and Acceptance, shall have the rights and obligations of a Lender under the
LoanDocuments, and (iii) the assigning Lendershall, to the extent that rights and obligations hereunder and under the other LoanDocuments have been assigned by it pursuant to such Assignment and Acceptance, relinquish its rights (except with nexpect to
Section 11.3 hereof) and be released from any future obligations under this Agreement (and in the cæe of an Assignment and Acceptance covering all or the remaining portion of an assigning Lender's rights and obligations under this Agreement and the
other Loan Documents, such Lendershall cesse to be a party hereto and thereto), and such assignment shall effect a novation between Bonower and the Assignee; provided, however, that nothing contained herein shall release any assigning Lender from
obligations that suvive the termination of this Agreement, including such assigning Lender's obligations under <u>Antide 16</u> and <u>Section 17.7</u> of this Agreement. The Assignee shall execute and deliver to the Agent and the Bourower the forms described in
Section

16.11 applicable to it.

- (c) The Agent, acting solely for this purpose as an on-fiduciary agent for the Borower, shall maintain at one of its offices acopy of each Assignment and Acceptance delivered to it and a register for the recordation of the names and addresses of the Lenders, and the Commitments of, and principal amount (and stated interest) of the Loans owing to, each Lender pursuant to the terms hereof from time to time (the "Register"). No assignment shall be effective unless it has been recorded in the Register approvised in this Section 141(c). The entries in the Register all ble conclusive, absent manifester curry, and the Borower, the Agent, and the Londers shall treatest Person whose name is neconclusified in the Register pursuant to the terms hereof-as a Lender here under for all purposes of this Agreement, not with standing notice to the contrary. This Section shall be construed so that the Loans are shall treate and in the gister with the part of any Lenders hall be available for inspection by such Londer at any reasonable time and from time to time upon reasonable prior notice.
- (d) By executing and delivering an Assignment and Acceptance, the assigning Lendertheaunder and the Assignee theaunder confirm to and agree with each other and the other parties here oas follows: (1) other than as provided in such Assignment and Acceptance, such assigning Lendermakes no representations want any and assumes no responsibility with respect to any statements, want arties or representations made in or incorrection with this Agreement or any other Loan Document furnished pursuant hereto, (2) such assigning Lendermakes no responsibility with respect to the financial condition of Bonower or the performance or observance by Bonower of any of its obligations under this Agreement or any other Loan Document furnished pursuant hereto, (3) such assignine confirms that it has received a copy of this Agreement, together with such other documents and information as it has deemed appropriate to make its own credit analysis

- 106 -137020431v13 and decision to enter into such Assignment and Acceptance, (4) such Assignee will, independently and without reliance upon Agent, such assigning Lender or any other Lender, and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under this Agreement, (5) such Assignee appoints and authorizes Agent to take such actions and to exercise such powers under this Agreement as are delegated to Agent, by the terms hereof, together with such powers as are reasonably incidental thereto, and (6) such Assignee agrees that it will perform all of the obligations which by the terms of this Agreement are required to be performed by it as a Lender.

- (e) Immediately upon Agent's receipt of the required processing fee payment and the fully executed Assignment and Acceptance, this Agreement shall be deemed to be amended to the extent, but only to the extent, necessary to reflect the addition of the Assignee and the resulting adjustment of the Commitments arising therefrom. The Commitment allocated to each Assignee shall reduce such Commitments of the assigning Lender to the extent of such assignment.
- (f) Any Lendermay at any time, with the written consent of Agent, sell to one commercial banks, financial institutions, or other Persons not Affiliates of such Lender (a "Patricipant") participating interests in its Obligations, the Commitment, and the other lights and interests of that Lender (the "Digitating Lender") here under and under the other Loan Documents, provided Invester, that (i) the Originating Lender shall remains "Lender" for all purposes of this Agreement and the other Loan Documents and the Originating Lender's obligations under this Agreement and the other Loan Documents and the Originating Lender's obligations under this Agreement and the other Loan Documents and the Originating Lender's obligations, and the Lender's hall remains obleve exponsible for the performance of such obligations, and the Lender's hall remains obligations under this Agreement and the other Loan Documents, (iv) no Lender's hall transfer or grant any participating interest under which the Participant has the right to approve any amendment to, or any consent or waiver with respect to this Agreement or any other Loan Document, except to the extent such amendment to, or consent or waiver with respect to this Agreement or any other Loan Document, except to the extent such amendment to, or consent or waiver with respect to this Agreement or any other Loan Document, except to the extent such amendment to, or consent or waiver with respect to this Agreement or any other Loan Document, except to the extent such amendment to, or consent or waiver with respect to this Agreement or any other Loan Document, except to the extent such and the final maturity date of the Obligations here under in which such Participant is participating. (C) release all or substantially all of the Obligations here under in which such Participant is participating. (C) release all or substantially all of the Collateral or grantiants (except to the extent expressly provided herein or in any of the Loan Documents) appraint give Documents or presyments or pr

- 107 -137020431v13 have any rights under this Agreement or the other Loan Documents or any direct rights as to the other Lenders, Agent, Borrower, the Collections of Borrower or its Subsidiaries, the Collateral, or otherwise in respect of the Obligations. No Participant shall have the right to participate directly in the making of decisions by the Lenders among themselves. Each Lender that sells a participation shall, acting solely for this purpose as an agent of Borrower, maintain a register on which it enters the name and address of each Participant and the principal amounts (and stated interest) of each Participant's interest in the Obligations (the "Participant Register"); provided that no Lender shall have any obligation to disclose all or any portion of the Participant Register (including the identity of any Participant or any information relating to a Participant's interest in any commitments, loans, letters of credit or its other obligations) to any Person except to the extent that such disclosure is necessary to establish that such commitment, loan, letter of credit or other obligation is in registered form under Section 5f.103-1(c) of the United States Treasury Regulations. The entries in the Participant Register shall be conclusive absent manifest error, and such Lender shall treat each Person whose name is recorded in the Participant Register as the owner of such participation for all purposes of this Agreement notwithstanding any notice to the contrary. Each Participant shall be entitled to the benefits and obligations of Section 16.11 (subject to the requirements and limitations of such section, including the requirements to provide tax documentation (it being understood that the documentation required under Section 16.11 (subject to the participant to subsection (a) of this Section 16.11 shall be delivered to the participating Lender)) to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to subsection (a) of this Section

- (g) Inconnection with any such assignment or participation or proposed assignment or participation, a Lendermay, subject to the provisions of Section 17.7, disclose all documents and information which it now or hereaftermay have relating to Bourower and its Subsidiaries and their respective businesses.
- (h) Any other provision in this Agreement not with standing, any Lender may at any time oceate a security interest in, or pledge, all or any portion of its rights under and interest in this Agreement in favor of any Federal Reserve Bank in accordance with Regulation A of the Federal Reserve Bank or U.S. Tiessury Regulation 31 CFR §20324, and such Federal Reserve Bank may enforce such pledge or security interest in any manner permitted under applicable law.
- 14.2 <u>Successors</u>. This Agreement shall bind and inure to the benefit of the respective successors and assigns of each of the parties; <u>provided</u>, <u>however</u>, that Borrower may not assign this Agreement or any rights or duties hereunder without the Lenders' prior written consent and any prohibited assignment shall be absolutely void *ab initio*. No consent to assignment by the Lenders shall release Borrower from its Obligations. A Lender may assign this Agreement and the other Loan Documents and its rights and duties hereunder and thereunder pursuant to <u>Section 14.1</u> hereof and, except as expressly required pursuant to <u>Section 14.1</u> hereof, no consent or approval by Borrower is required in connection with any such assignment.

#### 15. AMENDMENTS; WAIVERS.

15.1 Amendments and Waivers. No amendment or waiver of any provision of this Agreement or any other Loan Document (other than Bank Product Agreements), and no consent with respect to any departure by Borrower therefrom, shall be effective unless the same shall be

- 108 -

14.1

in writing and signed by the Required Lenders (or by Agent at the written request of the Required Lenders) and Borrower and then any such waiver or consent shall be effective, but only in the specific instance and for the specific purpose for which given; <u>provided</u>, <u>however</u>, that no such waiver, amendment, or consent shall, unless in writing and signed by all of the Lenders affected thereby and Borrower, do any of the following:

	(a)	incresse or extendany Commitment of any Lender, or
Document, or	(b)	postpone or delay any date fixed by this Agreement or any other Loan Document for any payment of principal, interest, fees, or other amounts due hereunder or under any other Loan Document for any payment of principal, interest, fees, or other amounts due here under any other Loan Document for any payment of principal, interest, fees, or other amounts due here under any other Loan Document for any payment of principal, interest, fees, or other amounts due here under any other Loan Document for any payment of principal, interest, fees, or other amounts due here under any other Loan Document for any payment of principal, interest, fees, or other amounts due here under any other Loan Document for any payment of principal, interest, fees, or other amounts due here under any other Loan Document for any payment of principal, interest, fees, or other amounts due here under any other Loan Document for any payment of principal, interest, fees, or other amounts due here under any other Loan Document for any payment of principal, interest, fees, or other amounts due here under any other Loan Document for any payment of principal and the pa
Document, or	(c)	reduce the principal of, or the rate of interest on, any loan or other extension of credit hereunder; or reduce any fees or other amounts payable hereunder or under any other Loan
	(d)	changethe Pro Rata Sharethat is required to take any action hereunder, or
	(e)	amend or modify this Section or any provision of the Agreement providing for consent or other action by all Lenders, or the consent of the Consent or other action by all Lenders, or the consent or other action by all Lenders, or the consent or other action by all Lenders, or the consent or other action by all Lenders, or the consent or other action by all Lenders, or the consent or other action by all Lenders, or the consent or other action by all Lenders, or the consent or other action by all Lenders, or the consent or other action by all Lenders, or the consent or other action by all Lenders, or the consent or other action by all Lenders, or the consent or other action by all Lenders, or the consent or other action by all Lenders, or the consent or other action by all Lenders, or the consent or other action by all Lenders, or the consent or other action by all Lenders, or the consent or other action by all Lenders are the consent or other action by all Lenders are the consent or other actions are the con
	<b>(f)</b>	other than as permitted by <u>Section 16.12</u> , release Agent's Lien in all or substantially all of the Collateral, or
	(g)	changethedefinition of "Required Lenders" or "Pro Rata Share", or
	(h)	increase the Maximum Commitment Amount permitted in <u>Section 2.15</u> , or
amendanyoftheprovis	(i) sions of <u>Section 16</u> .	other than as permitted by this Agreement or any other Loan Document, release Borrower or all or substantially all of the Guarantos from any obligation for the payment of money, or

and, <u>provided further</u>, <u>however</u>, that no amendment, waiver or consent shall, unless in writing and signed by Agent or Issuing Lender, as applicable, affect the rights or duties of Agent or Issuing Lender, as applicable, under this Agreement or any other Loan Document. The foregoing notwithstanding, any amendment, modification, waiver, consent, termination, or release of, or with respect to, any provision of this Agreement or any other Loan Document that relates only to the relationship of the Lender Group among themselves, and that does not affect the rights or obligations of Borrower, shall not require consent by or the agreement of Borrower.

#### 15.2 Replacement of Holdout Lender.

(a) If any action to be taken by the Lender Group or Agent hereunder requires the unanimous consent, authorization, or agreement of all Lenders, and a Lender ("<u>Hobbot Lender</u>") fails to give its consent, authorization, or agreement, then Agent, upon at least 5 Business Days prior inevocable notice to the Hobbot Lender; may permanently replace the Hobbot Lender with one or more substitute Lenders (each, a "<u>Replacement Lender</u>"), and the Hobbot Lendershall have no right to refuse to be replaced the reunder. Such notice to replace the

- 109 -137020431v13 Holdout Lender shall specify an effective date for such replacement, which date shall not be later than 15 Business Days after the date such notice is given.

- (b) Prior to the effective date of such replacement, the Holdout Lender and leach Replacement Lender shall execute and deliver an Assignment and Acceptance, subject only to the Holdout Lender being repaid its share of the outstanding Obligations (including an assumption of its Pro Rata Share of the Risk Paticipation Liability) without any premium or penalty of any kind what soever. If the Holdout Lender shall be executed and deliver any such Assignment and Acceptance The replacement of any Holdout Lender shall be executed and delivered such Assignment and Acceptance. The replacement of any Holdout Lender shall be made in a contained with the terms of Section 14.1. Until such time as the Replacement Lenders shall have acquired all of the Obligations, the Commitments, and the other rights and obligations of the Holdout Lender shall contain the advantage of the Risk Paticipation Liability of such Letter of Ceciti, in an amount equal to its Pro Rata Share of Advances and Delayed Draw Term Loan, and to purchase a paticipation in each Letter of Ceciti, in an amount equal to its Pro Rata Share of the Risk Paticipation Liability of such Letter of Ceciti, in an amount equal to its Pro Rata Share of the Risk Paticipation Liability of such Letter of Ceciti, in an amount equal to its Pro Rata Share of the Risk Paticipation Liability of such Letter of Ceciti.
- 15.3 No Waivers; Cumulative Remedies. No failure by Agent or any Lender to exercise any right, remedy, or option under this Agreement or any other Loan Document, or delay by Agent or any Lender in exercising the same, will operate as a waiver thereof. No waiver by Agent or any Lender will be effective unless it is in writing, and then only to the extent specifically stated. No waiver by Agent or any Lender on any occasion shall affect or diminish Agent's and each Lender's rights thereafter to require strict performance by Borrower of any provision of this Agreement. Agent's and each Lender's rights under this Agreement and the other Loan Documents will be cumulative and not exclusive of any other right or remedy that Agent or any Lender may have.

#### 16. AGENT; THE LENDER GROUP.

Appointment and Authorization of Agent. Each Lender hereby designates and appoints CNB as its representative under this Agreement and the other Loan Documents and each Lender hereby invocably authorizes Agent to execute and deliver each of the other Loan Documents on its behalf and to take such other action on its behalf under the provisions of this Agreement and each other Loan Document and to exercise such powers and perform such duties as are expressly delegated to Agent by the terms of this Agreement or any other Loan Document, together with such powers as are reasonably incidental thereto. Agent agrees to act as such on the express conditions contained in this Section 16. The provisions of this Section 16 are solely for the benefit of Agent, and the Lenders, and Borrower and its Subsidiaries shall have no rights as a third party beneficiary of any of the provisions contained herein. Any provision to the contrary contained elsewhere in this Agreement or in any other Loan Document notwithstanding, Agent shall not have any duties or responsibilities, except those expressly set forth herein, nor shall Agent have or be deemed to have any fiduciary relationship with any Lender, and no implied covenants, functions, responsibilities, duties, obligations or liabilities shall be read into this Agreement or any other Loan Document or otherwise exist against Agent; it being expressly understood and agreed that the use of the word "Agent" is for convenience only, that CNB is merely the representative of the Lenders, and only has the contractual duties set forth herein.

- 110 -137020431v13 Except as expressly otherwise provided in this Agreement, Agent shall have and may use its sole discretion with respect to exercising or refraining from exercising any discretionary rights or taking or refraining from taking any actions that Agent expressly is entitled to take or assert under or pursuant to this Agreement and the other Loan Documents. Without limiting the generality of the foregoing, or of any other provision of the Loan Documents that provides rights or powers to Agent, Lenders agree that Agent shall have the right to exercise the following powers as long as this Agreement remains in effect: (a) maintain, in accordance with its customary business practices, ledgers and records reflecting the status of the Obligations, the Collateral, the Collections of Borrower and its Subsidiaries, and related matters, (b) execute or file any and all financing or similar statements or notices, amendments, renewals, supplements, documents, instruments, proofs of claim, notices and other written agreements with respect to the Loan Documents, (c) make Advances, for itself or on behalf of Lenders as provided in the Loan Documents, (d) exclusively receive, apply, and distribute the Collections of Borrower and its Subsidiaries as provided in the Loan Documents, (e) open and maintain such bank accounts and cash management arrangements as Agent deems necessary and appropriate in accordance with the Loan Documents for the foregoing purposes with respect to the Collateral and the Collections of Borrower and its Subsidiaries, (f) perform, exercise, and enforce any and all other rights and remedies of the Lender Group with respect to Borrower, the Obligations, the Collateral, the Collections of Borrower and its Subsidiaries, or otherwise related to any of same as provided in the Loan Documents, and (g) incur and pay such Lender Group Expenses as Agent may deem necessary or appropriate for the performance and fulfillment of its functions and powers pursuant to the Loan Documents.

- 16.2 <u>Delegation of Duties</u>. Agent may execute any of its duties under this Agreement or any other Loan Document by or through agents, employees or attorneys in fact and shall be entitled to advice of counsel concerning all matters pertaining to such duties. Agent shall not be responsible for the negligence or misconduct of any agent or attorney in fact that it selects as long as such selection was made without gross negligence or willful misconduct.
- 16.3 Liability of Agent. None of the Agent-Related Persons shall (a) be liable for any action taken or omitted to be taken by any of them under or in connection with this Agreement or any other Loan Document or the transactions contemplated hereby (except for its own gross negligence or willful misconduct), or (b) be responsible in any manner to any of the Lenders for any recital, statement, representation or warranty made by Borrower or any Subsidiary or Affiliate of Borrower, or any officer or director thereof, contained in this Agreement or in any other Loan Document, or in any certificate, report, statement or other document referred to or provided for in, or received by Agent under or in connection with, this Agreement or any other Loan Document, or the validity, effectiveness, enforceability or sufficiency of this Agreement or any other Loan Document, or for any failure of Borrower or any other party to any Loan Document to perform its obligations hereunder or thereunder. No Agent-Related Person shall be under any obligation to any Lender to ascertain or to inquire as to the observance or performance of any of the agreements contained in, or conditions of, this Agreement or any other Loan Document, or to inspect the Books or properties of Borrower or the books or properties of any of Borrower's Subsidiaries or Affiliates.
  - 16.4 Reliance by Agent. Agent shall be entitled to rely, and shall be fully protected in relying, upon any writing, resolution, notice, consent, certificate, affidavit, letter, telegram,

- 111 -137020431v13 telefacsimile or other electronic method of transmission, telex or telephone message, statement or other document or conversation believed by it to be genuine and correct and to have been signed, sent, or made by the proper Person or Persons, and upon advice and statements of legal counsel (including counsel to Borrower or counsel to any Lender), independent accountants and other experts selected by Agent. Agent shall be fully justified in failing or refusing to take any action under this Agreement or any other Loan Document unless Agent shall first receive such advice or concurrence of the Lenders as it deems appropriate and until such instructions are received, Agent shall act, or refrain from acting, as it deems advisable. If Agent so requests, it shall first be indemnified to its reasonable satisfaction by the Lenders against any and all liability and expense that may be incurred by it by reason of taking or continuing to take any such action. Agent shall in all cases be fully protected in acting, or in refraining from acting, under this Agreement or any other Loan Document in accordance with a request or consent of the requisite Lenders and such request and any action taken or failure to act pursuant thereto shall be binding upon all of the Lenders.

- Notice of Default or Event of Default. Agent shall not be deemed to have knowledge or notice of the occurrence of any Default or Event of Default, except with respect to defaults in the payment of principal, interest, fees, and expenses required to be paid to Agent for the account of the Lenders and, except with respect to Events of Default of which Agent has actual knowledge, unless Agent shall have received written notice from a Lender or Borrower referring to this Agreement, describing such Default or Event of Default, and stating that such notice is a "notice of default." Agent promptly will notify the Lenders of its receipt of any such notice or of any Event of Default of which Agent has actual knowledge. If any Lender obtains actual knowledge of any Event of Default, such Lender promptly shall notify the other Lenders and Agent of such Event of Default. Each Lender shall be solely responsible for giving any notices to its Participants, if any. Subject to Section 16.4, Agent shall take such action with respect to such Default or Event of Default as may be requested by the Required Lenders in accordance with Section 9: provided, however, that unless and until Agent has received any such request, Agent may (but shall not be obligated to) take such action, or refrain from taking such action, with respect to such Default or Event of Default as it shall deem advisable.
- Credit Decision. Each Lender acknowledges that none of the Agent-Related Persons has made any representation or warranty to it, and that no act by Agent hereinafter taken, including any review of the affairs of Borrower and its Subsidiaries or Affiliates, shall be deemed to constitute any representation or warranty by any Agent-Related Person to any Lender. Each Lender represents to Agent that it has, independently and without reliance upon any Agent-Related Person and based on such documents and information as it has deemed appropriate, made its own appraisal of and investigation into the business, prospects, operations, property, financial and other condition and credit worthiness of Borrower and any other Person party to a Loan Document, and all applicable bank regulatory laws relating to the transactions contemplated hereby, and made its own decision to enter into this Agreement and to extend credit to Borrower. Each Lender also represents that it will, independently and without reliance upon any Agent-Related Person and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit analysis, appraisals and decisions in taking or not taking action under this Agreement and the other Loan Documents, and to make such investigations as it deems necessary to inform itself as to the business, prospects, operations, property, financial and other condition and credit worthiness of Borrower and any

other Person party to a Loan Document. Except for notices, reports, and other documents expressly herein required to be furnished to the Lenders by Agent, Agent shall not have any duty or responsibility to provide any Lender with any credit or other information concerning the business, prospects, operations, property, financial and other condition or creditworthiness of Borrower and any other Person party to a Loan Document that may come into the possession of any of the Agent-Related Persons.

- Costs and Expenses; Indemnification. Agent may incur and pay Lender Group Expenses to the extent Agent reasonably deems necessary or appropriate for the performance and fulfillment of its functions, powers, and obligations pursuant to the Loan Documents, including court costs, attorneys' fees and expenses, fees and expenses of financial accountants, advisors, consultants, and appraisers, costs of collection by outside collection agencies, auctioneer fees and expenses, and costs of security guards or insurance premiums paid to maintain the Collateral, whether or not Borrower is obligated to reimburse Agent or Lenders for such expenses pursuant to this Agreement or otherwise. Agent is authorized and directed to deduct and retain sufficient amounts from the Collections of Borrower and its Subsidiaries received by Agent to reimburse Agent for such out-of-pocket costs and expenses prior to the distribution of any amounts to Lenders. In the event Agent is not reimbursed for such costs and expenses from the Collections of Borrower and its Subsidiaries received by Agent, each Lender hereby agrees that it is and shall be obligated to pay to or reimburse Agent for the amount of such Lender's Pro Rata Share thereof. Whether or not the transactions contemplated hereby are consummated, the Lenders shall indemnify upon demnand the Agent-Related Persons (to the extent not reimbursed by or on behalf of Borrower and without limiting the obligation of Borrower to do so), according to their Pro Rata Shares, from and against any and all Indemnified Liabilities; provided, however, that no Lender shall be liable for the payment to any Agent-Related Person of any portion of such Indemnified Liabilities resulting solely from such Person's gross negligence or willful misconduct nor shall any Lender be liable for the obligations of any Defaulting Lender in failing to make an Advance, Delayed Draw Term Loan or other extension of credit hereunder: Without limitation of the foregoing, each Lender shall reimburse Agent upon demand for such
- 16.8 Agent in Individual Capacity. CNB and its Affiliates may make loans to, issue letters of credit for the account of, accept deposits from, acquire equity interests in, and generally engage in any kind of banking, trust, financial advisory, underwriting, or other business with Borrower and its Subsidiaries and Affiliates and any other Person party to any Loan Documents as though CNB were not Agent hereunder, and, in each case, without notice to or consent of the other members of the Lender Group. The other members of the Lender Group acknowledge that, pursuant to such activities, CNB or its Affiliates may receive information regarding Borrower or its Affiliates and any other Person party to any Loan Documents that is subject to confidentiality obligations in favor of Borrower or such other Person and that prohibit the disclosure of such

information to the Lenders, and the Lenders acknowledge that, in such circumstances (and in the absence of a waiver of such confidentiality obligations, which waiver Agent will use its reasonable best efforts to obtain), Agent shall not be under any obligation to provide such information to them. The terms "Lender" and "Lenders" include CNB in its individual capacity.

- Successor Agent. Agent may resign as Agent upon 45 days' notice to the Lenders. If Agent resigns under this Agreement, the Required Lenders shall appoint a successor Agent for the Lenders. If no successor Agent is appointed prior to the effective date of the resignation of Agent, Agent may appoint, after consulting with the Lenders, a successor Agent. If Agent has materially breached or failed to perform any material provision of this Agreement or of applicable law, the Required Lenders may agree in writing to remove and replace Agent with a successor Agent from among the Lenders. In any such event, upon the acceptance of its appointment as successor Agent hereunder, such successor Agent shall succeed to all the rights, powers, and duties of the retining Agent and the Required Lenders and the Lenders and the Lenders appoint and the retining Agent and the retining Agent and the Required Lenders and the Lenders a
- 16.10 Lender in Individual Capacity. Any Lender and its respective Affiliates may make loans to, issue letters of credit for the account of, accept deposits from, acquire equity interests in and generally engage in any kind of banking, trust, financial advisory, underwriting, or other business with Borrower and its Subsidiaries and Affiliates and any other Person party to any Loan Documents as though such Lender were not a Lender hereunder without notice to or consent of the other members of the Lender Group. The other members of the Lender Group acknowledge that, pursuant to such activities, such Lender and its respective Affiliates may receive information regarding Borrower or its Affiliates and any other Person party to any Loan Documents that is subject to confidentiality obligations in favor of Borrower or such other Person and that prohibit the disclosure of such information to the Lenders, and the Lenders acknowledge that, in such circumstances (and in the absence of a waiver of such confidentiality obligations, which waiver such Lender will use its reasonable best efforts to obtain), such Lender shall not be under any obligation to provide such information to them.

#### 16.11 <u>Withholding Taxes</u>.

(a) All payments made by Borowerhereunder or under any note or other Loan Document will be made without set off, counterclaim, or other defense. In addition, except as required by applicable law, all such payments will be made fire and clear of, and without deduction or withholding for, any present or future Taxes. If any applicable law, (as determined in the good faith discretion of the applicable withholding agent) requires the deduction or withholding of any Tax from any such payment by the Borower, then the applicable withholding agents hall be entitled to make such deduction or withholding and shall timely pay the full amount deducted or withheld to the relevant Governmental Authority in accordance with

- 114 -137020431v13 applicable law and, if such Tax is an Indemnified Tax, then the sum payable by the Borrower shall be increased as necessary so that after such deduction or withholding has been made (including such deductions and withholdings applicable to additional sums payable under this Section 16.11(a)) the applicable Recipient receives an amount equal to the sum it would have received had no such deduction or withholding been made. Borrower will furnish to Agent as promptly as possible after the date the payment of any Tax is due pursuant to applicable law certified copies of tax receipts evidencing such payment by Borrower.

- (b) Any Lenderthat is entitled to an exemption from one duction of withholding Tax with neget to payments made under any Loan Documentshall deliver to the Borrower and Agent, at the time or times reasonably requested by the Borrower or Agent, such properly completed and executed documentation reasonably requested by the Borrower or Agent as will permit such payments to be made withholding or at a reduced rate of withholding, in addition, any Lender, if reasonably requested by the Borrower or Agent, shall deliver such other documentation prescribed by applicable law or reasonably requested by the Borrower or Agent as will enable the Borrower or Agent as withholding or information reporting requirements. Not with standing anything to the contrary in the preceding two sentences, the completion, execution and submission of such documentation (other than such documentation set forth in paragaphs (b)(i)-
- (v) of this Section) shall not be required if in the Lender's reasonable judgment such completion, execution or submission would subject such Lender to any material unreimbursed cost or expense or would materially prejudice the legal or commercial position of such Lender. Without limiting the generality of the foregoing, each Lender agrees to deliver to Agent and Borrower:
- $(i) if such Lender claims an exemption from United States with holding tax pursuant to its portfolio interest exception, (A) a statement of the Lender substantially in the form of $\underline{Exhibit B-1}$, signed under penalty of penjury, that it is not a (I) a "bank" as described in Section 881 (c) (3) (A) of the IRC, (II) a <math>10\%$  shareholder of Bourover (within the meaning of Section 871 (h) (3) (B) of the IRC, or (III) a controlled foreign corporation related to Bourover within the meaning of Section 864 (d) (4) of the IRC, and (B) a properly completed and executed IRS Form W-8BEN-E, before receiving its first payment under this Agreement and at any other time reasonably requested by Agent or Bourover;
- (ii) if such Lenderclaims an exemption from, or a reduction of, with holding tax under a United States tax treaty, properly completed and executed IRS Form W-8BEN-E before receiving its first payment under this Agreement and at any other time reasonably requested by Agentor Borrower;
- (iii) if such Lender deinns that interest paid under this Agreement is even put from United States with holding tax because it is effectively connected with a United States trade or business of such Lender, two properly completed and executed copies of IRS Form W-RECI before receiving its first payment under this Agreement and at any other time reasonably requested by Agent or Borrower;
- (iv) to the extent such Lender is not the beneficial owner, executed copies of IRS Form W-8IMY, accompanied by IRS Form W-8ECI, IRS Form W-8EN, IRS Form W-8E

- 115 -137020431v13 W-9, and/or other certification documents from each beneficial owner, as applicable; provided that if such Lender is a partnership and one or more direct or indirect partners of such Lender are claiming the portfolio interest exemption, such Lender may provide a statement substantially in the form of <a href="Exhibit B-4">Exhibit B-4</a> on behalf of each such direct and indirect partner; or

- (v) IRS Form W-9, as may be required under the IRC corother laws of the United States as a condition to exemption from, or reduction of, United States with holding or backup with holding tax before receiving its first payment under this Agreement and at any other time reasonably requested by Agent or Borrower:
- (c) Any Foreign Lendershall, to the extent it is legally entitled to do so, deliver to Borrower and Agent (in such number of copies as shall be requested by the recipient) on or about the date on which such Lender becomes a Lender under this Agreement (and from time to time the reafter upon the reasonable request of Borrower or Agent), executed copies of any other from prescribed by applicable law as a basis for daiming exemption from or a reduction in U.S. federal with holding Tax, duly completed, together with such supplementary documentation as may be prescribed by applicable law to permit Borrower or Agent to determine the with holding or deduction required to be made.
- (d) If a payment made to a Lender under any Loan Document would be subject to U.S. federal withholding Tax imposed by FATCA if such Lender were to fail to comply with the applicable reporting requirements of FATCA (including those contained in Section 1471(b) or 1472(b) of the IRC, as applicable), such Lender shall deliver to Borrower and the Agent at the time or times prescribed by law and at such time or times prescribed by Borrower and the Agent active active and the Agent active active and the Agent active acti

Each Lender agrees that if any form or certification it previously delivered expires or becomes obsolete or inaccurate in any respect, it shall update such form or certification or promptly notify Borrower and Agent in writing of its legal inability to do so.

(e) The Borrowershall timely pay to the relevant Governmental Authority in accordance with applicable law, cratthe option of Agent timely reimburse it for the payment of, any Other Taxes.

(f) If the Agant is a U.S. Person, it shall deliver to the Borrower on or prior to the date on which it becomes the Agant under this Agreement (and from time to time the earlier upon the reasonable request of the Borrower). (A) two executed copies of Form W-BECI with respect to any amounts payable to the Agant for the account of others, certifying

- 116 -137020431v13 that it is a "U.S. branch" and that the payments it receives for the account of others are not effectively connected with the conduct of its trade or business within the United States and that it is using such form as evidence of its agreement with the Borrower to be treated as a U.S. Person with respect to such payments (and the Borrower and the Agent agree to so treat the Agent as a

U.S. Person with respect to such payments as contemplated by Section 1.1441-1(b)(2)(iv) of the United States Treasury Regulations).

- $(g) If any Lender is entitled to a reduction in the applicable with holding tax, Agant may with hold from any interest payment to such Lender an amount equivalent to the applicable with holding tax after taking into account such reduction. If the forms or other documentation required by <math display="block"> \frac{1}{2} = \frac{1}{2} \frac{1}{2} \frac{1}{2} = \frac{1}{2} \frac{1}{2} \frac{1}{2} = \frac{1}{2} \frac{1}{2} \frac{1}{2} = \frac{1}{2} \frac{1}{2} \frac{1}{2} = \frac{1}{2} \frac{1}$
- (h) If the IRS or any other Covernmental Authority of the United States or other jurisdiction asserts a daim that Agent did not propely withhold tax from amounts paid to or for the account of any Lender due to a failure on the part of the Lender (because the appropriate form was not delivered, was not properly executed, or because such Lender failed to motify Agent of a drange in circumstances which mendered the exemption from, or reduction of, withholding tax in effective, or for any other reason) such Lendershall indemnify and hold Agent harmless for all amounts paid, directly or indirectly, by Agent, as tax or otherwise, including parallies and interest, and including any taxes imposed by any jurisdiction on the amounts payable to Agent under this <u>Section 16.11</u>, together with all costs and expenses (including attorneys' fees and expenses). The obligation of the Lenders under this subsection shall survive the payment of all Obligations and the resignation or replacement of Agent.
- (i) If any party determines, in its sole discretion exercised in good faith, that it has received a refund of any Taxes as to which it has been in demnified pursuant to this Section (including by the payment of additional amounts pursuant to this Section), it shall pay to the indemnifying party an amount equal to such refund (but only to the extent of inclemnify payments made under this Section with respect to such refund), not of all out-of-prodet expenses (including Taxes) of such indemnified party and without intees (other than any interest paid by the relevant exting authority with respect to such refund). Such indemnifying party, upon the request of such indemnified party, shall repay to such indemnified party the amount paid over pursuant to this paragraph (to (plus any parallies, interest or other charges imposed by the relevant taxing authority) in the event that such indemnified party is required to repay such refund to such taxing authority. This paragraph is all most be constructed one quire any indemnified party to make available its Tax returns (or any other information retaining to its Taxes that it deems confidential) to the indemnificipanty or any other Person.

#### 16.12 Collateral Matters.

- (a) The Lendes heeby inevocably authorize Agent, at its option and in its sole discretion, to release any Lien on any Collateral (i) upon the termination of the Commitments and payment and satisfaction in full by Borrower of all Obligations,
- (ii) constituting property being sold or disposed of if a release is required or desirable in connection therewith and if Borrower certifies to Agent that the sale or disposition is permitted

- 117 -137020431v13 under Section 7.4 of this Agreement or the other Loan Documents (and Agent may rely conclusively on any such certificate, without further inquiry), (iii) constituting property in which Borrower or its Subsidiaries owned no interest at the time the Agent's Lien was granted nor at any time thereafter, or (iv) constituting property leased to Borrower or its Subsidiaries under a lease that has expired or is terminated in a transaction permitted under this Agreement. Except as provided above, Agent will not execute and deliver a release of any Lien on any Collateral without the prior written authorization of (y) if the release is of all or substantially all of the Collateral, all of the Lenders, or (z) otherwise, the Required Lenders. Upon request by Agent or Borrower at any time, the Lenders will confirm in writing Agent's authority to release any such Liens on particular types or items of Collateral pursuant to this Section 16.12; provided, however, , that (1) Agent shall not be required to execute any document necessary to evidence such release on terms that, in Agent's opinion, would expose Agent to liability or create any obligation or entail any consequence other than the release of such Lien without recourse, representation, or warranty, and (2) such release shall not in any manner discharge, affect, or impair the Obligations or any Liens (other than those expressly being released) upon (or obligations of Borrower in respect of) all interests retained by Borrower, including, the proceeds of any sale, all of which shall continue to constitute part of the Collateral.

(b) Agentshall have no obligation what sever to any of the Lenders to assure that the Collateral exists or is owned by Borrower or is cared for, protected, or insured or has been encombered, or that the Agent's Liens have been properly or sufficiently or knowledge protected, or enforced or accentified to any particular priority, or to exercise at all or in any particular manner or under any duty of case, disclosure of idelity, or to continue exercising, any of the rights, authorities and powers granted or available to Agent pursuant to any of the Loan Documents, it being understood and agreed that in respect of the Collateral, or any act, or is soin, or even the lateful threeto, subject to the terms and conditions contained herein, Agent may act in any manner it may deem appropriate, in its sole discretion given Agent's own interest in the Collateral in its capacity as one of the Lenders and that Agent shall have no other duty or liability what sever to any Lender as to any of the foregoing, except as otherwise provided herein.

#### 16.13 Restrictions on Actions by Lenders; Sharing of Payments.

- (a) Each of the Lenders agrees that its hall not, without the express written consent of Agent, and that its hall, to the extent it is lawfully entitled to do so, upon the written request of Agent, set off against the Obligations, any amounts owing by such Lender to Borrower or any deposit accounts of Borrowern own rhereafter maintained with such Lender. Each of the Lenders further agrees that its hall not, unless specifically requested to do so in writing by Agent, take or cause to be taken any action, including the commencement of any legal or equitable proceedings, to foreclose any Lienon, or otherwise enforce any security interest in, any of the Collateral.
- (b) If, at any time or times any Lendershall receive (i) by payment, foredooure, set off, or otherwise, any proceeds of Collateral or any payments with respect to the Obligations, except for any such proceeds or payments received by such Lender from Agent pursuant to the terms of this Agreement, or (ii) payments from Agent in excess of such Lender's ratable portion of all such distributions by Agent, such Lender promptly shall (1) turn the same

- 118 -137020431v13 over to Agent, in kind, and with such endorsements as may be required to negotiate the same to Agent, or in immediately available funds, as applicable, for the account of all of the Lenders and for application to the Obligations in accordance with the applicable provisions of this Agreement, or (2) purchase, without recourse or warranty, an undivided interest and participation in the Obligations owed to the other Lenders so that such excess payment received shall be applied ratably as among the Lenders in accordance with their Pro Rata Shares; provided, however, that to the extent that such excess payment received by the purchasing party is thereafter recovered from it, those purchases of participations shall be rescinded in whole or in part, as applicable, and the applicable portion of the purchase price paid therefor shall be returned to such purchasing party, but without interest except to the extent that such purchasing party is required to pay interest in connection with the recovery of the excess payment.

- 16.14 **Agency for Perfection.** Agent hereby appoints each other Lender as its agent (and each Lender hereby accepts such appointment) for the purpose of perfecting the Agent's Liens in assets which, in accordance with Article 8 or Article 9, as applicable, of the Code can be perfected only by possession or control. Should any Lender obtain possession or control of any such Collateral, such Lender shall notify Agent thereof, and, promptly upon Agent's request therefor shall deliver possession or control of such Collateral to Agent or in accordance with Agent's instructions.
- 16.15 Payments by Agent to the Lenders. All payments to be made by Agent to the Lenders shall be made by bank wire transfer of immediately available funds pursuant to such wire transfer instructions as each party may designate for itself by written notice to Agent. Concurrently with each such payment, Agent shall identify whether such payment (or any portion thereof) represents principal, premium, fees, or interest of the Obligations.
- 16.16 <u>Concerning the Collateral and Related Loan Documents</u>. Each member of the Lender Group authorizes and directs Agent to enter into this Agreement and the other Loan Documents. Each member of the Lender Group agrees that any action taken by Agent in accordance with the terms of this Agreement or the other Loan Documents relating to the Collateral and the exercise by Agent of its powers set forth therein or herein, together with such other powers that are reasonably incidental thereto, shall be binding upon all of the Lenders.
- 16.17 <u>Field Examinations and Examination Reports; Confidentiality; Disclaimers by Lenders; Other Reports and Information</u>. By becoming a party to this Agreement, each Lender:
- (a) is deemed to have requested that Agent furnish such Lender, promptly after it becomes available, acopy of each field examination or examination report (each a "Report" and collectively, "Reports") prepared by Agent, and Agent shall so furnish each Lender with such Reports,
- (b) expressly agrees and adknowledges that Agent does not (i) make any representation or warranty as to the accuracy of any Report, and (ii) shall not be liable for any information contained in any Report,

- 119 -137020431v13

- (c) expressly agrees and a knowledges that the Reports are not comprehensive audits or examinations, that Agent or other party performing any examination will inspect only specific information regarding Borrower and will bely significantly upon the Books, as well as on representations of Borrower's personnel, and the property of the Books and the Books as well as on representations of Borrower's personnel, and the Books are the Books as well as on representations of Borrower's personnel, and the Books are the Books as well as on representations of Borrower's personnel, and the Books are the Books as well as on representations of Borrower's personnel, and the Books are the Books as well as on representations of Borrower's personnel, and the Books are the Books as well as on representations of Borrower's personnel, and the Books are the Books as well as on representations of Borrower's personnel, and the Books are the Books as well as on representations of Borrower's personnel, and the Books are the Books are the Books as well as on representations of Borrower's personnel, and the Books are the Books are the Books as well as on the Books are the Books are the Books as well as on the Books are the Books are the Books as well as on the Books are the Books ar
- (d) agrees to keep all Reports and othermaterial, non-public information regarding Borrower and its Subsidiaries and their operations, assets, and existing and contemplated business plans in a confidential manner in accordance with <u>Section 17.7</u>, and
- (e) without limiting the generality of any other indemnification provision contained in this Agreement, agrees (i) to hold Agent and any other Lender preparing a Report harmless from any action the indemnifying Lender fail to take or any conclusion the indemnifying Lender has made or may make to Borrowe; or the indemnifying Lender's participation in, or the indemnifying Lender's purchase of, alconor borroof Borrowe; and (iii) to pay and protect, and indemnify, defend and hold Agent, and any such other Lender preparing a Report harmless from and against, the claims, actions, proceedings, damages, costs, expenses, and other amounts (including attorneys' fees and costs) incurred by Agent and any such other Lender preparing a Report as the direct or indirect result of any third parties who might obtain all or part of any Report through the indemnifying Lender.

In addition to the foregoing: (x) any Lender may from time to time request of Agent in writing that Agent provide to such Lender a copy of any report or document provided by Borrower to Agent that has not been contemporaneously provided by Borrower to such Lender, and, upon receipt of such request, Agent promptly shall provide a copy of same to such Lender, (y) to the extent that Agent is entitled, under any provision of the Loan Documents, to request additional reports or information from Borrower, any Lender may, from time to time, reasonably request Agent to exercise such right as specified in such Lender's notice to Agent, whereupon Agent promptly shall request of Borrower the additional reports or information reasonably specified by such Lender, and, upon receipt thereof from Borrower, Agent promptly shall provide a copy of same to such Lender, and (z) any time that Agent renders to Borrower a statement regarding the Loan Account, Agent shall send a copy of such statement to each Lender.

16.18 Several Obligations; No Liability. Notwithstanding that certain of the Loan Documents now or hereafter may have been or will be executed only by or in favor of Agent in its capacity as such, and not by or in favor of the Lenders, any and all obligations on the part of Agent (if any) to make any credit available hereunder shall constitute the several (and not joint) obligations of the respective Lenders on a ratable basis, according to their respective Commitments, to make an amount of such credit not to exceed, in principal amount, at any one time outstanding, the amount of their respective Commitments. Nothing contained herein shall confer upon any Lender any interest in, or subject any Lender to any liability for, or in respect of, the business, assets, profits, losses, or liabilities of any other Lender. Each Lender shall be solely responsible for notifying its Participants of any material responsible to the Loan Documents to the extent any such notice may be required, and no Lender shall have any other Lender. Except as provided in Section 16.7, no member of the Lender Group shall have any liability for the acts of any other member of the Lender Group. No Lender shall be responsible to Borrower or any other Person for any failure by any

- 120 -137020431v13 other Lender to fulfill its obligations to make credit available hereunder, nor to advance for it or on its behalf in connection with its Commitment, nor to take any other action on its behalf hereunder or in connection with the financing contemplated herein.

- 16.19 Bank Product Providers. Each Bank Product Provider shall be deemed a party hereto for purposes of any reference in a Loan Document to the parties for whom Agent is acting; it being understood and agreed that the rights and benefits of such Bank Product Provider under the Loan Documents consist exclusively of such Bank Product Provider's right to share in payments and collections out of the Collateral as more fully set forth herein. In connection with any such distribution of payments and collections, Agent shall be entitled to assume no amounts are due to any Bank Product Provider unless such Bank Product Provider has notified Agent in writing of the amount of any such liability owed to it prior to such distribution.
- 16.20 Lead Arranger. Lead Arranger, in such capacity, shall not have any right, power, obligation, liability, responsibility, or duty under this Agreement other than those applicable to it in its capacity as a Lender or as Agent, as applicable. Without limiting the foregoing, Lead Arranger, in such capacity, shall not have or be deemed to have any fiduciary relationship with any Lender or any Loan Party. Each Lender, Agent, Issuing Lender, and each Loan Party acknowledges that it has not relied, and will not rely, on the Lead Arranger in deciding to enter into this Agreement or in taking or not taking action hereunder. Lead Arranger, in such capacity, shall be entitled to resign at any time by giving notice to Agent and Borrower.
- 16.21 Legal Representation of Agent. In connection with the negotiation, drafting, and execution of this Agreement and the other Loan Documents, or in connection with future legal representation relating to loan administration, amendments, modifications, waivers, or enforcement of remedies, Katten Muchin Rosenman LLP ("KMR") only has represented and only shall represent CNB in its capacity as Agent and as a Lender. Each other Lender hereby acknowledges that KMR does not represent it in connection with any such matters.

#### GENERAL PROVISIONS.

- 17.1 **Effectiveness.** This Agreement shall be binding and deemed effective when executed by Borrower, Agent, and each Lender whose signature is provided for on the signature pages hereof.
- 17.2 Section Headings. Headings and numbers have been set forth herein for convenience only. Unless the contrary is compelled by the context, everything contained in each Section applies equally to this entire Agreement.
- 17.3 Interpretation. Neither this Agreement nor any uncertainty or ambiguity herein shall be construed against the Lender Group or Borrower, whether under any rule of construction or otherwise. On the contrary, this Agreement has been reviewed by all parties and shall be construed and interpreted according to the ordinary meaning of the words used so as to accomplish fairly the purposes and intentions of all parties hereto.
- 17.4 Severability of Provisions. Each provision of this Agreement shall be severable from every other provision of this Agreement for the purpose of determining the legal enforceability of any specific provision.

17.5 Counterparts; Electronic Execution. This Agreement may be executed in any number of counterparts and by different parties on separate counterparts, each of which, when executed and delivered, shall be deemed to be an original, and all of which, when taken together; shall constitute but one and the same Agreement. Delivery of an executed counterpart of this Agreement by telefacsimile or other electronic method of transmission shall be equally as effective as delivery of an original executed counterpart of this Agreement. Any party delivering an executed counterpart of this Agreement by telefacsimile or other electronic method of transmission also shall deliver an original executed counterpart of this Agreement but the failure to deliver an original executed counterpart shall not affect the validity, enforceability, and binding effect of this Agreement. The foregoing shall apply to each other Loan Document mutatis mutandis.

#### 17.6 Revival and Reinstatement of Obligations; Certain Waivers.

(a) If any member of the Lender Group or any Bank Product Providen epays, refunds, restores, on etums in whole or in part, any payment or property (including any proceeds of Collateral) previously paid or transferred to such member of the Lender Group or such Bank Broduct Providen in full or partial satisfaction of any Obligation coronaccount of any other obligation of any Loan Party Loan Document or any Bank Product Agreement, because the payment, transfer, or the incurrence of the obligations os satisfied is assented or declared to be void, voidable, or other visited in order any law relating to ordinos' rights, including provisions of the Bankuptcy Code relating to fault under the satisfied payment, transfer, or other voidable or recoverable obligations or transfers (each, a "<u>Voidable Tiansfer</u>"), or because such member of the Lender Group or Bank Product Provider elects to do so on the reasonable advice of its coursel in correction with a dain that the payment, transfer, or incurrence is ormany be a Voidable Tiansfer, then, as to any such Voidable Tiansfer, or the amount the evolution than the payment, transfer, or incurrence is ormany be a Voidable Tiansfer, then, as to any such Voidable Tiansfer, or the amount the evolution than the payment, transfer, or incurrence is ormany be a Voidable Tiansfer, then, as to any such Voidable Tiansfer, or the amount the evolution to the Londer Group or Bank Product Provider elects to repay.

restore, or return (including pursuant to a settlement of any claim in respect thereof), and as to all reasonable costs, expenses, and attorneys' fees of such member of the Lender Group or Bank Product Provider related thereto, (i) the liability of the Loan Parties with respect to the amount or property paid, refunded, restored, or returned will automatically and immediately be revived, reinstated, and restored and will exist and (ii) Agent's Liens securing such liability shall be effective, revived, and remain in full force and effect, in each case, as fully as if such Voidable Transfer had never been made. If, prior to any of the foregoing, (A) Agent's Liens shall have been released or terminated or (B) any provision of this Agreement shall have been terminated or cancelled, Agent's Liens, or such provision of this Agreement, shall be reinstated in full force and effect and such prior release, termination, cancellation or surrender shall not diminish, release, discharge, impair or otherwise affect the obligation of any Loan Party in respect of such liability or any Collateral securing such liability.

(b) Anything to the contrary contained hereinnot with standing, if Agent or any Lender accepts a guaranty of only aportion of the Obligations pursuant to any guaranty, Bourowerhereby waives its right under Section 2822(a) of the California Civil Code or any similar laws of any other applicable jurisdiction to designate the portion of the Obligations satisfied by the applicable guarantor's partial payment.

- 122 -137020431v13

- Confidentiality. Agent and Lenders each individually (and not jointly or jointly and severally) agree that material, non-public information regarding Borrower and its Subsidiaries, their operations, assets, and existing and contemplated business plans shall be treated by Agent and the Lenders in a confidential manner, and shall not be disclosed by Agent and the Lenders to Persons who are not parties to this Agreement, except: (a) to attorneys for and other advisors, accountants, auditors, and consultants to any member of the Lender Group, (b) to Subsidiaries and Affiliates of any member of the Lender Group (including the Bank Product Providers), provided that any such Subsidiary or Affiliate shall have agreed to receive such information hereunder subject to the terms of this Section 17.7. (c) as may be required by statute, decision, or judicial or administrative order, rule, or regulation, (d) as may be agreed to in advance by Borrower or its Subsidiaries or as requested or required by any Governmental Authority pursuant to any subpoena or other legal process, (e) as to any such information that is or becomes generally available to the public (other than as a result of prohibited disclosure by Agent or the Lenders), (f) in connection with any assignment, prospective assignment, sale, prospective sale, participation or prospective participations, or pledge or prospective pledge of any Lender's interest under this Agreement, provided that any such assignee, prospective assignee, purchaser, prospective purchaser, participant, prospective participant, pledgee, or prospective pledgee shall have agreed in writing to receive such information hereunder subject to the terms of this Section, and (g) in connection with any litigation or other adversary proceeding involving parties hereto which such litigation or adversary proceeding involves claims related to the rights or duties of such parties under this Agreement or the other Loan Documents. The provisions of this Section 17.7 shall survive for 2 years after the paymen
- 17.8 Integration. This Agreement, together with the other Loan Documents, reflects the entire understanding of the parties with respect to the transactions contemplated hereby and shall not be contradicted or qualified by any other agreement, oral or written, before the date hereof.
- Amendment and Restatement. This Agreement amends and restates in its entirety the Existing Loan Agreement effective as of the date hereof. Anything contained herein to the contrary notwithstanding, this Agreement is not intended to and shall not serve to effect a novation of the "Obligations" (as defined in the Existing Loan Agreement). Instead, it is the express intention of the parties hereto to reaffirm the indebtedness, obligations and liabilities created under the Existing Loan Agreement which is secured by the Collateral pursuant to the terms of the Loan Documents, as reaffirmed by the reaffirmations required hereunder. Each of Borrower and the other Loan Parties acknowledges and confirms (x) that the liens and security interests granted pursuant to the Loan Documents secure the applicable indebtedness, liabilities and obligations of the Loan Parties to the Agent and the Lenders under the Existing Loan Agreement, as amended and restated by this Agreement, (y) the Loan Documents shall continue in full force and effect in accordance with their terms unless otherwise amended by the parties thereto, and (z) that the term "Obligations" as used in the Loan Documents (or any other term used therein to describe or refer to the indebtedness, liabilities and obligations of the Loan

Parties to the Agent and the Lenders) includes, without limitation, the indebtedness, liabilities and obligations of the Loan Parties under this Agreement and any notes delivered hereunder, and under the Existing Loan Agreement, as amended and restated hereby, as the same may be further amended, modified, supplemented and/or restated from time to time. The Loan Documents and all agreements, instruments and documents executed or delivered in connection with any of the foregoing shall each be deemed to be amended to the extent necessary to give effect to the provisions of this Agreement. Each reference to the "Loan and Security Agreement" in any Loan Document shall mean and be a reference to this Agreement (as further amended, restated, supplemented or otherwise modified from time to time). Cross-references in the Loan Documents to particular section numbers in the Existing Loan Agreement shall be deemed to be cross-references to the corresponding sections, as applicable, of this Agreement. Upon the effectiveness of this Agreement, and on and after the Restatement Effective Date, each reference in the Existing Loan Agreement to "this Agreement", "hereof", "hereof", "herein", or words of like import, and each reference to the Existing Loan Agreement in any other related document, including any Loan Document as amended hereby, shall mean and be a reference to this Agreement.

17.10 Keepwell. If Borrower is a Qualified ECP Guarantor at the time the guarantee or the grant of a security interest under the Loan Documents by any Specified Loan Party becomes effective with respect to any Hedge Obligation, Borrower hereby absolutely, unconditionally and invevocably undertakes to provide such funds or other support to each Specified Loan Party as may be needed by such Specified Loan Party from time to time to honor all of its obligations under the Loan Documents in respect of such Hedge Obligation (but, in each case, only up to the maximum amount of such liability that can be hereby incurred by Borrower without rendering its obligations and undertakings under this Section 17.10 or otherwise under this Agreement voidable under applicable bankruptcy or insolvency laws, and not for any greater amount). The obligations and undertakings of Borrower under this Section shall remain in full force and effect until the Obligations have been paid and performed in full. Borrower intends this Section to constitute, and this Section shall be deemed to constitute, a "keepwell, support, or other agreement" for the benefit of each Specified Loan Party for all purposes of the Commodity Exchange Act.

17.11 Acknowledgement and Consent to Bail-In. Solely to the extent any Lender or Issuing Lender that is an EEA Financial Institution is a party to this Agreement and notwithstanding anything to the contrary in any Loan Document or in any other agreement, arrangement or understanding among any such parties, each party hereto acknowledges that any liability of any Lender or Issuing Bank that is an EEA Financial Institution arising under any Loan Document, to the extent such liability is unsecured, may be subject to the write-down and conversion powers of an EEA Resolution Authority and agrees and consents to, and acknowledges and agrees to be bound by:

(a) the application of any Wite-down and Conversion Powers by an EEA Resolution Authority to any such liabilities arising hereunder which may be payable to it by any Lender Issuing Bankthatis an EEA Financial Institution; and

 $\mbox{ the effects of any Bail-In Action on any such liability, including, if applicable:} \label{eq:Bail-In Action on any such liability, including if applicable:}$ 

- 124 -137020431v13 a reduction in full or in part or cancellation of any such liability;

(i)

- (ii) a conversion of all, or a portion of, such liability into shares or other instruments of ownership in such EEA Financial Institution, its parent undertaking, or a bridge institution that may be issued to it or otherwise conferred on it, and that such shares or other instruments of ownership will be accepted by it in lieu of any rights with respect to any such liability under this Agreement or any other Loan Document; or
  - (iii) the variation of the terms of such liability in connection with the exercise of the write-down and conversion powers of any EEA Resolution

Authority.

- 17.12 Patriot Act Notice. Each Lender subject to the Patriot Act hereby notifies Borrower and the other Loan Parties that, pursuant to the requirements of the Patriot Act, such Lender is required to obtain, verify and record information that identifies each Loan Party, which information includes the name and address of each Loan Party and other information that will allow such Lender to identify each Loan Party in accordance with the Patriot Act.
- 17.13 No Advisory or Fiduciary Responsibility. In connection with all aspects of each transaction contemplated hereby (including in connection with any amendment, waiver or other modification hereof or of any other Loan Document), each Loan Party acknowledges and agrees that: (i) (A) the arranging and other services regarding this Agreement provided by the Agent, the Lead Arranger, the Issuing Bank and the Lenders are am's-length commercial transactions between the Borrower and its Affiliates, on the one hand, and the Agent, the Lead Arranger, the Issuing Bank and the Lenders, on the other hand, (B) each Loan Party has consulted its own legal, accounting, regulatory and tax advisors to the extent it has deemed appropriate, and (C) each Loan Party is capable of evaluating, and understands and accepts, the terms, risks and conditions of the transactions contemplated hereby and by the other Loan Documents; (ii) (A) the Agent, the Lead Arranger, the Issuing Bank and the Lenders each is and has been acting solely as a principal and, except as expressly agreed in writing by the relevant parties, has not been, is not, and will not be acting as an advisor, agent or fiduciary for the Borrower or any of its Affiliates, or any other Person and (B) none of the Agent, the Lead Arranger, the Issuing Bank and the Lenders has any obligation to the Borrower or any of its Affiliates with respect to the transactions contemplated hereby except those obligations expressly set forth herein and in the other Loan Documents; and (iii) the Agent, the Lead Arranger, the Issuing Bank and the Lenders and their respective Affiliates entry by the engaged in a broad range of transactions that involve interests that differ from those of the Borrower and its Affiliates, and none of the Agent, the Lead Arranger, the Issuing Bank and the Lenders has any obligation to disclose any of such interests to the Borrower or its Affiliates. To the fullest extent permitted by law, each Loan Party hereby waives and releases any claims that it may have again

[Signature pages to follow.]

- 125 -137020431v13 **IN WITNESS WHEREOF,** the parties hereto have caused this Agreement to be executed and delivered as of the date first above written.

FRESHPET, INC., a Delaware Corporation

By: /s/ Richard Kassar
Name: Richard Kassar
Title: Chief Financial Officer

[Signature Page to Fourth Amended and Restated Loan and Security Agreement]

CITY NATIONAL BANK, a national banking association As Agent, Lead Arranger and as a lender

By: /s/ Garen Papazyan
Name: Garen Papazyan
Title: Senior Vice President

 $[Signature\ Page\ to\ Fourth\ Amended\ and\ Restated\ Loan\ and\ Security\ Agreement]$ 

### BANK OF AMERICA, N.A.,

as a lender

By: /s/ Edward Macclti
Name: Edward Macclti
Title: Senior Vice President

#### EXHIBITS AND SCHEDULES

	EATIDITS AND SCIED
Exhibit A-1	Form of Assignment and Acceptance
Exhibit B-1	Form of U.S. Tax Compliance Certificate
Exhibit B-2	Form of U.S. Tax Compliance Certificate
Exhibit B-3	Form of U.S. Tax Compliance Certificate
Exhibit B-4	Form of U.S. Tax Compliance Certificate
Exhibit C-1	Form of Compliance Certificate
Exhibit L-1	Form of LIBOR Notice
Exhibit N-1	Form of Notice of Borrowing
Schedule A-1	Agent's Account
Schedule C-1	Commitments
Schedule D-1	Designated Account
Schedule P-1	Permitted Liens
Schodulo D-2	Dormitted Holders

Schedule P-2 Schedule R-1 Permitted Holders Real Property Collateral Schedule 2.7(a) Cash Management Banks Schedule 2.7(a)
Schedule 5.5
Schedule 5.7(b)
Schedule 5.7(c)
5.7(d)
Schedule 5.8(c)
Schedule 5.8(d)
Schedule 5.10
Schedule 5.10 Locations of Inventory and Equipment States of Organization

States of Organization
Chief Executive Offices
Organizational Identification Numbers Schedule
Commercial Tort Claims
Capitalization of Borrower's Subsidiaries
Subscriptions, Options, Warrants or Calls Relating to Shares of Borrower's Subsidiaries' Capital Stock

Litigation Environmental Matters Intellectual Property Schedule 5.14 Schedule 5.16 Deposit Accounts and Securities Accounts Schedule Permitted Indebtedness

Schedule 5.18 5.20

#### **EXHIBIT A-1**

#### FORM OF ASSIGNMENT AND ACCEPTANCE AGREEMENT

This ASSIGNMENT AND ACCEPTANCE AGREEMENT ("Assignment

Agreement") is entered into as of , 20 between ("Assignor") and

("Assignee"). Reference is made to the Agreement described in <u>Annex I</u> hereto (the "Credit Agreement"). All initially capitalized terms used herein and not otherwise defined shall have the meanings ascribed to them in the Credit Agreement.

- 1. In accordance with the terms and conditions of Section 14.1 of the Credit Agreement, the Assignor hereby sells and assigns to the Assignee, and the Assignee hereby purchases and assumes from the Assignor; that interest in and to the Assignor; sights and obligations under the Loan Documents as of the date hereof with respect to the Obligations owing to the Assignor; and Assignor's portion of the Commitments, all to the extent specified on Annex I.
- 2. The Assignor (a) represents and warrants that (i) it is the legal and beneficial owner of the interest being assigned by it hereunder and that such interest is free and clear of any adverse claim and (ii) it has full power and authority, and has taken all action necessary, to execute and deliver this Assignment Agreement and to consummate the transactions contemplated hereby; (b) makes no representation or warranty and assumes no responsibility with respect to (i) any statements, representations or warranties made in or in connection with the Loan Documents, or (ii) the execution, legality, validity, enforceability, genuineness, sufficiency or value of the Loan Documents or any other instrument or document furnished pursuant thereto; (c) makes no representation or warranty and assumes no responsibility with respect to the financial condition of Borrower or any Guarantor or the performance or observance by Borrower or any Guarantor of any of their respective obligations under the Loan Documents or any other instrument or document furnished pursuant thereto, and
- (d) represents and warrants that the amount set forth as the Purchase Price on Annex I represents the amount owed by Borrower to Assignor with respect to Assignor's share of the loan assigned hereunder, as reflected on Assignor's books and records.
- 3. The Assignee (a) confirms that it has received copies of the Credit Agreement and the other Loan Documents, together with copies of the financial statements referred to therein and such other documents and information as it has deemed appropriate to make its own credit analysis and decision to enter into this Assignment Agreement; (b) agrees that it will, independently and without reliance upon Agent, Assignor, or any other Lender; based upon such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking any action under the Loan Documents;
- (c) appoints and authorizes the Agent to take such action as agent on its behalf and to exercise such powers under the Loan Documents as are delegated to Agent by the terms thereof, together with such powers as are reasonably incidental thereto; (d) agrees that it will perform in accordance with their terms all of the obligations which by the terms of the Loan Documents are required to be performed by it as a Lender; [and (e) attaches the forms prescribed by the Internal Revenue Service of the United States certifying as to the Assignee's status for purposes of

determining exemption from United States withholding taxes with respect to all payments to be made to the Assignee under the Credit Agreement or such other documents as are necessary to indicate that all such payments are subject to such rates at a rate reduced by an applicable tax treaty.]

- 4. Following the execution of this Assignment Agreement by the Assignor and Assignee, the Assignor will deliver this Assignment Agreement to the Agent for recording by the Agent. The effective date of this Assignment (the "Settlement Date") shall be the latest to occur of (a) the date of the execution and delivery hereof by the Assignor and the Assignee, (b) the receipt by Agent for its sole and separate account a processing fee in the amount of \$5,000, (c) the receipt of any required consent of Agent, and (d) the date specified on Annex I.
- 5. As of the Settlement Date (a) the Assignee shall be a party to the Credit Agreement and, to the extent of the interest assigned pursuant to this Assignment Agreement, have the rights and boligations of a Lender thereunder and under the other Loan Documents, and (b) the Assignment Sall, to the extent of the interest assigned pursuant to this Assignment Agreement, relinquish its rights and be released from its obligations under the Credit Agreement and the other Loan Documents, provided, however, that nothing contained herein shall release any assigning Lender from obligations that survive the termination of this Agreement, including such assigning Lender's obligations under Article 16 and Section 17.7 of the Credit Agreement.
- 6. Upon the Settlement Date, Assignee shall pay to Assignor the Purchase Price (as set forth in <u>Annex 1</u>). From and after the Settlement Date, Agent shall make all payments that are due and payable to the holder of the interest assigned hereunder (including payments of principal, interest, fees and other amounts) to Assignor for amounts which have accrued up to but excluding the Settlement Date and to Assignee for amounts which have accrued from and after the Settlement Date. On the Settlement Date, Assignor shall pay to Assignee an amount equal to the portion of any interest, fee, or any other charge that was paid to Assignor prior to the Settlement Date on account of the interest assigned hereunder and that are due and payable to Assignee with respect thereto, to the extent that such interest, fee or other charge relates to the period of time from and after the Settlement Date.
- 7. This Assignment Agreement may be executed in counterparts and by the parties hereto in separate counterparts, each of which when so executed and delivered shall be an original, but all of which shall together constitute one and the same instrument. This Assignment Agreement may be executed and delivered by telecopier or other facsimile transmission all with the same force and effect as if the same were a fully executed and delivered original manual counterpart.
- 8. THIS ASSIGNMENT AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF CALIFORNIA.

IN WITNESS WHEREOF, the parties hereto have caused this Assignment Agreement and Annex I hereto to be executed by their respective officers, as of the first date written above.

[NAME OF ASSIGNOR], as Assignor

Name: Title: By:

[NAME OF ASSIGNEE], as Assignee By:

Name: Title:

ACCEPTED THIS. DAY OF

CITY NATIONAL BANK,

a national banking association, as Agent

By: Name: Title:

[CONSENTED TO THIS\_, 20 DAY OF

FRESHPET, INC.,

a Delaware corporation, as Borrower

By:

Name: Title: ]1

1 Include to the extent required by Section 14.1(a) of the Credit Agreement.

#### ANNEX FOR ASSIGNMENT AND ACCEPTANCE ANNEX I

- 1. Borrower: FRESHPET, INC., a Delaware corporation
- 2. Name and Date of Credit Agreement:

Fourth Amended and Restated Loan and Security Agreement, dated as of May 15, 2019, by and among Borrower, the lenders identified on the signature pages thereof (such lenders, together with their respective successors and assigns, are referred to hereinafter each individually as a "Lender" and collectively as the "Lenders") and CITY NATIONAL BANK, a national banking association, as the lead arranger and administrative agent for the Lenders (in such capacity as administrative agent, together with its successors and assigns, if any, in such capacity, "Agent").

- 3. Date of Assignment Agreement:
- 4. Amounts:
  - Assigned Amount of Revolver Commitment\$
  - b. Assigned Amount of Delayed Draw Term Loan\$ Commitment

  - Assigned Amount of Advances\$ Assigned Amount of Delayed Draw Term Loans\$ c. d.
- Settlement Date: 5.
- 6. Purchase Price\$
- 7. Notice and Payment Instructions, etc.

Assignee: Assignor:

## FORM OF U.S. TAX COMPLIANCE CERTIFICATE (For Foreign Lenders That Are Not Partnerships For U.S. Federal Income Tax Purposes)

Reference is hereby made to the Fourth Amended and Restated Loan and Security Agreement, dated as of May 15, 2019 (as amended, supplemented or otherwise modified from time to time, the "Loan Agreement"), by and among Freshpet, Inc., a Delaware corporation (the "Borrower"), the lenders identified on the signature pages thereof (such lenders, together with their respective successors and assigns, are referred to hereinafter each individually as a "Lender" and collectively as the "Lenders") and City National Bank, a national banking association, as the lead arranger and administrative agent for the Lenders (in such capacity as administrative agent, "Administrative Agent").

Pursuant to the provisions of Section 16.11 of the Loan Agreement, the undersigned hereby certifies that (i) it is the sole record and beneficial owner of the Loan(s) (as well as any Note(s) evidencing such Loan(s)) in respect of which it is providing this certificate, (ii) it is not a "bank" within the meaning of Section 81(c)(3)(A) of the Code, (iii) it is not a "ten percent shareholder" of the Borrower within the meaning of Section 871(h)(3)(B) of the Code and (iv) it is not a "controlled foreign corporation" related to the Borrower as described in Section 881(c)(3)(C) of the Code.

The undersigned has furnished the Administrative Agent and the Borrower with a certificate of its non-U.S. Person status on IRS Form W-8BEN or IRS Form W-8BEN-E. By executing this certificate, the undersigned agrees that (1) if the information provided in this certificate changes, the undersigned shall promptly so inform the Borrower and the Administrative Agent, and (2) the undersigned shall have at all times furnished the Borrower and the Administrative Agent with a properly completed and currently effective certificate in either the calendar year in which each payment is to be made to the undersigned, or in either of the two calendar years preceding such payments.

Unless otherwise defined herein, terms defined in the Loan Agreement and used herein shall have the meanings given to them in the Loan Agreement.

[NAME OF LENDER]

Exhibit B-1 139215038v1

# FORM OF U.S. TAX COMPLIANCE CERTIFICATE (For Foreign Participants That Are Not Partnerships For U.S. Federal Income Tax Purnoses)

Reference is hereby made to the Fourth Amended and Restated Loan and Security Agreement, dated as of May 15, 2019 (as amended, supplemented or otherwise modified from time to time, the "Loan Agreement"), by and among Freshpet, Inc., a Delaware corporation (the "Borrower"), the lenders identified on the signature pages thereof (such lenders, together with their respective successors and assigns, are referred to hereinafter each individually as a "Lender" and collectively as the "Lenders") and City National Bank, a national banking association, as the lead arranger and administrative agent for the Lenders (in such capacity as administrative agent, together with its successors and assigns, if any, in such capacity, "Administrative Agent").

Pursuant to the provisions of Section 16.11 of the Loan Agreement, the undersigned hereby certifies that (i) it is the sole record and beneficial owner of the participation in respect of which it is providing this certificate, (ii) it is not a "bank" within the meaning of Section 881(c)(3)(A) of the Code, (iii) it is not a "ten percent shareholder" of the Borrower within the meaning of Section 871(h)(3)(B) of the Code and (iv) it is not a "controlled foreign corporation" related to the Borrower as described in Section 881(c)(3)(C) of the Code.

The undersigned has furnished its participating Lender with a certificate of its non-

U.S. Person status on IRS Form W-8BEN or IRS Form W-8BEN-E. By executing this certificate, the undersigned agrees that (1) if the information provided in this certificate changes, the undersigned shall promptly so inform such Lender in writing, and (2) the undersigned shall have at all times furnished such Lender with a properly completed and currently effective certificate in either the calendar year in which each payment is to be made to the undersigned, or in either of the two calendar years preceding such payments.

Unless otherwise defined herein, terms defined in the Loan Agreement and used herein shall have the meanings given to them in the Loan Agreement.

[NAME OF PARTICIPANT]

By:_ Title:		Name:
Date:_	, 20[]	
Exhibit B-2		

## FORM OF U.S. TAX COMPLIANCE CERTIFICATE (For Foreign Participants That Are Partnerships For U.S. Federal Income Tax Purposes)

Reference is hereby made to the Fourth Amended and Restated Loan and Security Agreement, dated as of May 15, 2019 (as amended, supplemented or otherwise modified from time to time, the "Loan Agreement"), by and among Freshpet, Inc., a Delaware corporation (the "Borrower"), the lenders identified on the signature pages thereof (such lenders, together with their respective successors and assigns, are referred to hereinafter each individually as a "Lender" and collectively as the "Lenders") and City National Bank, a national banking association, as the lead arranger and administrative agent for the Lenders (in such capacity as administrative agent, together with its successors and assigns, if any, in such capacity, "Administrative Agent").

Pursuant to the provisions of Section 16.11 of the Loan Agreement, the undersigned hereby certifies that (i) it is the sole record owner of the participation in respect of which it is providing this certificate, (ii) its direct or indirect partners/members are the sole beneficial owners of such participation, (iii) with respect such participation, neither the undersigned nor any of its direct or indirect partners/members is a "bank" extending credit pursuant to a loan agreement entered into in the ordinary course of its trade or business within the meaning of Section 881(c)(3)(A) of the Code, (iv) none of its direct or indirect partners/members is a "ten percent shareholder" of the Borrower within the meaning of Section 871(h)(3)(B) of the Code and

(v) none of its direct or indirect partners/members is a "controlled foreign corporation" related to the Borrower as described in Section 881(c)(3)(C) of the Code.

The undersigned has furnished its participating Lender with IRS Form W-8IMY accompanied by one of the following forms from each of its partners/members that is claiming the portfolio interest exemption: (i) an IRS Form W-8BEN or IRS Form W-8BEN-E or (ii) an IRS Form W-8IMY accompanied by an IRS Form W-8BEN or IRS Form W-8BEN-E from each of such partner's/member's beneficial owners that is claiming the portfolio interest exemption. By executing this certificate, the undersigned agrees that (1) if the information provided in this certificate changes, the undersigned shall promptly so inform such Lender and (2) the undersigned shall have at all times furnished such Lender with a properly completed and currently effective certificate in either the calendar year in which each payment is to be made to the undersigned, or in either of the two calendar years preceding such payments.

Unless otherwise defined herein, terms defined in the Loan Agreement and used herein shall have the meanings given to them in the Loan Agreement.

By:\_ Name: Title:

Exhibit B-3

[NAME OF PARTICIPANT]

Exhibit B-3 139215038v1

## FORM OF U.S. TAX COMPLIANCE CERTIFICATE (For Foreign Lenders That Are Partnerships For U.S. Federal Income Tax Purposes)

Reference is hereby made to the Fourth Amended and Restated Loan and Security Agreement, dated as of May 15, 2019 (as amended, supplemented or otherwise modified from time to time, the "Loan Agreement"), by and among Freshpet, Inc., a Delaware corporation (the "Borrower"), the lenders identified on the signature pages thereof (such lenders, together with their respective successors and assigns, are referred to hereinafter each individually as a "Lender" and collectively as the "Lenders") and City National Bank, a national banking association, as the lead arranger and administrative agent for the Lenders (in such capacity as administrative agent, together with its successors and assigns, if any, in such capacity, "Administrative Agent").

Pursuant to the provisions of Section 16.11 of the Loan Agreement, the undersigned hereby certifies that (i) it is the sole record owner of the Loan(s) (as well as any Note(s) evidencing such Loan(s)) in respect of which it is providing this certificate, (ii) its direct or indirect partners/members are the sole beneficial owners of such Loan(s) (as well as any Note(s) evidencing such Loan(s)), (iii) with respect to the extension of credit pursuant to this Loan Agreement or any other Loan Document, neither the undersigned nor any of its direct or indirect partners/members is a "bank" extending credit pursuant to a loan agreement entered into in the ordinary course of its trade or business within the meaning of Section 881(c)(3)(A) of the Code, (iv) none of its direct or indirect partners/members is a "ten percent shareholder" of the Borrower within the meaning of Section 871(h)(3)(B) of the Code and (v) none of its direct or indirect partners/members is a "controlled foreign corporation" related to the Borrower as described in Section 881(c)(3)(C) of the Code.

The undersigned has furnished the Administrative Agent and the Borrower with IRS Form W-8IMY accompanied by one of the following forms from each of its partners/members that is claiming the portfolio interest exemption: (i) an IRS Form W-8BEN or IRS Form W-8BEN-E or (ii) an IRS Form W-8IMY accompanied by an IRS Form W-8BEN-E from each of such partner's/member's beneficial owners that is claiming the portfolio interest exemption. By executing this certificate, the undersigned agrees that (1) if the information provided in this certificate changes, the undersigned shall promptly so inform the Borrower and the Administrative Agent, and (2) the undersigned shall have at all times furnished the Borrower and the Administrative Agent with a properly completed and currently effective certificate in either the calendar year in which each payment is to be made to the undersigned, or in either of the two calendar years preceding such payments.

Unless otherwise defined herein, terms defined in the Loan Agreement and used herein shall have the meanings given to them in the Loan Agreement.

[NAME OF LENDER]

Exhibit B-4 139215038v1 By:\_ Title:

Exhibit B-4 139215038v1

#### EXHIBIT C-1

#### FORM OF COMPLIANCE CERTIFICATE

[on Borrower's letterhead]

To: City National Bank 555 S. Flower

Street 21st Floor

Los Angeles, CA 90071 Attention: Garen Papazyan

Re: Compliance Certificate dated Ladies and

Gentlemen:

Reference is made to that certain **FOURTH AMENDED AND RESTATED LOAN AND SECURITY AGREEMENT** (the "Loan Agreement"), dated as of May 15, 2019, by and among **FRESHPET, INC.**, a Delaware corporation ("<u>Borrower</u>"), the lenders identified on the signature pages thereof (such lenders, together with their respective successors and assigns, are referred to hereinafter each individually as a "<u>Lender</u>" and collectively as the "<u>Lenders</u>") and **CITY NATIONAL BANK**, a national banking association, as the lead arranger and administrative agent for the Lenders (in such capacity as administrative agent, together with its successors and assigns, if any, in such capacity, "<u>Agent</u>"). All initially capitalized terms used herein shall have the meanings set forth in the Loan Agreement unless specifically defined herein.

Pursuant to Section 6.3 of the Loan Agreement, the undersigned chief financial officer of Borrower hereby certifies in his/her capacity as a chief financial officer of Borrower and not in his/her individual capacity that:

- 1. The financial information of Borrower and its Subsidiaries furnished in <u>Schedule 1</u> attached hereto (the "<u>Financial Statements</u>"), has been prepared in accordance with GAAP (except for year-end adjustments and the lack of footnotes), and fairly presents in all material respects the consolidated financial condition of Borrower and its Subsidiaries as of the date thereof or the consolidated results of operations of Borrower and its Subsidiaries for the period then ended.
- 2. Such officer has reviewed the terms of the Loan Agreement and has made, or caused to be made a review in reasonable detail of the transactions and financial condition of Borrower and its Subsidiaries during the accounting period covered by the Financial Statements.
- 3. Such review has not disclosed the existence on and as of the date of the Financial Statements, and the undersigned does not have knowledge of the existence as of the date of the Financial Statements, of any event or condition that constituted a Default or Event of Default, except for such conditions or events listed on <u>Schedule 2</u> attached hereto, specifying the nature and period of existence thereof and what action Borrower and its Subsidiaries have taken, are taking, or propose to take with respect thereto.

 $As of the \textit{ date} of the \textit{Financial Statements}, Borrower and its \textit{Subsidiaries} are in compliance with the applicable coverants contained in \underline{Section 7.18} of the Loan Agreement as \textit{Contained Statements}. As the \textit{Contained Statements} is a subsidiarie for the \textit{Contained Statements} is a subsidiarie for the \textit{Contained Statements}. As the \textit{Contained Statements} is a subsidiarie for the \textit{Contained Statements} is a subsidiarie for the \textit{Contained Statements}. As the \textit{Contained Statements} is a subsidiarie for the \textit{Contained Statements} is a subsidiarie for the \textit{Contained Statements}. As the \textit{Contained Statements} is a subsidiarie for \textit{Contained St$ 

4. demonstrated on <u>Schedule 3</u> hereof.

**FRESHPET, INC.,** a Delaware corporation

By: Name: Title: Chief Financial Officer

[SIGNATURE PAGE TO COMPLIANCE CERTIFICATE]

SCHEDULE 1

Financial Information

SCHEDULE 2

#### **SCHEDULE 3**

#### **Financial Covenants**

# 1. Adjusted EBITDA.

Borrower's and its Subsidiaries' TTM EBITDA, measured on a quarter-end basis for the twelve month period ending \$16,020,000 for the twelve month period ending June 30, 2019, (ii) \$19,620,000 for the twelve month period ending September 30, 2019, (iii) \$23,640,000 for the twelve month period ending December 31, 2019, (iv) \$28,938,000 for the twelve month period ending March 31, 2020, (v) \$33,741,000 for the twelve month period ending December 31, 2019, (iv) \$28,938,000 for the twelve month period ending December 31, 2019, (iv) \$28,938,000 for the twelve month period ending December 30, 2020, (vi) \$39,088,000 for the twelve month period ending December 30, 2020, and (vii) \$45,150,000 for the twelve month period ending December 31, 2020; provided that if Borrower delivers to the Agent a certificate of occupancy with respect to Property A, in form and substance satisfactory to the Agent, the testing of such financing covenant shall cease.

#### 2. Leverage Ratio.

Borrower's and its Subsidiaries' Leverage Ratio, measured on a quarter-end basis for the twelve month period ending ,\_\_\_is\_\_\_:\_\_\_, which amount [is/is not] less than or equal to (i) 3.00:1.00 for each twelve month period ending June 30, 2019 through and including the twelve month period ending September 30, 2021, (ii) 2.75:1.00 for each twelve month period ending December 31, 2021 through and including the twelve month period ending September 30, 2022, and (iii) 2.50:1.00 for each twelve month period ending December 31, 2022 and each twelve month period ending thereafter.

#### Fixed Charge Coverage Ratio.

Borrower's and its Subsidiaries' Fixed Charge Coverage Ratio, measured on a quarter-end basis for the 12 month period ending , \_\_is \_\_\_\_\_, which amount [is/is not] greater than or equal to 1.25:1.00.

#### 4. Capital Expenditures.

Borrower's and its Subsidiaries' Capital Expenditures during the fiscal year ending on the date of the Financial Statement is \$
, which amount [is/is not] less than or equal to (i) \$81,000,000 for the fiscal year 2019, (ii) \$69,000,000 for the

fiscal year 2020, (iii)

\$13,000,000 for the fiscal year 2021, (iv) \$13,000,000 for the fiscal year 2022, (v) \$23,000,000 for the fiscal year 2023, and (vi) \$25,000,000 for the fiscal year 2024; provided that if the amount of Capital Expenditures permitted to be made in any fiscal year as set forth above is greater than the actual amount of the Capital Expenditures (excluding the amount, if any, of Capital Expenditures made with Net Cash Proceeds reinvested pursuant to the proviso in Section 2.4(e)(ii) of the Loan Agreement) actually made in such fiscal year (the amount by which such permitted Capital Expenditures for such fiscal year exceeds the actual amount of Capital Expenditures for such fiscal year, the "Excess Amount"), then 100% of such Excess Amount (the "Carry-Over Amount") may be carried forward to the next succeeding Fiscal Year (the "Succeeding Fiscal Year"), so long as no Default or Event of Default shall have occurred and be

continuing or would result therefrom; <u>provided further</u> that the Carry-Over Amount applicable to a particular Succeeding Fiscal Year may not be used in that Fiscal Year until the amount permitted above to be expended in such Fiscal Year has first been used in full and the Carry- Over Amount applicable to a particular Succeeding Fiscal Year may not be carried forward to another fiscal year.

#### EXHIBIT L-1 FORM OF LIBOR NOTICE

City National Bank 555 S. Flower Street 21st Floor Los Angeles, CA 90071 Attention: Garen Papazyan

Ladies and Gentlemen:

Reference is made to that certain **FOURTH AMENDED AND RESTATED LOAN AND SECURITY AGREEMENT** (the "Loan Agreement"), dated as of May 15, 2019, by and among **FRESHPET, INC.**, a Delaware corporation ("<u>Borrower</u>") the lenders identified on the signature pages thereof (such lenders, together with their respective successors and assigns, are referred to hereinafter each individually as a "<u>Lender</u>" and collectively as the "<u>Lenders</u>") and **CITY NATIONAL BANK**, a national banking association, as the lead arranger and administrative agent for the Lenders (in such capacity as administrative agent, together with its successors and assigns, if any, in such capacity, "<u>Agent</u>"). All initially capitalized terms used herein shall have the meanings set forth in the Loan Agreement unless specifically defined herein.

This LIBOR Notice represents Borrower's request to elect the LIBOR Option with respect to [all] [a portion of] the [Advances][Delayed Draw Term Loans][Incremental Term Loans] in the amount of \$\_\_\_\_\_ (the "LIBOR Rate Loan")[, and is a written confirmation of the telephonic notice of such election given to Lender].

The LIBOR Rate Loan will have an Interest Period of [1, 2, 3 or 6] month(s) commencing on\_

Borrower represents and warrants that no Event of Default has occurred and is continuing on the date hereof, nor will any thereof occur after giving effect to the request above.

FRESHPET, INC., a Delaware corporation, as Borrower

By Name: Title:

### **EXHIBIT N-1**

#### FORM OF NOTICE OF BORROWING

City National Bank 555 S. Flower Street 21st Floor Los Angeles, CA 90071 Attention: Garen Papazyan

**RE:** Notice of Borrowing under that certain **FOURTH AMENDED AND RESTATED LOAN AND SECURITY AGREEMENT** (the "Loan Agreement"), dated as of May 15, 2019, by and among **FRESHPET, INC.**, a Delaware corporation ("Borrower") the lenders identified on the signature pages thereof (such lenders, together with their respective successors and assigns, are referred to hereinafter each individually as a "Lender" and collectively as the "Lenders") and **CITY NATIONAL BANK**, a national banking association, as the lead arranger and administrative agent for the Lenders (in such capacity as administrative agent, together with its successors and assigns, if any, in such capacity, "Agent").

## Ladies and Gentlemen:

Reference hereby is made to the Loan Agreement. Capitalized terms used herein, and not otherwise defined herein, have their respective meanings given them in the Loan Agreement.

Pursuant to Section 2.3 of the Loan Agreement, the Borrower hereby gives you irrevocable notice that the Borrower hereby requests [a Delayed Draw Term Loan][an Advance][an Incremental Term Loan] under the Loan Agreement, and sets forth below the information relating to such proposed Advance (the "Proposed Borrowing") as required by Section 2.3 of the Loan Agreement.

- a. The Business Day of the Proposed Borrowing is\_.
- b. The amount of the Proposed Borrowing is \$\_.
- c. The Proposed Borrowing will be [a Delayed Draw Term Loan][an Advance][an Incremental Term Loan].
- d. The Proposed Borrowing will be a [Base Rate Loan] [LIBOR Rate Loan] with an Interest period of [1][2][3][6] month(s).
- e. The Proposed Borrowing is to be made pursuant to the instructions set forth on <u>Exhibit A</u> attached hereto.

The Borrower hereby further certifies that on and as of the date for the Proposed Borrowing, and after giving effect to the Proposed Borrowing:

- a. the representations and warranties contained the Loan Documents are true and correct in all material respects on and as of the date of such extension of credit, as though made on and as of such date (except to the extent that such representations and warranties relate solely to an earlier date),
  - b. no Default or Event of Default shall have occurred and be continuing on the date of such extension of credit, nor shall either result from the making thereof,
- c. no injunction, writ, restraining order, or other order of any nature restricting or prohibiting, directly or indirectly, the extending of such credit shall have been issued and remain in force by any Governmental Authority against Borrower, Lender, or any of their Affiliates,
  - d. no Material Adverse Change has occurred[.][, and]
  - e. [all actions and conditions specified in Section 6.17 of the Loan Agreement shall have been taken or satisfied, as applicable, with respect to such Real Property[.], and]];
- f. [after giving pro forma effect to any such Delayed Draw Term Loan, the Leverage Ratio of Borrower and its Subsidiaries shall not exceed the Leverage Ratio required by Section 7.18(a)(ii) of the Loan Agreement minus 0.25, and Borrower shall have delivered to the Agent a Compliance Certificate demonstrating compliance with such condition.]

Very truly yours,

**FRESHPET, INC.,** a Delaware corporation

By:\_ Name: Title:

<sup>2</sup> Include if any portion of the proceeds of such Advance or Delayed Draw Term Loan will be used to acquire any Real Property.

3 Include for Delayed Draw Term Loans.

# EXHIBIT A WIRING INSTRUCTIONS

Payee	Wiring Instructions
	Bank: [City/State] ABA #
	Account # Ref:

Schedule A-1 Agent's Account

An account at a bank designated by Agent from time to time as the account into which Borrower shall make all payments to Agent under this Agreement and the other Loan Documents; unless and until Agent notifies Borrower to the contrary, Agent's Account shall be that certain deposit account bearing account number 101306674 and maintained by Agent at City National Bank, Los Angeles, N.A., ABA # 122016066, 2100 Park Place, El Segundo, CA 90245.

# Schedule C-l Commitments

Lender	Revolver	Delayed Draw Term	Total Commitment
	Commitment	Loan Commitment	
City National Bank	\$17,500,000	\$27,500,000	\$45,000,000
Bank of America, N.A.	\$17,500,000	\$27,500,000	\$45,000,000
All Lenders	\$35,000,000	\$55,000,000	\$90,000,000

Schedule D-I Designated Account

Account number 123380541 of Borrower maintained with Borrower's Designated Account Bank, or such other deposit account of Borrower (located within the United States) that has been designed as such, in writing, by Borrower to Agent.

"Designated Account Bank" means City National Bank, Los Angeles, N.A., whose office is located at 2100 Park Place, El Segundo, CA 90245, and whose ABA number is 122016066.

# **Schedule P-2** Permitted Holders

- 1. MidOcean Partners III, L.P.
- 2. MidOcean Partners III-A, L.P.
- 3. MidOcean Partners III-D, L.P.

# Schedule R-l

# Real Property Collateral

146 North Commerce Way, Hanover Township PA. 176 North Commerce Way, Hanover Township PA

# Schedule 2.7(a) Cash Management Banks

# • Bank of America

100 West 33rd Street New York, NY 10001 ABA number - 0260-0959-3

 City National Bank, Los Angeles, N.A., 2100 Park Place, El Segundo, CA 90245, ABA number is 122016066.

# Schedule 5.5

## Location of Inventory and Equipment

- 1. 146 North Commerce Way, Bethlehem, PA 18017
- 2. 176 North Commerce Way, Bethlehem, PA 18017
- 3. 7132 Ruppsville Road, Allentown, PA 18106
- 4. Turners LTD Fordam Road, Newmarket, SFK CB8 7NR, United Kingdom
- 5. Agro Merchants Barneveld Harselaarseweg 133, 3771MA Barneveld, The Netherlands
- 6. Agro Forelgracht 15 6642 Ed Beuningen, The Netherlands
- 7. 2410 Broadhead Road, Bethlehem, PA 18020
- 8. 787 California Road, Quakertown, PA 18951
- 9. 303 West Marquette Avenue, Oak Creek, WI 53154

# Schedule 5.7(a) States of Organization

Entity	Jurisdiction of Organization
Freshpet, Inc.	Delaware
FP Foods Realty PA, LLC	Delaware
Professor Connor's Canada Inc.	Ontario, Canada
Freshpet Europe LTD	England and Wales
Freshpet NE B.V.	The Netherlands

# Schedule 5.7(b) Chief Executive Offices

Entity	Chief Executive Office
Freshpet, Inc.	400 Plaza Drive FL 1
	Secaucus, New Jersey 07094
FP Foods Realty PA, LLC	400 Plaza Drive FL 1
-	Secaucus, New Jersey 07094
Professor Connor's Canada Inc.	502-500 Avenue Road, Toronto, Ontario, Canada M4V 2J6
Freshpet Europe LTD	Oak House, 2 Woodward Close Ipswich United Kingdome IP2 0EA
Freshpet NE B.V.	Forelgracht 15 Beuningen 6642 ED Netherlands

Schedule 5.7(c) Organizational Identification Numbers

<u>Schedule 5.7(c)</u> Organization			
Entity	Organizational Identification Number	Authority	
Freshpet, Inc.	3880284	Delaware Certificate of Incorporation	
Freshpet, Inc.	20-1884894	Employer Identification Number	
FP Foods Realty PA, LLC	5763148	Delaware Certificate of Incorporation	
FP Foods Realty PA, LLC	38-3972891	Employer Identification Number	
Professor Connor's Canada Inc.	001690726	Ontario Corporate Number	
Freshpet Europe LTD	10545785	Companies House Number	
Freshpet NE B.V.	70842582	Trade Register Chamber of Commerce	

None.

Schedule 5.8(c) Capitalization of Borrower's Subsidiaries

	Capitalization of Do		
Subsidiary	Jurisdiction of Organization	Authorized Stock	Held by Borrower
FP Foods Reality PA, LLC	Delaware	N/A	100% membership interest
Professor Connor's Canada Inc.	Ontario, Canada	Class A Common - 200 shares Class B Common - 200 shares	200 Class A common shares (100%) 200 Class B common shares (100%)
Freshpet Europe LTD	England and Wales	N/A	100% membership interest
Freshpet NE B.V.	The Netherlands	Common -100 shares	Common -100 shares (100%)

<u>Schedule 5.8(d)</u>

Subscriptions, Options, Warrants or Calls Relating to Shares of Borrower's Subsidiaries' Capital Stock

None.

# Schedule 5.10 Litigation

None.

(a) None. (b) None. (c) None. (d) None.

# **Schedule 5.16** Intellectual Property

# Trademarks — Registered & Pending

Status	Country	Mark	Appl. No./ Date Filed	Reg. No. / Date Registered	Owner
Registered	Canada	DELI FRESH	1237682 11/12/2004	TMA741181 6/1/2009	Freshpet, Inc.
Registered	Canada	DOGNATION	1541048 8/24/2011	TMA866353 11/29/2013	Freshpet, Inc.
Pending	Canada	FRESH CUTS	1831505 4/6/2017		Freshpet, Inc.
Registered	Canada	Freshpet (logo)	1310701 7/26/2006	TMA745960 8/24/2009	Freshpet, Inc.
Registered	Canada	Freshpet (new logo)	1834776 4/27/2017	TMA1004048 8/31/2018	Freshpet, Inc.
Registered	Canada	FRESHPET	1541043 8/24/2011	TMA843186 2/13/2013	Freshpet, Inc.
Registered	Canada	FRESHPET DOG JOY TREATS	1541045 8/24/2011	TMA865936 11/26/2013	Freshpet, Inc.
Registered	Canada	FRESHPET STEWS	1797079 8/23/2016	TMA1018988 4/9/2019	Freshpet, Inc.
Registered	Canada	FRESHPET VITAL	1582000 6/13/2012	TMA856664 7/31/2013	Freshpet, Inc.
Registered	Canada	HOMESTYLE SELECT	1310322 7/26/2006	TMA722403 8/7/2008	Professor Connors, Inc.
Registered	Canada	NATURE'S FRESH	1541044 8/24/2011	TMA866727 12/5/2013	Freshpet, Inc.
Registered	Canada	PETS EAT FRESH	1497098 9/23/2010	TMA813823 12/12/2011	Freshpet, Inc.
Registered	Canada	PLATS GRILLES (MC) (ROASTED MEALS)	1547896 10/14/2011	TMA838832 12/20/2012	Freshpet, Inc.
Registered	Canada	PROFESSOR CONNOR'S DELIFRESH PET FOODS	1217119 5/12/2004	TMA654589 12/7/2005	Freshpet,

Status	Country	Mark	Appl. No./ Date Filed	Reg. No. / Date Registered	Owner
		Professor Connor's DeliFresh Pet			
		Foods (logo)	1219143	TMA654095	Freshpet,
Registered	Canada		5/31/2004	12/1/2005	Inc.
Registered	Canada	PROFESSOR CONNOR'S HOSPITAL DIET	1235841 10/27/2004	TMA654174 12/1/2005	Freshpet, Inc.
Registered	Canada	ROASTED MEALS	1547894 10/14/2011	TMA838834 12/20/2012	Freshpet, Inc.
Pending	Canada	FRESHPET BRAND DOG FOOD SELECT SLICE & SERVE ROLL (& Design)	1949890 3/6/2019	1	Freshpet,
Pending	Canada	REAL FOOD FRESH FROM THE FRIDGE NON GMO (& Design)	1949891 3/6/2019	1	Freshpet,
Pending (Allowed)	Canada	THE POWER OF FRESH	1864574 10/25/2017		Freshpet, Inc.
Pending	Canada	VITAL FRESH CUTS	1948256 2/26/2019		Freshpet, Inc.
Registered	International Register (European Union)	DELI FRESH	1306466 6/27/2016	1306466 6/27/2016	Freshpet,
Registered	European Union	DOGNATION	10218683 8/25/2011	10218683 1/25/2012	Freshpet, Inc.
Registered	European Union	Freshpet (logo) (color)	14368922 7/16/2015	14368922 11/6/2015	Freshpet,
Registered	European Union	FRESHPET	10218741 8/25/2011	10218741 1/7/2012	Freshpet, Inc.
Registered	European Union	FRESHPET DOG JOY TREATS	10218642 8/25/2011	10217642 1/7/2012	Freshpet, Inc.
Registered	European Union	FRESHPET ROASTED MEALS	10324713 10/7/2011	10324713 2/16/2012	Freshpet, Inc.
Registered	International Register (European Union)	FRESHPET STEWS	1311488 8/19/2016	1311488 8/19/2016	Freshpet, Inc.

Status	Country	Mark	Appl. No./ Date Filed	Reg. No. / Date Registered	Owner
Registered	European Union	NATURE'S FRESH	10218717 8/25/2011	10218717 1/8/2012	Freshpet, Inc.
Registered	European Union	PETFRESH	10992808 6/26/2012	10992808 3/26/2013	Freshpet, Inc.
	European	VITAL, BRAND DOG FOOD, NATURALLY FRESH, INSPIRED BY NATURE (includes bowl/product image)	10538619	10538619	Freshpet,
Registered	Union		1/3/2012	5/31/2012	Inc.
Registered	Germany	Freshpet (logo)	307489620 7/25/2007	30748962 10/16/2007	Professor Connors Inc.
Registered	Germany	Frisches Haustier	3020181029094 3/14/2018	302018102909 4/16/2018	Freshpet, Inc.
Registered	UK	FRESH PETS	2269182 5/3/2001	2269182 4/26/2002	Professor Connor's, Inc. also t/a FreshPet
Registered	UK	FRESH FROM THE FRIDGE	3265924 10/25/2017	3265924 1/19/2018	Freshpet, Inc.
Registered	UK	FRESHPET FRESH FROM THE FRIDGE	3265925 10/25/2017	3265925 1/19/2018	Freshpet, Inc.
Registered	UK	PETS PEOPLE PLANET	3252401 8/24/2017	3252401 11/17/2017	Freshpet, Inc.
Registered	UK	THE POWER OF FRESH	3265920 10/25/2017	3265920 1/19/2018	Freshpet, Inc.
Registered	US	BARK FOR A CURE	85966194 6/21/2013	4633387 11/4/2014	Freshpet, Inc.
Registered	US	CELEBRATION COOKIE	85854618 2/20/2013	4456600 12/24/2013	Freshpet, Inc.
Registered	US	CHILLY WAGS	85439504 10/5/2011	4239027 11/6/2012	Freshpet, Inc. DBA FreshPet
Registered (Renewed)	US	DELI FRESH	78526899 12/3/2004	3097552 5/30/2006	Freshpet, Inc.
Registered	US	Deli Fresh (logo)	86928841 3/4/2016	5063073 10/18/2016	Freshpet, Inc.
Registered	US	DOGNATION	85306353 4/27/2011	4163353 6/26/2012	Freshpet, Inc.

Status	Country	Mark	Appl. No./ Date Filed	Reg. No. / Date Registered	Owner
Registered	US	FRESH CUTS	87310241 1/23/2017	5444443 4/10/2018	Freshpet, Inc.
Registered	US	FRESH FOOD FOR FIDO	85646134 6/7/2012	4322208 4/16/2013	Freshpet, Inc.
Registered	US	FRESH FROM THE FRIDGE	87535893 7/20/2017	5406652 2/20/2018	Freshpet, Inc.
Registered	US	FRESH FROM THE KITCHEN	86346669 7/24/2014	4701478 3/10/2015	Freshpet, Inc.
Registered	US	FRESH INGREDIENTS. REAL HONEST FOOD.	86521129 2/2/2015	4875862 12/22/2015	Freshpet, Inc.
Registered (Renewed)	US	Freshpet (logo)	78803099 1/31/2006	3383087 2/12/2008	Freshpet, Inc.
Registered	US	FRESHPET	85245285 2/17/2011	4028462 9/20/2011	Freshpet, Inc.
Registered	US	FRESHPET DOG JOY	85306337 4/27/2011	4163352 6/26/2012	Freshpet, Inc.
Registered	US	FRESHPET PETS PEOPLE PLANET	86287260 5/21/2014	4800707 8/25/2015	Freshpet, Inc.
Registered	US	FRESHPET PURE	87462157 5/24/2017	5439688 4/3/2018	Freshpet, Inc.
Registered	US	FRESHPET STEWS	87056477 6/1/2016	5068840 10/25/2016	Freshpet, Inc.
Registered	US	FRESHPET VITAL	85647320 6/8/2012	4318889 4/9/2013	Freshpet, Inc.
Registered	US	HEALTHY MIXERS (Supplemental Register)	87126926 8/4/2016	5220074 6/6/2017	Freshpet, Inc.
Registered	US	NATURE'S FRESH	78778585 12/21/2005	3779569 4/20/2010	Freshpet, Inc.
Registered	US	PETFRESH	85659531 6/22/2012	4564737 7/8/2014	Freshpet, Inc.
Registered	US	PETS EAT FRESH	85659548 6/22/2012	4564738 7/8/2014	Freshpet, Inc.
Registered	US	PETS PEOPLE PLANET	87430073 4/28/2017	5403376 2/13/2018	Freshpet, Inc.

Status	Country	Mark	Appl. No./ Date Filed	Reg. No. / Date Registered	Owner
Registered	US	ROASTED MEALS (Supplemental Register)	85306362 4/27/2011	4127101 4/10/2012	Freshpet, Inc.
Registered	US	SIMPLE DELIGHTS	86314629 6/19/2014	4729209 4/28/2015	Freshpet, Inc.
Registered	US	VETERINARY NUTRITION	87377288 3/20/2017	5333853 11/14/2017	Freshpet, Inc.
Registered	US	VITAL	85306374 4/27/2011	4490116 3/4/2014	Freshpet, Inc.
Registered	US	VITAL COMPLETE MEALS	85306384 4/27/2011	4210970 9/18/2012	Freshpet, Inc. DBA FreshPet
Registered	US	VITAL PURE	86966534 4/6/2016	5116082 1/3/2017	Freshpet, Inc.
Registered	US	VITAL WHOLE BLENDS	86838339 12/3/2015	5097020 12/6/2016	Freshpet, Inc.
Registered	US	WAG FACTOR	85439509 10/5/2011	4306121 3/19/2013	Freshpet, Inc.
Registered	US	WAG-GURT	85439497 10/5/2011	4239026 11/6/2012	Freshpet, Inc. DBA FreshPet
Pending	US	FRESH REAL FOOD (Supplemental Register)	88013486 6/25/2018		Freshpet, Inc.
Not filed yet	US	FOOD FOR DOGS, NOT DOG FOOD			
Registered	US	HOMESTYLE CREATIONS	87947679 6/4/2018	5735344 4/23/2019	Freshpet, Inc.
Registered	US	THE NEXT BEST THING TO A HOME- COOKED MEAL	87851296 3/27/2018	5629554 12/11/2018	Freshpet, Inc.
Registered	US	VITAL PIONEERING REAL FOOD	87800650 2/16/2018	5651338 1/8/2019	Freshpet, Inc.
Registered	US	ROASTED MEALS	87726487 12/19/2017	5508599 7/3/2018	Freshpet, Inc.

# Licenses

None. **Patents** None. **Copyrights** None.

# Schedule 5.18 Deposit Accounts and Securities Accounts

Entity	Type of Account	Account Number	Name and Address Associated with the Account
Freshpet, Inc.	Primary inbound account	0095-0517-5245	Bank of America 100 West 33rd Street New York, NY 10001
Freshpet, Inc.	Main AP Account	3810-1069-2460	Bank of America 100 West 33rd Street New York, NY 10001
Freshpet, Inc.	FSA Account	0038-1882-2370	Bank of America 100 West 33rd Street New York, NY 10001
Freshpet, Inc.	Plant Petty cash	0038-1259-2758	Bank of America 100 West 33rd Street New York, NY 10001
Freshpet, Inc.	Savings Account	0038-1257-8028	Bank of America 100 West 33rd Street New York, NY 10001
Freshpet, Inc.	Money Market Account	0038-1257-8073	Bank of America 100 West 33rd Street New York, NY 10001
Professor Connors Canada Inc.	US Dollar Account	0617-7318566	TD Canada Trust
Professor Connors Canada Inc.	TD Account	0617-5300778	TD Canada Trust
Freshpet Inc. / FP Europe LTD	GBP Account	6008-7828-8018	Bank of America London UK
Freshpet Inc. / FP Europe LTD	Euro Account	6022-5642-4016	Bank of America Dublin Ireland
Freshpet Inc. / FP NE B.V.	Euro Account	6004-2052-0016	Bank of America Amsterdam Netherlands

None.

#### CERTIFICATIONS

#### I, William B. Cyr, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Freshpet, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the

audit committee of the registrant's board of directors (or persons performing the equivalent functions):

(a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

(b) over financial reporting.

Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control

Date: August 6, 2019

<u>/s/ William B. Cyr</u>
William B. Cyr
Chief Executive Officer

#### CERTIFICATIONS

#### I, Richard Kassar, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Freshpet, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the

audit committee of the registrant's board of directors (or persons performing the equivalent functions):

(a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

(b) over financial reporting.

Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control

Date: August 6, 2019

/s/ Richard Kassar Richard Kassar Chief Financial Officer

#### CERTIFICATION PURSUANT TO 18 U.S.C. § 1350, AS ADOPTED PURSUANT TO § 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the filing of the Quarterly Report on Form 10-Q of Freshpet, Inc., a Delaware corporation (the "Company"), for the quarter ended June 30, 2019, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), each of the undersigned officers of the Company certifies, pursuant to 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, that, to such officer's knowledge:

1. The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and

2. The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company as of the dates and for the periods expressed in the Report.

Date: August 6, 2019

<u>/s/ William B. Cyr</u>
William B. Cyr
Chief Executive Officer

/s/ Richard Kassar Richard Kassar Chief Financial Officer

The foregoing certification is being furnished solely pursuant to 18 U.S.C. § 1350 and is not being filed as part of the Report or as a separate disclosure document.